INFORMATION TO USERS

This manuscript has been reproduced from the microfilm master. UMI films the text directly from the original or copy submitted. Thus, some thesis and dissertation copies are in typewriter face, while others may be from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleedthrough, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by sectioning the original, beginning at the upper left-hand corner and continuing from left to right in equal sections with small overlaps.

Photographs included in the original manuscript have been reproduced xerographically in this copy. Higher quality 6" x 9" black and white photographic prints are available for any photographs or illustrations appearing in this copy for an additional charge. Contact UMI directly to order.

ProQuest Information and Learning 300 North Zeeb Road, Ann Arbor, MI 48106-1346 USA 800-521-0600

UM

NOTE TO USERS

This reproduction is the best copy available.

UMI

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

University of Alberta

Tissue Oxygenation and Blood Volume and Whole-body Metabolic Responses During Low-back Intensive Occupational Tasks

by



A dissertation submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Rehabilitation Science

Faculty of Rehabilitation Medicine

Edmonton, Alberta 2002

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



National Library of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque nationale du Canada

Acquisitions et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

Your Ne Votre rélérence

Our file Notre référence

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission. L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

0-612-68600-0

Canadä

University of Alberta

Library Release Form

Name of Author:	Rammohan Venkata Maikala
Title of Thesis:	Tissue Oxygenation and Blood Volume and Whole-body Metabolic Responses During Low-back Intensive
	Occupational Tasks

Degree: Doctor of Philosophy

Year this Degree Granted: 2002

Permission is hereby granted to the University of Alberta to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly or scientific research purposes only.

This author reserves all other publication and other rights in association with the copyright in the thesis, and except as herein before provided, neither the thesis nor any substantial portion thereof may be printed or otherwise reproduced in any material form whatever without the author's prior written permission.

April 10, 2002

H.No. 25-85, Ashok Road Mancherial, Dist. Adilabad Andhra Pradesh, INDIA 504208

Faculty of Graduate Studies and Research

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research for acceptance, a thesis entitled *Tissue Oxygenation and Blood Volume and Whole-body Metabolic Responses During Low-back Intensive Occupational Tasks* submitted by *Rammohan Venkata Maikala* in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Rehabilitation Science.

Dr. Y. Bhambhani (Co-Supervisor)

Dr. M. upervisor)

Dr. R. Jones

P. Gervais

A. Hargens (External Examiner)

16th April, 2002

Dedication

To my beloved brother 'Karunamohan Maikala' who is the spine of my life. I am not sure what I will do without you. Your kindness, patience, unselfishness, and generosity brought our family to what it is today. You are everything one brother can wish for. Many thanks for helping me throughout my academics and encouraging me to follow my heart. Also to my parents, 'Padma' and 'Mohan Rao' for teaching me the importance of education, sharing and caring. You both are my almighty god. Your love and compassion carried my dreams throughout my educational pursuits. Thanks to both of you for giving me such a great brother, and lovely and good hearted sister

`Jyotsna'.

ABSTRACT

This research examined tissue and whole-body physiological responses during seated whole-body vibration [WBV] (*Study #1*) and repetitive pushing-pulling/arm cranking (*Study #2*) in healthy men and women. The acute physiological responses were measured using open circuit spirometry and tissue oxygenation was monitored using near infrared spectroscopy.

Study #1: Each subject was exposed to 3 Hz, 4.5 Hz, and 6 Hz WBV in random order on three separate days. The protocol included a six-minute baseline, eight minutes of WBV 'With' and 'Without' backrest, with four-minute recovery periods following each exposure. In each WBV session the subject performed maximal rhythmic handgrip contractions during the final minute.

In both genders, oxygenation and blood volume monitored on the right pre-frontal cortex increased significantly during WBV and were not influenced by the dose 'With' and 'Without' a backrest. In contrast, oxygenation and blood volume on the right lumbar erector spinae significantly decreased with exposure to WBV. Cerebral oxygenation and blood volume were not influenced by backrest support, whereas muscle blood volume was significantly higher without backrest support in both genders. Addition of hangrip contractions significantly increased the cerebral oxygenation and blood volume with concomitant decreases in these variables in the erector spinae muscle. These changes were not related to grip strength in both genders. Exposure to WBV significantly increased oxygen uptake, heart rate and ventilation in both genders but did not elicit significant changes in cardiac output and stroke volume. These differences were more evident in the absence of a backrest.

Study #2: During incremental exercise to voluntary fatigue, the peak oxygen uptake, power output, ventilation rate and oxygen pulse were significantly higher during arm cranking compared to pushing-pulling, with men showing greater values than women. Despite the lower cardiorspiratory responses during pushing-pulling, localized muscle oxygenation was significantly higher, suggesting a greater peripheral limitation during this task. Significant deoxygenation was observed in the right lumbar erector spinae during both these exercise modes, but the values were not significantly different. These findings strengthen the argument that occupational fitness screening should focus on task specific rather than standardized upper body exercise protocols.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

PREFACE

The nature of light is unknown and yet light itself is the key to explain many unknown things"

Hammamatsu Photonics

My interests in "optical investigations of physiology" started the day I joined Dr. Yagesh Bhambhani's Work Physiology Research Laboratory at the Faculty of Rehabilitation Medicine, University of Alberta in 1996. As a result of his mentorship, my personal interests became directed more into understanding of *Photon Propagation in Tissue* than just application of the Near Infra-red Spectroscopy (NIRS) per se. Being an engineer, I was fascinated by the theory and principles behind NIRS, and spellbound by its enormous potential in its noninvasiveness and user-friendliness (of course, the burns on my hands and low-back might prove otherwise). My appreciation of Physiology strengthened with studying its fundamentals, mainly the infamous *Fick equation*, which I believe is so simple but at the same time crucial, as it defines blood flow in the vascular system or fluid mechanics in engineering.

For this doctoral work, I combined my main research interest - Industrial Ergonomics, with the toughest field I have ever come across - Work Physiology. I sincerely hope you enjoy reading this doctoral work of an engineer who attempted to explain occupational physiology in its simplest form.

This dissertation is divided into seven chapters. Chapter 1 introduces the rationale of studying low-back intensive occupational tasks. There is a brief review of injury statistics related to occupations susceptible to mechanical loading and low-back complaints, specifically Whole-body Vibration and Pushing-Pulling. The Introduction extends to various physical risk factors, and provides the impetus of studying both central and peripheral physiological variables. The final pages give details of the NIRS, and conclude with a brief description of this technique to both cerebral and skeletal muscle tissue.

In the context of investigating the role of vibration dose, backrest support, work during vibration, and the influence of gender, Chapters 2,3, and 4 describe a variety of research methodologies. Chapter 2 explores the application of NIRS in studying the cerebral region during exposure to Whole-body Vibration. Chapter 3 evaluates the lumbar erector spinae muscle physiology with NIRS during exposure to Whole-body Vibration. Chapter 4 describes the whole-body physiological responses using the Open Circuit Spirometry during exposure to Whole-body Vibration. These chapters also examine the role of upper body aerobic fitness of the exposed population on their physiological responses.

Chapter 5 examines both biceps brachii (oxygenation and blood volume) responses and cardiorespiratory responses in healthy men and women during repetitive pushing-pulling, and compares them with a standardized arm cranking protocol till exhaustion.

Chapter 6 studies the role of lumbar erector spinae muscle oxygenation and blood volume responses in healthy men and women during both repetitive pushing-pulling and standardized arm cranking protocols.

Chapter 7 is a general discussion of the overall research, and finally the Appendices contain further details of study protocols, and statistical adjustments.

Have fun surfing!

Acknowledgements

I would like to thank,

Dr. Yagesh Bhambhani, the Human Calculator and respected 'Guru' who guided me during every step of my doctoral program. Even though my dreams of becoming a sportsman were curtailed during my childhood, your encouragement in getting me involved in sports-related projects gave me immense satisfaction. Thanks also for involving me in the Adapted Physical Activity-related projects. I will never forget your statement "Quality Not Quantity". Certainly, our joint publications in various disciplines are an excellent indication of your brilliant ideas and our productivity. You taught me "Work Physiology is more than just understanding VO₂". Your introduction to NIRS has been the strength of my doctoral studies, and the application of NIRS continues to be my focus. Because of my nocturnal habits, special thanks, also, for organizing most of our meetings during my stay in Edmonton at 10:30 AM!

Dr. Michelle Battié, who has given me her valuable time when and as needed. Your input during the beginning of my doctoral program in addition to solving various administrative issues [courtesy: my rocky start in September 1995] as a chair of Physical Therapy will never be forgotten. Thanks for the moral support throughout my doctoral program!

Dr. Gordon Bell, the ever-friendly professor who always encouraged me to think "what am I asking for" in a research project and, subsequently, "how should I interpret the physiological data". Thanks for showing me in 1996 how muscle biopsies are done and analyzed.

When I bought the book *Principles of Exercise Testing and Interpretation* by Karl Wasserman et al., I was not sure If I was going to read that book thoroughly. However, Dr. Richard Jones! your advice before my Candidacy made me read it and understand the Bohr Curve. Thanks for teaching me the importance of interpreting correlations [during my Candidacy] and the importance of presenting and explaining the graphs.

The one course that stands out in my doctoral program is Seminars in Biomechanics. Dr. Pierre Gervais! you made the course simple but challenging. I am not sure how often a student can say "the course I took is worth the money", and certainly your course was worth every penny. Thanks for your advice on how to interpret the vibration data.

To all of you! Umpteen times I showed up at your door without prior notice and still every one of you had time for me. That indicates how much you care for your students' work. I am touched by such humbleness and sincerity regarding my future.

Dr. Alan Hargens, a big thanks for being my external examiner. I am honored to have a person of your stature on my committee. Your research on various species including humans is just unimaginable. Your input on my doctoral work is very much appreciated, in particular the questions on "intramuscular pressure". Thanks for extending the post-doctoral opportunity at the UCSD. Considering your great credentials, you are a very down-to-earth man! Special thanks to Dr. Gita Murthy for taking time to come to *The Cold Place* and for asking important and thought-provoking questions.

1 also would like to thank Dr. Herbie Rochet [past Associate Dean] and Dr. Al Cook [Dean] of the Faculty of Rehabilitation Medicine, University of Alberta, for going to great lengths in accommodating my needs and helping during every step of my doctoral program. Dr. Cook, thanks for your advice during my turbulent period.

Anita Moore, my Canadian mother, what can I say! You have done so much for me that I can't express in words. If I stayed at U of A, it is not only because of my mentors but also due to your persistence, kindness, and love. In every step of my Corbett life, your presence is visible. Last, but not least your editorial skills are amazing! Thanks for taking your time to teach me the importance of determination and dedication.

My thanks are also extended to Martha Roxburgh of the Occupational Analysis and Performance Unit for letting me use the laboratory and computers at free will. My thanks also to Adele Colon and Vicki Ross for making my life at Corbett Hall smoother and simpler. I would be remiss if I did not thank AI Fleming, the technical genius who, instead of laughing at my ideas, made the vibration project a great success. It's nice knowing you, Al! Thanks also to Brian Henderson and Ashutosh Muni for software programming as well as to Dr. Bonnie Dobbs and Dr. Todd Rodgers for statistical help. Dr. Nanda Kittusamy [NIOSH, Spokane, WA], for helping and sharing scientific ideas throughout my ergonomics career. You are an excellent buddy and a great colleague to work with.

I am very grateful to Gurpreet Sidhu who is very special to me. You have helped me in both good and bad times. You are a sweetheart and I learned a lot from you. Thanks to your family who took care of me as if I were a part of your family.

Many thanks to my first friends in Edmonton, Dr. Corinna Andiel, Dr. Srinivas 'Al' Padmanabhuni and Gagan Deep Singh for the good times. Special thanks to Rajeev Nair and Arees Rauf. Rajeev, you are one friend who is sincere, kind, calm, and caring. Your friendship is truly missed. Arees, your genuine friendship and ever-helping nature is the best one can wish for. Keep up the good work.

To the 'Desi' gang: Ramesh Kadali the trouble-shooter, Venkat Reddy, Trivikram Chebrulu, and Raghu Dronamraju for helping in troubled times; Prabhu and Pallavi Reddy, Rajat and Shilpee Bhargava, Hanumanth and Pallavi Reddy, Prasanna, Bramha, Srinath, and Sridhar for sharing their software fortunes.

I would also like to express my gratitude to Mamdouh Farag, Winnie Ting, and Geetanjali Kashyap for their assistance during my doctoral projects. Thanks for the good times Mamdouh! you are fun to work with. Many thanks to my very good and intelligent friends, Elle Stewart, the sweet and most 'worldly' knowledgeable lady I have ever met, Dr. Sharla King for being such a good friend, Dr. Wonita Janzen [The Statistical Encyclopedia], and Denyse Hayward, the NSERC lady from downunder. Thanks also to Mike Kennedy, Eric Parent, and Doug Gross for being good doctoral buddies.

Thanks to the Liberty Mutual Research Center staff for preparing me for the doctoral defense. Last, but not least, thanks to all the human volunteers who went through my torture with a smile. Special thanks to Tyler Boake and Allison Black for letting me use their pictures for this doctoral work.

Table of Contents

CHAPTER 1

Introduction	1
Injury Epidemiology	1
Whole-body Vibration	3
Influential Factors	6
Vibration at Resonance	8
ISO-2631 document (ISO 2631/1-Part 1, 1985 and 1997)	8
Vibration from Seat to Head	9
Role of Vibration Dose on Cardiorespiratory Responses	10
Role of Postures Adopted During Whole-body Vibration	15
Effect of Combined Stresses	16
Influence on Physiological Responses	17
Role of Gender During Whole-body Vibration	19
Low-Back Muscle Fatigue During Whole-body Vibration	21
Role of Physical Fitness related to Low-back Intensive Tasks	24
Repetitive Pushing-Pulling	26
Aerobic protocols	28
Role of Gender	29
Central and Peripheral Factors Influencing Physical Work	
Capacity	31
Maximal Aerobic Power (VO _{2max})	31
Energy expenditure	32
Muscle Metabolism	32
Tissue Oxygenation and Blood Volume	
Cerebral Physiology	34
Muscle Physiology	37
Regional Distribution of Blood Flow	39

Review of Near Infrared Spectroscopy			
Modified Beer's Law applied to the Human Tissue			
Theoretical Basis	45		
Validity of Near Infrared Spectroscopy			
Reliability of Near Infrared Spectroscopy	51		
Applications of Near infrared spectroscopy			
Monitoring Cerebral Oxygenation and Blood Volume	52		
Skeletal Muscle Oxygenation and Blood Volume	54		
Role of Adipose Tissue Thickness	57		
Summary of Introduction	58		
Objectives	59		
References	62		
CHAPTER 2			
Cerebral Oxygenation and Blood Volume Responses during Exposure to Seated			
Whole-body Vibration	85		
Materials and Methods	88		
Results	93		
Discussion	97		
Conclusion	108		
References	110		
CHAPTER 3			
Lumbar Erector Spinae Oxygenation and Blood Volume Responses duri	ng		
Exposure to Seated Whole-body Vibration	129		
Materials and Methods	132		
Results	136		
Discussion	139		
Conclusion	149		
References	151		

CHAPTER 4

Acute Physiological Responses during Exposure to Seated Whole-body	
Vibration	165
Materials and Methods	167
Results	171
Discussion	177
Conclusion	185
References	187
CHAPTER 5	
Cardiorespiratory and Biceps Brachii Oxygenation and Blood Volume	
Comparisons during Arm Cranking and Task-specific Pushing-Pulling	198
Materials and Methods	200
Results	208
Discussion	213
Conclusion	227
References	228
CHAPTER 6	
Lumbar Erector Spinae Oxygenation and Blood Volume Comparisons du	ring Arm
Cranking and Task-specific Pushing-Pulling	249
Materials and Methods	251
Results	257
Discussion	261
Conclusion	272
References	274
CHAPTER 7	
General Discussion	289

	305
Consent Forms	
Whole-body Vibration Study	306
Repetitive Pushing-Pulling	310
Physical Activity Readiness Questionnaire	314
APPENDIX B	315
Bonferroni Adjustments	316

List of Tables

Table 2.1A.	Demographics of subjects (mean \pm SD)	116
Table 2.1B.	Demographics of subjects (mean ± SD)	116
Table 2.2.	Postural and acceleration differences during WBV at	
	Different doses in healthy men and women (mean \pm SD)	117
Table 2.3	Cerebral oxygenation and blood volume differences during	
	WBV at different doses in healthy men and women	
	(mean ± SD)	118
Table 2.4	Pearson correlations between angles measured during	
	sitting and acceleration at C6	119
Table 2.5	Pearson correlations between cerebral responses and	
	selected variables during sitting 'With' backrest	119
Table 2.6	Pearson correlations between maximal rhythmic handgrip	
	contractions and cerebral responses	120
Table 2.7	Pearson correlations between cerebral responses and	
	relative aerobic power	120
Table 3.1	Demographics of subjects (mean \pm SD)	156
Table 3.2	Erector spinae oxygenation and blood volume differences	
	during WBV at different doses in healthy men and women	
	(mean \pm SD)	157
Table 3.3	Pearson correlations between maximal rhythmic handgrip	
	contractions and erector spinae responses	158
Table 3.4	Spearman correlations between erector spinae responses	
	and adipose tissue thickness	158
Table 3.5	Pearson Correlations between erector spinae responses	
	and relative aerobic power	159
Table 4.1A	Demographics of subjects (mean \pm SD)	191
Table 4.1B	Demographics of subjects (mean \pm SD)	191
Table 4.2	Metabolic and respiratory responses during whole-body	
	vibration (mean \pm SD)	192

Table 4.3	Cardiovascular responses during whole-body vibration	
	(mean ± SD)	193
Table 4.4	Pearson correlations between metabolic responses and	
	maximal rhythmic handgrip contractions	194
Table 5.1	Demographics of subjects (mean \pm SD)	235
Table 5.2	Peak physiological, perceptual, and biomechanical	
	Responses during incremental pushing-pulling and arm	
	cranking in healthy men and women (mean \pm SD)	236
Table 5.3	Biceps brachii oxygenation and blood volume responses	
	during incremental pushing-pulling and arm cranking	
	in healthy men and women (mean \pm SD)	237
Table 5.4	Pearson correlations between Biceps brachii oxygenation	
	and blood volume responses among three protocols	238
Table 5.5	Spearman correlations between Biceps brachii oxygenation	
	and blood volume responses and adipose tissue thickness in	n
	healthy men and women	239
Table 6.1	Demographics of subjects (mean \pm SD)	279
Table 6.2	Erector spinae oxygenation and blood volume responses	
	during incremental pushing-pulling and arm cranking	
	in healthy men and women (mean \pm SD)	280
Table 6.3A	Pearson correlations between Erector spinae oxygenation	
	and blood volume responses among three protocols	281
Table 6.3B	Pearson correlations of erector spinae oxygenation	
	and blood volume responses among three protocols in	
	healthy men and women	282
Table 6.4	Spearman correlations between erector spinae oxygenation	
	and blood volume responses and adipose tissue	
	thickness in healthy men and women	283

List of Figures

Simultaneous cerebral and muscle oxygenation and	
blood volume trends in a typical subject	82
Cuff ischemia protocol for biceps brachii oxygenation	
changes	83
Maximal isometric extension for erector spinae muscle	
oxygenation changes	83
Use of Belardinelli equation for incremental arm cranking	
exercise	84
Vibration Simulator	121
Near Infrared Spectroscopy unit	122
Near Infrared Spectroscopy sensor placement on the head	123
Tensor bandages on the head	123
Subject sitting 'Without' backrest	124
Subject sitting 'Without' backrest combined with handgrip	
contractions	124
Subject setup during sitting 'With' backrest	125
Cerebral oxygenation trends in a typical male and female	126
Cerebral blood volume trends in a typical male and female	126
Interaction between gender and work – Mean cerebral	
oxygenation changes	127
Interaction between gender and work – Mean cerebral	
blood volume changes	127
Interaction between gender and dose – Mean cerebral	
oxygenation changes	128
Interaction between gender and dose – Mean cerebral	
blood volume changes	128
NIRS sensor placement on the low-back at Lumbar erector	
spinae	160
Tensor bandage around the low-back	160
	Simultaneous cerebral and muscle oxygenation and blood volume trends in a typical subject Cuff ischemia protocol for biceps brachii oxygenation changes Maximal isometric extension for erector spinae muscle oxygenation changes Use of Belardinelli equation for incremental arm cranking exercise Vibration Simulator Near Infrared Spectroscopy unit Near Infrared Spectroscopy sensor placement on the head Tensor bandages on the head Subject sitting 'Without' backrest Subject sitting 'Without' backrest Subject sitting 'Without' backrest Cerebral oxygenation trends in a typical male and female Cerebral blood volume trends in a typical male and female Interaction between gender and work – Mean cerebral blood volume changes Interaction between gender and dose – Mean cerebral blood volume changes Interaction between gender and dose – Mean cerebral blood volume changes Interaction between gender and dose – Mean cerebral blood volume changes Interaction between gender and dose – Mean cerebral blood volume changes Interaction between gender and dose – Mean cerebral blood volume changes NIRS sensor placement on the low-back at Lumbar erector spinae Tensor bandage around the low-back

Figure 3.2A	Lumbar erector spinae oxygenation trends in a typical male	
	and female	161
Figure 3.2B	Lumbar erector spinae oxygenation trends in a typical male	
	and female	161
Figure 3.3A	Interaction between gender and work - Mean Lumbar erector	٥r
	spinae oxygenation changes	162
Figure 3.3B	Interaction between gender and work - Mean Lumbar erector	or
	spinae blood volume changes	162
Figure 3.4A	Interaction between gender and dose - Mean Lumbar erector)r
	spinae oxygenation changes	163
Figure 3.4B	Interaction between gender and dose - Mean Lumbar erector	or i
	spinae blood volume changes	163
Figure 3.5A	Interaction between gender and backrest – Mean Lumbar	
	erector spinae oxygenation changes	164
Figure 3.5B	Interaction between gender and backrest – Mean Lumbar	
	erector spinae blood volume changes	164
Figure 4.1A	Subject setup for metabolic measurements	195
Figure 4.1B	Noninvasive cardiac output using CO2 rebreathing	
	Technique	195
Figure 4.2A	Interaction between gender and work – Absolute oxygen	
	uptake changes	196
Figure 4.2B	Interaction between gender and work – Relative oxygen	
	uptake changes	196
Figure 4.3	Interaction between gender and work – Oxygen pulse	
	changes	197
Figure 5.1	BTE Work simulator	240
Figure 5.2	NIRS sensor placement on the biceps brachii	241
Figure 5.3A	Subject setup for cuff ischemia	242
Figure 5.3B	Subject setup during Arm Cranking	242
Figure 5.4	Subject stance during Pushing-pulling	243
Figure 5.5A	Subject during Pulling phase	244

•

Figure 5.5B	Subject during Pushing phase	244
Figure 5.6A	Biceps brachii oxygenation trends in a typical female during	
	Cuff ischemia, incremental arm cranking, and incremental	
	Pushing-pulling	245
Figure 5.6B	Biceps brachii blood volume trends in a typical female during	l
	Cuff ischemia, incremental arm cranking, and incremental	
	Pushing-pulling	245
Figure 5.7A	Biceps brachii oxygenation trends in a typical male during	
	Cuff ischemia, incremental arm cranking, and incremental	
	Pushing-pulling	246
Figure 5.7B	Biceps brachii blood volume trends in a typical male during	
	Cuff ischemia, incremental arm cranking, and incremental	
	Pushing-pulling	246
Figure 5.8	Relationhsip between Biceps brachii deoxygenation and	
	absolute oxygen uptake in healthy men and women during	
	incremental arm cranking, and incremental pushing-pulling	247
Figure 5.9A	Correlation between adipose tissue thickness of Biceps	
	brachii and maximum oxygenation during recovery (pushing-	•
	pulling)	248
Figure 5.9B	Correlation between adipose tissue thickness of Biceps	
	brachii and maximum blood volume during recovery	
	(pushing-pulling)	248
Figure 6.1A	Subject setup for the Sorensen test	284
Figure 6.1B	Securing the subject on the plinth	284
Figure 6.2	Subject performing the maximal isometric extension test	285
Figure 6.3A	Lumbar erector spinae oxygenation trends in a typical female	e
	during Sorensen test, incremental arm cranking, and	
	incremental Pushing-pulling	286
Figure 6.3B	Lumbar erector spinae blood volume trends in a typical fema	le
	during Sorensen test, incremental arm cranking, and	
	incremental Pushing-pulling	286

Figure 6.4A	Lumbar erector spinae oxygenation trends in a typical male	
	during Sorensen test, incremental arm cranking, and	
	incremental Pushing-pulling	287
Figure 6.4B	Lumbar erector spinae blood volume trends in a typical male	•
	during Sorensen test, incremental arm cranking, and	
	incremental Pushing-pulling	287
Figure 6.5	Correlation between adipose tissue thickness of Lumbar	
	erector spinae and maximum oxygenation during	
	Sorensen test	288

٠

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

CHAPTER 1 Introduction

Injury Epidemiology:

Gatchel et al. (1995) reported that industrial back injuries are costing employers and insurers billions of dollars in lost time, compensation wages, litigation, and rehabilitation. According to the International Social Security Association, four to seven percent of all employees in the US, Canada, and some European countries are exposed to potential harmful whole-body vibration (Bovenzi 1996). In Great Britain, Palmer et al. (2000) reported that 7,200,000 men and 1,800,000 women at the workplace were exposed to whole-body vibration in a one week period. National health survey comparisons from the United States of America and Canada found that prevalence of low-back complaints in workers exposed to vibration was higher than that estimated in the male working population of USA (Guo et al. 1995), and also compared to controls not exposed to vibration (Liira et al. 1996). Aircrew and helicopter pilots (Simon-Arndt et al. 1997) and drivers of earth-moving vehicles (Miyashita et al. 1992), agricultural tractor drivers (Boshuizen et al. 1990, Bovenzi and Betta 1994), and buses and transit drivers (Bongers et al. 1988) are some of the occupational groups who are at a greater risk for low-back complaints than the control groups unexposed to whole-body vibration. However, the effects of whole-body vibration on the lower back and its relationship with low-back pain are not well understood.

Manual material handling is defined as the application of human effort (including lifting, lowering, pushing, pulling or carrying) for transporting or supporting a load such as an object, person or animal (Dickinson 1995). Injuries resulting from manual material handling are mostly due to overexertion when workers exceed their ability to perform the task (Genaidy et al. 1992). According to the Annual Cumulative Trauma Disorders (CTD) News Cost & Incident Survey, industrial workplaces in the United States of America account for more than 81% of CTD compensation costs and 55% of recordable CTD cases (CTD News 1998). This article also reported that, between the years 1995 and 1997, manual material handling cases cost employers about \$2.56 billion annually in direct compensation. In the definition of cumulative trauma disorders, *cumulative* indicates that these injuries develop gradually over periods of weeks, months, or even years as a result of repeated stresses on a particular body part; *trauma* signifies bodily injury from mechanical stresses; and the term *disorders* refers to physical ailments or abnormal conditions (Putz-Anderson 1990). It should be noted, however, that the concept of cumulative trauma to explain the growth with ill-defined soft tissue disorders is controversial.

In the province of Alberta, during 1995 and 1996, twenty nine-percent of total compensation claims were related to the lower back with lifting, pushing or pulling most frequently cited as the cause. Industries that showed the highest lost-claim rates were Agriculture and Forestry, Manufacturing and Processing, Construction and Transportation, and Utilities (Alberta Labor 1997). In the first quarter of 1998, more than 40% of the clients attending the Millard Rehabilitation Center at the Workers Compensation Board of Alberta were suffering from lower back injuries, followed by upper extremity claims representing 22% (Graham In the state of Washington, USA, from 1989 to 1996, workers in 1998). industries characterized by manual material handling were at the highest risk for back and shoulder disorders. Back disorders represented 55% of direct compensation costs and 12.8 million lost work days whereas upper extremity claims constituted 36% resulting in 12 million lost workdays. The percentage of female claimants increased over this period. According to the National Institute for Occupational Health and Safety (NIOSH), overexertion injury, defined as a temporary or permanent disruption of normal musculoskeletal function due to the demands of a job which exceed the worker's capacity, is the greatest concern of any industry (NIOSH 1991). In the United States of America, 60% of the lowback related patients claimed overexertion as the cause of low back pain (NIOSH 1981). Based on type of accidents, Statistics Canada (1994) showed that for the years 1991 to 1993 overexertion injuries represented 29% of all the injuries reported.

In a two-year prospective Canadian study on nurses at a tertiary care hospital, lifting (22.6%) and transferring of patients with assistance (23.3%) were the two most common risk factors for back injury (Yassi et al. 1995). McCoy et al. (1997) studied 437 patients with severe disabling low-back pain due to work-related injuries and concluded that lifting was the most common cause of injury (66%), followed by pushing and pulling (13%). Damkot et al. (1984) also indicated that pushing and pulling of loads were important risk factors for low-back pain. In a long-distance transport company, Bari-Gingras and Lortie (1995) studied the activities of handlers and reported that pulling (26%) and pushing (22.2%) were the most frequent types of efforts. Unlike lifting, pulling was performed in two-thirds of the handlings (62.3%) and pushing 44.2% of the time. Thus, these statistics show pulling and pushing related injuries are serious problems encountered in the workplace.

Whole-body Vibration:

Whole-body vibration is the vibration transmitted to the entire body through a supporting structure such as a vehicle seat in a car, tractor, ship, aircraft, and snowmobile or through lifts, escalators in buildings, and from the shop floor. Epidemiological investigations (Boushuizen et al. 1990, Bovenzi and Betta 1994, Burdorf and Sorock 1997, Frymoyer et al. 1980, Pope et al. 1998) and literature reviews (Bernard and Fine 1997, Bovenzi 1996, Bovenzi and Hulshof 1999, Dupuis and Zerlett 1986, Hulshof and van Zanten 1987, Lings and Leboeuf-Yde 2000, Pope 1989, Seidel and Heide 1986) have indicated that human exposure to whole-body vibration is an occupational risk factor for the development of low-back disorders.

Using magnetic resonance imaging, Luoma et al. (1998) studied the risk factors of lumbar disc degeneration with special emphasis on occupational load and back accidents. In this cross-sectional study of 53 machine drivers, 51 construction carpenters, and 60 municipal office workers, they identified posterior disc bulges in the carpenters and anterior disc bulges in the machine drivers. Their study concluded that disc degeneration was not related to body height,

overweight, smoking, or the frequency of physical exercise and cautioned the smaller sample size of this study might have affected such results. They attributed the signs of disc degeneration to a history of back accidents and car driving reported by the subjects.

Between 1991 and 1993, a case-control study by Simon-Arndt et al. (1997) on US navy pilots and aircrew members, showed that aircrew members have a higher risk of diagnosed back problems than pilots. This conclusion was contrary to their hypothesis which predicted higher back complaints in pilots. They suggested that this phenomenon may be due to frequent carrying of heavy equipment by aircrew than pilots or greater tendency of aircrew members reporting problems to a flight surgeon than pilots. However, back problems related to spinal curvature (classification based on International Classification of Diseases code: 737.1 - 737.99) were higher in pilots, suggesting that different factors of their work environment or their lifestyle might have contributed to these complaints.

Bovenzi and Betta (1994) investigated the occurrence of low-back pain in 1155 tractor drivers exposed to whole-body vibration and 220 office staff (control group). They used, both a questionnaire (to identify work- and subject-related risk factors) and vibration field measurement methods to quantify the exposure. They also calculated vibration dose for each tractor driver in terms of vibration magnitude and duration of exposure. Effect of postural stress was also assessed in terms of frequency and/or duration of awkward postures at work. Their logistic modeling showed that low-back disorders were significantly associated with both vibration dose and postural load independently. This study also showed that the duration of exposure to vibration was more highly related to low-back pain than was equivalent vibration magnitude. Even though vibration dose was measured and postural load was subjective, this is the only study so far which has looked at the effect of both postural load and vibration dose on the occurrence of back pain complaints.

Using a questionnaire approach, Johanning (1991) studied back pain complaints of subway train operators associated with whole-body vibration. The results indicated that operators had a higher prevalence of back problems than control group (switch-board operators), especially in the cervical and low-back region. Even though this study couldn't establish a dose-response relationship, the author recommended improved working conditions and a decrease in exposure of vibration to the operator. Boshuizen et al. (1990a) analyzed selfreported back pain in tractor drivers exposed to whole-body vibration. They observed a 10% higher prevalence of back pain in the tractor drivers compared to a control group of workers not exposed to vibration. They showed that back pain increased with the duration of exposure, but not with the estimated mean magnitude of vibration. In another study (Boshuizen et al. 1990b), the authors found that long-term absenteeism due to back disorders was high in the tractor drivers exposed to whole-body vibration compared to a control group.

Bongers et al. (1988) showed that crane operators with more than 5 years of exposure had almost three times the risk of occurrence of disability due to intervertebral disc disorders compared to a control group (floor operators and maintenance workers). Using incidence density ratio, they suggested that crane operators with long-term (15-20 years) exposure will have a greater overall risk for spinal disorders. They attributed these ill effects to the combination of exposure to whole-body vibration, awkward postures, and climatic conditions. They also cited an American study that investigated the long-term effects of whole-body vibration on interstate drivers and a control group of office workers and drivers with less than 5 years of professional driving experience. This American study also concluded the combined effect of posture, off-balanced movements due to fatigue, and whole-body vibration exposure resulted in spinal disorders.

Based on interviews and medical examinations, a cross-sectional study of earth-moving machine operators (with three years of exposure to whole-body vibration) was conducted by Dupuis and Zerlett (1987). The percentage of operators reporting spinal discomfort (68.7% lumbar column, 6.8% thoracic column, and 18.2% cervical column) was much higher than a non-exposed control group of the same age category exposed to same working conditions of

cases except vibration. A dose-response relationship was not clear in this study. However, the researchers identified a trend indicating that operators exposed to higher vibration loads had an increased prevalence of pathological lumbar spinal disorders.

In their explanation of pathogenesis of backaches in helicopter pilots, Delahaye et al. (1982) reasoned synergistic action of two main factors for their back pain: poor flying posture due to the constant and coordinated use of all four limbs during flying; and vibration of the helicopter itself. Frymoyer et al. (1980) studied 3920 patients during a three year interval in which 11% of males and 9.5% females reported an episode of low-back pain. Their low-back complaints were significantly (p<0.001) related to truck driving, lifting, carrying, pulling, pushing and twisting and also non-driving exposure to vibration. However, this study couldn't quantify risk factors in the causation of low-back pain.

Establishing dose-response relationships in humans is important, as driving has become an integral part of life for everybody including those specific occupational groups mentioned. However, most of the above studies couldn't establish a quantitative relationship between exposure (dose) and effect (response). One of the reasons may be due to concomitant exposure of individuals to whole-body vibration, prolonged sitting, awkward postures adopted, frequent twisting of the spine during work, and frequent heavy manual material handling associated with their job. From these points, it is understood that individuals engaged in the occupational activities are prone to more than a single risk factor associated with back complaints.

Influential Factors:

Many factors influence human response to whole-body vibration - the source (vibrating base, vehicle dynamics), interface such as seat cushion and finally the host (humans). The host's physical characteristics (e.g. anthropometry), psychological differences (e.g. experience), and physiological differences (e.g. fitness) play a vital role in the performance and response during exposure to vibration (Boff and Lincoln 1988). The effect of whole-body vibration

on human exposure depends on the direction of exposure (linear motion in x, y, and z axis and also angular motion), duration of exposure, its intensity (in terms of frequency [cycles/second or Hz], and acceleration [expressed in gravitational units 'g': meters/sec² or in decibels]).

Low-frequency vibration below 20 Hz and high amplitude vibration energy is more prevalent and harmful especially to the low-back region of tractor drivers, military tank operators, aviation pilots, machinery personal and passengers in cars (Cornelius et al. 1994, Guinard 1985, Ramos et al. 1996, Wikstrom et al. 1994, Zagorski et al. 1976). The most sensitive frequency range in the vertical 'z' direction for the whole-body is from 4 to 8 Hz with the transfer of maximum mechanical energy to the spine resulting at about 5 Hz (Griffin 1990, Panjabi et al. 1986). The first resonant frequency of the seated human was in the range from 4.5 to 5.5 Hz due to compression of the buttocks tissue and the interaction of the vertical response with rotational motion of the pelvis (Pope et al. 1989b and 1990). At this frequency range, the lumbar spine stretches the most resulting in the internal amplification thus increasing the effects of vibration (Wasserman et al. 1997). Unfortunately, it was also identified that most of the vehicles (trucks, buses, tractors, etc.) produce frequencies between 3 to 6 Hz (Wilder et al.1982).

According to International Labor Office Codes of Practices (ILO 1984) on *Protection of workers against noise and vibration in the working environment*, low-frequency whole-body vibration (between 1 and 20 Hz) causes a wide range of pathological phenomena including lumbago, sciatica, neck ache, disc herniation and other spinal disorders. These manifestations may occur after a period of exposure in persons who are initially in good health. The common vibration exposure at the workplace is in the 1- to 10-Hz range except in aeronautics, which lies between 2 and 20 Hz (Delahaye et al. 1982, Kjellberg and Wikstrom 1987). Frequency of above 20 or 30 Hz applied to the whole-body may not significantly affect morbidity in the workplace. These higher frequencies do not transmit substantial amounts of vibration to humans and are effectively attenuated by human body surface, by seating and floor material (Guinard 1985).

Vibration at Resonance:

The frequency of greatest amplification, also known as resonance frequency, depends on the individuals exposed, different parts of human body, and changes with the body posture we adopt. These resonance frequencies transfer maximum mechanical energy, resulting in an optimum tuning between the vibration source and the host, together with an amplification of 1.5 to 3 times the exposed vibration. The natural frequency is determined by the acceleration transmissibility, in which the output acceleration (e.g. at the host) in the simple structure is compared with the input acceleration (e.g. at the source of vibration), and at resonance, the ratio of input to output exceeds unity (Pope et al. 1990).

Griffin (1975) and Pope et al. (1987 and 1990) established that resonant frequencies for the seated operator occur between 4 and 6Hz due to vertical vibration and also between 10 and 14 Hz, indicating a bending vibration of the upper torso with respect to lumbar spine, resulting in a back slap against a back rest. According to Dupuis (1989), lumbar spine stretches the most in the frequency between 2 and 6 Hz with a maximum at 4 Hz. In vivo measurements in the spinal column showed that resonance in the lumbar region is at 4.5 Hz (Panjabi et al. 1986). Hence, the *recommended limits* in ISO 2631/1 for exposure to WBV take into account the direction, magnitude, duration, and frequency (to avoid resonance) of the harmful vibration (Thalheimer 1996).

International Organization for Standardization - 2631 document (ISO 2631/1-Part 1, 1985 and 1997):

International Organization for Standardization (ISO) document 2631-Part 1, is applied to human subjects exposed to whole-body vibration transmitted through a supporting surface such as the feet of a standing worker in the vicinity of machinery, buttocks of a seated person in a car or supporting area of a reclining sick or injured person in an ambulance (ISO 2631/1 1985). In 1995, the American Conference of Governmental Industrial Hygienists (ACGIH) published an official document based on ISO 2631/1, incorporating whole-body vibration exposure into their annual recommended guidelines: *Threshold Limit Values and* Biological Exposure Indices. These ISO limits are also adopted all over the world with some minor changes in Europe. Other ISO documents are also in existence for human exposure to vibration and shock in buildings, and crew exposure to vibration on board sea-going ships.

This ISO 2631/1-part 1 document, contains numerical values for limits of exposure to vibration, specified according to the direction of exposure (linear movements in x, y, z coordinates), intensity (frequency of 0-80 Hz) and duration of exposure (1min-24 hours). These limits are set to define three boundaries known as the *Reduced Comfort boundary* (for preserving comfort of passengers), the *Fatigue-decreased Proficiency boundary* (to maximize performance of drivers or machine operators who are exposed to vibration), and finally the *Exposure limit* (for safety and health). These limits are based on the data available from both practical experience and laboratory experimentation on humans. Some of the limitations of this document include:

- 1. Insufficient information on the inter- and intra-subject variability of the populations exposed (Griffin, 1990).
- 2. No information on the human physiological responses for various values of vibration dose.
- 3. Development of these ISO limiting curves on the basis of research conducted primarily on men (Bovenzi 1996, Seidel 1993).
- 4. Limited data pertaining to the extent of vibration transmissibility through the body (Lundstrom et al. 1998).

Vibration from Seat to Head:

According to Boff and Lincoln (1988), transmission of seat vibration to the head is affected by posture, muscle tension, body size, head position, sustained acceleration, and attachment of extra masses such as helmets to the head. Minor variations in the orientation of the lower back and the angle of the head can cause substantial changes in the vibration transmitted through the spine to the head (Griffin 1990). In Australian farmers, Scutter et al. (1997) showed a prevalence of headache (79.2%) and neck pain (77.7%) during exposure to

driving tractors. They attributed these symptoms to combination of whole-body vibration and frequent rotated neck postures during farming. Their study did not show any time-dependent exposure relationship to the frequency of neck pain or headache. They hypothesized that vibration induced tonic contractions in the jaw muscles and their relationship to the temperomandibular disorders may lead to the development of symptoms of face and head pain in farmers.

Wilder et al. (1982) showed that transmissibility of vibration from the buttocks to head was amplified 1.79 times, i.e., 1 g of acceleration at the buttocks will result in 1.79 g at the head. Griffin et al. (1982) showed that vertical transmission to the head was slightly greater in women than in men at frequencies > 5 Hz and slightly greater in men than in women at frequencies < 5 Hz. Boff and Lincoln (1988) showed that vibration transmitted through the body to the head and eyes reduced visual acuity by causing images to move on the retina.

While explaining vibration interference with the performance of a continuous manual control task, McLeod and Griffin (1994) suggested one of the mechanisms that may contribute to this cause is the *central effect of cognitive process*. Arousal, motivation, anxiety, attention were some of the examples cited in their study. Their group also showed that performance decrements during vibration (between 1.6 and 10 Hz) were mainly due to visual impairments arising from relative translational movement between the subject's eyes and display.

Role of Vibration Dose on Cardiorespiratory Responses:

Vibration exposure in the region of 2 to 20 Hz at intensities of 0.1 to $0.5g_{ms}$ elicits cardiopulmonary responses similar to moderate exercise, with variable increases in heart beat, respiration rates, cardiac output, pulmonary ventilation, and oxygen uptake (Griffin 1990, Guinard 1985). Hoover and Ashe (1962) studied respiratory responses in five healthy volunteers exposed to whole-body vibration. Subjects, in a sitting posture, experienced vibration frequencies = 2, 4, 6, 8, 11 Hz, at acceleration of 0.03 to 2.88 g, and amplitude = 0.062 and 0.125 inches. Experimental protocol consisted of 20 min control period, 20 min

exposure period, and finally 15 min recovery. Their results showed changes in the respiratory rate and/or tidal volume. At lower amplitude, respiratory rate was affected and at higher amplitude tidal volume was affected and these changes in tidal volume were more than adequate to provide ventilation. They suggested that the response at lower amplitude may be due to correction of ventilation to decrease tidal volume. Hoover and Ashe (1962) concluded that results obtained in this study may be similar to those seen in hyperventilation. However, they did not indicate the gender of these volunteers participated in the study. The same group, in another study (Gaeuman et al. 1962) suggested that, at vibration levels of 6-15 Hz (near the primary and secondary body resonant points), the increase in oxygen consumption was nearly linear with increasing frequency of vibration, indicating such an increase in metabolic load may fatigue the subjects over extended periods of exposure to such vibration frequencies.

Duffner et al. (1962) showed no interaction between the subjects, frequencies and acceleration intensities during exposure to whole-body vibration. Ten healthy males were subjected to a 4 min control, exposure (frequency = 2 - 7 Hz and acceleration 0.15 g and 0.35 g), and recovery phases respectively. Their results showed the greatest increase in alveolar ventilation during the first minute of vibration and the increase observed was greater at 0.35g than at the lower level. Ventilation values showed a decline toward 'control' levels more slowly at frequencies of 4 - 5 Hz, with the most prolonged hyperventilation occurring at 4 and 5 Hz (at both intensities). After the exposure period, alveolar ventilation immediately returned to control period or near control period values, with a further decline during the 2^{nd} , 3^{rd} , and 4^{th} min of the recovery period.

As large vertical displacements were needed to obtain forces of 0.15g and 0.35g at these low frequencies, Duffner et al. (1962) showed a greater increase in oxygen consumption at lower rather than resonant frequencies (4-5 Hz for a seated human body), which may be a reflection of the physical effort essential to maintain the state of equilibrium and posture of the unrestrained body. However, the increase in oxygen consumption (in this study) did not correlate with the degree of hyperventilation observed at different frequencies of vibration.

According to the authors, the drop in resting mid position (for end-expiratory level) during vibration, probably reflected both the increased tidal volume associated with hyperventilation, and increased abdominal muscular tension needed to maintain posture. The return of effective ventilation toward control period values during the vibration period after an initial marked elevation may be the result of normal physiological control. They concluded that a decrease in ventilation observed during vibration in this study proves the concept of a physiological adaptation to vibration.

Hood et al. (1966) studied four subjects sitting with helmets in a semi supine position on an adjustable table with vibration transmitted in the direction of x-axis. Experimental protocol consisted of a control period of 3 min, exposure period of 7 min (at frequencies 2, 4, 6, 8, 10, 12 Hz at accelerations 0.6g and 1.2g), and finally a recovery period of 15 min. They showed changes in mean arterial blood pressure, heart rate, cardiac output, and oxygen consumption, at a higher level of acceleration and in the frequency range of 8 - 10 Hz. These maximal changes were accompanied by increased arterio-venous oxygen difference, decreased peripheral vascular resistance index, and a minor increase in stroke index, thus indicating a similar physiological response pattern of an 'exercise'.

At the lower intensity of vibration (0.6g) and at frequencies on either side of the 8-10 Hz range, the small increase in cardiac index without increased heart rate or metabolic rate indicates the presence of over perfusion relative to the control period (Hood et al. 1966). This suggests that at lower intensities of vibration a venous pump mechanism is usually activated. This study also indicated increase in ventilation during vibration. Their findings also supported the hypothesis that the physiological changes observed are due to maximal stretching of muscle spindles and reflex muscular contraction under optimal vibratory conditions. They also indicated that vigorous mechanical movements of the whole body, particularly movement of the extremities, act as a mild circulatory stress suggesting that patients with severe cardiovascular disease should avoid rough means of transportation.
Benett et al. (1977) showed that, while performing a motor task (tracking) during exposure to vibration, the level of performance may be effected by the physiological cost (heart rate, and oxygen consumption). Based on a locus of control scale, twelve males were divided into internal and external groups. This scale measures a person's general belief about his/her ability to influence events. Subjects, while sitting on a vibration chair, were exposed to both sinusoidal (frequency of 6 Hz), and random vibration (0 - 6 Hz with 125 msec shocks superimposed at regular intervals) at accelerations 0.21, 0.28, and 0.35g r.m.s. Exposure period was for 30 min. Compared to external subjects, internal subjects experienced smaller performance decrements under all levels of vibration, with greater responses resulting during higher acceleration levels and random vibration. Oxygen uptake measurements showed small increments at vibration onset, however, no differences were related to locus of control. Heart rate also increased in all the subjects at the vibration onset. However, a highly significant difference in heart rate was shown in internal than external subjects, with more predominant values during random and high acceleration vibration levels.

In the real world, vibration conditions are usually of a random rather than a sinusoidal nature, and therefore individuals may show different physiological and/or psychological responses. However, Cole and Withey (1977) found that, exposure to different wave forms didn't result in significant physiological differences in humans. They investigated the human responses in 12 young males while exposed to both sinusoidal and random vibration of 6 Hz in a seated upright posture on a hydraulic vibrator. Experimental protocol consisted of 10 min control period, 30 min exposure period (with accelerations at 0.21, 0.28, and 0.35g r.m.s), and finally 10 min recovery period. No statistical significant differences were observed in oxygen consumption, pulmonary ventilation, tidal volume, and heart rate values during sinusoidal and random vibration at all the acceleration levels. These investigators concluded that only sinusoidal wave input studies are valuable in establishing exposure limits for human vibration.

Magnusson et al. (1987) studied the relationship of the oxygen consumption and posture during exposure to whole-body vibration. Five male subjects, sat on a flat, rigid, wooden plate bolted to a vibration machine without a back rest, for a period of 10 min while exposed to a constant frequency of 5 Hz and acceleration 2.0 m/sec2. Experimental protocol started by monitoring oxygen uptake during sitting (without vibration) in a relaxed symmetric posture, sitting erect without vibration (erect posture - adopting pelvis, thoracic spine and occiput in a plane perpendicular to the seat), axial trunk rotation to the left without vibration, and vibration plus axial rotation respectively. Their results showed an increase in oxygen uptake during the erect posture compared to the relaxed posture with the highest increase (32.1%) observed during twisting and vibration exposure. They concluded that occupations such as farming, forestry, driving fork-lift trucks involve these coupling motions, thus resulting in higher oxygen uptake values.

Pope et al. (1990b) also evaluated the effect of posture, intermittent (at 5Hz) and continuous (0, 2.5, 5, and 8 Hz) sinusoidal vibration on oxygen uptake in five healthy men and women. For each combination of posture and frequency, subjects were exposed for a period of 30 min. They also monitored subjects' discomfort using the Visual Analog Scale. Intermittent vibration was given every minute for a period of ten seconds. Both relaxed and erect postures were adopted during the experimental protocol. Their results showed increase in oxygen uptake during vibration when compared to sitting at rest without vibration. However, they did not observe increase in oxygen uptake with the increase in the frequency of vibration. Their results did not identify gender differences clearly. In intermittent vibration, they did not find any differences in the oxygen uptake values during erect and relaxed postures. They indicated a positive linear relationship between frequency and discomfort ratings. Based on the results, the authors concluded that changes in oxygen uptake during vibration resembles light machining work and attributed such an increase in oxygen uptake to postural maintenance in erect position.

Further, Pope et al. (1990b) suggested that monitoring physiological variables such as oxygen uptake during whole-body vibration is useful and relevant to the assessment of the energy demands of populations exposed in the industries. The review of Seidel and Heide (1986) stressed that whole-body

vibration cannot be ruled out as an additional risk factor for ischemic heart disease. However, the literature review of Wikstrom et al. (1994) suggested that exposure to vibration doesn't significantly increase the risk of diseases in the cardiovascular system. With such mixed opinions of various authors, it is essential to evaluate both the central physiological responses and peripheral limitations of human system during exposure to whole-body vibration.

Role of Postures adopted during Whole-body Vibration:

Posture also has a great influence on the amount of vibration transmitted to a person either in a standing, sitting or reclining position. In a sitting position, the effect of postural change increases with increasing frequency (Griffin 1990). Effects of sitting posture are widely studied in office ergonomics and human vibration. It is the primary concern of most vibration-induced complaints, distraction and injuries to health (Dupuis and Zerlett 1986). Sitting is defined as a body position in which the weight of the whole-body is transferred to a supporting area such as a seat or chair mainly by the ischial tuberosities of the pelvis and their supporting tissues (Chaffin and Andersson 1991). Some percent of weight is also transferred to the floor, backrest and armrests based on the posture adopted and type of seat.

With a vibration exposure in the sitting position, the load applied along the moment arm (created by the anterior offset of the ischial tuberosities) may be transmitted by the spine (Wilder 1993). Such transmission induces an additional rocking motion, thus amplifying the vibration motion transmitted to the spine. Since the seated human has a resonance frequency close to those frequencies encountered in an occupational environment, the possibility of failure of the spine mechanism vibrating at this resonance frequency is serious (Wilder 1993). Wilder et al. (1988) also reported that in vivo, in vitro, and modeling studies have shown that pressure within the intervertebral disc increases during compression, flexion, and also during sitting posture. While explaining the vibration induced low-back fatigue, Bongers (1990) in her doctoral dissertation cited a study that

showed prolonged sitting in a constrained posture decreases the nutrient supply of the disc and may disturb the balance of metabolites.

During driving of tractors or earth-moving vehicles, twisting of trunk combined with whole-body vibration may result in excessive compressive load and shear stress on soft and bone tissues of the spine (Bovenizi and Betta 1994). While explaining the pathogenesis of backache, Delahaye et al. (1982) reported that theoretically the thoracic and lumbar spine should be firmly held against the seat back by straps of restraint harness. However, during flying, pilots need continuous visibility which results in pilots not tightening the suspension straps and supporting themselves by leaning forward on the straps. Thus, the back is separated from the seat and cervical spine is hyper-extended. and the resulting hyper-extension becomes even worst due to the placement of instrument panels on the top, or if the pilot is shorter. They also reported that unchanging and asymmetrical posture leads to involuntary muscular contractions in some parts of the limb, and as the spine is laterally twisted this posture increases the lumbar spine's sensitivity to the vibrations of the aircraft. This phenomenon is aggravated in landing, take-off, hovering flight, and at lowspeeds.

Effect of combined stresses during Exposure to Whole-body Vibration:

Previous studies have suggested that most of the occupants, especially drivers and pilots, often have to spend their job in prolonged sitting and manually operating controls (Boff and Lincoln 1988, Delahaye et al. 1982, Griffin 1990). The controls generally used in vehicles are steering wheels, push-buttons, hand-operated levers, rotating knobs, joysticks, foot pedals, etc. (Boff and Lincoln 1988). For example, a pilot uses the right arm to operate the cyclic pitch control resulting in movement in translation; uses the left arm controlling the lift which operates the collective pitch lever; and uses the feet to operate the rudder bars, thus turning the helicopter and allowing the flight heading to be selected (Delahaye et al. 1982).

Operating controls during prolonged sitting, in addition to whole-body vibration can also be taxing on the forearm muscles. Research on hand-control interface or static work such as usage of isometric or isotonic controls during flying, and while driving vehicles, such as tractors, fork-lifts and snow-removal vehicles, is not reported in the literature so far. Multi-axis moving of controls (e.g. joy stick controls) can also be one of the activities of a driver to accomplish certain tasks during his/her job. The wrist movements of flexion, extension, deviation (ulnar/radial) depend on the situation desired, and the holding time may last from couple of seconds to minutes and may be hours. Maintaining a posture during this static activity for a prolonged period of time also depends on the level, duration of effort, and variation of the load, task performed which in turn increase the risk of acute discomfort, thus resulting in cumulative trauma disorders (Miedema et al. 1997).

Griffin and coworkers (Griffin 1990, McLeod and Griffin 1994) pointed out that transmission of vibration from the seat to the hands could impede performance during manual control. These investigators, as well as Boff and Lincoln (1988) published extensive reports on human performance during tracking targets (with supported and unsupported arms), keyboard digit punching, navigational plotting, data entry, etc., and showed that in the frequency ranges of 3 - 8 Hz, manual tracking is most sensitive to disruptions during whole-body vibration. They also showed that force-operated, isometric controls may be more vulnerable to frequencies > 5 Hz than conventional springcentered controls due to greater amounts of vibration-induced control activity, and at 4 Hz of vibration rotary controls behave the same way as that of joysticks. However, all these studies investigated psychological responses and did not consider the effects of force of exertion or effect of holding period of these controls on the wrist or forearm muscles.

Role of Combined Stressors on Physiological Responses:

Grether et al. (1971 and 1972) studied the effect of heat (120F), noise (105dB) and vibration (5Hz at peak 0.5g) stressors, individually and/or in

combination, on psychological and physiological responses. They tested ten male military personnel while sitting in a temporary heat enclosure constructed around the subject's seat on a mechanical vibration table. Noise, voice communication inputs, displays and controls were also provided, thus reflecting a real world situation. The physiological measures monitored were: body temperature (17 locations on the skin, rectal), heart rate, weight loss, and biochemical urine analysis, and performance measures monitored were: tracking, choice reaction time, voice communication, mental arithmetic, visual acuity, and subjective severity ratings on a 7 point scale. They also measured transmissibility of vibration by placing an accelerometer at the right shoulder. Except mental arithmetic tests and subjective ratings, their results showed lower values in physiological and performance measurements during exposure to combined stressors. They concluded that additive or synergistic interactions among the three stressors are absent. However, subjective stress severity ratings increased with the number of stressors in the combination.

Manninen (1984) studied the effect of noise, whole-body vibration (sinusoidal and random) and dynamic muscular work (2 W, 4 W, 8 W) on human performance. The author attempted to establish a relationship between the changes in the heart rate and the temporary changes in the hearing threshold. Ninety male student volunteers, sitting in a vibration chair, were exposed to combined stresses inside a special chamber controlled with noise (no noise and a stable, broad band noise of 90 dbA), vibration (sinusoidal frequency of 5 Hz and random frequency of 2.8-11.2 Hz), and dynamic task arrangements. The experimental protocol consisted of a control period of 30 min, exposure of 60 min (three consecutive exposure periods of 16 min, each followed by a 4 min postexposure interval), and a 15 min recovery period. All the subjects were exposed to vibration alone, noise alone or a combination of the two during dynamic work. The results showed increase in heart rate values during all exposures and in all exposure combinations than during control period values, suggesting the change in heart rate was not only accomplished with the dynamic work but also with the quality of noise and vibration exposure. This study was an important contribution to the existing whole-body vibration literature because these combined stressors represent traffic and practical work situations.

Manninen (1985) showed that psychic load caused by 'competition' increased the temporary hearing threshold, the heat rate, and systolic blood pressure when subjects were simultaneously exposed to noise or a combination of noise and vibration. The author classified 108 male students into competitors and non-competitors and exposed them to a combination of noise and vibration at dry bulb temperatures of 20°C and 30°C while working on a choice reaction apparatus. Subjects were seated in a chamber and exposed to the same experimental protocol as mentioned in Manninen (1984). The author monitored hearing threshold, heart rate, R wave amplitude, systolic and diastolic blood pressure, pulse pressure, and reaction time. Relatively high correlation coefficients between the hearing threshold values at 4 kHz and heart rate values and between the hearing threshold values at 6 kHz and the systolic and diastolic blood pressure values emphasized that hypertension in the systemic circulation contributes to the changes in the cranial circulation and further in the metabolism of the inner ear. A combination of sinusoidal vibration and noise reduced the R wave amplitude values than random vibration and noise exposures which showed an increase at 20°C.

Thus, contrary to the study conducted by Cole et al. (1977), Manninen (1984) showed significant differences in exposure to sinusoidal and random vibration. The author also stressed that as the control chamber reflects the driver's working environment, in the future research, attention should be given to various design factors such as: exposure combinations; prolonged exposure periods; and investigating how the reactions of the organism of young, previously non-exposed persons differ from those of people who have worked as drivers for some time.

Role of Gender during Whole-body Vibration:

Besides vibration dose variability (intensity, direction, duration and input position) in general, responses and performance of the inter- and intra-subjects exposed, is important to understand and prevent any harmful effects on humans.

Based on psychophysical experiments, it was reported that women are relatively more sensitive than men to frequencies of vertical seat vibration in the range 3.15 - 5Hz, but no significant difference was found between standing males and females in the range of 2.4 - 60 Hz (Griffin 1990).

In a recent investigation, Lundstrom et al. (1998) stressed the need for differential guidelines for men and women. They also argued that ISO 2631 document gives a poor description about the amount of vibration actually being transmitted to the body, as it gives only acceleration magnitude on the vibrating surface. The authors suggested a new concept i.e., employing the amount of vibration energy either absorbed or exchanged between the source and body, thus giving a better measure of physical stress imposed on the body. They evaluated absorbed power (also known as transmitted power to any structure, which is a sum of absorbed part of the power and the elastic power of any structure) on 15 males and 15 females during exposure to vertical whole-body vibration in a sitting posture. Lundstrom et al. (1998) showed the maximal absorption of vibration energy at lower frequencies for women than men. They hypothesized that these results may be due to differences in the structure of their body and/or due to a higher fat-mass/muscle-mass ratio of women. This higher ratio resulted in a decrease in the stiffness-to-mass ratio which eventually lowers the resonance frequency, thus, frequency for maximal power absorption. They also suggested in general, a higher fat content in women gives more damping. and thus, more absorption of power of vibration input.

Surprisingly, there are no controlled studies so far which investigated both central and peripheral physiological differences simultaneously in men and women to whole-body vibration. Also, as Bovenzi (1996) pointed out most of the epidemiological studies and conclusions are limited to only men, thus requiring a comprehensive understanding of gender responses and their differences to whole-body vibration. The author reported some studies that looked into the disorders of the female reproductive organs and adverse effects on pregnancy, however, the reasons for such an increase was were not given in the paper. The author also reported on an epidemiological survey study of coal miners that was conducted in Poland. According to this survey, women reported a more elevated occurrence of low-back pain than men. Even though this Polish study did not include a control group nor did it control for other confounding factors, it gave a new direction to whole-body vibration research due to the rapid rise of women in the workforce.

Bovenzi (1996) and Seidel (1993) also criticized the widely applied ISO 2631 document which doesn't take women in to consideration for the exposure limits. Review of Seidel and Heide (1986), reported a study which examined the effects of whole-body vibration on the blood supply of the organs of the minor pelvis in women. The results indicated that whole-body vibration exposure induced a distinct increase of blood volume during the phases of ovulation and menstruation. From this sub-section, it is evident that studies on women, exposed to whole-body vibration and related low-back studies are lacking. Research comparing the physiological responses of women and men are vital for developing gender specific standards in order to protect the general health of the workforce exposed to whole-body vibration.

Low-Back Muscle Fatigue during Exposure to Whole-body Vibration:

Compared to a rested or healthy worker, a worker fatigued from lack of sufficient rest or one who is recovering from an illness may be at a greater risk of developing a cumulative trauma disorder (Putz-Anderson 1990). While oxygen consumption and heart rate are used to assess whole-body work, these measurements are insufficient to monitor the localized load on either specific muscles or muscle groups. Muscle fatigue is defined as either "inability to sustain a specified force and power output during exercise" or a transient loss of work capacity due to the preceding work irrespective of whether or not the current performance is effected (Massen and Schneider 1997). Electromyography (EMG) is one of the techniques by which we can evaluate the localized stresses in terms of the muscular electrical activity either in a single muscle or group of muscles. For the last two decades, studies of physiological responses of the spinal musculature under whole-body vibration, in particular EMG studies, have been widely reported (Andersson et al. 1977, Hansson et al. 1981, Hosea et al. 1986, Seidel et al. 1986, Seroussi et al. 1989, Wilder et al. 1982, Zimmerman et al. 1993).

Wilder et al. (1982) evaluated the stiffness, impedance, and resonant characteristics of 30 males and 15 females. EMG measurements were taken in subjects with neutral, neutral and valsalva position, forward flexion (5°), extension (5°), left and right lateral bend (5°), maximal left/right axial rotation, from both the right erector spinae and external oblique muscle groups at the Lumbar 3 level. They found that greatest transmissibility of vibratory input occurs at 5 Hz. During an exposure period of 30 min (1-20 Hz frequency sweep), subjects (without a back support throughout this study), showed a frequency shift of raw EMG signals from higher to lower values indicating a fatigue of both muscles under vibration. They did not find any statistically significant changes in the EMG activity due to postural variations. However, Hosea et al. (1986) evaluated the responses of myoelectrical activities of 12 paraspinal muscles during a 3.5 h period of car driving on a road. They did not find any EMG evidence of fatigue (measured by decrease in mean power frequency and increase in amplitude for equal function) to the duration of exposure. No vibration magnitude and frequency values were reported.

Seidel et al. (1986) estimated the strain in the lumbar spine during exposure to sinusoidal whole-body vibration (1-15 Hz in steps of 0.5 Hz and at accelerations 1.5 and 3.0 m.sec⁻² r.m.s). The exposure duration was 30 sec (1-8 Hz) or 15 sec (9-15 Hz). Accelerometers were placed on the head (bite-bar in the mouth), shoulder, at 5th thoracic, and on the seat, respectively. Surface EMG recordings were registered from the back muscles (at rectus abdominus and at 12th thoracic/1st lumbar region). They estimated highest compressive forces at 7.5, 8 and 4.5 Hz and suggested the possibility of fatigue failures at the end plates of lumbar vertebrae after intense long-term exposure to whole-body vibration.

Hansson et al. (1991) studied the EMG activity of erector spinae muscles in six males during two types of sitting (forward bent position of 20° and carrying

an extra weight of 4 kg on the front of chest), with and without whole-body vibration (of 5 Hz and 0.2 g acceleration). During 5 min of testing, they showed that vibration increases both the speed and the magnitude of the development of muscle fatigue. The authors hypothesized that the onset of muscle fatigue may be due to the constriction of the arterial supply to the back muscles.

Zimmerman et al. (1993) measured the effects of three different unsupported seated postures (neutral upright, forward lean, and posterior lean) on the EMG responses of the erector spinae muscle in 11 healthy men. Data were collected 30 sec pre- and post-vibration and during 2 min of vertical vibration (at 4.5 Hz and 6.12 m.sec⁻² ms). The EMG magnitude was greater for the anterior (highest) and neural postures than for the posterior lean posture, indicating greater cyclic compression of the spine secondary to increased muscle activity results during anterior lean posture. They hypothesized that this compression may result in decreased nutrient diffusion, thus leading to earlier onset of erector spinae muscle fatigue during prolonged periods in this position.

Wilder et al. (1994) studied the effect of posture (leaning forward; seated upright; and seated back against the back rest), and seat suspension (gas spring suspension and standard spring seat) design on discomfort and back muscle fatigue during simulated truck driving. Six males were vibrated for a period of 10 min (1-20 Hz), during which magnitude and phase of acceleration transfer functions both from the vibrating base to the seat pan and from the seat pan to the bite bar were calculated. They used visual analog pain scale for subjective estimation of discomfort. Muscle fatigue was monitored using surface EMG, and is defined in terms of shift of median frequency to lower value. EMG results showed that upright sitting causes more fatigue than during any other posture. Seat suspension did not effect the EMG responses. They suggested that these results may be reflective of testing for only short-duration.

Thus, above electromyographical studies demonstrated that exposure to whole-body vibration in the seated posture accelerated the occurrence of back muscle fatigue, possibly due to a reduced blood flow to the muscles.

Role of Physical Fitness related to Low-back Intensive Tasks:

It has been often mentioned in the vibration-related literature, that physical fitness plays an important role in the effects of vibration on humans exposed to vibration (Boff and Lincoln 1988, Griffin 1990). However, there are no established studies so far which have examined this phenomenon. Most of the above studies showed that monitoring the cardiorespiratory system provide useful information on the acute effects of whole-body vibration exposure. To this effect, several manual material handling studies reported conflicting evidence of the role of aerobic fitness in low-back complaints. Investigations of the correlation of workers' cardiovascular capacity to work-related musculoskeletal injuries are challenging but potentially important in understanding injury risk.

Scientific research in determining the effects of aerobic capacity and cardiovascular risk factors on injury occurrence in manual material handling tasks has been an on going endeavor for years (Battié 1989, Cady et al. 1979, Craig et al. 1998, Kujala et al. 1996). In an epidemiological study of industrial back injury claims conducted at an air-craft industry, Battie et al. (1989) showed that aerobic capacity is not a risk factor for the back pain complaints. When controlling for age and sex, the maximal aerobic capacity (VO_{2max}), the best index of cardiovascular fitness, was not a predictor for future back pain complaints. However, for jobs of high fitness requirements such as fire fighters, Cady et al. (1979) showed a strong correlation between the physical fitness and back injury. They evaluated muscular and aerobic fitness and the subsequent occurrence of back injuries. Their results showed that firefighters with high physical fitness had both fewer and less costly back injury claims. But their conclusions were based on the sum of strength and aerobic fitness measurements which makes it difficult to attribute any specific measure such as aerobic fitness as the sole predictor for back pain complaints.

In a five-year Finnish prospective study, Kujala et al. (1996) showed that baseline aerobic power was not predictive of subsequent back pain. However, heavy occupational musculoskeletal loading, defined as work involving daily heavy lifting loads more than 35 kg and more frequent bending and twisting, and

physically heavy manual work predicted future back pain. During a highfrequency manual material handling tasks (10-20 lifts/lowers per minute), Craig et al. (1998) analyzed the correlation between injury occurrences, absolute and relative maximal aerobic power, and body composition. Their results indicated no significant relationship between injury occurrence and absolute maximal aerobic power or percentage of body fat. However, relative maximal aerobic power values suggested significant relationship with occurrence of injury. Since a higher percentage of the body part affected in this study was the back, the investigators hypothesized that a higher aerobic capacity was associated with a lower injury severity. Their results were in contrast to the conclusions of Battie et al. (1989) in which a wide variety of work tasks were evaluated. Craig et al. (1998) suggested that this discrepancy might have been due to the high frequency of lifting. Also, it is difficult to compare these findings with those of Kujala et al. (1996)'s because the frequency of lifts per day was not quantified in their study.

From these findings we can say that highly strenuous work such as reported in Cady et al. (1979) and Craig et al. (1998), may elicit different cardiovascular responses compared to either light or medium work. Also, the aerobic protocols of Battie et al. (1989), Cady et al. (1979), Craig et al. (1998) and Kujala et al. (1996) are standardized testing protocols and not task-specific thus, making it difficult to compare differences in their aerobic responses. Also, conclusion of these studies is only based on central physiological responses and not peripheral ones, which rises an important question: *Are individuals peripherally limited during repetitive manual material handling tasks?* Such conflicting results in the scientific literature led me to investigate the role of central (oxygen consumption, heart rate, etc.) and peripheral (tissue oxygenation and blood volume) physiological responses and their relationship during manual material handling.

Repetitive Pushing-Pulling:

Pushing and pulling activity is defined as the application of one or two hand force, with feet either stationary (Chaffin et al. 1983, Gagnon et al. 1992. Garg et al. 1978) or moving (Lee et al. 1991, Snook and Ciriello 1991), on an object or person. The application of hand force can be either in the horizontal (Esmail et al. 1995), vertical (Warwick et al. 1980) or in the inclined direction (Garg and Beller 1988, Imrhan and Ramakrishnan 1992). For example, cart pushing or pulling can be achieved with feet moving or walking (Lee et al. 1991, Mack et al. 1995, Resnick and Chaffin 1995), whereas floor mopping or shovelling of snow can be treated as effort applied with feet stationary (Franklin et al. 1995, Søgaard et al. 1996). Similar to repetitive lifting activity, pulling and pushing is also accompanied by isometric muscle activity in the trunk and upper extremities, thus influencing the cardiovascular system (Hoozemans et al. 1998). Sanchez et al. (1979) showed that during pushing and pulling without walking. both heart rate and oxygen uptake increased linearly with the load. Hence by understanding the combined static and dynamic responses (Sanchez et al. 1979, Garcin et al. 1996), we can develop guidelines for the pushing and pulling mechanism.

Past and present research has shown that pushing and pulling without feet moving depends on posture adopted, shoe/floor friction, height of force applied and individual's anthropometry (Ayoub and McDaniel 1974, Chaffin et al. 1983, Gagnon et al. 1992, Garg and Beller 1990, Imrhan and Ramakrishnan 1992, Kromer 1974, Kumar 1994a). Kromer (1974) studied horizontal push and pull forces exerted on various surfaces during standing in different positions. The author found that forces exerted varied from 100 N to 750 N for pushing and pulling on surfaces whose coefficient of friction varied from 0.2 to 0.9. Ayoub and McDaniel (1974) evaluated the effects of posture on the strength of subjects in isometric pushing and pulling against a wall and found an optimal height of hands at about 70-80% of shoulder height, approximately 36-45 inches from the floor. Using biomechanical analysis, they also estimated torque and compressive forces on the low-back. Pushing forces increased with increase in the foot

distance and subject weight, whereas both foot distance and handle height contributed equally to the increase in pulling forces with change in the body weight of the subjects. Chaffin et al. (1983) studied the effects of volitional postures during maximal pushing and pulling in the sagittal plane at three different heights (67, 109 and 152 cm) and showed that foot displacement, handle height, and body postures play a crucial role in the application of strength. Maximal strength of males was observed at the lowest height, resulting in high compressive forces on the low-back.

Gagnon et al. (1992) also studied low-back loading during pushing a 22 kg box onto shelves of different heights (58 cm, 99 cm, and 141 cm) and showed that lowest position resulted in high back compressive forces and larger amounts of energy. Kumar (1994a) showed that males and females were stronger in isometric pulling action at a handle height of 100 cm than at the heights of 35 cm and 150 cm respectively. The isokinetic strength values observed were lower than isometric strengths. When isometric pushing strength values were normalized against mean pulling strengths of males at medium height, strength values ranged between 41% and 68%, and between 27% and 44% for males and females respectively. However, on the same population, Kumar (1994b) reported the back compressive forces during pushing to be significantly higher (129% to 627%) than that of the corresponding pull conditions. Also in this study, peak external moments on the back were much higher in pushing compared to pulling. The author concluded that even though one is capable of exerting much less force in pushing, this activity generates much higher back compressive forces than pulling action.

In a one-hand dynamic pulling activity, Garg and Beller (1990) studied the effects of pulling speed (2.5, 3.6 and 3.96 km/hr), handle height (40%, 50%, 60% and 70% of shoulder height) and angle of pull (15°, 25° and 35°) from the horizontal plane on the pulling strength. They showed that strengths decreased with an increase in handle height from 100% at 40% shoulder height to 83% at 70% of shoulder height, with strengths highest at the medium angle of pull. The optimal height and the angle of pull suggested were at 50%-60% of shoulder

height and 25° respectively. Imrhan and Ramakrishnan (1992) examined the effects of arm elevation (45°, 90° and 153° with respect to the shoulder), direction of pull (toward and across the body), and speed of pull (1.58 and 2.38 km/hr) on isokinetic pull strength. Their results showed that pulling toward the body was 9% stronger than pulling across it and the speed of pulling on the strength depends on the body posture adopted.

Only few physiological studies have been reported in terms of the energy expenditure during pushing and pulling with feet stationery (Esmail et al. 1995, Garg et al. 1978, Sanchez et al. 1979). Garg et al. (1978) developed regression equations for pushing and pulling of loads at different heights. In their regression model, body weight, gender, force applied, horizontal distance were some of the dependent variables and found significant difference between genders during pushing and pulling. Sanchez et al. (1979) reported static, dynamic and combined static and dynamic activities results. For loads (6 to 24 kg) pushed. pulled or held without feet moving, the oxygen uptake and heart rate increased linearly with the load. Esmail et al. (1995) established biomechanical and physiological normative data in healthy males and females during a push-pull task on a work simulator for a period of 4 min. Their results showed that significant differences exist between genders in their relative VO_{2mer} and energy expenditure per unit amount of work done (men showed 19% lower values than women). It is evident from the above details that physiological studies related to repetitive pushing and pulling are few.

Aerobic protocols:

In the field of Occupational Health and Safety, it is still a standard practice to evaluate the worker's fitness for work using traditional incremental aerobic protocols on a treadmill, cycle ergometer or arm ergometer, instead of taskspecific maximal aerobic test (VO_{2max}) protocols. VO_{2max} is dependent on age, sex, heredity, and body composition of the individual, state and type of training or work (McArdle et al. 1996). These diagnostic protocols, though informative, may or may not identify the individual's actual physical capacity to do the job due to the different movements involved in the tasks compared to cycling, running or arm cranking (Garg et al. 1992, Khalil et al. 1985, Sharp et al. 1988, Waters et al. 1994, Wenger 1991). The NIOSH guidelines recommend that 33% of the maximum energy expenditure rate should not be exceeded for an eight-hour workday, however, the testing mode to determine VO_{2max} is not clearly addressed.

A recent study by Nindl et al. (1998) showed a modest correlation between repetitive lifting and treadmill running peak oxygen uptake values (r = 0.73 and r = 0.60 for men and women, respectively). Sharp et al. (1988) estimated VO_{2max} of repetitive lifting as 78% of treadmill, 89% of cycle and 125% of arm cranking values whereas for various lifting tasks, Khalil et al. (1985) showed VO_{2mex} as 91% of cycle values. In Petrofsky's study (1978a), depending on the weight of the load lifted or lowered, and rate of the activity performed, average VO_{2max} values were 19% to 47% lower than the bicycle values. These exercise and task specific aerobic values indicate that there is a considerable variation between them. Also, to our knowledge aerobic protocols and VO_{2mex} values during maximal or submaximal pushing or pulling have not been reported in the literature to date. These protocol results indicate that any exercise mode selected to measure aerobic capacity should simulate the specific task performed. Hence it is imperative that task-specific protocols for various occupational activities be developed (Sharp et al. 1988). To our knowledge aerobic protocols and peak aerobic power values during maximal pushing or pulling have not been reported in the literature to date.

Role of Gender:

It has been reported that there are several physiological differences between men and women during manual material handling that cause differences in aerobic capacity, energy expenditure and functional work capacity (Evans et al. 1983, Esmail et al. 1995, Sharp 1994). For the last two decades, the number of women involved in the manual material handling workforce has increased and industries need more comprehensive research investigations in establishing work protocols. Physiologically, women tend to have lower VO_{2max} values than men due to the smaller fat-free mass, lower hemoglobin content and lower maximum cardiac output values (Astrand and Rodahl 1986, Sharp 1994).

However, a study reported by Evans et al. (1983) demonstrated that women are comparable to men in physical capacity to work. In their study of self-paced hard work while carrying loads on four different terrains, both genders showed similar and constant relative energy expenditure values close to 45% VO_{2max} . From the various military investigations reported by Sharp (1994), it was identified that for the tasks that do not require maximal effort or during self-paced work, women tend to perform at the level of man's standards. In a report submitted by Singh et al. (1991) to the National Defence Head Quarters of Canada, women showed lower scores than men on laboratory and field task measures. Absolute aerobic power values were 64.3% of male scores. During anaerobic tests, women showed greater relative leg peak power (61.9% of male scores) than arm peak power tests (49.6% of male scores). Their static and dynamic muscular strengths ranged from 59.3% to 77.8% of that of men. Some of the women soldiers performed similar to that of men soldiers in both laboratory and field investigations. Esmail et al. (1995) also showed no gender differences in the energy expenditure of simulated tasks when expressed relative to the VO_{2max} of the subjects. However, women were less efficient than men when the energy expenditure was expressed per unit of work accomplished.

Nindl et al. (1998) compared the gender differences in peak oxygen uptake between treadmill running and repetitive lifting. Their results indicated significant differences in gender, mode of exercise, and interaction effects for peak oxygen uptake. They reported lower physiological responses in women (20%) than in men (26%), and also for repetitive lifting than for treadmill running. In a pushing and pulling study, Hoozemans et al. (1998) reported that on average, male postal workers showed a higher absolute level of energy expenditure but a lower HR than females. Thus, comparing physiological work capacity in men and women during such repetitive manual material handling activities enhance our understanding of the gender responses at the workplace.

Central and Peripheral Factors Influencing Physical Work Capacity:

The well-known Fick equation in physiology expresses oxygen consumption or uptake as the product of blood flow and arterio-venous oxygen difference. Cardiac output, a primary determinant of person's aerobic capacity, is defined as the amount of blood pumped per minute from the left side of the heart, and is not studied extensively in the field of ergonomics (Eastman Kodak, 1986). In other words, blood flow in the Fick equation or cardiac output is essential in evaluating the individual's capability to meet the increased oxygen needs of working muscles (Antonutto et al. 1995).

Cardiac output is also a product of stroke volume and heart rate where stroke volume is defined as the volume of blood ejected into the main artery by each ventricular beat. Cardiac output and stroke volume are fundamental descriptors of cardiovascular function. Arterio-venous oxygen difference in the Fick equation, is the difference in oxygen content between the blood entering and that leaving the pulmonary capillaries (Astrand and Rodahl 1986).

Maximal Aerobic Power (VO2mer)

Maximal aerobic power is defined as the maximal oxygen consumption per unit time, and is also known as maximal oxygen uptake per unit time. Garg (1997) suggested the following important factors that affect aerobic capacity: *age:* holds an inverse relationship (30% lower at age 65 compared to age at 25), *gender:* females, on average have 30% lower aerobic capacity than men,

physical fitness: aerobic capacity increases with physical fitness,

whole body vs upper-body work: on average, aerobic capacity is 30% lower for arm work because of small muscle mass involved,

nature of work: each task has its own maximum aerobic capacity as determined primarily by number of muscles available for work (bicycle, treadmill, arm, lifting, vibration, pushing/pulling etc.).

As Wagner (1988) pointed out, there is no single factor either controlling or limiting VO_{2max} , however there is a strong correlation between the compound variable oxygen delivery (QO_2) and VO_{2max} . Oxygen delivery (through convection process) is the product of blood flow and arterial oxygen content, which is also represented as the product of blood flow, hemoglobin concentration and arterial oxygen saturation. This simple relationship reveals that change in any of the determinants of QO_2 (blood flow, hemoglobin, or saturation) result in change in VO_{2max} . Intramuscular factors such as oxygen in the venous blood exiting the exercising muscle (through diffusion process) also play an important role in determining the VO_{2max} . Thus, VO_{2max} can be equated to both convection and diffusion processes.

Energy expenditure:

Evaluating energy expenditure (in terms of oxygen uptake) during a physical activity has been widely applied and reported both in exercise and occupational physiology (Astrand & Rodahl 1986, Eastman Kodak 1986). Prediction models of energy expenditure during manual material handling, developed by Garg et al. (1978), suggest that average energy expended per day varies from task to task. This variation can be attributed to the amount of musculature involved during a specific task. In addition to energy expenditure, cardiovascular and respiratory measurements such as heart rate, cardiac output, blood pressure and blood lactate are also essential to understand the physiological stress on the worker (Ayoub and Mital 1989). Many occupational activities such as lifting, pushing, pulling, and carrying are combinations of combined static and dynamic muscular work. Since these two types of muscle contractions elicit different physiological responses, it is important that these responses be examined when these muscular efforts are combined. Sanchez et al. (1979) reported that energy cost during combined work is higher than the additive cost of static and dynamic work separately.

Muscle Metabolism:

The dominant source of energy for skeletal muscle during exercise or work is oxidative metabolism and is evaluated by the Fick equation as shown below (Ferrari et al. 1997): oxygen consumption (VO₂) = blood flow * arterio-venous (a-v)O₂ difference O₂ availability to the tissue is either measured invasively such as blood sampling, biopsy, with laser fluorimetry or noninvasively such as Nuclear Magnetic Resonance (NMR) and NIRS (Sahlin 1992). Need of adequate O₂ to tissues is possible by an increase of O₂ delivery (DO₂) and by an increase of O₂ extraction from oxyhemoglobin (HbO₂) making sure oxidative metabolism proceed at a rate equal to the energy demands of the cell (Simonson and Piantadosi 1996). Muscle oxygenation, defined as the relative saturation of oxyhemoglobin (HbO₂) and oxymyoglobin (MbO₂) depends on the balance between O₂ delivery and metabolic rate or VO₂ (Belardinelli et al. 1995a). During circulation, O₂ is transported to tissues by blood flow (known as convection) and extracted by mitochondrion via diffusion (Piantadossi 1993).

Myoglobin is an O_2 binding protein that is available exclusively in skeletal muscle. This chromophore serves as an extra intracellular source of O_2 and helps in the transfer of O_2 to mitochondrion (Mancini et al. 1994a). Oxygen is delivered to the tissue capillary as HbO₂ where it diffuses into the muscle cell and binds to myoglobin (Mb). It is subsequently accepted by the terminal cytochrome - cytochrome oxidase (Cyt a,a₃) in the electron-transport chain, discharging its electrons directly to an oxygen atom (Simonson et al. 1996). Oxidation is complete when two electrons and two protons (H+) join with oxygen to form water (H₂O). The redox state (oxidation of hydrogen and subsequent reduction of oxygen) of Cyt a,a₃ in mitochondrion is determined by the flow of electrons through the electron-transport chain and by the availability of oxygen indicating the overall activity of oxidative metabolism in the cell (Simonson and Piantadosi 1996). During any physical activity, the limitation of aerobic ATP production results in inadequate O₂ delivery to mitochondrion leading to cellular hypoxia (Mancini et al. 1994a).

Tissue Oxygenation and Blood Volume

A brief description of both brain and muscle oxygenation and blood volume relevant to the present thesis are explained below.

Cerebral Physiology:

The brain is the primary organ that demands greater blood supply than any other organ of the body (Mchdelishvili 1986). Due to its high metabolic rate, the local hematocrit in the brain tissue is higher than either the working or the resting muscle. According to Nioka et al. (1998), during exercise, the brain relies primarily on mitochondrial oxidative phosphorylation for energy, and is more prone to hypoxic conditions than skeletal muscle. Hypoxic limitation in the brain results in less capacity for anaerobic metabolism (Mchedlishvili 1986). Also, in the brain during normal physiological conditions, about 95% of the adenosine triphosphate is produced from the aerobic pathway, and the rest of the energy (0-5%) comes from lactate production that is not further oxidized, but is lost from the brain as it is carried away with the venous blood (Roland 1993). However, lde and Secher (2000) commented that the brain, including neurons and the astrocytes, has the ability to use lactate as an energy source.

Glucose from the blood is the major respiratory fuel in the brain that is used for energy metabolism and has an average respiratory quotient of 0.99 (Roland 1984). Since glucose consumption increases during cerebral activation, measuring its regional glucose consumption over a period of time is equivalent to a measurement of energy expenditure for that period (Roland 1993). Additionally, oxygen is essential for oxidizing glucose into carbon dioxide and water. Therefore, measuring regional oxygen consumption is also important to understand oxygenation during functional activation of the brain. However, during this metabolic process of extra demand on the glucose and oxygen supply, dilation of pre-capillaries occurs, resulting in the opening of additional capillaries. Consequently, regional cerebral blood volume and blood flow are elevated. Hence, monitoring regional cerebral blood volume can also be used as a marker for cerebral activation during a specified activity (Roland 1993). According to the *Cortical field activation hypothesis* (Roland 1985a), the cerebral cortex facilitates the brain work of awake humans by activating multiple cortical fields, and each activated field occupies an area of a few square centimeters. The author further defines 'activation' as an increase of the biochemical activity of the synapses and neurons in such a field, thus leading to increases in transmembraneous ion transport and the rate of regional cerebral metabolism or regional cerebral blood flow (Roland 1993). During functional activation of brain, it was also shown that in the normal (healthy) brain there is a significant positive correlation between regional cerebral oxygen consumption and regional blood flow (Raichle et al. 1976, Heistad and Kontos 1983), with the increase in regional cerebral blood flow being greater than the regional metabolic oxygen demand (Buxton and Frank 1997, Ide and Secher 2000).

The amount of oxygen delivered to the brain depends mainly on the concentration of Hb and its affinity for oxygen, oxygen transport, characteristics of the capillary blood, and cerebral blood flow (Buxton and Frank 1997, Wardle et al. 2000). Even though blood flow is a function of perfusion pressure and vascular resistance, in the cerebral region, flow is independent of perfusion pressure, and under physiological conditions, it is predominantly regulated by changes in the resistance of cerebral arteries (Wahl and Schilling 1993). Thus, any variation in perfusion pressure results in concomitant opposite changes in cerebrovascular resistance, thus regulating the cerebral blood flow [known as autoregulation] within a constant range of 80 and 180 mmHg (Wahl and Schilling 1993). A decrease in the cerebral blood volume, and an increase in the resistance results in a reduction in the cerebral blood flow and cere

During physical exercise, the (global) cerebral blood flow is acutely affected by various physiological factors such as cardiac output, heart rate, blood pressure, changes in pH, arterial CO_2 tension, etc. (Doering et al. 1998, Jorgensen 1995, Olesen 1971). However, based on the techniques adopted, conflicting evidence of variability [increase or decrease or no change] in cerebral

blood flow has been shown by several authors (Friedman et al. 1992, Globus et al. 1983, Ide and Secher 2000, Olesen 1971, Pott et al. 1996, Roland and Larsen 1976, Zobl et al. 1965).

During cycling exercise, Globus et al. (1983) demonstrated no significant increase in the cerebral blood flow, but demonstrated an increase in the cerebrovascular resistance and a concomitant decrease in the peripheral vascular resistance in humans. In addition to increase in blood pressure and heart rate responses, Olesen (1971) demonstrated a significant increase in the regional cerebral blood flow in humans during arm work. Friedman et al. (1992) found a 20% increase in the regional cerebral blood flow in humans during arm work. Friedman et al. (1992) found a 20% increase in the regional cerebral blood flow in the sensorimotor area of humans during rhythmic handgrip contractions. Interestingly, Zobl et al. (1965) concluded no significant role for the brain during muscular exercise in humans. These authors showed a minor increase (not statistically significant) in the cerebral blood flow accompanied by a greater increase in cardiac output during cycling exercise. Cerebral vascular resistance remained constant and peripheral vascular resistance decreased during this activity. Zobl et al. (1965) further postulated that the brain behaves as a steady-state organ during vigorous physical work with minor changes in global cerebral circulation and metabolism.

Jorgensen et al. (1992) hypothesized that a combination of muscle, tendon, and joint activation during cycling are associated with an increase in cerebral perfusion that is linked to the exercise intensity. In describing the importance of circulation to the brain, Jorgensen et al. (1999) suggested that during both rest and exercise, the cerebral blood flow has to compete with the perfusion of capillary beds of various tissues including muscle. During leg cycling, Hollmann et al. (1990) suggested that cardiac neurohormones affect regulatory processes in the peripheral hemodynamics and muscle metabolism. They further demonstrated an increase in the regional cerebral blood flow during cycling at different intensities, and suggested such an increase in blood flow may be due to the speedy transport of neurotransmitters. Such variations in the cerebral blood flow during exercise are consistent with the hypothesis of Sokolff et al. (1955) who postulated that the increase in regional blood flow and metabolic rate in one region of the brain is balanced by a decrease in regional blood flow and metabolic rate in some other region of the brain.

Regional brain activity induces local arteriole vasodilation, resulting in an increase in blood volume and blood flow (Villringer and Chance 1997). Neuronal activation during an activity is coupled with an increase in regional cerebral blood flow, and is accompanied by an increase in cerebral blood volume either with volumetric expansion in blood vessels already perfused or with an increase in recruitment of vessels actually perfused (Narayan et al. 1995). Cerebral blood volume is only 3-4% of the total brain volume, which is defined as the blood volume from the internal carotid and vertebral arteries to the internal jugular veins (Keyeux et al. 1995). Effectively, it is a summation of: 75% relative venous blood volume (due to its wide lumina), 20% of blood in the arterioles, and 5% in the capillaries (Mchedlishvili 1986).

Muscle Physiology:

In human skeletal muscle, the motor units are classified into two broad categories according to their structural, functional, and metabolic characteristics (Fox et al. 1988, McArdle et al. 1996). These are slow-twitch (Type I) and fasttwitch fibers (Type II). Slow-twitch fibers are rich in blood supply, and have relatively greater aerobic and smaller anaerobic capacity compared to fast-twitch fibers (Fox et al. 1988). Due to larger quantities of mitochondrion, capillaries, and myoglobin, these fibers demonstrate a greater oxidative capacity than the fasttwitch fibers. Further, the vascular beds of slow-twitch fibers consist of greater blood flow at rest, and do not demonstrate greater increase in blood flow during exercise because of a limited vasodilatory capacity. These fibers are preferentially recruited during postural and endurance type of activities, thus demonstrating a greater resistance to fatigue (Fox et al. 1988). Based on the energy production, fast twitch fibers are further divided into two categories. These are the fast-oxidative-glycolitic (Type IIa), and fast-glycolitic (Type IIb). However, Fox et al. (1988) reported the presence of a third category known as fast-unclassified (Type IIc) whose fiber distribution range from 0 to 4% in all

muscles. Fast-twitch fibers are preferentially recruited during high intensity activities, and have a greater vasodilatory capacity compared to slow-twitch. These authors suggested an oxidative hierarchy of the different fiber types as: slow-twitch > fast-oxidative-glycolitic > fast-glycolitic.

Since the present thesis is concentrated on low-back intensive tasks, a brief description of spine and back muscle physiology is given here. Jorgensen (1997) reported that paraspinal muscles are predominantly slow twitch, consisting of small fiber cross-sectional areas associated with a well-developed capillary network. Further, these muscles are supplied with blood from dorsal branches of the lumbar and intercostals arteries and veins. The author also demonstrated that the isometric endurance time of the trunk extensor muscles was similar to the soleus muscle and significantly greater than other skeletal muscles, hence its higher fatigue resistance properties. Moreover, at a higher percentage of the maximum voluntary contraction, paraspinal muscles are able to maintain a higher steady state level of oxygenation than muscles with a lower capillary density and a lower percentage of slow twitch fibers (Jensen et al. 1999). Interestingly, Jorgensen (1997) found anaerobic capacity (glycolytic activity) of these muscles to be significantly higher than other skeletal muscles. According to Haldeman (1999), a strong relationship between the muscles and the paraspinal ligaments and fascia forms a physiological corset and behaves as a primary defense mechanism against trauma. This author further emphasized that these paraspinal muscles consist of unencapsulated nerve endings and

From the spinal cord, special capillary loops and blood pools reach the hyaline cartilage of the vertebral disc, thus providing nutritional supply and (waste removal) to the intervertebral disc through diffusion, and eventually extensive networks of veins that run in the spinal cord facilitating the venous return (Junghanns 1990). Interestingly, although the intervertebral disc is avascular in nature, it still depends on the nutritional diffusion through the intervertebral disc also consists of fine nerve endings in the outer one third of the annulus, and these

mechanoreceptors that have been attributed to the muscular pain experienced.

nerve endings seem to be immunoreactive to pain-related neuropeptides such as substance P, calcitonin-gene-related peptide, and vasoactive intestinal peptide (Haldeman 1999). Impulses from these nerve endings and the adjacent longitudinal ligaments travel via a number of sensory nerves to the spinal cord. Thus, the intervetebral disc can also be a source of low-back pain.

For example, in a typical motorized vehicle when an occupant is exposed to whole-body vibration, the impact is absorbed by the spine and its muscle corset, and with long term exposure of relatively high intensity vibration, this stimulus may further damage the intervertebral motion segment (Junghanns 1990). Further, the author emphasized the importance of the relationship between the spinal movements and diffusion of nutrition to the intervertebral discs. The author theorized that exposure to whole-body vibration and prolonged sitting (without movement) may lead to muscle fatigue in the low-back and eventually damage the intervertebral body-intervertebral disc boundary. Lack of motion due to prolonged sitting may slow down the pumping mechanism, resulting in inefficient removal of waste metabolites and thus exacerbating the intervertebral disc disorders.

Regional Distribution of Blood Flow:

Like any other tissue, efficiency in the skeletal muscle function is disturbed if there is a mismatch between oxygen supply (delivery) and oxygen demand (consumption) (Pittman 2000). Further, oxygen extraction is defined as the ratio of oxygen demand and oxygen supply. In a healthy tissue, decrease in oxygen delivery is always matched by increase in oxygen extraction, thus meeting the necessary oxygen demand. In the skeletal muscle, Pittman (2000) reported that oxygen extraction varies from approximately 0.25 (resting condition) to 0.75 (contraction). According to Sibbald et al. (2000), blood flow is regulated at three levels of the circulatory system: (a) centrally by the cardiac output, (b) regionally by the distribution of blood flow between organs, and (c) microvascular distribution of blood flow within organs. In explaining the

importance of tissue oxygenation and blood circulation, Sibbald et al. (2000) stated:

"when anemia coexists in critical illness, the reduction in the content in arterial blood is compensated for by (a) increasing the cardiac output, (b) altering the distribution of blood flow between organs, and (c) altering the distribution of blood flow within organs, thereby to increase tissue oxygen extraction. When the capacity of these compensatory mechanisms is exhausted, tissue injury supervenes."

Similarly, Calbet (2000) proposed that during submaximal work, fluctuations in the content of arterial blood are compensated by equal changes in muscle blood flow and cardiac output, thus maintaining constant oxygen delivery to the tissue. They indicated that increased arterial content shows no significant affect on muscle blood flow or cardiac output, however, during maximal work with reduced content of arterial blood and under hypoxia, muscle blood flow is decreased with the maximal cardiac output of the heart.

Cardiac output increases linearly with exercise intensity until maximal aerobic power is attained. Depending upon the intensity of exercise, the majority of the increase in cardiac output is directed towards the muscles directly (e.g. leg, arms) and indirectly (e.g. heart and respiratory muscles) involved in the exercise. However, blood flow to the relatively inactive organs decrease significantly (McAllister 1997). As well, there is a significant increase in blood flow to the brain during exercise (McArdle et al. 1996). However, when expressed in relative terms (as a percentage of maximal cardiac output) blood flow to all the exercise, whereas in the brain, it actually decreases (global not regional) from 14% to 4% (McArdle et al. 1996). This 'redistribution' of blood flow is achieved through vasoconstriction in these inactive organs and vasodilation in the active muscles.

In an example of 'total muscle blood flow' in humans, Rowell (1988) stated that:

"in a 75-kg sedentary individual with 30 kg of muscle mass and a maximal cardiac output of 25 l/min, the maximal total muscle blood flow would be 22 l/min and rest of the blood (3 l/min) is directed to the heart, brain, visceral organs, and skin".

Further, this author suggested that limited cardiac pumping capacity in a human is exceeded by the greater ability of skeletal muscle to vasodilate. This example given by Rowell (1988) suggests that certain amount of blood is distributed to the brain as well, and such a distribution may depend on the work intensity.

During incremental cycling exercise till exhaustion, Bhambhani et al. (2000) monitored both cerebral and vastus lateralis muscle simultaneously (Figure 1.1). These authors demonstrated that the pre-frontal cortex region was activated and both oxygenation and blood volume increased systematically until volitional fatigue. However, muscle oxygenation decreased during the incremental exercise and blood volume leveled off until approximately 50% to 60% of peak oxygen uptake. Similarly Nioka et al. (1998) hypothesized that the ability of skeletal muscle to metabolize energy anaerobically via glycolytic pathway gives the muscle an advantage over the brain during the hypoxic conditions. However, this reverse phenomenon may not be always true. During hypovolemic shock and fainting, Jorgensen et al. (2000) demonstrated a decrease in the cerebral blood flow (subsequently oxygenation) while blood flow in the muscle increased at the same time.

Review of Near Infrared Spectroscopy:

In a recent conference on tissue oxygenation in acute medicine, Sibbald et al. (2000) emphasized the needs of studying tissue oxygenation, and stated:

"...measurements of global hemodynamics and oxygen-related parameters are considered to be poor surrogates for the quality of tissue oxygenation. The pulmonary artery carries blood from all vascular beds of the organism that previously was in equilibrium with tissue in individual organs. Parameters derived from *mixed venous blood* might therefore serve as a parameter of global tissue oxygenation but cannot reflect the adequacy of tissue oxygenation in critical organs. Changes in *mixed venous oxygen saturation* are therefore indicative only of possible changes in the relationship between whole-body oxygen delivery and consumption."

To this effect, Near Infrared Spectroscopy (NIRS) is a noninvasive optical technique that has been used to study tissue physiology (oxygenation and blood volume) during rest and work in the skeletal muscle (Bhambhani et al. 1998a, Boushel 1998, Chance et al. 1992, Murthy et al. 1997, 2001) and brain (Fallgatter and Strik 1998, Obrig et al. 1996, Obrig and Villringer 1997, Villringer et al. 1994). NIRS is based on the principle that light energy in the near-infrared region is differentially absorbed by the oxygenated and deoxygenated forms of hemoglobin and myoglobin (Jobsis 1977, Belardinelli et al. 1995b). Light in the near infra-red region is able to measure the relative *changes* in oxyhemoglobin and oxymyoglobin concentration rather than the *absolute* concentration with much less scattering than that possible at the shorter wavelengths of light (Simonson and Piantadosi 1996, Taylor and Simonson 1996).

In a typical investigation using NIRS technique, a known amount of incident near infra-red light is impinged on the tissue to be studied, e.g. vastus lateralis muscle during leg cycling, and is recovered by a receiving optode. Part of this light energy transmitted is either absorbed by chromophores in the tissue or scattered in the tissue. The amount of light recovered during this process depends on both absorption and scattering properties. Any changes monitored in the recovered light reflect changes in the relative amounts of chromophores in the illuminated region.

Modified Beer's Law applied to the Human Tissue:

When light illuminates a live human tissue, its transmission characteristics depend on a combination of absorption, scattering, and reflectance properties. Absorption and scattering are wavelength dependent and reflectance is represented by the inclination of light beam to the tissue surface. As the wavelength increases, scattering decreases thus benefiting the transmission of

infra-red light (Jobsis 1977). Propagation of light into a highly scattering medium such as living tissue follows Modified Beer's Law, thus converting the measured *changes in the absorbance* or attenuation to *changes in the absolute chromophore concentration* (Delpy et al. 1988, Simonson and Piantadosi 1996). Three tissue chromophores: hemoglobin (Hb), myoglobin (Mb), and cytochrome c oxidase (Cyt a, a_3) are the only biological compounds that show variable oxygen-dependent absorption spectra in the near-infrared range (Simonson and Piantadosi 1996).

The amount of light recovered (resulting from the reflectance) from an illuminated tissue depends on the intensity of incident light on the tissue, the physical separation of optodes, the degree of light scattered by tissue, and the amount of tissue absorbency due to the concentration of chromophores in the tissue. This relationship (DeBlasi et al. 1994, Simonson and Piantadosi 1996, Taylor and Simonson 1996) is expressed as follows:

$$OD = a c L B + G$$
 or $\Delta OD \cong \Delta c$

where OD = absorption of light expressed as the optical density

a = extinction coefficient of chromophore for the wavelength in $\mu M^{1} cm^{-1}$

c = chromophore concentration in μM

L = distance between the point of light entry and exit or simply a physical optode spacing on the skin surface being tested in *cm*

B = pathlength factor of light through the tissue accounting for scattering (also known as differential pathlength factor)

G = a geometric correction factor to account for the geometry of tissue and optode positioning

 ΔOD = change in the absorbance or optical density

 $\Delta c =$ change in the chromophore concentration

Factor 'a' is a physical constant and is usually estimated from *in vitro* (Simonson and Piantadosi 1996) measurements. Extinction coefficients of Hb, Mb, and the oxidized minus reduced Cyt a_{a_3} were obtained from the brains of hemoglobin-free rats whose blood had been exchanged with a fluorocarbon blood substitute (Edwards et al. 1993, Piantadosi et al. 1986). Inter-optode separation between the source and detector is always a function of optical pathlength and depends on the degree of light scatter in a tissue. Optical pathlength used in the Beer's

Law must be increased to consider the effects of multiple scattering in the human living tissue which is considered in factor B of the equation. Because of the usage of continuous wavelength monitoring, the mean pathlength of the photons cannot be simultaneously measured (Obrig and Villringer 1997). However, in the transmitted phase calculation of minimum optical pathlength known as interoptode separation is feasible, thus the identification of upper limit on the chromophore concentration measurements (Delpy et al. 1988).

For estimating change in the chromophore concentration, some researchers used approximate or average pathlengths of photons through various configurations such as neonatal brain, adult brain, or forearm which varied between 4-6 times the inter-optode distance (Simonson and Piantadosi 1996. Taylor and Simonson 1996). Due to the scattering effect, the distance traveled by the light is greater than L (by a factor B) and equal to the product of L and B (Edwards et al. 1993). Factor B is estimated based on the photon migration models or measured separately by time resolved spectroscopy. Since factor B also depends on individual skin pigmentation, skull thickness (while monitoring brain activation) and doesn't take individual anatomy into consideration, the volume illuminated may differentiate between subjects and between different brain areas (Obrig and Villringer 1997). As G varies with both the type of tissue studied and the optode geometry it is very difficult to quantify. Even though theoretical research in this area suggests the feasibility of calculating absolute concentration changes in tissue chromophores in an illuminated region, currently this remains a challenging task. Assuming 'a, L, B and G' as constants, changes in the absorbance of light (ΔA) thus can be attributed only to the changes in the chromophore concentration in a tissue (Δc) and trends can be calculated relative to a starting point considered as zero (De Blasi et al. 1994, Simonson and Piantadosi 1996). In some experimental settings, change in absorbance is expressed as a percentage of the full optical signal (Simonson and Piantadosi 1996, Taylor and Simonson 1996).

Light in the near infra-red (NIR) region is able to measure these changes rather than the absolute concentration with much less scattering than that is

possible at the shorter wavelengths of light (Simonson and Piantadosi 1996, Taylor and Simonson 1996). According to Villringer and Chance (1997), increased activity of brain cells (e.g. during cognitive, visual, or motor stimulation) is associated with an increase in glucose and oxygen consumption. Also, the increase in cerebral blood flow exceeds the increase in oxygen consumption leading to an increase in intravascular hemoglobin oxygenation. Thus, any changes in these chromopheres during functional brain activation can be monitored by the modified Beer's law. Hence by applying the modified Beer's Law to human living tissue, near infra-red spectroscopy can be successfully applied *in vivo* to quantify the range or change in the chromophore concentrations continuously during rest, exercise, or recovery, and also in the clinical, research and occupational investigations (Belardinelli et al. 1995a, Bhambhani et al. 1997&1998a, Maikala and Bhambhani 1999b, Mancini et al. 1994a, Mohler et al. 1997, Murthy et al. 1997 & 2001, Simonson and Piantadosi 1996, Villringer and Chance 1997).

Theoretical Basis:

The relative transparency of biological tissue such as brain, skin, bone, and muscle to the light in the near-infrared region (700-1000 nm) results in much less scattering and less absorbency thus more penetration (up to 6 cm) than at the shorter wavelengths of light such as visible light (first 1 to 2mm) and first 500 μ m of tissue in the ultraviolet range (Edwards et al. 1993, Piantadossi 1993, Simonson and Piantadosi 1996). The principle of NIRS is that light energy in the near-infrared region can transmit through tissues with an average path length of 7-8 cm and at specified wavelengths it is differentially absorbed by the oxygenated and deoxygenated forms of Hb and Mb (Belardinelli et al. 1995b). With the availability of oxygen, variability in the absorption spectra of the iron (heme) or copper centers of the chromophores can be seen and the relative change in the chromophores can be measured using the modified Beer's Law explained in the above section (Simonson and Piantadosi 1996). NIRS bases its principle on this law and was developed to monitor changes in tissue

oxygenation noninvasively during rest or physical activity. Tissue oxygenation, defined as the relative saturation of oxyhemoglobin (HbO_2) and oxymyoglobin (MbO_2) depends on the balance between oxygen delivery and metabolic rate (Belardinelli et al. 1995a).

Absorption of light in the NIR range monitors changes in tissue oxygenation at the levels of small blood vessels (arterioles, capillaries, and venules) and intracellular sites of oxygen consumption. Since near infra-red light propagation through a single red blood cell is minimal, majority of the light is absorbed at the level of the arterioles, capillaries, and venules. In arteries with greater than 1mm diameter NIR is more than 90% absorbed (Manicini et al. 1994b).

The possibility of using NIRS for measuring muscle cytochrome Cyt a,a₃ is still questionable (Ferrari et al. 1997) due to the overlapping spectra in the NIR region and the low intensity of oxidized copper (Cu_a) band of cytochrome Cyt a,a₃ (Piantadosi 1993). Also in the NIR range, the absorption spectra of Mb is not distinguishable from that of Hb as it is an insignificant contributor [the ratio of Hb: Mb = 10:1 in the human skeletal muscle] (Belardinelli et al. 1995a, Mancini et al. 1994b, Piantadossi 1993). However, the concentrations of (HbO₂+MbO₂) can be distinguished from (Hb+Mb). Hemoglobin absorbs light differentially while it is carrying O₂ and blood changes from deep red to bright red as it becomes saturated with O₂ (Mohler et al. 1997). Mancini et al. (1994a) showed a higher degree of HbO₂ without any concomitant MbO₂ during exercise. They also showed that HbO₂ changes precede MbO₂, suggesting that NIRS monitors mainly muscle Hb and not Mb deoxygenation.

The absorption band of water in the NIR range can be monitored at about 1000 nm, and therefore changes in water content in the muscle do not affect the NIR signals in the range 700 to 900 nm (Piantadossi 1993). Assuming (MbO_2+Mb) as constant, we can attribute the changes in the concentration of $[(HbO_2+MbO_2)$ and (Hb+Mb)] from the observed changes in the optical density to the change in HbO₂ and Hb alone entering or exiting the capillary bed of exercising muscle (Piantadossi 1993). Hence changes in the tissue total

hemoglobin volume (tBV) which normally amounts to only about 5% of the total tissue volume can also be equated simply to the changes in the sum $[(HbO_2+MbO_2) + (Hb+Mb)]$ (Piantadossi 1993).

Millikan (1937) demonstrated deoxygenation in cat soleus muscle using visible light (400-650 nm) during contraction and ischemia but this study was limited by poor penetration to the tissues. NIR absorption spectrum of a human muscle was first produced by Norris in the 1960s at the United States Department of Agriculture (Smith 1977). In 1977, Jobsis demonstrated the changes in the concentrations of cerebral (cat and human) oxy, deoxy hemoglobin, cerebral Cyt a, a₃ and blood volume in the near infra-red range. So far numerous NIRS techniques have been developed to investigate tissue oxygenation. They differ in *the type of light source adopted* such as lamps, lasers and light emitting diodes and *the modality of the light sources* such as continuous wave, continuous wave spatially resolved, pulsed or time resolved spectroscopy and phase modulation (Ferrari et al. 1997).

One of the very popular and reliable NIRS tools, known as a continuous dual wave spectrometer was developed by Britton Chance. It measures the reflected light at wavelengths of 760 and 850 nm (MicroRunman, NIM Inc., PA). At 760 nm, Hb absorbs more light whereas HbO₂ absorbs more light at 850 nm, suggesting that as Hb is oxygenated the absorption of light at 760 nm decreases while the absorption at 850 nm increases. As mentioned earlier, Mb also absorbs light at these wavelengths but is known to be insignificant (Mancini et al. 1994b). The difference between the amount of light reflected at these two wavelengths indicates a change in the concentration of HbO₂. The sum of the signals at these wavelengths indicates a change in the blood volume (Mohler et al. 1997, Belardinelli et al. 1995a). Further, it has been shown that NIRS signal predominantly reflects the oxygenation state of Hb in the mixed [cerebral or muscle] venous blood in the 'focal' optical volume [not 'regional' as correctly suggested by Cruz (1993)] of the tissue under study (Mancini et al. 1994b).

The changes in tissue oxygenation during ischemic conditions or varying levels of physical activity are scaled to the overall change in the signal which has

been termed as Total Labile Signal (TLS) [(Piantadossi et al. 1986, Simonson and Piantadosi 1996, Taylor and Simonson 1996)] or 'physiological calibration' [Wilson et al. 1989]. TLS, is the difference in absorbence of a chromophore (e.g. Hb) between fully reduced conditions created through restricting the O₂ supply by cuff ischemia or arterial clamping and maximum oxidation obtained by increasing the O₂ delivery to the tissue (hyperoxygenation) or breathing simply room air (Piantadossi et al. 1986, Simonson and Piantadosi 1996, Taylor and Simonson 1996). Depending on the muscle group studied, one can use cuff ischemia for establishing TLS. An example of vascular occlusion to the biceps brachii at suprasystolic pressure of 250 mm Hg is illustrated in Figure 1.2 (Maikala and Bhambhani 1999a). However, it is not possible to occlude blood flow to trunk muscles using this protocol. Recently, Maikala et al. (2000) established minimum and maximum oxygenation and blood volume limits for the lumbar extensor muscles during maximal trunk extension in both sitting and standing postures (Figure 1.3). Jensen et al. (1999) reported submaximal limits for this lumbar muscle using local compression. McGill et al. (2000) also demonstrated submaximal limits for this muscle. Another alternative is the static isometric back endurance approach called the Sorensen test (Biering-Sorensen 1984). Yoshitake et al. (2001) studied maximal trunk extension for a brief period with this approach, and reported both minimum and maximum values for oxygenation and blood volume.

However, one can also establish minimum and maximum muscle oxygenation during incremental exercise till exhaustion using an equation developed by Belardinelli et al. (1995a). This equation states:

Percent Tissue oxygenation during exercise = $\frac{y-M}{m-M}$ * 100

where $y = O_2$ saturation value observed during exercise (in milli Volts)

M = minimal saturation or minimum NIRS value at the peak exercise (in milli Volts)

m = maximum saturation value during recovery of the exercise (in milli Volts)
This equation applied during incremental arm cranking exercise till exhaustion (Bhambhani et al. 1998a) is illustrated in Figure 1.4. Thus, NIRS only assesses directional *changes* in tissue oxygenation from one phase (e.g. rest) to another phase (e.g. incremental exercise).

Validity of Near Infrared Spectroscopy:

The validity of NIRS in evaluating changes in cerebral oxygenation and blood volume has been well established in humans (Holzschuch et al. 1997, Pollard et al. 1996, Punwani et al. 1998, Quaresima et al. 2000). Correlation of brain tissue oxygen pressure (measured in the frontal white matter with a Clark-type electrode) and hemoglobin oxygenation (on the forehead using NIRS) were studied in severe head injury patients (Holzschuch et al. 1997). These authors showed that under the conditions of a stable NIRS signal and a diffuse brain lesion, changes of oxygen pressure were highly correlated with NIRS (r > 0.7). Cerebral oxygen saturation was also validated against jugular venous saturation (r=0.88-0.99) in healthy volunteers during hypoxia (Pollard et al. 1996). Further, measured tissue oxygenation and calculated cerebral venous oxygen saturation were compared in healthy volunteers during partial venous occlusion (Quaresima et al. 2000). They showed that oxygen saturation measured at the right and left forehead using the spatial NIRS technique reflects mainly the intracranial venous compartment saturation (r=0.76).

NIRS is a valid technique in measuring muscle venous oxygen saturation under a variety of experimental conditions in humans (Boushel et al. 1998, Hamoka et al. 2000, Mancini et al. 1994, Sako et al. 2001, Tran et al. 1999). During dynamic handgrip exercise, Sako et al. (2001) obtained a significant correlation of 0.97 for muscle oxidative metabolism between continuous wave NIRS spectroscopy and magnetic resonance spectroscopy measurements. During exercise and cuff ischemia, Tran et al. (1999) showed NIRS signal in the gastrocnemius muscle closely matches with nuclear magnetic resonance spectra (no correlation values reported). They demonstrated a decline in the oxygenation (obtained by NIRS) and myoglobin desaturation (obtained with nuclear magnetic resonance spectra), and suggested that NIRS signal monitors myoglobin desaturation as well. During ischemia and dynamic handgrip exercise, Boushel et al. (1998) showed significant oxygen saturation correlations ranging from 0.60 (ischemia at rest) to 0.56 (rhythmic handgrip exercise at 30% maximal voluntary contraction) between continuous wave NIRS method and venous blood sampling measurements. Mancini et al. (1994b) also measured oxygen desaturation during both work and cuff ischemia in the forearm flexors. They obtained a significant correlation (0.82 to 0.97) between the absorption band and venous blood oxygen saturation.

Despite the abundance of research that supports the validity of NIRS both in vivo and in vitro under variety of experimental conditions, some researchers have questioned the validity of this technique during exercise in humans. Using continuous wave NIRS during prolonged submaximal cycling exercise, Costes et al. (1996) reported a significant correlation (r=0.55) between the NIRS signal and femoral venous oxygen saturation under hypoxic conditions but not under normoxic conditions. Further, MacDonald et al. (1999) showed a significantly higher correlation between NIRS signal and femoral venous oxygen saturation (r value ranging between 0.53 and 0.61) during kicking exercise under hypoxic conditions only. During isometric contractions of forearm flexors, Hicks et al. (1999) compared muscle oxygenation measured with NIRS and venous oxygen saturation obtained from direct blood sampling from the deep forearm vein. At 10% of maximum voluntary contraction during both normoxic and hypoxic conditions, the NIRS signal did not show any significant change in muscle oxygenation. However venous blood oxygen saturation was significantly reduced in normoxia, with greater reduction obtained in hypoxia. At 30% of maximum voluntary contraction, oxygenation in the muscle represented by the NIRS signal showed significant reduction in normoxia, however, this signal was not influenced during hypoxia. In contrast, blood oxygen saturation demonstrated a significant reduction in both normoxia and hypoxia.

Majority of the above studies (Costes et al. 1996, Hicks et al. 1999, McDonald et al. 1999) used continuous wave NIRS technique to determine the

validity. Recently, Harnoka et al. (2000) stressed the importance of using a timeresolved technique to demonstrate the validity of NIRS. Minimum oxygenation saturation obtained during cuff ischemia in the resting forearm muscles with time-resolved spectroscopy was $24.1 \pm 5.6\%$, and direct measurement of venous blood hemoglobin saturation was $26.2 \pm 6.4\%$. However, these authors did not report any correlation values.

Reliability of Near Infrared Spectroscopy Measurements:

The reliability of NIRS in evaluating cerebral oxygenation and blood volume during rhythmic handgrip exercise has been recently reported (Bhambhani et al. 2000b). Eight men and five women performed intermittent maximal rhythmic handgrip contractions for a period of one minute with two minutes of recovery between each trial. These authors reported significant correlations between the two trials: 0.86 for oxygenation; and 0.89 for blood volume. Totaro et al. (1998) demonstrated significant correlations for cerebral oxyhemoglobin (0.68), deoxyhemoglobin (0.76), and flow velocity (0.60) reactivity index measurements with NIRS during breathing different concentrations of carbon dioxide.

Under ischemic conditions induced by 250 mm Hg blood pressure for a period of 8 min in the vastus lateralis at rest, Bhambhani et al. (1998b) reported correlations of 0.88 to 0.96 (between sessions), and 0.95 to 0.97 (within sessions) for oxygenation measurements. Using a similar protocol for biceps brachii at rest, Maikala and Bhambhani (1999) reported correlations of 0.89 to 0.94 (between sessions), and 0.98 to 0.99 (within sessions) for oxygenation measurements. Maikala et al. (2000) established the reliability of NIRS in the lumbar erector spine muscle during maximal back extension for a period of 2 min. These authors reported significant correlation coefficients of 0.83 and 0.84 for minimum oxygenation [measured in milli volts] during sitting and standing respectively, and 0.99 for the minimum blood volume during both postures. The higher reliability of these studies emphasized the fact that with tissue

oxygenation and blood volume measured by NIRS are fairly stable over time and therefore, repeated measurements are not mandatory.

Applications of Near Infra-red Spectroscopy:

In the last fifteen years application of NIRS has increased rapidly. In this section, only few relevant applications have been reported. However, each following chapters cited specific NIRS studies pertinent to the respective tissue under investigation.

Monitoring Cerebral Oxygenation and Blood Volume

In evaluating the application of NIRS to study cerebral oxygenation levels, Simonson and Piantadosi (1996) stated that:

"near infrared spectroscopy provides a continuous noninvasive window to the central nervous system in a way not otherwise possible."

In normal, awake humans the hemodynamic trends have shown increases in both 'focal' cerebral oxygenation and blood volume trends during functional activation, suggesting an increase in the neuronal activity in the 'focal' area monitored by the NIRS sensor (Fallgatter and Strik 1998, Obrig et al. 1996, Obrig and Villringer 1997, Villringer et al. 1994). In a typical NIRS investigation on a 'focal' cerebral region, these authors showed that there is an increase in the concentration of oxy-Hb, a decrease in the cocnetration of deoxy-Hb, and an increase of total Hb (an indicative of increase in cerebral blood volume). Villringer and Dirnagl (1995) attributed such an increase in oxy-Hb molecules to arteriolar vasodilation and increase in velocity, leading to a decrease in deoxy-Hb. While clarifying such a decrease in deoxy-Hb, these authors stated:

"...if recruitment of erythrocytes into previously only plasma perfused capillaries occurs, the increase in total blood volume may counteract this tendency and prevent deoxy-Hb changes within the capillary space. The drop in deoxy-Hb may then best be explained by an increase in venous oxygenation. Hence, the interpretations of NIRS findings leads to different conclusions on the underlying hemodynamic changes depending whether

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

recruitment of erythrocyte flux is a major mechanism of cerebral blood flow increases during stimulation or not".

Using NIRS, Villringer et al. (1994) monitored cerebral oxygenation during cognitive (performing calculations) and visual tasks (observing a picture) for a duration of one minute. They evaluated the feasibility of testing whether NIRS measurements were sensitive enough to monitor changes in the brain hemodynamics and oxygenation due to physiological changes in the brain function, and also to determine its assessment during pathological changes in the brain. They used a multi-wavelength spectrometer (Hamamatsu NIRO 500) with optodes placed at a distance of 3.5 to 7 cm at various positions of the head. Cognitive and visual simulations were monitored in 17 healthy subjects and three patients. Their results showed two interesting trends. Majority of the subjects showed the usual increase in local oxygenated hemoglobin (HbO₂) and blood volume, and a decrease in deoxygenated hemoglobin (Hb), suggesting that during cognitive stimulation, the increase in cerebral blood flow is larger than the increase in cerebral oxygen consumption. Some of them also showed an increase in Hb (even though its concentration relative to HbO2 overall is decreased). This trend was also true with visual stimulation. The authors concluded that NIRS was a useful tool to assess brain function.

In a review article, Obrig and Villringer (1997) reported that most of the studies published showed an increase in blood volume in response to the stimulus applied, indicating an evidence of cortical activation. They also suggested that an increase in blood volume is usually caused by an increase in HbO₂. However, the changes observed in Hb varied between their studies. During a performance of sequential motor task by 56 subjects at 1, 2, and 3 Hz, Obrig et al. (1996) demonstrated an increase in oxygenated hemoglobin concentration. These trends increased in amplitude with increase in the frequency rate. Fallgatter and Strik (1997) demonstrated significant cerebral activation in both left and right frontal regions during a continuous cognitive performance test. Thus, using the optical properties of different tissue chromophores, NIRS seems

to be a sensitive technique for monitoring 'focal' changes in the 'activated area' of a specific cerebral region. Although there is a considerable database of cerebral NIRS measurements during various psychophysical activities (e.g. visual, cognitive, auditory, sensory motor), to date, no studies have monitored cerebral oxygenation and blood volume responses during whole-body vibration.

Skeletal Muscle Oxygenation and Blood Volume

Several studies have reported the assessment of human skeletal muscle physiology using NIRS during various physical activities. Murthy et al. (1997) demonstrated that NIRS can monitor changes in extensor carpi radialis brevis muscle oxygenation at low levels of isometric maximum voluntary contraction. Mean tissue oxygenation reduced from 100% to 47% as they reached the 50% Maximal Voluntary Contraction (MVC). In their recent work (Murthy et al. 2001), a significant correlation (r=0.78) was reported between the extensor carpi radialis brevis muscle twitch force and muscle oxygenation, suggesting force reduction in the muscle depends on the reduced muscle oxygenation. Based on the percent change in muscle oxygenation, they hypothesized that ishemia associated with a reduced muscle oxygenation (7% or greater) results in muscle fatigue. Kahn et al. (1998) studied isometric elbow flexions from 25% to 100% MVC and showed that maximal deoxygenation of the brachi radialis muscle occurs at 50% MVC and not at 100% MVC. They concluded that at 50% MVC, there is a greater activation of slow and fast oxidative muscle fibers coupled with a reduced arterial inflow, thus, resulting in maximal deoxygenation. During rhythmic handgrip contractions at 15% and 30% MVC, Boushel et al. (1998) demonstrated that deoxygenation in the forearm flexor was proportional to the %MVC.

Bhambhani et al. (1998a) established a significant exponential relationship (r = ->0.93) between muscle oxygenation and maximal oxygen uptake during arm cranking for both genders. The percent change in biceps muscle oxygenation was significantly lower in women compared to men indicating that the women deoxygenate to a greater degree than the men at the same relative exercise intensity. They concluded that the rate of decline in

percent change in muscle oxygenation for a given increase of VO_2 between 30% and 90% of the peak VO_2 was independent of exercise mode and gender. During moderate arm cranking Ogata et al. (2001) demonstrated a decrease in both oxygenated hemoglobin and muscle blood volume in the inactive vastus latearlis muscle. However, when arm cranking was combined with leg exercise, such a decrease in both variables diminished in the active vastus latearlis muscle, suggesting metabolites in the active muscles cause vasodilation and "counteract sympathetic vasoconstriction" (Ogata et al. 2001).

Jensen-Urstad et al. (1995) studied the effect of hypoxia (12% O_2 in N_2) and normoxia (21% O_2 , N_2) conditions on muscle oxygenation and metabolism in six healthy men during submaximal arm exercise. All subjects performed an arm exercise at an absolute work load (54% of their peak VO_2) for 30 min in both conditions. Results showed that increase in VO_2 at the onset of exercise was slower in hypoxic compare to normoxic conditions. In both conditions, NIRS measurements showed an initial decrease in oxygenation in the biceps and reversed at a slower rate in hypoxia (than normoxia), thus reaching steady state values at the end of exercise. The authors concluded that higher anaerobic energy production at the onset of exercise was associated with both an increase in O_2 extraction, and decrease in muscle oxygenation.

Rundell et al. (1996) studied the effect of posture on Hb/Mb desaturation in the capillary bed of five muscle groups (vastus lateralis, vastus medialis, rectus femoris, biceps femoris and gluteus maximus) during speed skating. These authors demonstrated that oxygenation was greater during low-skating position compared to upright position. Based on the lower blood volume changes during low-skating position, these authors concluded a compromised blood flow and limited oxygen delivery in the low-speed skating position. In a recent investigation, Szmedra et al. (2000) demonstrated a similar reduction in blood volumes in giant slalom during alpine skiing. Whole-body passive postural variations (from -10° to 75° in increments of 15°) in subjects restrained to a bed and changes in calf muscle oxyhemoglobin and deoxyhemoglobin values were monitored by Ferrari et al. (1997) and Binzoni et al. (1997). They showed that changes in deoxyhemoglobin and total blood volume increased as the bed angle increased and returned to base line when the bed position is changed to initial set up, thus suggesting that near infrared spectroscopy can be sensitive enough to monitor postural changes.

These above near infrared spectroscopy studies indicate that by monitoring the specified muscles during exercise, we can evaluate local muscle physiology. However, muscle oxygenation and blood volume studies in the back muscles during either occupational or exercise activities are few to date (Jensen et al. 1999, McGill et al. 2000, Maikala et al. 2000, Maronities et al. 2000, Yoshitake et al. 2001). Yoshitake et al. (2001) used simultaneous EMG, mechanomyography, and NIRS to study low-back muscle fatigue during subjects lying in the prone position on a table. They recorded all the three measurements from the erector spinae muscle at the 3rd lumbar level during back extension, and showed a rapid decrease in muscle oxygenation and blood volume suggesting anaerobic muscular activity results in the low-back muscles. However, they did not report any correlations between these three physiological measurements. Based on their results, Yoshitake et al. (2001) concluded that restriction of blood flow due to high intramuscular mechanical pressure is one of the primary factors in developing low-back muscle fatigue.

Other researchers arrived to similar conclusions as well (Jensen et al. 1999, McGill et al. 2000, Maikala et al. 2000). Jensen et al. (1999) demonstrated that during brief and prolonged submaximal trunk extension in an erect position, oxygenation in parvertebral muscles is significantly reduced, and local homeostatic adjustment over a period may assist in establishing a constant capillary oxygen tension, thus preventing a further decrease in oxygenation in the erector spinae muscles. Also, Maikala et al. (2000) showed a similar rapid decrease in oxygenation and blood volume during the first minute of maximal back extension in both sitting and standing positions. Further decreases in both blood volume and oxygenation ceased after the minute of extension. However, during recovery from maximal extension, both oxygenation and blood volume trends reached either baseline or increased further. Yoshitake et al. (2001)

reported similar observations in blood volume responses as well. During submaximal back extension in a seated posture, McGill et al. (2000) monitored both NIRS and EMG responses, and showed a similar decrease in oxygenation values.

From the above studies it is evident that near infra-red spectroscopy (NIRS) can be a reliable and valid investigative tool to study the deoxygenation of trunk muscles in various activities. During repetitive lifting, Maronitis et al. (2000) showed no significant correlation between oxygenation (from NIRS measurements) and either right (-0.09) or left erector spinae EMG values (-0.31). However, application of NIRS to study local tissue physiology during occupational activities is still at infancy.

Role of Adipose Tissue Thickness on NIRS measurements:

Absorbency of NIRS signals from the measurement site on the muscle is inversely related to the skinfold thickness (Feng et al. 2001, Homma et al. 1996, van Beekvelt et al. 2001, Yamamoto et al. 1998), implying that the sensitivity of the NIRS signals reduce with increasing adipose tissue thickness. Sensitivity also increases with increase in the source-detector distance (Feng et al. 2001). In other words, penetration depth of light is lower (thus less light absorption) if the thickness of adipose tissue is greater. van Beekvelt et al. (2001) reported that adipose tissue metabolism is lower than muscle metabolism, hence muscle oxygenation determined using NIRS is underestimated. During cuff ischemia, McCully and Hamaoka (2000) demonstrated a greater change in the physiological range in a subject with less subcutaneous fat compared to a smaller change observed in subjects with higher subcutaneous fat.

In the flexor digitorum super-ficialis muscle during rest and sustained isometric handgrip contractions, van Beekvelt et al. (2001) demonstrated a significant decrease in muscle oxygen consumption in both men and women with increase in the thickness of adipose tissue (r=-0.70, P<0.05). These authors further showed significantly lower muscle oxygenation in women than in men during both rest and submaximal isometric contractions. Additionally, these

authors found a significant but a weak correlation of 0.29 between forearm blood flow and adipose tissue thickness, suggesting an increase in blood flow with increase in the thickness of adipose tissue.

Another factor that should be considered is the role of skin blood flow in the contamination of NIRS blood volume signal. With source-to-detector separation of 40-50 mm, Hampson and Piantadosi (1988) showed that NIRS measurements were not affected by skin blood flow, and attributed such findings to a small volume of skin relative to muscle between the NIRS optodes, and the low concentration of cyctochorme oxidase in skin. Other authors also reported negligible influence of skin blood flow on NIRS signals (Mancini et al. 1994b, Piantadosi et al. 1986). Further, Feng et al. (2001) showed that for an adipose thickness of 8 mm, the optimal source-detector distance was 45 mm. These authors also identified the optimal distance of 35-40 mm for an adipose tissue thickness range of 2.5-5 mm, and 50 mm for a thickness of 17 mm.

Summary of Introduction

Since motor vehicle driving has become an integral part of life for everyone, investigating the influence of vibration dose on physiological responses in both men and women is important. Further, confounding factors such as prolonged sitting, awkward postures adopted, and frequent manual material handling associated with their job, makes whole-body vibration one of the major questionable risk factors for low-back disorders. Additionally, repetitive pushing and pulling is also considered to be one of the causal factors for workrelated musculoskeletal disorders affecting both the low-back and upper extremities. It is still common practice in occupational fitness screening programs to use standardized aerobic protocols that might not be a true representative of the specific job assigned. Thus, this doctoral research investigates the physiological responses during these two low-back intensive occupational tasks in a simulated laboratory setting. Open circuit spirometry was utilized to evaluate the whole body physiological responses while Near Infra-red Spectroscopy was applied simultaneously to examine the peripheral responses in a non-invasive manner.

Objectives:

Much of the scientific research reported in the physiological capacity of human performance during exposure to whole-body vibration and manual material handling (repetitive pushing-pulling) has concentrated more on measurements such as whole-body oxygen consumption, heart rate, ventilation volume and surface EMG studies. Investigations of central and peripheral limitations have been conducted but have not been extensively researched during either of these activities. This thesis consists of two main studies. Study #1 evaluates the whole-body physiological, tissue oxygenation and blood volume responses during seated whole-body vibration. Study # 2 compares the cardiorespiratory, local muscle oxygenation and blood volume responses during and task-specific repetitive pushing-pulling protocols.

Specific objectives of Study # 1 are to:

- evaluate the effects of whole-body vibration on the central (cardiac output, stroke volume, and heart rate) and peripheral (arterio-venous oxygen difference, change in muscle and cerebral oxygenation and blood volume) physiological responses in healthy males and females performing rhythmic handgrip exercise in different postures,
- 2. determine the relationship between vibration dose and the above physiological responses, and
- 3. investigate the role of physical fitness on the responses to whole-body vibration.

The results of this study are presented as three independent papers which are outlined below.

Paper 1 Cerebral Oxygenation and Blood Volume Responses during Exposure to Seated Whole-body Vibration

The purposes of this study were to examine the:

- (1) effects of 3 Hz, 4.5 Hz, and 6 Hz WBV doses on the relative changes in oxygenation and blood volume of the prefrontal cortex region in healthy men and women;
- (2) alterations in these responses during exposure to whole-body vibration'With' and 'Without' a backrest;
- (3) addition of rhythmic handgrip contractions on these responses during whole-body vibration; and
- (4) differences in these responses between genders.

Paper 2 Lumbar Erector Spinae Oxygenation and Blood Volume Responses during Exposure to Seated Whole-body Vibration

The purposes of this study were to examine the:

- effects of 3 Hz, 4.5 Hz, and 6 Hz vibration dose on the relative changes in oxygenation and blood volume in the lumbar erector spinae muscle in healthy men and women;
- (2) differences in these trends 'With' and 'Without' a backrest exposure to WBV;
- (3) alterations in these responses with the addition of rhythmic handgrip contractions during WBV; and
- (4) differences in these responses between genders.

Paper 3 Acute Whole-body Physiological Responses during Exposure to Seated Whole-body Vibration

The purposes of this study were to examine the effects of 3 Hz, 4.5 Hz, and 6 Hz vibration doses on the:

- (1) acute physiological responses in young healthy men and women;
- (2) changes in these physiological responses during exposure to WBV 'With' and 'Without' backrest;
- (3) alterations in these responses with addition of maximal rhythmic handgrip contractions during WBV; and

(4) differences in these responses between genders.

Specific objectives Study #2 are to:

- 1. evaluate the cardiorespiratory, muscle oxygenation and blood volume responses during standardized arm cranking and task-specific repetitive pushing and pulling,
- 2. examine the inter-relationship amongst these variables during the two protocols; and
- 3. to identify the gender differences during these two aerobic protocols.

The results of this study are presented as two independent papers which are outlined below.

Paper 4 Cardiorespiratory and Biceps Brachii Oxygenation and Blood Volume Comparisons during Arm Cranking and Task-specific Pushing-Pulling

The purposes of this study were to compare the:

- (1) whole-body physiological responses during incremental pushing-pulling (with feet stationary) and incremental arm cranking till volitional exhaustion;
- (2) oxygenation and blood volume responses of the biceps brachii muscle during both these protocols; and
- (3) differences in these responses between genders.

Paper 5 Lumbar Erector Spinae Oxygenation and Blood Volume Comparisons during Arm Cranking and Task-specific Pushing-Pulling

The purposes of this study were to compare the:

- (1) lumbar erector spinae oxygenation and blood volume trends during incremental pushing-pulling (with feet stationary) and incremental arm cranking till volitional exhaustion; and
- (2) differences in these responses between genders.

REFERENCES

- 1. Alberta Labor (1997) Occupational injury and diseases in Alberta Lost-time claims and Claim rates 1996 Survey. Alberta Labor, Edmonton, CANADA.
- 2. American Conference of Governmental Industrial Hygienists (1995-1996): Threshold Limit Values (TLV's) and Biological Exposure Indices (BEI's). ACGIH Worldwide, USA.
- 3. Andersson GBJ, Ortengen R, and Nachemson A (1977) Intradiscal pressure, intra-abdominal pressure and myoelectric back muscle activity related to posture and loading. *Clinc Ortho* 129: 156-164
- 4. Antonutto G, Girardis M, Tuniz D, and Di Prampero PE (1995) Noninvasive assessment of cardiac output from arterial pressure profiles during exercise. *Eur J Appl Physiol* 72: 18-24
- 5. Astrand P and Rodahl K (1986) Textbook of work physiology. McGraw-Hill, Inc. New York, USA.
- 6. Ayoub M and Mital A (1989) Manual Material Handling. Taylor and Francis. New York, USA.
- 7. Ayoub MM and McDaniel JW (1974) Effects of operator stance on pushing and pulling tasks. *AIIE Transactions* 185-195
- 8. Bari-Gingras G and Lortie M (1995) The handling of objects other than boxes: univariate analysis of handling techniques in a large transport company. *Ergonomics* 38(5): 905-925
- Battie MC, Bigos SJ, Fisher LD, Hansson TH, Nachemson AL, Spengler DM, Wortley MD, and Zeh JH (1989) A prospective study of the role of cardiovascular risk factors and fitness in industrial back pain complaints. Spine 14(2): 141147
- 10. Belardinelli R, Barstow TJ, Porszasz J, and Wasserman K (1995a) Change in skeletal muscle oxygenation during incremental exercise measured with NIRS. *Eur J Appl Physiol* 70: 487-492
- Belardinelli R, Georgiou D, and Barstow TJ (1995b) NIRS and changes in skeletal muscle oxygenation during incremental exercise in chronic heart failure: A comparison with healthy subjects. *G Ital Cardiol* 25: 715-724

- 12. Belardinelli R, Barstow TJ, Porszasz J, and Wasserman K (1995c) Skeletal muscle oxygenation during constant work rate exercise. *Med Sci Sports Exerc* 27: 512-519
- 13. Bennett MD, Webb R, and Withey WR (1977) Personality, performance and physiological cost during vibration. Proceedings of the Physiological Society: 75-76
- Bernanrd B and Fine LJ (1997) Musculoskeletal disorders and workplace factors. (ed) Beranrd B, US Dept. of Health and Human Services, Public Health Services, CDC, DHHS (NIOSH) Publication # 97-141, Cincinnati, OHIO, USA.
- 15. Bhambhani Y, Maikala R, Farag M, and Ting W (2000a) Simultaneous Muscle and Cerebral Oxygenation/Blood Volume trends during Incremental Cycle Exercise Med Sci Sports Exerc S247, 32:5
- 16. Bhambhani Y, Maikala R, Farag M, and Ting W (2000b) Reliability of cerebral oxygenation measurements during rhythmic handgrip contractions using near infra-red spectroscopy (Abstract). Canadian Society of Exercise Physiology Annual Scientific Conference, October 25-28, Canmore, CANADA
- 17. Bhambhani Y, Maikala R, and Buckley S (1998a) Muscle oxygenation during incremental arm and leg exercise in men and women. *Eur J Appl Physiol* 78:422-431
- Bhambhani Y, Maikala R, Jeon J, and Bell G (1998b) Reliability of tissue deoxygenation during cuff ischemia in healthy males. Annual Scientific Conference-Canadian Society for Exercise Physiology: *Abstract* page 3.
- Bhambhani Y, Buckley SM, and Susaki T (1997) Detection of ventilatory threshold using NIRS in men and woman. *Med Sci Sports Exerc* 29(3): 402-409
- 20. Biering-Sørensen F (1984) Physical measurements as risk indicators for low back trouble over a one-year period. Spine 9: 106-119
- 21. Binzoni T, Cerretelli P, Ferrari M, and Delpy D (1997) Metabolic studies of human skeletal muscle by NIRS: possible applications in space research. Int J Sports Med 18: S312-S314
- 22. Boff KR and Lincoln JE (1988) Vibration and display perception. Engineering Data Compendium: Human Perception and Performance. AAMRL, Wright-Patterson Air Force Base, OH.

- 23. Bongers PM (1990) Back disorders and Whole-body vibration at work. Unpublished Doctoral Dissertation. University of Amerstadam, Netherlands.
- 24. Bongers PM, Boshuizen HC, Hulshof CTJ, and Koemeester AP (1988) Longterm sickness absence due to back disorders in crane operators exposed to whole-body vibration. Int Arch Occup Environ Health 61: 59-64
- 25. Boshuizen HC, Bongers PM, and Hulshof CTJ (1990a) Self-reported back pain in tractor drivers exposed to whole-body vibration. Int Arch Occup Environ Health 62: 109-115
- 26. Boshuizen HC, Hulshof CTJ, and Bongers PN (1990b) Long-term sick leave and disability pensioning due to back disorders of tractor drivers exposed to whole-body vibration. Int Arch Occup Environ Health 62: 117-122
- Boushel R, Pott F, Madsen P, Radegran G, Nowak M, Quistroff B, and Secher N (1998) Musice metabolism from near infrared spectroscopy during rhythmic handgrip in humans. *Eur J Appl Physiol* 79: 41-48
- Bovenzi M and Hulshof CTJ (1999) An updated review of epidemiologic studies on the relationship between exposure to whole-boy vibration and low back pain (1986-1997). Int Arch Occup Environ Health 72: 351-365
- 29. Bovenzi M and Betta A (1994) Low-back disorders in agricultural tractor drivers exposed to whole-body vibration and postural stress. *Appl Ergon* 25(4): 231-241
- 30. Bovenzi M (1996) Low back pain disorders and exposure to whole-body vibration in the workpalce. *Sem Perinatolgy* 20(1): 38-53
- 31. Burdorf A and Sorock G (1997) Positive and negative evidence of risk factors for back disorders. Scand J Work Environ Health 23: 243-256
- 32. Buxton RB and Frank LR (1997) A model for the coupling between cerebral blood flow and oxygen metabolism during neural stimulation. *J Cereb Blood Flow Metab*17: 64-72
- 33. Cady LD, Bischoff DP, O'Connell ER, Thomas PC, and Allan JH (1979) Strength and fitness and subsequent back injuries in firefighters. J Occ Med 21(4): 269-272
- 34. Calbet JAL (2000) Oxygen content and limb blood flow. Acta Physiol Scand 2000 168: 465-472
- 35. Chaffin D, Andres RO and Garg A (1983) Volitional postures during maximal push/pull exertions in the sagittal plane. *Human Factors* 25(5): 541-550

- Chaffin DC and Andersson GBJ (1991) Occupational Biomechanics. John Wiley & Sons, Inc.
- Chance B, Dait M, Zhang C, Hamoka T, and Hagerman F (1992) Recovery from exercise induced desaturation in the quadriceps muscles of elite competitive rowers. *Am J Physiol* 262: C766-C775
- 38. Cold GE (1990) Measurement and cerebral blood flow and oxygen consumption, and the regulation of cerebral circulation. In Cerebral Blood Flow in Acute Head Injury Acta Neurochir Suppl 49:2-14
- 39. Cole SH and Withey WR (1977) Human responses to whole-body vibration of different wave forms. Proceedings of the Physiological Society: 76-77
- 40. Cornelius KM, Redfern MS, and Steiner LJ (1994) Postural stability after wholebody vibration exposure. *Int J Ind Ergon* 13: 343-351
- Costes F, Barthelemy JC, Feasson L, Busso T, Geyssant A, and Denis C (1996) Comparison of muscle near-infrared spectroscopy and femoral blood gases during steady-state exercise in humans. *J Appl Physiol* 80(4): 1345-1350
- 42. Craig BN, Congleton JJ, Kerk CJ, Lawler JM, and McSweeney KP (1998) Correlation of injury occurrence data with estimated maximal aerobic capacity and body composition in a high-frequency manual material handling task. *Am Ind Hyg Assoc J* 59:25-33
- 43. Cruz J (1993) Cerebral oxygenation monitoring and management. Acta Neurochir 59:86-90
- 44. CTD News (1998) Industry accounts for lion's share of CTD costs, cases. LRT Publications, PA, page 1, June Issue.
- 45. Damkot DK, Pope MH, Lord J, and Frymoyer JW (1984) The relationship between work history, work environment and low-back pain in men. *Spine* 9(4): 395-399
- 46. DeBlasi RA, Cope M, Elwell C, Safone F, and Ferrari M (1993) Non invasive measurement of human forearm oxygen consumption by NIRS. *Eur J Appl Physiol* 67:20-25
- DeBlasi RA, Ferrari M, Natali A, Conti G, Mega A, and Gasparetto A (1994) Noninvasive measurement of forearm blood flow and oxygen consumption by NIRS. J Appl Physiol 76: 1386-1393
- 48. Delahaye RP, Auffret R, Metges PJ, et al. (1982) Physiopathology and pathology of spinal injuries in aerospace medicine, 2nd ed. Neuilly-sur-seine,

France: NATO Advisory Group for Aerospace Research and Development. AGARD-AG-250, England: 48-53, 226-60, 276-80, 293-296

- Delpy DT, Cope M, Van Der Zee P, Arridge S, Wray S, and Wyatt J (1988) Estimation of optical pathlength through tissue from direct time of flight measurement. *Phys Med Biol* 33(12): 1233-1442
- 50. Dickinson CE (1995) Proposed manual handling: International & European Standards. *Appl Ergon* 26(4): 265-270
- Doering TJ, Resch KL, Steuernagel B, Brix J, Schneider B, and Fischer GC (1998) Passive and active exercises increase cerebral blood flow velocity in young, healthy individuals. *Am J Phys Med Rehab* 77: 490–493
- 52. Duffner LR, Hamilton LH, and Schmutz MA (1962) Effects of whole-body vibration on respiration in human subjects. *J Appl Physiol* 17(6): 913-916
- 53. Dupuis H and Zerlett G (1986) The Effects of Whole-body Vibration. Springer-Verlag.
- 54. Dupuis H (1989) Biodynamic behavior of the trunk and the abdomen during whole-body vibration. Acta Anaesthesiol Scand: Suppl 90: 34-38
- 55. Eastman Kodak (1986) Ergonomic design for people at work. Volume 2. Van Nostrand Rienhold, New York.
- 56. Edwards AD, Richardson C, Van der Zee P, Elwell C, Wyatt JS, Cope M, Delpy DT, and Reynolds EOR (1993) Measurement of Hb flow and blood flow by NIRS. *J Appl Physiol* 75: 884-1889
- 57. Esmail S, Bhambhani Y, and Brintnell S (1995) Gender differences in work performance on the Baltimore Therapeutic Equipment work simulator. *Am J Occu Therapy* 49: 405-411
- 58. Evans OM, Zerbib Y, Faria MH, and Monod H (1983) Physiological responses to load holding and load carriage. *Ergonomics* 26(2): 161-171
- 59. Fallgatter AJ and Strik WK (1998) Frontal brain activation during the Wisconsin card sorting test assessed with two-channel near-infrared spectroscopy. *Eur Arch Psychiatry Clin Neurosci* 248: 245-249
- 60. Fallgatter AJ and Strik WK (1997) Right frontal activation during the continuous performance test assessed with near-infrared spectroscopy in healthy subjects. *Neuroscience Letters*. 223(2): 89-92

- 61. Feng W, Haishu D, Fenghua T, Jun Z, Qing X, and Xianwu T (2001) Influence of overlying tissue and probe geometry on the sensitivity of a near-infrared tissue oximeter. *Physiol Meas* 22: 201-208
- 62. Ferrari M, Binzani T, and Quaresima V (1997) Oxidative metabolism in muscle. Phil Trans R Soc Lond B 352: 677-683
- 63. Fox EL, Bowers RW, and Foss ML (1988) The physiological basis for exercise and Sport. WCB Brown & Benchmark Publishers, Indiana, USA.
- Franklin BA, Hogan P, Bonzheim K, Bakalyar D, Terrien E, Gordon S, and Timmis GC (1995) Cardiac demands of heavy snow shoveling. JAMA 273(11): 880-882
- Friedman DB, Friberg L, Payne G, Mitchell JH, and Secher NH (1992) Effects of axillary blockade on regional blood flow during dynamic hand contractions. J Appl Physiol 73(5): 2120-2125
- 66. Frymoyer JW, Pope MH, Costanza MC, Rosen JC, Goggin JE, and Wilder DG (1980) Epidemiologic studies of low-back pain. *Spine.* 5(5): 419-23
- 67. Gaeuman JV, Hoover GN, and Ashe WF (1962) Oxygen consumption during human vibration exposure. *Aerospace Med* 468-474
- Gagnon M, Beaugrand S, and Authier M (1992) The dynamics of pushing loads onto shelves of different heights. Int J Ind Ergon 9: 1-13
- 69. Garcin M, Cravic JY, Vandewalle H, and Monod H (1996) Physiological strains while pushing or hauling. *Eur J Appl Physiol* 72: 478-482
- 70. Garg A, Chaffin DB, and Herrin GD (1978) Prediction of metabolic rates for manual material handling jobs. *Am Ind Hyg Assoc J* 39(8): 661-674
- Garg A and Beller D (1990) One-handed dynamic pulling strength with special reference to speed, handle-height and angles of pulling. *Int J Ind Ergon* 6: 231-240
- 72. Garg A, Rodgers SH, and Yates JW (1992) The physiological basis for manual lifting. In: Kumar S (eds). Advances in Industrial Ergonomics and Safety IV, Taylor & Francis.
- 73. Garg A (1997) Manual Material Handling The Science: In: Nordin M, Anderssson GBJ, Pope MH (eds). Musculoskeletal disorders in the workplace. Principles and Practice. Mosby, New York, pp 85-119

- 74. Gatchel RJ, Polatin PB, and Kinney RK (1995) Predicting outcome of chronic back pain using clinical predictions of psychopathology: a prospective analysis. *Health Psychol* 14(5): 415-420
- 75. Genaidy AM, Karwowski W, Guo L, Hidalgo J, and Garbu H (1992) Physical training: A tool for increasing work tolerance limits of employees engaged in manual handling tasks. *Ergonomics* 35 (9): 1081-1102
- 76. Globus M, Melamed E, Keren A, Tzivoni D, Granot C, Lavy S, Stern S (1983) Effect of exercise on cerebral circulation. *J Cereb Blood Flow Metab* 3:287-290
- 77. Graham K (1998) Program evaluation report First quarter, WCB Millard Rehabilitation Center, May 1998
- 78. Grether W F, Harris C S, Mohr G C, Nixon C W, Shalburn M Sommer H C, Thaler V H, and Reghte J H (1971) Effects of combined heat, noise and vibration stress on human performance and physiological functions. *Aerospace Med* 42(10): 1092-1097
- 79. Grether W F, Harris C S, Ohlbaum M, Sampson PA, and Guignard JC (1972) Further study of combined heat, noise and vibration stress. *Aerospace Med* 43(6): 641-645
- 80. Griffin MJ (1975) Vertical vibration of seated subjects: effects of posture, vibration level, and frequency. *Aviat Space Environ Med* 46: 269-276
- 81. Griffin MJ, Whitham EM, and Parsons KC (1982) Vibration and comfort I: translational set vibration. *Ergonomics* 26: 603-630
- 82. Griffin MJ (1990) Handbook of Human Vibration. Academic Press Limited.
- 83. Guinard JC (1985) Vibration. Chapter 15. In: Patty's Industrial Hygiene and Toxicology. Vol. 3b: Biological Responses. Wiley.
- 84. Guo HR, Tanaka S, Cameron LL, Seligman PJ, Beherens VJ, Ger J, Wild DK, and Putz-Anderson V (1995) Back pain among workers in the United States: national estimates and workers at high risk. Am J Ind Med 28: 591-602
- 85. Haldeman S (1999) Low back pain current physiologic concepts. *Neurological Clinics of North America* 17(1): 1-15
- 86. Hamoka T, Katsumura T, Murase N, Nishio S, Osada T, Sako T, Higuchi H, Kurosawa Y, Shimomitsu T, Miwa M, and Chance B (2000) Quantifiation of ischemic muscle deoxygenation by near infrared time-resolved spectroscopy. J Biomed optics 5(1): 102-105

- Hampson NB and Piantadosi CA (1988) Near infrared monitoring of human skeletal muscle oxygenation during forearm ischemia. J Appl Physiol 64(6): 2449-2457
- Hansson T, Magnusson M, and Broman H (1991) Back muscle fatigue and seated whole-body vibrations: an experimental study in man. *Clin Biomech* 6(3): 173-178
- Heistad DD and Kontos HA (1983) Cerebral Circulation. In: Handbook of Physiology. The Cardiovascular System. Peripheral Circulation and Organ Blood Flow. Bethesda, MD: Am. Physiol. Soc., 1984, sect. 2., vol. III, part 2, chapt. 27, p.137-182.
- 90. Hicks A, McGill S, and Hughson RL (1999) Tissue oxygenation by near infrared spectroscopy and muscle blood flow during isometric contractions of the forearm. Can J App Physiol 24(3): 216-230
- 91. Hollmann W, de Meirleir K, Herholz K, Heiss WD, Bittner H, Volker K, and Forssmann WG (1990) Regional cerebral blood flow, neurotransmitters, and cardiac hormonal reactions during exercise. In: International perspectives in exercise physiology. Edited by Nazar K, Human Kinetics Books, Champaign, IL, pp: 150-157
- 92. Holzschuch M, Woertzen C, Metz C, and Brawanski A (1997) Dynamic changes of cerebral oxygenation measured by brain tissue oxygen pressure and near infrared spectroscopy. *Neurol Res* 19: 242-246
- 93. Homma S, Fukunaga T, and Kagaya A (1996) Influence of adipose tissue thickness on near infrared spectroscopic signal in the measurement of human muscle. *J Biomed Opt* 1: 418-424
- 94. Hood WB, Murray RH, Urschel CW, Bower JA, and Clark JG (1966) Cardio pulmonary effects of whole-body vibration in man. *J Appl Physiol* 21(6): 1725-1731.
- 95. Hoover GN, and Ashe WF (1962) Respiratory response to whole body vertical vibration. *Aerospace Med:* 980-984
- 96. Hoozemans MJM, Van Der Beek AJ, Frings-Dresen MHW, Van Dijk JH, and Van Der Woude LHV (1998) Pushing and pulling in relation to musculoskeletal disorders: a review of risk factors. *Ergonomics* 41(6): 757-781

- 97. Hosea TM, Sheldon RS, Delatizky J, Wong MA, and Hsieh C (1986) Myoelectric analysis of the paraspinal musculature in relation to automobile driving. *Spine* 11(9): 928-936
- 98. Hulshof CTJ and van Zanten VOBA (1987) WBV and low-back pain A review of epidemiological studies. Int Arch Occup Environ Health 59: 205-220
- 99. Ide K and Secher NH (2000) Cerebral blood flow and metabolism during exercise. *Prog Neurobiol* 61: 397-414
- 100.ILO (1984) Protection of workers against noise and vibration. International Labor Office, Geneva.
- 101. Imrhan SN and Ramakrishnan U (1992) The effects of arm elevation, direction of pull and speed of pull on isokinetic pull strength. *Int J Ind Ergon* 9: 265-273
- 102. International Labor Office (1984) Protection of workers against noise and vibration in the working environment. ILO Geneva, Switzerland.
- 103. International Organization for Standardization 2631/1 (1985) Evaluation of human exposure to whole-body vibration-Part 1. General requirements. ISO Geneva, Switzerland.
- 104. International Organization for Standardization (ISO) 2631/1 (1997) Evaluation of human exposure to whole-body vibration-Part 1. General requirements. ISO Geneva, Switzerland.
- 105. Jensen BR, Jorgensen K, Hargens AR, Nielsen PK, and Nicolaisen T (1999) Physiological response to submaximal isometric contractions of the paravertebral muscles. Spine 24: 2332-2338
- 106. Jensen-Urstad M, Hallback I, and Sahlin K (1995) Effect of hypoxia on muscle oxygenation and metabolism during arm exercise in humans. *Clin Physiol* 15: 27-37
- 107. Jobsis FF (1977) Noninvasive, infrared monitoring of cerebral and myocardial oxygen sufficiency and circulatory parameters. *Science* 198:1264-1266
- 108. Johanning E (1991) Back disorders and health problems among subway train operators exposed to whole-body vibration. *Scand J Work Environ Health* 17: 414-419
- 109. Jorgensen K (1997) Human trunk extensor muscles physiology and ergonomics. Acta Physiol Scand Suppl 160: 1-58
- 110. Jorgensen LG, Nowak M, Ide K, and Secher NH (1999) Cerebral blood flow and metabolism. In: Saltin B, Secher NH, Mitchell JH, Boushel R (Eds.),

Exercise and circulation in health and disease. Human Kinetics, Champaign, IL. Pp: 105-115

- 111. Jorgensen LG (1995) Transcranial doppler ultrasound for cerebral perfusion. Acta Physiol Scand Suppl 154: 1-44
- 112. Jorgensen LG, Perko M, Hanel B, Schroder TV, and Secher NH (1992) Middle cerebral artery flow velocity and blood flow during exercise and muscle ischemia in humans. *J Appl Physiol* 72: 1123-1132
- 113. Junghanns H (1990) Clinical applications of normal biomechanical stresses on spinal function. English Language edition. Edited by Hans J Hager. Aspen Publishers, Inc., Maryland, USA.
- 114. Kahn JF, Jouanin JC, Bussiere JL, Tinet E, Avrillier S, Ollivier JP, and Monod H (1998) The isometric force that induces maximal surface muscle deoxygenation.
 Eur J Appl Physiol 78(2): 183-7
- 115. Keyex A, Ochrymowicz-Bernelmans D, and Charlier AA (1995) Induced response to hypercapnia in the two-compartment total cerebral blood volume: Influence on brain vascular reserve and flow efficiency. J Cereb Blood Flow Metab 15: 1121-1131
- 116. Khalil TM, Genaidy AM, Asfour SS, and Vinciguerra T (1985) Physiological limits in lifting. *Am Ind Hyg Assoc J* 46: 220-224
- 117. Kiellberg A (1990) Psychological aspects of occupational vibration. Scan J Work Envi Health16: 39-43
- 118. Kjellberg A and Wikstrom BO (1987) Acute effects of whole-body vibration -Stabilography and electrogastrography. *Scand J Work Envi Health* 13(3): 243-6
- 119. Kroemer KH (1974) Horizontal push and pull forces. Appl Ergon 5(2): 94-102
- 120. Kujala UM, Taimela S, Viljanen T, Jutila H, Viitasalo JT, Videman T, and Battie MC (1996) Physical loading and performance as predictors of back pain in healthy adults A 5-year Prospective study. *Eur J Appl Physiol* 73:452-458
- 121.Kumar S (1994a) The back compressive forces during maximal push-pull activities in the sagittal plane. *J Human Ergology* 23: 133-150
- 122. Kumar S (1994b) Upper body push-pull strength of normal young adults in sagittal plane at three heights. *Int J Ind Ergon* 15: 427-436
- 123. Lee KS, Chaffin DB, Herrin GD, and Waikar, AM (1991) Effect of handle height on lower-back loading in cart pushing and pulling. *Appl Ergon* 22(2): 117-123

- 124. Liira JP, Shannon HS, Chambers LW, and Haines TA (1996) Long term back problems and physical work exposures in the 1990 Ontario health Survey. *Am J Public Health* 86: 382-387
- 125. Lings S and Leboeuf -Yde C (2000) Whole-body vibration and low back pain: a systematic, critical review of the epidemiological literature 1992-1999. Int Arch Occup Environ Health 73: 290-297
- 126. Lundstrom R, Holmlund P, and Lindberg L (1998) Absorption of energy during vertical WBV exposure. J Biomech 31: 317-326
- 127. Luoma K, Riihimaki H, Raininko R, Luukkonen R, Lamminen A, and Viikari-Juntura E (1998) Lumbar disc degeneration in relation to occupation. Scand J Work Environ Health 24(5): 358-366
- 128. McAllister RM (1997) Adapatations in control of blood flow with training: splanchnic and renal blood flows. *Med Sci Sports Exerc* 30(3): 375-381
- 129. McArdle WD, Katch FI, and Katch VL (1996) Exercise Physiology. Fourth Edition, Williams & Wilkins, Baltimore
- 130. McCoy CE, Hadjipavlou AG, Overman T, Necessary JT, and Wolf C (1997) Work-related low back injuries caused by unusual circumstances. *J Ortho Sport Phys Ther* 26(5): 260-265
- 131. McCully KK and Hamaoka T (2000) Near-infrared spectroscopy: what can it tell us about oxygen saturation in skeletal muscle? *Exerc Sport Sci Rev* 28: 123-127
- 132. McGill SM, Hughson RL, and Parks K (2000) Lumbar erector spinae oxygenation during prolonged contractions: implications for prolonged work. *Ergonomics* 43(4): 486-493
- 133. McLeod RW and Griffin MJ (1994) Mechanisms of vibration-induced interference with manual control performance. *Ergonomics* 38(7): 1431-1444
- 134. Maassen N and Schneider G (1997) Mechanism of fatigue in small muscle groups. Int J Sports Med 18(4): S320-S321
- 135. McDonald MJ, Tarnopolsky MA, Green HJ, and Hughson RL (1999) Comparison of blood gases and muscle near-infrared spectroscopy at exercise onset in humans. *J Appl Physiol* 86(2): 687-693
- 136. Mack K, Haslegrave CM, and Gray MI (1995) Usability of manual handling aids for transporting materials. *Appl Ergon* 26(5): 353-364

- 137. Magnusson M, Pope MH, Lindell V, Svensson M, and Andersson G (1987) The metabolic cost of occupational exposures. Ergonomics 30 (1): 55-60
- 138. Maikala R, Farag M, and Bhambhani Y (2000) Low-back muscle oxygenation trends during simple postural variations – Intersession reliability using NIRS. Proceedings of International Ergonomics Conference, San Diego, USA: pp 5.9-5.12
- 139. Maikala R and Bhambhani Y (1999a) Biceps brachii deoxygenation during cuff ischemia and recovery: Intra and inter session reliability. Med Sci Sport Exerc S245 31:5 (Abstract)
- 140. Maikala R and Bhambhani Y (1999b) Muscle Oxygenation Trends using Near Infrared Spectroscopy (NIRS). In: Starker L and Pollock C (Eds.,). CD-ROM Proceedings of CybErg 1999. The Second International Conference on Ergonomics. The International Ergonomics Association Press, Curtin University of Technology, Perth, Australia, pp.542-557
- 141. Mancini, DM, Wilson JR, Bolinger L, Li H, Kendrick K, Chance B, and Leigh JS (1994a) In Vivo Magnetic Resonance Spectroscopy Measurement of deoxymyoglobin during exercise in patients with heart failure. *Circulation* 90(1): 500-508
- 142. Mancini, DM, Bolinger L, Li H, Kendrick K, Chance B, and Wilson JR (1994b) Validation of near-infrared Spectroscopy in Humans. *J Appl Physiol* (77): 2740-2747
- 143. Manninen O (1984) Hearing threshold and heart rate in men after repeated exposure to dynamic muscle work, sinusoidal Vs stochastic whole body vibration and stable broad band noise. *Int Arch Occup Environ Health* 54: 19-32
- 144. Manninen O (1985) Cardiovascular changes and hearing threshold shifts in men under complex exposures to noise, whole-body vibrations, temperature and competition type psychic load. *Int Arch Occup Environ Health* 56: 251-274
- 145. Maronitis AB, Kovacs KM, Splittstoesser RE, and Marras (2000) The effectiveness of whole-body and localized measures of fatigue. Proceedings of the IEA 2000/HFES 2000 Congress, San Diego, USA, pp 5-91-5-94
- 146. Mchedlishvili G (1986) Cerebral arterial behavior providing constant cerebral blood flow, pressure, and volume. In: *Arterial behavior and blood circulation in the brain* (Bevan JA, ed), Plenum Press, New York, pp 55-60

- 147. Miedema MC, Douwes M, and Dul J (1997) Recommended maximum holding times for prevention of discomfort of static standing postures. Int J Ind Ergon 19: 9-18
- 148. Millikan GA (1937) Experiments on muscle hemoglobin in vivo: the instantaneous measurement of muscle metabolism. *Proc R Soc Lond B* 123: 218-241
- 149. Mital A and Ayoub MM (1981) Effect of task variables and their interactions in lifting and lowering loads. *Am Ind Hyg Assoc* J 42: 134-142
- 150. Miyashita K, Morioka I, Tanabe T, Iwata H, and Takeda S (1992) Symptoms of construction workers exposed to whole-body vibration and local vibration. Int Arch Occup Environ Health 64: 347-351
- 151. Mohler LR, Styf JR, Pedowitz RA, Hargens AR, and Gershuni DH, (1997) Intramuscular deoxygentaion during exercise in patients who have chronic anterior compartment syndrome of the leg. *J Bone Joint Surgery* 79-A(6): 844-849
- 152. Murthy G, Kahan NJ, Hargens AR, and Rempel DM (1997) Forearm muscle oxygenation decreases with low levels of voluntary contraction. *J Ortho Res* (15): 507-511
- 153. Murthy G, Hargens AR, Lehman S, and Rempel DM (2001) Ischemia causes muscle fatigue. *J Ortho Res* 19: 436-440
- 154. Narayan SM, Esfahani P, Blood AJ, Sikkens L, and Toga AW (1995) Functional increase in cerebral blood volume over somatosensory cortex. *J Cereb Blood Flow Metab* 15: 754-765
- 155. Nindl BC, Sharp MA, Mello RP, Rice VJ, Murphy MM, and Patton JF (1998) Gender comparison of peak oxygen uptake: repetitive box lifting versus treadmill running. *Eur J Appl Physiol* 77: 112-117
- 156. Nioka S, Moser D, Lech G, Evengelisti M, Verde T, Chance B, and Kuno S (1998) Muscle deoxygenation in aerobic and anaerobic exercise. *Adv Exp Med Biol* 454:63-70
- 157.NIOSH (1981) Work practices guide for manual lifting. National Institute for Occupational Safety and Health. Technical Report: 81-122
- 158. NIOSH Technical Report (1990) Work practices guide for manual lifting Draft, 2nd Edition. US Dept. Of Health & Human Services.

- 159.NIOSH (1991) A work practices guide for manual lifting. US Dept. Of Health & Human Services, Technical Report: 81-122
- 160. Obrig H and Villringer A (1997) Near-infrared spectroscopy in functional activation studies. In: Villinger and Dirnagh (eds) Optical Imagery of Brain Function and Metabolism II. Plenum Press, New York.
- 161. Obrig H, Hirth C, Junge-Hulbsing JG, Doge C, Wolf T, Dirnagl U, and Villringer A (1996) Cerebral oxygenation changes in response to motor stimulation. J Appl Physiol 81(3): 1174-1183
- 162. Ogata H, Yunoki T, and Yano T (2001) Effect of arm cranking on the NIRSdetermined blood volume and oxygenation of human inactive and exercising vastus lateralis muscle. *Eur J Appl Physiol* 2001DOI 10.1007/s00421-001-0527-7
- 163. Olsen J (1971) Contralateral focal increase of cerebral blood flow in man during arm work. *Brain* 94: 635-646
- 164. Palmer K, Griffin MJ, Bendall H, Pannett B, and Coggon D (2000) Prevalence and pattern of occupational exposure to whole body vibration in Great Britain: findings from a national survey. *J Occup Environ Med* 57: 229-236
- 165. Panjabi MM, Andersson GBJ, Jorneus L, Hult E and Mattsson L (1986) In vivo measurement of spinal column vibrations. *J Bone Joint Surg* 68A(5): 695-703
- 166. Petrofsky JS and Lind AR (1978) Comparison of metabolic and ventilatory responses of men to various lifting tasks and bicycle geometry. *J Appl Physiol* 45: 60-63
- 167. Piantadosi CA, Hemstreet TM, and Jobsis-Vandervliet FF (1986) Near-Infrared spectrophotometric monitoring of oxygen distribution to intact brain and skeletal muscle tissues. *Crit Care Med* 14(8): 698-706
- 168. Piantadosi CA (1993) Absorption spectroscopy for assessment of mitochondrial function in vivo. *Methods in Toxicology* vol. 2: Academic Press, pp.107-126
- 169. Pollard V, Prough DS, DeMelo AE, Deyo DJ, Uchida T, and Stoddart HF (1996) Validation in volunteers of a near-infrared spectroscope for monitoring brain oxygenation in vivo. *Anesth Analg* 82(2): 269-77
- 170. Pope MH, Magnusson M, and Wilder D (1998) Kappa Delta Award Low back pain and whole-body vibration. *Clin Orthop Research* 354: 241-248
- 171. Pope MH, Broman H, and Hansson T (1990a) Factors affecting the dynamic response of the seated subject. *J Spinal Disorders* 3: 135-142

- 172. Pope MH, Magnusson M, Lindell V, Svensson M, and Andersson GBJ (1990b). The measurement of oxygen uptake under whole-body vibration. *Iowa Orthop J* 10: 85-88
- 173. Pope M (1989) Risk indicators in low-back pain. Annal Med 21(5): 387-392
- 174. Pope MH, Broman H, and Hansson T (1989) The dynamic response of a subject seated on various cushions. *Ergonomics* 32(10): 1155-1166
- 175. Pope MH, Wilder DG, Jorneus L, Broman H, Svensson M, and Andersson GBJ (1987) The response of the seated human to sinusoidal vibration and impact. *J Biomech Eng* 109: 279-284
- 176. Pittman RN (2000) Oxygen supply to contracting skeletal muscle at the microcirculatory level: diffusion Vs convection. *Acta Physiol Scand* 168: 593-602
- 177. Pott F, Jensen K, Hansen H, Christensen NJ, Lassen NA, and Secher NH (1996) Middle cerebral artery blood velocity and plasma catecholamines during exercise. Acta Physiol Scand 158: 349-356
- 178. Punwani S, Ordidge RJ, Cooper CE, Amess P, and Clemence M (1998) MRI measurements of cerebral deoxyhaemoglobin concentration [dHb]-correlation with near infrared spectroscopy (NIRS). *NMR Biomed* 11(6): 281-9
- 179. Putz-Anderson V (1990) Cumulative trauma disorders. A manual for musculoskeletal diseases of the upper limbs. Taylor & Francis, New York.
- 180. Quaresima V, Sacco S, Totaro R, and Ferrari M (2000) Noninvasive measurement of cerebral hemoglobin oxygen saturation using two near infrared spectroscopy approaches. *J Biomed Optics* 5: 201-205
- 181. Raichle ME, Grubb RL, Gado MH, Eichling JO, and Ter-pogossian MM (1976) Correlation between regional cerebral blood flow and oxidative metabolism. *Arch Neurol* 33: 523-526
- 182. Ramos D, Schoenmann L, Scott D, and Trent J (1996) The effects of vibration on workers. Work 6: 127-132
- 183. Resnick M and Chaffin D (1995) An ergonomic evaluation of handle height and load in maximal and submaximal cart pushing. *Appl Ergon* 20: 173-178
- 184. Roland PE and Larsen B (1976) Focal increase of cerebral blood flow during stereognostic testing in man. Arch Neurol 33: 551-558
- 185. Roland PE (1984) Metabolic measurements of the working frontal cortex in man. *Trends Neurosci:* 430-435

- 186. Roland PE (1985a) Application of imaging of brain blood flow to behavioral neurophysiology: The cortical field activation hypothesis. In Brain Imaging and Brain Function, L. Sokoloff, ed., pp. 87-104, Raven, New York.
- 187. Roland PE (1985b) Cortical organization of voluntary behavior in man. *Human* Neurobiol 4: 155-167
- 188. Roland PE (1993) Brain Activation. Wiley-Liss, New York, pp. 105-140
- 189. Rowell LB (1988) Muscle blood flow in humans: how high can it go. Med Sci Sports Exerc 20: S97-S107
- 190. Rowell LB (1984) Cardiovascular adjustments to thermal stress. In: Handbook of Physiology. The Cardiovascular System. Peripheral Circulation and Organ Blood Flow. Bethesda, MD: Am. Physiol. Soc., 1984, sect. 2., vol. III, part 2, chapt. 27, p.967-1023.
- 191. Rundell KW, Nioka S, and Chance B (1997) Hemoglobin/ Myoglobin desaturation during speed skating. *Med Sci Sports Exerc* 29: 248-258
- 192. Sahlin K (1992) Non-invasive measurements of oxygen availability in humans skeletal muscle with near-infrared spectroscopy. Int J Sports Med 13: 5157-5160
- 193. Sako T, Hamaoka T, Higuchi H, Kurosawa Y, and Katsumura T (2001) Validity of NIRS spectroscopy for quantitatively measuring muscle oxidative metabolic rate in exercise. *J Appl Physiol* 90: 338-344
- 194. Sanchez J, Monod H, and Chabaud F (1979) Effects of Dynamic, Static and Combined work on Heart Rate and Oxygen Consumption. *Ergonomics* 8: 935-943
- 195. Scutter S, Turker KS, and Hall R (1997) Headaches and neck pain in farmers. Aust J Rural health 5: 2-5
- 196. Seidel H (1993) Selected health risks caused by long-term, whole-body vibration. *Am J Ind Med* 23: 589-604
- 197. Seidel H and Heide R (1986) Long-term effects of whole-body vibration. A critical survey of the literature. Int Arch Occup Environ Health 58: 1-29
- 198. Seidel H, Bluethner R, and Hinz B (1986) Effects of sinusoidal WBV on the lumbar spine: the stress-strain relationship. Int Arch Occup Environ Health 57: 207-233
- 199. Seroussi RE, Wilder DG, and Pope MH (1989) Trunk Muscle Electromyography and Whole Body Vibration. J. Biomech 22: 219-229

- 200. Sibbald WJ, Messmer K, and Fink MP (2000) Roundtable conference on tissue oxygenation in acute medicine, Brussels, Belgium, 14-16 March 1998. Int Care Med 26: 780-791
- 201. Sharp M (1994) Physical fitness and occupation performance of women in the US Army. Work 4(2): 80-92
- 202. Sharp MA, Harman E, Vogel JA, Knapik JJ, and Legg SJ (1988) Maximal aerobic capacity for repetitive lifting: Comparison with three standard exercise testing modes. *Eur J Appl Physiol* 57:753-760
- 203. Simon-Arndt, Yuan, H, and Hourani LL (1997) Aircraft type and diagnosed back disorders in US Navy pilots and aircrew. Aviat Space Environ Med 68(11): 1012-1018
- 204. Simonson SG and Piantadosi CA (1996) NIRS: Clinical Applications. Crit Care Clin 12(4): 1019-1029
- 205. Singh M (1991) Army physical fitness standards for field units. Final Report to Canadian Military. Faculty of Physical Education and Recreation. University of Alberta. Pages 571
- 206. Smith KC (1977) New topics in Photobiology. In: Smith KC (eds) The Science of Photobiology. Plenum Press, New York, pp 397-417
- 207. Snook SH and Ciriello VM (1991) The Design of Manual Handling Tasks: Revised Tables of Maximum Acceptable Weights and Forces. *Ergonomics* 34(9): 1197-1213
- 208. Søgarrd K, Fallentin N, and Nielsen J (1996) Work load during floor cleaning. The effects of cleaning methods and work technique. *Eur J Appl Physiol* 73: 73-81
- 209. Sokolff L, Mangold R, Wechsler RL, Kennedy C, and Kety S (1955) The effect of mental activity on cerebral circulation and metabolism. *J Clin Invest* 34: 1101-1108
- 210. Statistics Canada (1994) Work Injuries, Cat No: 72-708. Statistics Canada.
- 211. Szmedra L, Im J, Nioka S, Chance B, and Rundell KW (2001) Hemoglobin/myoglobin oxygen desaturation during alpine skiing. *Med Sci* Sports Exerc 33(3): 232-236
- 212. Taylor DE and Simonson SG (1996) Use of near-infrared spectroscopy to monitor tissue oxygenation. New Horizons 4(4): 420-425

- 213. Thalheimer E (1996) Practical exposure to measurement and evaluation of exposure to whole-body vibration in the workplace. Sem Perinat 20: 77-89
- 214. Totaro R. Barattelli G. Quaresima V. Carolei A, and Ferrari M (1998) Evaluation of potential factors affecting the measurement of cerebrovascular reactivity by near-infrared spectroscopy. *Clini Sci* 95(4): 497-504
- 215. Tran TK, Saliasuta N, Kreutzer U, Hurd R, Chung Y, Mole P, Kuno S, and Jue T (1999) Comparative analysis of NMR and NIRS measurements of intracellular PO2 in human skeletal muscle. *Am J Physiol* 276 (45): R1682-R1690
- 216. van Beekvelt MCP, Borghuis MS, van Engelen BGM, Wevers RA, and Colier WNJM (2001) Adipose tissue thickenss affects in vivo quantitative near-IR spectroscopy in human skeletal muscle. *Clin Sci* 101: 21-28
- 217. Villringer A, Planck J, Hock C, Schleinkofer L, and Dirnagl U (1993) NIRS: a new tool to study hemodynamic changes during activation of brain function in human adults. *Neurosci Lett* 154: 101-104
- 218. Villriger A, Planck J, Stodieck S, Botzel K, Schleinkofer L, and Dirnagl U (1994) Noninvasive assessment of cerebral hemodynamics and tissue oxygenation during activation of brain cell function in human adults using NIRS. *Adv Exp Med Biol* 345: 559-565
- 219. Villringer A and Dirnagl U (1995) Coupling of brain activity and cerebral blood flow: basis of functional neuroimaging. *Cerebrovascular and Brain Metabolism Reviews* 7: 240-276
- 220. Villringer A and Chance B (1997) Noninvasive optical spectroscopy and imaging of human brain function. *Tech Neurosci* 20(10): 435-442
- 221. von Gierke HE and Goldman DE (1988) Effects of shock and vibration on man. In: Harris, CM (Eds)., Shock and Vibration Hand Book, Pages 44-1 – 44-56.
- 222. Wagner PD (1988) The determinants of VO2max. Ann Sports Med 4(4): 196-202
- 223. Wahl M and Schilling L (1993) Regulation of cerebral blood flow- a brief review. Acta Neurochir Suppl 59: 3-10
- 224. Wardle SP, Yoxall WC, Weindling MA (2000) Determinants of cerebral fractional oxygen extraction using Near Infrared Spectroscopy in preterm neonates. J Cereb Blood Flow Met 20:272-279

- 225. Warwick D, Novak G, and Schultz A (1980) Maximum voluntary strengths of feamle adults in some lifting, pushing and pulling activities. *Ergonomics* 23 (1): 49-54
- 226. Wasserman DE, Wilder DG. Pope MH, Magnusson M, Aleksiev AR, and Wasserman JF (1997) Whole-body vibration exposure and occupational workhardening [Review] J Occ Envi Med 39(5): 403-407
- 227. Waters TR, Putz-Anderson V, Garg A, and Fine LJ (1993) Revised NIOSH equation for the design and evaluation of manual lifting tasks. *Ergonomics* 36:749-776
- 228. Waters TR, Putz-Anderson V, and Garg A (1994) Applications Manual for the Revised Lifting Equation. USDHHS (NIOSH) Publication No. 94-110
- 229. Webb RPG, Bennett MD, Farmillo B, Cole SH, Page SJ, and Withey WR (1981) Personality and inter-subject differences in performance and physiological cost during whole-body vibration. *Ergonomics* 24: 245-255
- 230. Wenger NK (1991) Ability, disability, and the functional capacity of patients with cardiovascular disease. *Transactions of the Association of life insurance medical directors of America* 74: 78-91
- 231. Wikstrom BO, Kjellberg A, and Landstrom U (1994) Health effects of long-term occupational exposure to WBV: a review. *Int J Ind Ergon* 14: 273-292
- 232. Wilder D, Magnusson M, Fenwick J, and Pope M (1994) The effect of posture and seat suspension design on discomfort and back muscle fatigue during simulated truck driving. *Appl Ergon* 25(2): 66-76
- 233. Wilder DG (1993) The biomechanics of vibration and low back pain. *Am J ind Med* 23: 577-588
- 234. Wilder DG, Pope MH, and Frymoyer JW (1988) The biomechanics of lumbar disc herniation and the effect of overload and instability. *J Spinal Disord* 1: 16-32
- 235. Wilder D, Woodworth BB, Frymoyer JW, and Pope MH (1982) Vibration and the human spine. *Spine* 7(3): 243-254
- 236. Wilson JR, Mancini DM, McCully KK, Ferrato N, Lanoce V, and Chance B (1989) Noninvasive detection of skeletal muscle underperfusion with nearinfrared spectroscopy in patients with heart failure. *Circulation* 80: 1668-1703

- 237. Yassi A, Khokhar J, Tate R, Cooper J, Snow C, and Vallentyne S (1995) The epidemiology of back injuries in nurses at a large canadian tertiary care hospital: implications for prevention. *Occ Med* 45(4): 215-20
- 238. Yoshitake Y, Ue H, Miyazaki M, and Moritani T (2001) Assessment of lowerback muscle fatigue using electromyography, mechanomyography, and nearinfrared spectroscopy. *Eur J Appl Physiol* 84: 174-179
- 239. Zagorski J, Jakubowski R, Solecki L, Sadeo A, and Kasperek W (1976) Studies on the transmission of vibrations in human organsim exposed to low-frequency whole-body vibration. *Acta Physiol Pol* 27: 347-354
- 240. Zimmerman CL, Cook TM, and Goel VK (1993) Effects of seated posture on erector spinae EMG activity during whole body vibration. *Ergonomics* 36(6): 667-675
- 241.Zobl GE, Talmers FN, Christensen RC, and Baer LJ (1965) Effect of Exercise on the Cerebral Circulation and Metabolism. *J Appl Physiol* 20: 1289-1293



Load Increments 0.5 Kg every 2 minutes







BL - Baseline Rest for 2 minutes WU - Warmup at No Load for 2 minutes AR - Active Recovery for 2 minutes PR - Passive Recovery for 2 minutes

Load Increments 25 Watts every 2 minutes

For example, based on the above figure:

y = oxygen saturation at 75 Watts of workload = -100 mV

M = minimum saturation or maximum deoxygenation at the final workload

m = maximum saturation during recovery = 10 mV

Percent muscle oxygenation at 75 Watts of workload

$$= (y - M)/(m - M)$$

= $[(-100) - (-130)]/[10]$

= [(-100) -(-130)]/[10-(-130)]

= 21%
CHAPTER 2

Cerebral Oxygenation and Blood Volume Responses during Exposure to Seated Whole-body Vibration

Whole-body vibration (WBV) is vibration transmitted to the entire human body from a vibrating surface such as the seat in a heavy equipment vehicle, truck, bus or aircraft. Thus, humans are exposed to WBV in a variety of occupations. The human response to WBV depends on the source (vibrating base, vehicle dynamics), interface (such as a seat cushion) and the characteristics of the individual. The most dominant frequency dose identified in the motorized vehicles ranges from 3 to 6 Hz (Wilder et al. 1982), with the transfer of maximum mechanical energy to the spine resulting at about 4.5 to 5 Hz (Panjabi et al. 1986, Pope et al. 1998, Wilder et al. 1982).

It has been demonstrated that the transmissibility of vibration from the buttocks to the head in a seated posture increases in an exponential manner (Wilder et al. 1982). Minor variations in the orientation of the lower back and the angle of the head can cause substantial changes in vibration transmitted though the spine to the head (Griffin 1990). Vertical transmission to the head is slightly greater in women than in men at frequencies > 5 Hz with the trend being reversed at frequencies < 5 Hz (Griffin et al. 1982). Also, leaning against a backrest ['With' Backrest] increases the transmission of vibration to the head when compared to the response 'Without' a backrest support (Boff and Lincoln 1988, Magnusson et al. 1992, Paddan and Griffin 1988a).

Compared to exposure to WBV alone, physical factors such as handling controls, static work, prolonged sitting, attention, mental stress, in association with environmental factors such as noise, light, temperature, etc. might have additive or variable effects on WBV exposed populations (Manninen 1988, Seidel et al. 1988). However, the role of physical work (e.g., handgrip) on cerebral activity during WBV, and its influence on human performance has not been investigated. Hence, it is necessary to study regional cerebral oxygenation and blood volume responses due to rhythmic handgrip contractions during exposure to WBV to more accurately simulate the occupational conditions. According to Bovenzi (1996), comparison studies on vibration exposure and physiological responses during seated WBV between genders are limited. Additionally, Boff and Lincoln (1988) and Griffin (1990) suggested that physical fitness may play an important role in minimizing the effects of the WBV on human performance. However, to date, no studies have objectively tested this hypothesis.

Near infra-red spectroscopy (NIRS) is a non-invasive optical technique capable of measuring changes in oxygenation and blood volume in the brain (Chance et al. 1988, Colier et al. 1999, Fallgatter and Strik 1998, Ide and Secher 2000, Villringer et al. 1994). Light in the near infra-red region is differentially absorbed by the oxygenated and deoxygenated forms of hemoglobin in the tissue. NIRS identifies tissue oxygenation and blood volume changes continuously in real-time during rest and work. Tissue oxygenation, defined as the relative saturation of oxyhemoglobin, depends on the balance between oxygen delivery and localized metabolic rate (Simonson and Piantadosi 1996). NIRS indicates the relative change in tissue oxygenation within the illuminated area by measuring the difference in tissue absorbency between the oxygenated and deoxygenated states of hemoglobin in the near infra-red range (700 to 1000nM). Sum of these two states of hemoglobin reflects the changes in the cerebral blood volume of the illuminated region of the tissue.

Most NIRS instrumentation uses a continuous wave technique to evaluate relative changes in tissue oxygenation and blood volume. Absolute changes in tissue oxygenation and blood volume cannot be quantified using continuous wave NIRS because the optical path length of the photons is difficult to determine. The validity of NIRS in evaluating changes in cerebral oxygenation and blood volume has been well established in humans (Holzschuch et al. 1997, Pollard et al. 1996, Punwani et al. 1998, Quaresima et al. 2000). Correlation of brain tissue oxygen pressure (measured in the frontal white matter with a Clarktype electrode) and hemoglobin oxygenation (on the forehead using NIRS) were studied in severe head injury patients (Holzschuch et al. 1997). These authors showed that under the conditions of a stable NIRS signal and a diffuse brain lesion, changes of oxygen pressure were significantly correlated with NIRS (r > 0.7). Cerebral oxygen saturation was also validated against jugular venous saturation (r=0.88-0.99) in healthy volunteers during hypoxia (Pollard et al. 1996).

Measured tissue oxygenation and calculated cerebral venous oxygen saturation were compared in healthy volunteers during partial venous occlusion (Quaresima et al. 2000). They showed that oxygen saturation measured at the right and left forehead using the spatial NIRS technique reflects mainly the intracranial venous compartment saturation (r=0.76). The reliability of this technique in evaluating cerebral oxygenation and blood volume during rhythmic handgrip exercise has been recently reported (Bhambhani et al. 2000). To date, cerebral functional activity research using NIRS has been limited to cognitive demand, motor activation, and visual stimulation studies (Colier et al. 1999. Fallgatter and Strik 1998, Hock et al. 1997, Hoshi and Tamura, 1993, Obrig et al. 1996, Obrig and Villringer 1997, Villringer et al. 1994). Nakamura et al. (1988) studied the effect of WBV on cerebral metabolites in rats and demonstrated that vibration activates the pituitary adrenal system, thus causing an imbalance in the homeostasis of the organism. However, research pertaining to changes in cerebral oxygenation and blood volume during WBV in humans has not been conducted.

The purposes of this study were to examine the effects of discrete 3 Hz, 4.5 Hz, and 6 Hz vibration doses on the: (1) accelerations at the sixth cervical region, 'With' and 'Without' a backrest in healthy young men and women; and (2) relative changes in oxygenation and blood volume of the prefrontal cortex region under the above conditions with and without the addition of rhythmic handgrip contractions. It was hypothesized that cerebral oxygenation and blood volume responses will be similar in both men and women 'With' and 'Without' backrest, and will not be influenced by the exposure to increase in vibration dose from 3 to 6 Hz during rhythmic handgrip contractions. Similarly, it was hypothesized that during exposure to vibration doses within the range of 3 to 6 Hz, accelerations generated at the cervical region will be similar in both men and women 'With' and 'Without' backrest.

MATERIALS AND METHODS

Subjects

Written informed consent was obtained from thirteen healthy men (age 24.7 ± 3.9 yrs, mass 70.8 ± 12.1 kg, height 171 ± 7 cm) and fourteen healthy women (age 23.9 ± 3.51 yrs, mass 60.3 ± 9.0 kg, height 164 ± 10 cm). Subjects were undergraduate and graduate students recruited from the university population (Table 1). They all completed the Revised Physical Activity Readiness Questionnaire (Canadian Society for Exercise Physiology, 1994) to identify contraindications for exercise. All subjects were right hand dominant. The test protocols utilized were approved by the Human Ethics Review Board at this institution (APPENDIX A.1).

Vibrating Base

A steel vibrating base that housed an electro-mechanical motor connected to a cam mechanism (Advanced Therapy Products, Inc., VA, USA) was modified with an external DC speed control device. Varying the speed of this motor from 0 to 100% corresponded to vibrating frequencies of 0 Hz to 6.2 Hz. These frequencies were verified with an oscilloscope (Hitachi Digital Oscilloscope VC-6050, JAPAN). The vibrating base (Figures 2.1A and 2.1B) was also attached with a displacement transducer (Intertechnology, Inc., ON, CANADA) to measure the cam linear displacement during vibration. The average linear displacement with a subject sitting on the base during vibration was 6.1 mm at 3 Hz, 6.4 mm at 4.5 Hz and 6.6 mm at 6 Hz. The vibrating base was fitted with a seat that had an internal spring-suspension system with a weight-adjustable knob. This seat was adjustable to three seating positions: upright, 3° and 6°.

Placement and Calibration of Accelerometers

Two tri-axial silicon micro-machined accelerometers (Crossbow, Inc., CA, USA) were used to measure the acceleration during vibration: one was glued on

to the vibrating base, and the second one was attached on the subject's neck at the level of the sixth cervical vertebrae (C6). These accelerometers were interfaced with an analog/digital board and customized software was used to convert the acceleration from volts to 'g' units. Accelerations values were then transformed into units of m/sec^2 (1g = 9.81 m/sec²) at each interface (base and C6), and were converted into root mean square values over the total vibration period.

Each accelerometer (for each axis) was placed on a horizontal surface that corresponded to +1g to obtain a more positive reading on a voltmeter (FLUKE, USA) than the zero-g voltage. It was then rotated by 180° so as to correspond to -1g resulting in a more negative reading on the voltmeter than the zero-g voltage. The regression equations between voltage output and 'g' units in the Z-axis for the two accelerometer positions were as follows: (1) Head accelerometer = 1.998 volts/g * (output voltage -2.492); and (2) Base accelerometer = 2.004 volts/g * (output voltage -2.544).

Test Protocols

Each subject completed an arm cranking exercise test in the first session to determine peak aerobic capacity, followed by three WBV tests with each dose (3 Hz, 4.5 Hz, and 6 Hz) on a separate day in a randomized order.

(1) Assessment of the Peak Oxygen Uptake

In the first testing session, the subjects completed an aerobic fitness test on an arm cranking ergometer (Cybex, MET 300, USA). The test was initiated with a two-minute rest period. It was followed by two minutes of arm cranking at no load and subsequent increments of 25 watts every 2 minutes until voluntary exhaustion, or attainment of two or more of the following end points suggested by the American College of Sports Medicine (1993): (1) leveling off in the oxygen consumption (increase of \leq 100 ml.min⁻¹) with increasing load, (2) age predicted maximal heart rate, equated to 220-age (in years), (3) respiratory exchange ratio of \geq 1.10, and (4) rate of perceived exertion \geq 18 (Borg, 1982).

(2) Vibration Tests

The test protocol on each day lasted 30 minutes, and included initial sixminute rest without vibration, eight-minute vibration 'With' (or 'Without') a backrest, four-minute recovery, eight-minute vibration 'Without' (or 'With') a backrest, and four-minute recovery. The subject was randomly assigned to a posture 'With' or 'Without' a backrest (Figures 2.4A and 2.5), and was instructed to adopt a comfortable posture when tested 'Without' a backrest. Throughout the test, subjects sat in the vibrating chair with the eyes open. A footrest was provided for subjects to place their feet at a comfortable level. During sitting 'With' and 'Without' a backrest, angles at the knee joint and low-back were measured by a goniometer (JAMAR, Clifton, NJ, USA).

(3) Rhythmic Handgrip Measurements

Before the vibration session, right hand grip size was adjusted on the dynamometer (JAMAR, CA, USA) to a position that was comfortable for each subject. During the fifth minute of each vibration condition, the subject performed rhythmic maximal contractions for one minute at a contraction rate of one every five seconds. Contractions were performed with the right hand with the forearm in a horizontal position and the elbow at an angle of 180° (Figures 2.4B). Similar to a method suggested by Schmidt and Toews (1970), throughout the study the dynamometer was calibrated with known weights to check its accuracy identified through the dial reading on the dynamometer.

(4) Oxygenation and Blood Volume Measurements

A continuous dual wave NIRS unit (MicroRunman, NIM Inc., PA, USA) was used to evaluate relative changes in cerebral oxygenation and blood volume during the vibration tests. This unit consists of: a superficial probe consisting of a tungsten lamp placed at a distance of 4 cm from two silicone diodes which absorb light at 760nm and 850 nm; a display unit that amplifies the absorbency signal generated (Figure 2.2). Before calibration, the sensor was placed on the anterior right frontal lobe approximately 2-4 cm from the mid-line of the forehead

and 2 cm above the supra-orbital ridge (Fallgatter and Strik 1998, Obrig et al. 1996). Using a linear scale, the probe placement on each subject was measured and was used for all the vibration sessions (Figure 2.3A). A piece of clear plastic was wrapped around the sensor to prevent sweat from distorting the absorbency signal. Further, a dark elastic bandage was wrapped around the sensor to secure it in place and minimize any loss of light (Figure 2.3B).

As per the manufacturer's specifications, the sensor was calibrated at 760 nm and 850 nm wavelengths before the test for each subject. The vertical penetration depth of light was set to 4 cm. The difference in absorbency between these two wavelengths indicated the change in oxygenation saturation and the sum of these absorbencies indicated the change in total blood volume. Both oxygenation and blood volume were measured in terms of optical density (OD) which is defined as the logarithmic ratio of intensity of light calibrated at each wavelength to the intensity of measured light at the same wavelength (MicroRunman User Manual, NIM Inc., PA, USA, 1999). Real time data were recorded using the NIRCOM software provided by the manufacturer.

(5) NIRS Data Analysis

The real time NIRS values were averaged every 20 seconds using a customized Microsoft Excel software program specially designed for this study. Baseline oxygenation and blood volume values were monitored during initial rest or recovery from each vibration dose. Maximum values were identified for each dose during vibration with and without rhythmic handgrip contractions, and 'With' and 'Without' a backrest. The change (delta) in oxygenation and blood volume for each condition of vibration dose (3Hz, 4.5 Hz, and 6 Hz), backrest ('With' and 'Without' a backrest), workload (vibration alone and vibration combined with rhythmic handgrip contractions), was calculated as the difference between the maximum and baseline values identified for each phase of the test session. Oxygenation and blood volume responses 30 seconds before the beginning of each WBV session were identified as baseline values, and were used to

calculate delta values. All statistical analyses were performed using the delta values calculated for each phase of the study.

Statistical Analysis

Two separate four-way analysis of variance (ANOVA), with gender as a between-subjects factor, and (1) three within-subject factors (dose, backrest, and acceleration) as repeated measures with a fully crossed design was used to evaluate the differences in acceleration measured at the base and C6 in m/sec²; and (2) three within-subject factors (dose, backrest, and angle) as repeated measures with a fully crossed design was used to evaluate the differences in knee and low-back angles that were measured in degrees, respectively. A fourway analysis of covariance (ANCOVA), using the peak oxygen uptake as the covariate, gender as a between-subjects factor, and three repeated measures factors (dose, backrest, and workload) with a fully crossed design was used to evaluate the differences in the oxygenation and blood volume responses (measured in OD units). Since peak oxygen uptake is considered to be one of the best indicators of aerobic fitness, it will be used to examine whether the fitness level of the subjects has any influence on the cerebral oxygenation and blood volume responses.

To minimize the violation of the assumption of homogeneity of variance, the 'Greenhouse-Geiser' adjustment was used. Since multiple comparisons were made for each dependent variable in the present study, the Bonferroni adjustment for a P value of 0.05 was applied to minimize Type 1 error. The Bonferroni adjustment was calculated as the ratio of 0.05 and number of statistical comparisons (APPENDIX B.1). Thus, statistical significance was considered at P <0.01. Significant F ratios were further analyzed with the Scheffe post hoc multiple comparison test. Pearson product moment correlations were used to examine the relationship between low-back angles and acceleration at C6. Also, the relationship between selected anthropometric variables and oxygenation and blood volume responses were evaluated using Pearson product moment correlations. The Statistical Package for Social Sciences SPSS (version 10) was used for all statistical analyses (SPSS Inc., Chicago, USA).

RESULTS

Biodynamic and Postural Differences

Effects of WBV on acceleration values at the base and C6 levels are summarized in Table 2.2. No significant four-way or three-way interactions were obtained in the acceleration values. This implies that in both genders, the changes observed in the acceleration values at different WBV doses were similar 'With' and 'Without' a backrest. However, significant two-way interactions were observed for interface and dose; interface and backrest; and interface and gender. It should be noted that comparisons of these interactions were based on the pooled mean values of other remaining independent variables.

Significant differences were observed in the acceleration values at C6 only between 4.5 Hz and 6 Hz (9.40 \pm 1.45 m.sec⁻²_{r.m.s} Vs 8.60 \pm 1.24 m.sec⁻²_{r.m.s}, P=0.003), but not at 3 Hz and 4.5 Hz (8.75 \pm 1.04 m.sec⁻²_{r.m.s} Vs 9.40 \pm 1.45 m.sec⁻²_{r.m.s}, P=0.059) and 3 Hz and 6 Hz (8.75 \pm 1.04 m.sec⁻²_{r.m.s} Vs 8.60 \pm 1.24 m.sec⁻²_{r.m.s}, P=0.072). Acceleration values at the base for all doses were not significant (P>0.05). Further, acceleration values at C6 'With' backrest were significantly higher than those 'Without' backrest ('With' backrest: 9.04 \pm 1.25 m.sec⁻²_{r.m.s} Vs 8.77 \pm 1.33 m.sec⁻²_{r.m.s}, P=0.000), but there was no difference observed at the base ('With' backrest: 9.00 \pm 0.09 m.sec⁻²_{r.m.s} Vs 9.00 \pm 0.08 m.sec⁻²_{r.m.s}, P=0.381). At C6, men had significantly higher accelerations than women (men: 9.36 \pm 1.27 m.sec⁻²_{r.m.s} Vs 8.48 \pm 1.18 m.sec⁻²_{r.m.s}, P=0.015), but there was no gender difference observed at the base (men: 9.03 \pm 0.09 m.sec⁻² r.m.s Vs 9.01 \pm 0.08 m.sec⁻²_{r.m.s}, P=0.640).

The postural differences during the WBV experimental conditions in both genders are presented in Table 2.2. No significant four-way or three-way interactions were obtained in the postural values. This implies that in both genders during 'With' and 'Without' a backrest, the changes in low-back and

knee angles were similar across all three vibration doses. A significant interaction was found between backrest and angle. However, knee angles were not significantly different between 'With' and 'Without' a backrest conditions (99.6° \pm 7.5° Vs 99.2° \pm 6.4°, p=0.488), and low-back angles were significantly higher during 'With' and 'Without' a backrest conditions (107.6° \pm 7.1° Vs 95.2° \pm 5.9°, p=0.000). It should be noted that comparisons of the significant two-way interaction were based on the pooled mean values at the three different doses and both genders.

Cerebral Oxygenation and Blood Volume Trends in Men and Women

A typical profile of the cerebral oxygenation and blood volume trends at 4.5 Hz, 'With' and 'Without' a backrest in a male and female subject is shown in Figures 2.6A and 2.6B respectively. It is evident from both figures that as soon as the vibration started 'Without' a backrest, oxygenation and blood volume responses increased from the baseline (resting) value during the four-minute period. During vibration combined with rhythmic handgrip contractions for one minute, oxygenation values increased to a higher level than vibration alone. As soon as the rhythmic handgrip contractions stopped, oxygenation (Figure 2.6A) and blood volume (Figure 2.6B) trends dropped to the initial vibration level. In the recovery from vibration exposure phase, the trends dropped further toward baseline values (recovery 1). Similar trends were observed in the oxygenation and blood volume responses 'With' a backrest, however with the values increasing to higher levels than that of 'Without' a backrest.

The cerebral oxygenation and blood volume differences during the WBV experimental conditions in both genders are presented in Table 2.3. Statistical analysis of cerebral oxygenation and blood volume values for each dose during the different experimental conditions in both men and women revealed: (1) the covariate, namely, peak oxygen uptake, had no significant influence on the cerebral oxygenation and blood volume responses, (2) the four-way and three-way interactions were not significant, implying that both genders responded similarly to the three doses of WBV during 'With' and 'Without' a backrest, and

with and without the rhythmic handgrip contractions, (3) the interaction between gender and workload was significant for the oxygenation and blood volume responses, and (4) the main effects of dose and workload were significant. It should be noted that the comparisons of the significant two-way interaction were based on the pooled mean values at the three different doses and both genders. Similarly comparison of the main effects was also based on the pooled mean values of the remaining three independent variables in the experimental design. The results of these significant main effects and interaction are presented below.

Comparison Between Doses

Significant differences were observed in the mean values of the mean oxygenation between 3 Hz and 4.5 Hz (-0.009 \pm 0.07 OD Vs 0.050 \pm 0.17 OD, P=0.010) and 3 Hz and 6 Hz (-0.009 \pm 0.07 OD Vs 0.054 \pm 0.09 OD, P=0.017). Corresponding comparisons for the blood volume changes were also significant: between 3 Hz and 4.5 Hz (0.028 OD \pm 0.12 Vs 0.114 \pm 0.10 OD, P=0.000) and 3 Hz and 6 Hz (0.028 \pm 0.12 OD Vs 0.117 \pm 0.20 OD, P=0.012). However, no significant difference was observed in oxygenation (P=0.975) and blood volume (P=0.945) mean values between 4.5 Hz and 6 Hz.

Comparison 'With' and 'Without' a Backrest Support

No statistical differences were found in the values of mean oxygenation and blood volume between conditions 'With' and 'Without' a backrest. 'With' backrest mean delta values for oxygenation were higher but not significant compared to 'Without' backrest condition $(0.0326 \pm 0.12 \text{ OD Vs } 0.0311 \pm 0.13 \text{ OD}, P=0.419)$. Further, mean delta values for blood volume were also higher 'With' backrest but not significant compared to 'Without' a backrest condition $(0.092 \pm 0.16 \text{ OD Vs } 0.080 \pm 0.15 \text{ OD}, P=0.055)$.

Comparison Between Vibration Alone and Vibration Combined with Rhythmic Handgrip Contractions

Significant differences in mean oxygenation and blood volume responses were observed between vibration alone and vibration combined with rhythmic handgrip contractions. Mean oxygenation values were significantly higher during vibration combined with rhythmic handgrip contractions than vibration alone $(0.054 \pm 0.14 \text{ OD Vs} 0.010 \pm 0.09 \text{ OD, P=}0.000)$. Also, mean blood volume values were significantly higher during vibration combined with rhythmic handgrip contractions than vibration alone (0.116 \pm 0.17 OD Vs 0.057 \pm 0.13 OD. P=0.000). Significant interaction was obtained between workload and gender. suggesting effect of gender on work performance (Figures 2.7A and 2.7B). During vibration alone, men showed less mean oxygenation changes than women (0.0005 \pm 0.11 OD Vs 0.019 \pm 0.07 OD, P=0.036). However, during vibration with rhythmic handgrip contractions, this trend was reversed (0.070 \pm 0.19 OD Vs 0.042 ± 0.07 OD, P=0.036). During vibration alone, men showed more mean blood volume changes than women $(0.066 \pm 0.17 \text{ OD Vs} 0.05 \pm 0.06)$ OD. P=0.001). A similar trend was observed during vibration with rhythmic handgrip contractions (0.160 \pm 0.20 OD Vs 0.08 \pm 0.12 OD, P=0.001).

Comparison Between Genders

No significant differences in oxygenation and blood volume were found between men and women for all conditions of dose, workload and backrest. In men, mean delta values for oxygenation were not significantly different from women (0.033 ± 0.16 OD Vs 0.030 ± 0.07 OD, P=0.894). Blood volume values were higher in men but not significantly different from women (0.111 ± 0.20 OD Vs 0.063 ± 0.10 OD, P=0.068). Further, no significant interaction between gender and dose was observed in both oxygenation and blood volume responses (Figures 2.8A and 2.8B).

Correlations between Selected Variables and Oxygenation and Blood Volume

Pearson correlations obtained between angles measured during sitting and accelerations at C6 were not significant (P>0.05), suggesting that greater accelerations reported 'With' backrest were not influenced by the greater angles measured at either low-back or knee during sitting 'With' backrest (Table 2.4). Further, acceleration at C6 and angles measured at the low-back during 'With' and 'Without' backrest did not correlate (P>0.05) with cerebral oxygenation and blood volume responses (Table 2.5). Anthropometric variables such as diameter at C6, neck length measured from C6 to the occipital region were also not significantly correlated with cerebral oxygenation and blood volume responses (Table 2.5).

Pearson correlations were also calculated between the maximal handgrip contractions performed during WBV and changes in oxygenation and blood volume responses, however, gender differences in the right handgrip strength did not influence the increase observed in oxygenation and blood volume responses (Table 2.6). Additionally, relationship between the relative peak aerobic power during arm cranking and changes in oxygenation and blood volume responses were also not significantly correlated (Table 2.7) (P>0.05).

DISCUSSION

Biodynamic and Postural Differences during Sitting

In the current investigation, acceleration values measured at C6 in both genders was significantly highest at 4.5 Hz compared to the other doses. However, Griffin et al. (1982) found highest transmissibility in women above 5 Hz. Skin-mounted placement of the accelerometer, and anatomical region selected (C6) in the current study might have contributed to this discrepancy. Pope et al. (1986) reported a significant artifact due to surface- or skin-mounted accelerometers, and Magnusson et al. (1992) recommended a pin-mounted method for the best results. Further, optimal measurement of acceleration and

transmissibility at the head can be obtained by a bite-bar method (Pope et al. 1989, Griffin 1990).

Men showed 9.4% higher acceleration values at C6 than women, suggesting gender influence in acceleration responses at the head. Such response might be due to the greater girth of neck in men measured at C6 and Occipital regions (Table 2.1B). Griffin (1990) and Paddan and Griffin (1988a,b) found that transmission of vertical vibration to seated subjects depends on the vibration of the backrest. However, acceleration of the backrest attached to the vibration shaker was not measured in the current investigation. On the contrary, Magnusson et al. (1992) found use of a backrest and its inclination had minor effects on the attenuations of WBV. Thus, they concluded that a backrest does not necessarily reduce the effect of vibration on spinal loading. They also did not report the acceleration values of the backrest in their study. In the present study, use of a backrest played a small but significant role in acceleration measured at C6, with acceleration during vibration 'With' backrest being 3% higher than 'Without' backrest (P=0.003).

Furthermore, vibration accelerations reported in the present study are within the range of 1g reported by Wilder et al. (1982). Human tolerance limits for WBV between 4 and 8 Hz was also reported in the ranges of 1.5g to 2.0g r.m.s (Magid et al. 1960). Similar to the present study, Hoover and Ashe (1962) and Gaeuman et al. (1962) used an electromechanical vibration shaker, and reported different displacements and accelerations at the vibration base with increases in the vibration dose. Thus, higher accelerations of up to 1g at the vibration base and C6 in the present study might be due to the type of vibration shaker used in the current study may limit comparison to other mentioned studies using different methodology.

In the present study, knee and low-back angles were greater while using a backrest (Table 2.2). In this posture, subjects tend to transfer most of their body weight to the backrest, thus reducing load on the low-back caused by the upper body weight (Andersson et al. 1991). However, based on the non-significant

correlation values reported in the Table 2.4, postural adjustment might not have resulted in greater transmission of vibration to the C6 region.

Cerebral Oxygenation and Blood volume Responses during Whole-Body Vibration

To the best of our knowledge, no studies to date have examined cerebral oxygenation and blood volume responses during WBV in any species. Schoenberger (1972) defined vibration as a diffuse stimulus that can stimulate various kinds of neural receptors that include touch, pressure, vision, body position and spatial orientation, and pain receptors. The response of a particular receptor on the human body depends on the type and degree of vibration exposed. Cortical activation is possible with different stimuli to these receptors in the body (Landstrom and Lundstrom 1985). Based on strength and explosive power-training studies, Bosco et al. (1998) suggested that exposure to WBV may result in increased neural potentiation either via spinal or cortical reflexes. In another investigation (Bosco et al. 2000), these researchers demonstrated an increase in plasma concentrations of testosterone and growth hormone, and a decrease in cortisol as a result of 10 min exposure to WBV.

The typical NIRS response over an activated cortical area during a motor task results in an increase in oxyhemoglobin and total hemoglobin (cerebral blood volume), with a simultaneous decrease in deoxyhemoglobin (Obrig et al. 1996). These authors, with a hand tapping activity demonstrated that cerebral oxygenation exceeds the increase in oxygen demand during motor stimulation. In the present study, vibration as a stimulus increased the frontal cortical oxygenated signal (850 nm signal) with a concomitant but smaller decrease in the deoxygenated signal (760 nm signal), thereby resulting in an increase in total blood volume (Figure 2.6B). Regional brain activity induces a local arteriolar vasodilation, resulting in an increase in blood volume and blood flow (Villringer and Chance 1997). In the present study, blood volume trends (Figure 2.6B) were almost parallel and showed greater increase than that of oxygenation trends

(Figure 2.6A), suggesting a close coupling between oxygenation and blood volume responses during exposure to WBV.

Based on animal studies, Nakamura et al. (1994) suggested that pain transmitters such as hippocampal vasoactive intestinal polypeptides might play a significant physiological role in regulating cerebral blood flow during WBV. They also found a decrease in vasoactive intestinal polypeptide like immunoreactivity in the frontal cortex after exposure to WBV. However, they did not observe a significant difference in the frontal cortex values during control and WBV conditions. Weinstein et al. (1988) studied the effects of WBV on neuropeptides. substance P (pain mediator) and vasoactive intestinal polypeptide neuropeptides in rabbits' dorsal root ganglion, a major output structure of the frontal cortex (Goldman-Rakic 1984). Weinstein et al. (1988) demonstrated a localized decrease in substance P and an increase in vasoactive intestinal polypeptide following low-frequency vibration, suggesting an increase in axonal transport due to direct stimulation of the brain similar to the mechanism occurring during peripheral nerve injury. Even though the present study did not study these pain transmitters, the current evidence from exposure to WBV in different experimental conditions suggest that, increased neuronal activity subsequently results in increased perfusion to the pre-frontal cortex.

Increases in oxygenation and blood volume immediately at the end of vibration exposure for a brief period were observed in most of the subjects (Figures 2.6A and 2.6B). Such a response may be due to the sudden 'stop' of vibration exposure, thus increase in oxygenation and blood volume changes. Additionally, in the present study the exact percentage of hemoglobin in the cortical size of the frontotemporalis region investigated was not determined. Also, results from the present study were based on a single cortical volume (in specific, a sample optical volume from the right side of the forehead), hence these values were representative of regional differences in that a specific cerebral region was monitored rather than the global cerebral area. With these limitations in mind, one has to be cautious in relating the results obtained in the blood flow studies reported to blood volume interpretations.

Role of Whole-Body Vibration Dose

Griffin et al. (1982) reported that vertical vibration transmission to the head was greater in men at frequencies below 5 Hz but greater in women at frequencies above 5 Hz. The present study demonstrated a similar trend in terms of cerebral oxygenation and blood volume changes, however they were not statistically significant. Men demonstrated highest cerebral oxygenation and blood volume values at 4.5 Hz (P>0.05) while women peaked at 6 Hz (P>0.05) [Figure 2.8A and 2.8B]. Such stimulus frequency-dependent changes in cerebral blood flow responses during a variety of cortical non-WBV activities have been previously reported using positron emission tomography (PET) (Fox and Raichle 1984, Sadato et al. 1996, Vafaee et al. 1999).

Fox and Raichle (1984) studied changes in blood flow responses in human striate cortex as a function of visual stimulus rate and found that maximal responses occurred at 7.8 Hz with a subsequent drop in blood flow as the frequency further increased. Overall, they observed a non-linear relationship between blood flow and stimulus rate. They indicated that this pattern of response may be due to the region being monitored and the subsequent neuronal demand resulting from its activation, and "the methodological necessity of accumulating tissue counts over an interval of time". Sadato et al. (1996) also showed a non-linear increase in regional blood flow in humans as a function of repetitive flexion movement of the right index finger against the thumb. They found a greater increase in regional blood flow between the slow (0.75 and 1 Hz) and fast movement repetitive rates (2 and 2.5 Hz), with flow saturation occurring at the very fast rates (3 and 4 Hz). Vafaee et al. (1999) also found that cerebral metabolic rate of oxygen in human striate cortex increased with increase in frequency of visual stimulus and peaked at 4 Hz (for both genders) and dropped considerably as the frequency increased from 4 Hz. They treated frequency as an indicator of neuronal activity and regional cerebral metabolic rate as an indicator of neuronal metabolism.

Even though none of the studies cited above pertain to WBV, the current NIRS observations of pre-frontal cortex oxygenation and blood volume trends

suggest that these responses are dependent upon the vibration dose as well. Even though not statistically significant, the decrease in cerebral oxygenation and blood volume responses in men after 4.5 Hz, and a further increase in women at 6 Hz of vibration in the present study supports the non-linear observations of Vafaee et al. (1999). These researchers concluded that the nonlinear frequency-dependent cerebral oxygen metabolic responses could be due to two possible limitations.

A physiological limitation known as finite oxygen diffusibility in the brain, resulting from limited oxygen transfer across the cerebral capillaries. This phenomenon can be overcome by an increase in blood flow such that it can elevate the average oxygen tension in capillary blood, thus increasing oxygen delivery to the tissue. However, this limitation does not seem to always increase the oxygen consumption to the region monitored, thus leading to a secondary limitation known as enzymatic mechanism (Vafaee et al. 1999). This mechanism causes the brain not to fully sustain the increase in cerebral metabolic rate that was possible by an increase in blood flow. This secondary limitation can be overcome by increasing the stimulus load (in the present study – may be the dose and its intensity). From these limitations, they hypothesized that once the frequency is increased beyond 4 Hz, neuronal work no longer requires the metabolic rate to rise. They concluded that the "stimulus load imposed on the brain tissue must exceed a certain threshold before glycolysis is augmented by increased oxidative metabolism" (Vafaee et al. 1999).

It is hypothesized that a similar limitation might have occurred in the present study, in light of the fact that beyond 4.5 Hz neuronal work did not require cerebral oxygenation (and blood volume) to rise in men, whereas women demonstrated linear increases with the increasing dose. However, these conclusions should be interpreted cautiously, as Vafaee et al. (1999) exposed their subjects to increasing doses of visual stimuli in the same session, whereas in the present study the subjects were exposed to three different WBV doses on three separate days. Also, the cerebral regions monitored in the studies reported

(Fox and Raichle 1984, Sadato et al. 1996, Vafaee et al. 1999) were entirely different from the present study.

Role of Sitting 'With' and 'Without' a Backrest during Whole-Body Vibration

Griffin (1990) summarized the results of various studies pertaining to the effect of backrest during vibration and concluded that leaning against the backrest, in general, increases the vibration transmission to the head. Oxygenation and blood volume values obtained in this present study during 'With' backrest were 5% and 14% higher than during 'Without' backrest condition (P>0.05). Griffin (1990) also reported that postural variations at the low back and head would substantially change the vibration transmission. In the present study, leaning against the backrest resulted in back angles that were significantly higher (11.5%) 'With' backrest compared to 'Without' backrest condition (Table 2.2). Also, in the present study accelerations obtained at the C6 during 'With' backrest were 3% higher than 'Without' backrest. However, the differences reported at the low-back angles and acceleration at C6 were not significantly correlated to the oxygenation and blood volume values (Table 2.5).

Mehagnoul-Schipper et al. (2000) demonstrated that the change of posture from supine to standing resulted in a decrease in cerebral oxygenation and blood volume changes in elderly subjects but not in young men and women. Moraine et al. (2000) reported that as the head elevates from 0 to 45°, cerebral blood flow decreases in a gradual manner in comatose patients. They concluded that arterio-venous pressure gradient was the major determinant for regional cerebral blood flow in postural variations. However, in the present study posture of the head during WBV was not measured.

Role of Workload during Whole-Body Vibration

Significant differences in oxygenation and blood volume were observed during vibration alone and during vibration combined with rhythmic handgrip contractions. As the exposure to dose varied from 3 to 6 Hz, oxygenation and blood volume values during vibration alone increased. However, during vibration combined with rhythmic handgrip contractions, maximum delta values were attained at 4.5 Hz. During vibration combined with rhythmic handgrip contractions, delta oxygenation and blood volume values were 81% and 51% higher than that of vibration alone, respectively.

In elite cyclists, Pott et al. (1996) showed that during rhythmic handgrip contractions, mean cerebral blood volume (obtained with transcranial Doppler technique) increased by 21% over resting conditions. Additionally, Friedman et al. (1992) suggested that blood flow changes in the cortical regions (during dynamic right hand contractions before and after right side axillary blockade) monitored were an indirect measure of neuronal activity within these regions, and thus neuronal feedback was essential for the changes observed. Using single photon computer tomography, these authors demonstrated that smaller afferents (Groups III and IV) were mainly responsible for the feedback signals that were essential for an increase in motor sensory regional blood flow. Also, a reduction in motor strength during anesthesia correlated with the decrease in motor sensory regional blood flow, thus demonstrating group I and II fibers might also influence the afferent process (Friedman et al. 1992).

Williamson et al. (1996) showed that regional blood flow to the motor sensory areas during left hand dynamic contraction was not influenced by the afferent stimulation of the Group Ia or III and IV metabolically sensitive fibers. However, these authors postulated that this type of contraction might have resulted in cerebral activation either due to the combination of increase in central command and muscle afferent input or due to activation of Group IIIa muscle mechanosensitive fibers or Ib Golgi tendon organs. Nowak et al. (1999) also studied regional cerebral blood flow during handgrip squeezing of a force transducer before and after axillary blockade to investigate if activated areas were responsible for the increase in cardiovascular variables. They concluded that cortical activation in the sensory motor area after regional anaesthesia was not indicative of central command that influenced the cardiovascular variables such as heart rate and blood pressure. The oxygenation and blood volume results of the present study are based on rhythmic maximum voluntary contractions over a period of one minute. Also, in the present study NIRS trends during rhythmic handgrip contractions without vibration were not evaluated. However, it can be hypothesized that the increased oxygenation and blood volume changes during vibration combined with rhythmic handgrip contractions observed in the present study will be greater than that obtained while performing rhythmic handgrip contractions without vibration. Also, Uzuner et al. (2000) pointed out that investigating the stimulated motor cortex area during "just thinking of gripping one's hands" rather than "actual gripping activity" may also give different but lower responses.

Drivers and pilots often have to spend their workday not only in prolonged sitting postures but also simultaneously operating manual controls. This necessitates isometric work of the forearm muscles at varying percentages of maximal voluntary contraction. The present study has shown that cerebral responses during prolonged sitting (without WBV) are much less than the values observed during WBV alone. However, delta values observed during WBV combined with rhythmic handgrip contractions showed the greatest demand of oxygenation and blood volume changes than any other condition of the study. Thus, the present study demonstrates the importance of investigating the combined effects of physical factors during WBV.

Role of Gender

Resonance frequencies associated with WBV in the Z-axis (foot-to-head) for humans range from doses of 4 to 8 Hz (International Organization for Standardization 1997). Interestingly, in the present study women showed a systematic increase in oxygenation and blood volume measurements with increases in vibration dose up to 6 Hz (P>0.05), while the responses in men peaked at 4.5 Hz (P>0.05). Even though gender differences in the current study were statistically not significant, men demonstrated higher oxygenation and blood volume delta values than women. However, accelerations measured at C6 were significantly different between men and women, with men's accelerations 9.4%

times higher than women's. At this point, this discrepancy between difference in acceleration values at the C6 and their non-significant oxygenation and blood volume values at the cerebral region are not clear.

Additionally, even though statistically not significant, interaction of dose and gender suggests that both men and women respond differently below and within the resonance frequency range (Figures 2.8A and 2.8B). Gender played a significant role in work performance as well. During vibration alone, women showed higher values in oxygenation and blood volume responses than men. However, during vibration combined with rhythmic handgrip contractions, men showed higher values than women (Figure 2.7A and 2.7B). This phenomenon suggests that, men show greater cerebral oxygenation and blood volume responses than women during motor simulation when being exposed to WBV. In other words, men demand more oxygen (hence greater blood flow to the cerebral region) during rhythmic handgrip contractions compared to women. Maximal grip strength of men are greater than women (Table 2.1), however, non-significant correlations reported between grip strength values during WBV and cerebral oxygenation and blood volume changes (Table 2.6), implies that men's greater strength did not influence the observed differences.

Bilateral hand gripping in both men and women showed a significant blood flow velocity increase in bilateral middle cerebral arteries compared to rest condition (Uzuner et al. 2000). These authors used transcranial Doppler sonography in studying the blood flow velocity and reported no significant differences between genders. Mehagnoul-Schipper et al. (2000) demonstrated that cerebral oxyhemoglobin changes were smaller in women than in men during postural variations, however, there were no significant gender differences in oxygenation and blood volume measurements. Levin et al. (1988) used functional magnetic resonance imaging to demonstrate significant sex differences in blood-oxygen-level-dependent signals when subjects were exposed to photic stimulation. The values were 38% lower in women than men, and the researchers attributed this to the type of stimulus adopted. Since hemoglobin was shown as a contrast in the magnetic resonance signal and its

concentration is generally lower in women compared to men, it was postulated that this could be a reason for this gender difference.

Rodriguez et al. (1988) reported that the regional blood flow in the frontal cortex was higher in women compared to men during resting. However, hemispheric comparisons indicated no significant differences in women, but significantly higher values in the right compared to the left hemisphere in men. They speculated that this difference between men and women might be due to differences in functional organization of the cerebral cortex or gender related differences in the functional state of brain during the testing sessions. During resting and seated conditions, Misra et al. (1998) showed sex, age, height. weight, and skin colour did not influence regional cerebral oxygen saturation. Using a blood-oxygen-level-dependent technique, Kastrup et al. (1999) showed higher signals in women than men in the visual cortex during sustained checkerboard stimulation. The researchers suggested that this gender difference in the vascular response to focal neuronal activation may be due to differences in visual physiology. With an isotope clearance technique, Gur et al. (1982) showed a higher rate of blood flow per unit weight of brain in women than in men during resting and cognitive activities. It should be noted that, none of the studies cited were related to WBV, thereby making it difficult to compare and interpret the present gender differences.

Role of Fitness

In the present study, the level of aerobic fitness in the subjects did not influence the oxygenation and blood volume responses during WBV (Table 2.7). Using the peak oxygen uptake obtained from incremental arm cranking test till exhaustion as a covariate (Table 2.1A), the results suggested that cerebral oxygenation and blood volume responses in men and women were not dependent upon the level of aerobic fitness in the subjects of the present study. These results are in contradiction to the hypothesis of Boff and Lincoln (1988) and Griffin (1990) who suggested that fitness might influence the subjects

exposed to WBV. However, their assumption might be based on the whole-body cardiorespiratory responses.

Dustman et al. (1990) showed that men of high fitness perform better in neurocognitive tasks, and have a greater visual sensitivity than low fit men. These authors divided the subjects into high and low fit groups based on their maximal oxygen uptake values obtained during an incremental treadmill exercise test till exhaustion. Based on the results, they hypothesized that oxygen availability for cerebral metabolism was greater in high fit groups. Even though Dustman et al. (1990)'s study is not related to WBV, their approach towards combined psychological demand and exercise can be related to occupational performance of transit drivers or aviation pilots who perform various cognitive and motor functions in addition to just driving (Johanning et al. 1991, Hankins and Wilson 1998).

CONCLUSION

In the present study, the role of vibration dose, backrest support, workload, and gender on cerebral oxygenation and blood volume responses during seated WBV were investigated. Compared to sitting without WBV, an increase in cerebral activity (implied through an increase in both blood volume and oxygenation trends) was observed in both men and women during WBV when exposed to discrete doses of 3 Hz, 4.5 Hz, and 6 Hz. Vibration alone showed a much lower increase in oxygenation (and blood volume) changes than vibration combined with rhythmic handgrip contractions, suggesting variable physiological effects during man-environment interactions with WBV. During vibration alone, women had higher values for mean oxygenation and blood volume responses than men. However, during vibration combined with rhythmic handgrip contractions, men demonstrated higher values than women. These responses were not significantly correlated to greater maximal grip strength of men. This suggests that, men demand greater cerebral oxygenation and blood volume responses to WBV.

Based on the vibration dose range of 3-6 Hz tested, men demonstrated the highest oxygenation and blood volume responses at 4.5 Hz and women peaked at 6 Hz. However, these results were not statistically significant. Subjects sitting 'With' and 'Without' a backrest showed similar cerebral responses, indicating the presence of a backrest support may not influence cerebral oxygenation and blood volume responses. The level of aerobic fitness in the subjects investigated did not influence the cerebral responses for both genders.

Significant secondary findings in biodynamics and postures during sitting were also observed. Accelerations at C6 were higher in men than women, with highest accelerations being observed at 4.5 Hz than any other dose. Greater girth of neck in men measured at C6 and Occipital regions might be the reasons for such a gender difference. Further, accelerations obtained at C6 during 'With' backrest were significantly higher than 'Without' backrest for all doses. Such significantly higher values during 'With' backrest may suggest the role of backrest vibration during WBV. Considering postural differences during seated vibration, angles adopted during 'With' backrest were higher than 'Without' backrest values, with low-back angles during 'With' backrest showed to be higher than 'Without' backrest values.

REFERENCES

- 1. American College of Sports Medicine (1993) Resource manual for guidelines for exercise testing and prescription. Lea & Febiger, Philadelphia, USA.
- Andersson GBJ, Chaffin DB, and Pope MH (1991) Occupational biomechanics of the lumbar spine. In Occupational low back pain. Assessment, Treatment, and Prevention. Edited by Pope MH, Andersson GBJ, Frymoyer JW, Chaffin DB. Mosby Year Book, Chicago, pp: 20-43.
- 3. Bhambhani Y, Maikala R, Farag M, and Ting W (2000) Reliability of cerebral oxygenation measurements during rhythmic handgrip contractions using near infra-red spectroscopy (Abstract). Canadian Society of Exercise Physiology Annual Scientific Conference, October 25-28, Canmore, CANADA
- 4. Boff KR and Lincoln JE (1988) Vibration and display perception. In: Engineering Data Compendium: Human Perception and Performance. AAMRL, Wright-Patterson Air Force Base, OH, pp 2064-2066
- 5. Borg G (1982) Psychophysical bases of perceived exertion. *Med Sci Sports Exerc* 29: 402-409
- Bosco C, Cardinale M, Tsarpela O, Colli R, Tihanyi J, Von Duvillard SP, and Viru A (1998) The influence of whole body vibration on jumping performance. *Biol* Sport 15: 157-164
- Bosco C, Iacovelli M, Tsarpela O, Cardinale M, Bonifazi M, Tihanyi J, Viru M, De Lorenzo A, and Viru A (2000) Hormonal responses to whole-body vibration in men. *Eur J Appl Physiol* 81: 449-454
- 8. Bovenzi M (1996) Low back pain disorders and exposure to whole-body vibration in the workpalce. Semin Perinatol 20(1): 38-53
- Chance B, Leigh JS, Miyake H, Smith DS, Nioka S, Greenfeld R, Finander M, Kaufmann K, Levy W, Young M, Cohen P, Yoshioka H, and Boretsky R (1988) Comparison of time-resolved and --unresolved measurements of deoxyhemoglobin in brain. *Proc Natl Acad Sci USA* 85: 4971-4975
- 10. Colier WNJM, Quaresima V, Oeseburg B, and Ferrari M (1999) Human motorcortex oxygenation changes induced by cyclic coupled movements of hand and foot. *Exp Brain Res* 129: 457-461
- 11. Canadian Society for Exercise Physiology (1994) Revised Physical Activity Readiness Questionnaire – RPARQ. Gloucester, NB, CANADA.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

- 12. Dustman RE, Emmerson RY, Ruhling RO, Shearer DE, Steinhaus LA, Johnson SC, Bonekat HW, and Shigeoka JW (1990) Age and fitness effects on EEG, ERPs, Visual sensitivity, and Cognition. *Neurobiology of Aging* 11(3): 193-200
- 13. Fallgatter AJ and Strik WK (1998) Frontal brain activation during the Wisconsin card sorting test assessed with two-channel near-infrared spectroscopy. *Eur Arch Psychiatry Clin Neurosci* 248: 245-249
- 14. Friedman DB, Friberg L, Payne G, Mitchell JH, and Secher NH (1992) Effects of axillary blockade on regional blood flow during dynamic hand contractions. *J Appl Physiol* 73(5): 2120-2125
- 15. Fox PT and Raichle ME (1984) Stimulus rate determines regional brain blood flow in straite cortex. *Ann Neurol* 17: 303-305
- 16. Gaeuman JV, Hoover GN, and Ashe WF (1962) Human vibration exposure. Aerospace Med 33: 469-474
- 17. Giller CA, Giller AM, Cooper CR, and Hatab MR (2000) Evaluation of the cerebral hemodynamic response to rhythmic handgrip. *J Appl Physiol* 88: 2205-2213
- 18. Goldman-Rakic PS (1984) The frontal lobes: uncharted provinces of the brain. *Trends Neurosci:* 425-429
- 19. Griffin MJ, Whitham EM, and Parsons KC (1982) Vibration an comfort I: Translational set vibration. *Ergonomics* 26: 603-630
- 20. Griffin MJ (1990) Vibration and discomfort. In: Handbook of Human Vibration. New York, Academic Press, pp 43-123
- 21. Gur RC, Gur RE, Obrist WD, Hungerbuhler JP, Younkin D, Rosen AD, Skolnick BE, and Reivich M (1982) Sex and handedness difference in cerebral blood flow during rest and cognitive activity. *Science* 217: 659-661
- 22. Hankins TC and Wilson GF (1998) A comparison of heart rate, eye activity, EEG and subjective measures of pilot mental workload during flight. *Aviat Space Environ Med* 69(4): 360-367
- 23. Hansson T, Magnusson M, and Broman H (1991) Back muscle fatigue and seated whole-body vibrations: an experimental study in man. *Clin Biomech* 6(3): 173-178
- 24. Hock C, Villringer K, Heekeren H, Hofmann M, Wenzel R, Villringer A, and Muller-Spahn F (1997) A role for near infrared spectroscopy in psychiatry? Adv Exp Med Biol 413: 105-23

- 25. Holzschuch M, Woertzen C, Metz C, and Brawanski A (1997) Dynamic changes of cerebral oxygenation measured by brain tissue oxygen pressure and near infrared spectroscopy. *Neurol Res* 19: 242-246
- 26. Hoover GN and Ashe WF (1962) Respiratory response to whole-body vertical vibration. *Aerospace Med* 33: 980-984
- 27. Hoshi Y and Tamura M (1993) Detection of dynamic changes in cerebral oxygenation coupled to neuronal function during mental work in man. *Neurosci* Lett 150: 5-8
- 28. Ide K, Horn A, and Secher NH (1999) Cerebral metabolic response to submaximal exercise. *J Appl Physiol* 87(5): 1604-1608
- 29. Ide K and Secher NH (2000) Cerebral blood flow and metabolism during exercise. *Prog Neurobiol* 61: 397-414
- 30. International Organization for Standardization 2631 Part 1 (1997) Mechanical vibration and shock Evaluation of human exposure to whole-body vibration Part 1: General Requirements. ISO 2631/1-1997 (E)
- 31. Johanning E, Wilder DG, Landrigan PJ, and Pope MH (1991) Whole-body vibration exposure in subway cars and review of adverse health effects. *J Occup Med* 33(5): 605-612
- 32. Kastrup A, Li TQ, Glover GH, Kruger G, and Moseley ME (1999) Gender differences in cerebral blood flow and oxygenation response during focal physiologic neural activity. *J Cereb Blood Flow Metab* 19: 1066-1071
- 33. Keyex A, Ochrymowicz-Bernelmans D, and Charlier AA (1995) Induced response to hypercapnia in the two-compartment total cerebral blood volume: Influence on brain vascular reserve and flow efficiency. J Cereb Blood Flow Metab 15: 1121-1131
- 34. Kurth CD and Uher B (1997) Cerebral hemoglobin and optical pathlength influence near-infrared spectroscopy measurement of cerebral oxygen saturation. *Anesth Analg* 84: 1297-1305
- 35. Landstrom U and Lundstrom R (1985) Changes in wakefulness during exposure to whole-body vibration. *Electroencephalogr Clin Neurophysiol* 61: 411-415.
- 36. Levin JM, Ross MH, Mendelson JH, Mello NK, Cohen BM, and Renshaw PF (1998) Sex differences in blood-oxygenation-level-dependent functional MRI with primary visual stimulation. *Am J Psychiatry* 155: 434-436

- 37. Magid EB, Coerman RR, and Ziegenruecker GH (1960) Human tolerance to hwole body sinusoidal vibration. *Aero Med* 31: 915-924
- 38. Magnusson M, Pope MH, Lindell V, Svensson M, and Andersson G (1992) Effect of backrest inclination on the transmission of vertical vibrations through the lumbar spine. *Clin Biomech* 8: 5-12
- 39. Manninen O (1988) Changes in hearing, cardiovascular functions, haemodynamics, upright body sway, urinary catecholamines and their correlates after prolonged successive exposures to complex environmental conditions. Int Arch Occup Environ Health 60: 249-272
- 40. Mchedlishvili G (1986) Cerebral arterial behavior providing constant cerebral blood flow, pressure, and volume. In: Arterial behavior and blood circulation in the brain (Bevan JA, ed), Plenum Press, New York, pp 55-60
- 41. Mehagnoul-Schipper DJ, Vloet LCM, Colier WNJM, Hoefnagles WHL, and Jansen RWMM (2000) Cerebral oxygenation declines in healthy elderly subjects in response to assuming the upright position. *Stroke* 31: 1615-1620
- 42. MicroRunMan User Manual (1998), NIM Inc., PA, USA.
- 43. Misra M, Stark J, Dujovny M, Widman R, and Ausman JI (1998) Transcranial cerebral oximetry in random normal subjects. *Neurol Res* 20: 137-141
- 44. Moraine JJ, Berre J, and Melot C (2000) Is cerebral perfusion pressure a major determinant of cerebral blood flow during head elevation in comatose patients with severe intracranial lesions? *J Neurosurg* 92: 606-614.
- 45. Nakamura H, Moroji T, Nagase H, Okazawa T, and Okada A (1994) Changes of cerebral vasoactive intestinal polypeptide and somatostatin-like immunoreactivity induced by noise and whole-body vibration in the rat. *Eur J Appl Phyisiol* 68: 62-67
- 46. Nakamura H, Okada A, Nohara S, and Nakamura H (1988) The effect of whole body vibration on the cerebral metabolites in rats. Aerospace Science – Proceedings of the 1st Nihon University International Symposium on Aerospace Medicine, Nihon University Press, Japan, pp 153-156
- 47. Nowak M, Olsen KS, Law I, Holm S, Paulson OB, and Secher NH (1999) Command-related distribution of regional cerebral blood flow during attempted handgrip. J Appl Physiol 86(3): 819-824

- 48. Obrig H, Hirth C, Junge-Hulbsing JG, Doge C, Wolf T, Dirnagl U, and Villringer A (1996) Cerebral oxygenation changes in response to motor stimulation. *J Appl Physiol* 81(3): 1174-1183
- 49. Obrig H and Villringer A (1997) Near-infrared spectroscopy in functional activation studies. Adv Exp Med Biol 413: 113-127
- 50. Paddan GS and Griffin MJ (1988a) The transmission of translational seat vibration to the head I. Vertical seat vibration. *J Biomech* 21(3): 191-197
- 51. Paddan GS and Griffin MJ (1988b) The transmission of translational seat vibration to the head II. Horizontal seat vibration. *J Biomech* 21(3): 199-206
- 52. Panjabi MM, Andersson GBJ, Jorneus L, Hult E, and Mattsson L (1986) In vivo measurement of spinal column vibrations. *J Bone Joint Surg* 68A(5): 695-703
- 53. Pollard V, Prough DS, DeMelo E, Deyo DJ, Uchida T, and Stoddart HF (1996) Validation in volunteers of a near-infrared spectroscope for monitoring brain oxygenation in vivo. *Anesth Analg* 82: 269-277
- 54. Pope MH, Svensson M, Broman H, and Andersson GBJ (1986) Mounting of the transducers in measurement of segmental motion of the spine. *J Biomech* 19: 675-677
- 55. Pope MH, Broman H, and Hansson T (1989) The dynamic response of a subject seated on various cushions. *Ergonomics* 32(10): 1155-1166
- 56. Pope MH, Magnusson M, and Wilder D (1998) Kappa Delta Award Low back pain and whole-body vibration. *Clin Orthop Research* 354: 241-248
- 57. Pott F, Jensen K, Hansen H, Christensen NJ, Lassen NA, and Secher NH (1996) Middle cerebral artery blood velocity and plasma catecholamines during exercise. Acta Physiol Scand 158: 349-356
- 58. Punwani S, Ordidge RJ, Cooper CE, Amess P, and Clemence M (1998) MRI measurements of cerebral deoxyhaemoglobin concentration (dHb) – correlation with near infra-red spectroscopy (NIRS). NMR Biomed 11(6): 281-289
- 59. Quaresima V, Sacco S, Totaro R, and Ferrari M (2000) Noninvasive measurement of cerebral hemoglobin oxygen saturation using two near infrared spectroscopy approaches. *J Biomed Optics* 5: 201-205
- 60. Rodriguez G, Warkentin S, Risberg J, and Rosadini G (1988) Sex differences in regional cerebral blood flow. *J Cereb Blood Flow Metab* 783-789
- 61. Roland PE (1984) Metabolic measurements of the working frontal cortex in man. Trends Neurosci: 430-435

- 62. Sadato N, Ibanez V, Deiber MP, Campbell G, Lenoardo M, and Hallett M (1996) Frequency-dependent changes of regional cerebral blood flow during finger movements. J Cereb Blood Flow Metab 16: 23-33
- 63. Schmidt RT and Toews JV (1970) Grip strength as measured by the Jamar dynamometer. Arch Phys Med Rehab 51: 321-327
- 64. Schoenberger RW (1972) Human response to whole-body vibration. Percep Motor Skills 34: 127-160.
- 65. Seidel H, Harazin B, Pavlas K, Sroka C, Richter J, Bluthner R, Erdmann U, Grzesik J, Hinz B, and Rothe R (1988) Isolated and combined effects of prolonged exposures to noise and whole-body vibration on hearing, vision and strain. *Int Arch Occup Environ Health* 61: 95-106
- 66. Simonson SG, and Piantadosi CA (1996) NIRS: Clinical Applications. Crit Care Clin 12(4): 1019-1029
- 67. Uzuner N, Yalcinbas O, Gucuyener D, and Ozdemir G (2000) Hand gripping effect on cerebral blood flow in normal subjects. *Eur J Ultrasound* 11: 147-150
- 68. Vafaee MS, Meyer E, Marrett S, Paus T, Evans AC, and Gjedde A (1999) Frequency-dependent changes in cerebral metabolic rate of oxygen during activation of human visual cortex. *J Cereb Blood Flow Metab* 19: 272-277
- 69. Villringer A and Chance B (1997) Non-invasive optical spectroscopy and imaging of human brain function. *Trends Neurosci* 20(10): 435-442.
- 70. Villringer A, Planck J, Stodieck S, Botzel K, Schleinskofer L, and Dirnagl U (1994) Noninvasive assessment of cerebral hemodynamics and tissue oxygenation during activation of brain cell function in human adults using near infrared spectroscopy. Adv Exp Med Biol 345: 559-565
- 71. Weinstein J, Pope M, Schmidt R, and Seroussi R (1988) Neuropharmacologic effects of vibration on the dorsal root ganglion An Animal Model. Spine 13(5): 521-525
- 72. Wilder D, Woodworth BB, Frymoyer JW, and Pope MH (1982) Vibration and the human spine. *Spine* 7(3): 243-254
- 73. Willamson JW, Fridman DB, Mitchell JH, Secher NH, and Friberg L (1996) Mechanisms regulating regional cerebral activation during dynamic handgrip in humans. J Appl Physiol 81(5): 1884-1890

Gender	Age	Body Mass	Sitting Height	Standing Height	Body Mass Index Standing
	Yrs	Kgs	m	m	Kg. m ⁻²
Men N=13	24.7 ± 3.9	70.8 ± 12.1 •	0.86 ± 0.05	1.71 ± 0.07*	24.0 ± 3.2
Women N=14	23.9 ± 3.5	60.3 ± 9.0	0.84 ± 0.08	1.63 ± 0.10	23.0 ± 3.7

Table 2.1A: Demographics of subjects (Mean ± SD)

^a – significant difference between Men and Women (p < 0.05)

Table 2.1B: Demographics of subjects (Mean ± SD)

Gender		Neck Dimer	nsion	Maximal Right Hand	Peak Oxygen Uptake Arm Cranking	
	Length	C6 ^Ψ	$Occipital^{\Psi}$	Grip Strength		
	Cm	Cm	Cm	Kgs	mL.Kg ⁻¹ .min ⁻¹	
Men N=13	12.6 ± 1.4	37.8 ± 3.2*	37.5 ± 2.8 *	34.2 ± 9.1*	24.5 ± 6.6	
Women N=14	12.3 ± 1.5	32.4 ± 1.7	31.5 ± 1.8	22.5 ± 3.6	21.4 ± 4.6	

 $^{\Psi}$ Diameter measured at the placement of accelerometer at C6 and at Occipital regions

* -- significant difference between Men and Women (p < 0.05)

GENDER	BACKREST	DOSE	LOW-BACK	KNEE ANGLE	Acceleration at C6	Acceleration at BASE
		Hz	Degrees	Degrees	m.sec ³ r.m.s	m.sec ³ r.m.s
Men	ON	3	106.2 ± 5.6"	100.7 ± 9.4	9.26 ± 1.164	8.94±0.1
	OFF	3	93.4 ± 5.2	99.5 ± 6.0	8.99 ± 1.33	
	ON	4.5	106.2 ± 7.2*	100.2 ± 6.1	9.68 ± 1.34 tal	8.99 ± 0.1
	OFF	4.5	94.7 ± 4.6	100.2 ± 7.0	9.69±1.49	
	ÔN	6	107.2 ± 7.8*	99.5 ± 7.1	9.29 ± 1.03W	9.11 ± 0.1
	OFF	6	94.1 ± 7.2	101.1 ± 6.8	9.06 ± 1.19	
Women	ÔN	3	108.6±5.8°	99.3 ± 6.3	8.62 ± 0.57*	8.94 ± 0.1
	OFF	3	95.6±6.2	97.4 ± 5.2	8.22 ± 0.75	
	ON	4.5	106.1 ± 6.7*	99.1 ± 7.2	9.24 ± 1.43 ^{la}	8.99±0.1
	OFF	4.5	97.4 ± 6.4	99.7 ± 6.2	8,85 ± 1.46	
	ÓN	6	108.6 ± 9.4*	98.8 ± 9.3	8.06 ± 1.16	9.11±0.1
	Off	6	96.0 ± 5.6	97.2 ± 7.1	7.93 ± 1.08	

Table 2.2: Postural and acceleration differences during WBV at different doses in healthy men and women (Mean ±SD)

*- significant difference between back-on and back-off for the low-back angle at the same dose (P<0.01)

*- significant difference between back-on and back-off for acceleration at the C6 at the same dose (P<0.01)

*- significant difference between 4.5 Hz and 6 Hz for acceleration at the C6 (P<0.01)

⁴-significant difference between men and women for acceleration at the C6 (P<0.01)

GENDER	BACKREST	DOSE	∆ Oxyge	nation ^Ψ		volume ⁴
		Frequency	WBV	WBV	WBV	WBV
		in Hz	Without Grip	With Grip	Without Grip	With Grip
Men	ON	3	-0.034 ± 0.11^{abcd}	0.007 ± 0.09^{abd}	0.046± 0.10 abod	0.124 ± 0.13^{abd}
	OFF	3	-0.044 ± 0.07 abod	-0.005 ± 0.08 abd	-0.04 ± 0.11 abod	0.028 ± 0.14^{abd}
	ON	4.5	0.029 ± 0.18^{cd}	0.114 ± 0.27 ^{ad}	0.111 ± 0.09 ^{cd}	0.197 ± 0.08 ^{ad}
	OFF	4.5	0.020 ± 0.11 ^{cd}	0.138 ± 0.34 ad	0.123 ± 0.13 ^{cd}	0.207 ± 0.13 ad
	ON	6	0.018 ± 0.09 [∞]	0.072 ± 0.10 ^{bd}	0.071 ± 0.32 ^{cd}	0.189 ± 0.35 ^{bd}
	OFF	6	0.015 ± 0.11 ^{cd}	0.068 ± 0.12 ^{bd}	0.083 ± 0.16^{cd}	0.193 ± 0.21 ^{bd}
Women	ON	3	$-0.004 \pm 0.05^{\text{abc}}$	0.015 ± 0.06 *	$0.016 \pm 0.03^{\text{abc}}$	0.045 ± 0.05 *
	OFF	3	-0.013 ± 0.02^{abc}	0.003 ± 0.03 ^{sb}	0.016 ± 0.03 ^{abc}	-0.008 ± 0.23 *
	ON	4.5	0.022 ± 0.06 °	0.043 ± 0.07*	0.051 ± 0.06 °	0.096 ± 0.06 •
	OFF	4.5	0.008 ± 0.04 ^c	0.033 ± 0.04 ^a	0.050 ± 0.05 °	0.088 ± 0.05*
	ON	6	0.044 ± 0.10°	0.065 ± 0.09	0.061 ± 0.1 °	0.114 ± 0.1
	OFF	6	0.056 ± 0.10 ^c	0.095 ± 0.07	0.094 ± 0.07 °	0.135 ± 0.08

Table 2.3: Cerebral Oxygenation and Blood Volume differences in Optical Density units (Mean ± SD) in healthy men and women during WBV at three different doses

* A is defined as the difference between 'maximum optical density' [during seated WBV with or without rhythmic handgrip contraction] and 'minimum optical density' [during initial rest without WBV or after recovery from WBV session with a backrest condition (recovery 1)]. NOTE: All subjects were exposed to 3, 4.5 and 6 Hz doses.

a - significant difference between doses 3 Hz and 4.5 Hz (p < 0.01)

b - significant difference between doses 3 Hz and 6 Hz (p < 0.01) c - significant difference between vibration alone and vibration combined with rhythmic handgrip (p < 0.05)

d - significant interaction between work and gender (for oxygenation - p < 0.05); for blood volume - p < 0.01)

DOSE	BACKREST	ME	MEN		IEN
Frequency in Hz		Low-back	Knee	Low-back	Knee
3	WITH	0.56	-0.16	0.33	-0.36
	WITHOUT	0.06	-0.42	-0.07	-0.27
4.5	WITH	-0.54	-0.32	-0.01	-0.35
	WITHOUT	-0.19	-0.39	-0.12	-0.15
6	WITH	0.14	0.38	-0.41	0.04
	WITHOUT	0.46	0.51	-0.10	-0.24

Table 2.4: Pearson Correlations between Angles measured during Sitting and Acceleration at C6

Table 2.5: Pearson Correlations between Cerebral [Oxygenation and Blood Volume] responses and Selected Variables during Sitting 'With' Backrest

GENDER	DOSE	Selected Variables	Oxyger	nation	Blood v	olume
	Frequency		WBV	WBV	WBV	WBV
	in Hz		Without Grip	With Grip	Without Grip	With Grip
Men		Acceleration at C6	0.49	0.15	-0.04	-0.11
	3	Diameter at C6	0.05	-0.06	-0.28	-0.19
		Neck Length	-0.06	-0.20	0.22	0.02
		Low-back Angle	0.33	0.11	-0.04	0.01
		Acceleration at C6	-0.15	-0.37	-0.05	0.09
	4.5	Diameter at C6	-0.30	0.04	0.01	0.04
		Neck Length	0.02	-0.14	0.49	0.52
		Low-back Angle	-0.11	0.01	0.37	0.32
		Acceleration at C6	0.13	0.31	-0.34	-0.36
	6	Diameter at C6	0.16	0.29	0.37	0.40
	}	Neck Length	-0.03	0.30	-0.03	-0.09
		Low-back Angle	-0.03	0.32	0.18	-0.03
Women		Acceleration at C6	0.19	0.16	-0.13	-0.17
	3	Diameter at C6	0.07	-0.05	-0.15	-0.25
ļ		Neck Length	-0.06	-0.13	-0.16	-0.19
l		Low-back Angle	-0.01	0.16	0.33	0.03
		Acceleration at C6	-0.07	-0.11	0.25	0.15
1	4.5	Diameter at C6	-0.25	-0.18	-0.37	-0.33
		Neck Length	-0.34	-0.24	-0.52	-0.24
	L	Low-back Angle	-0.08	-0.25	0.23	0.21
		Acceleration at C6	0.15	-0.01	0.10	-0.04
}	6	Diameter at C6	0.49	0.51	0.48	0.44
ł		Neck Length	-0.11	-0.15	-0.25	-0.28
		Low-back Angle	0.19	0.25	-0.09	-0.02

Table 2.6: Pearson Correlations between Maximal Rhythmic Handgrip Contractions and Cerebral [Oxygenation and Blood Volume] responses

DOSE	BACKREST	N	IEN	WOMEN		
Frequency in Hz		Oxygenation	Blood Volume	Oxygenation	Biood Volume	
3	WITH	-0.46	0.03	-0.33	0.15	
	WITHOUT	0.55	0.13	0.32	0.45	
4.5	WITH	0.02	-0.01	0.21	0.43	
	WITHOUT	-0.55	-0.10	0.25	-0.16	
6	WITH	0.15	0.52	-0.27	-0.34	
	WITHOUT	0.24	-0.04	-0.47	-0.11	

Table 2.7: Pearson Correlations between Cerebral [Oxygenation and Blood Volume] responses and Relative Peak Aerobic Power

GENDER	BACKREST	DOSE	Oxyge	nation	Blood \	Blood Volume	
	}	Frequency	WBV	WBV	WBV	WBV	
		in Hz	Without Grip	With Grip	Without Grip	With Grip	
Men	WITH	3	0.22	-0.01	0.48	0.51	
	WITHOUT	3	0.14	0.20	-0.24	-0.26	
	WITH	4.5	0.41	0.22	-0.21	-0.08	
	WITHOUT	4.5	-0.38	-0.45	-0.10	0.10	
	WITH	6	-0.33	0.14	0.31	0.44	
	WITHOUT	6	0.04	0.10	0.01	0.06	
Women	WITH	3	-0.12	-0.05	-0.11	-0.06	
	WITHOUT	3	-0.25	-0.01	-0.09	-0.17	
	WITH	4.5	-0.06	-0.06	0.14	0.13	
	WITHOUT	4.5	0.14	0.27	-0.07	-0.05	
	WITH	6	-0.46	-0.52	-0.57	-0.52	
	WITHOUT	6	-0.44	-0.47	-0.41	-0.48	


Figure 2.1A Vibration Simulator



Figure 2.1B Vibration Simulator



Figure 2.2 Near Infra-red Spectroscopy Unit



Figure 2.3A NIRS Sensor Placement on the Right Forehead

(The sensor was positioned approximately 3 cm from midline and 1 cm above supraorbital ridge)



Figure 2.3B Tensor Bandage on the Head



Figure 2.4A Subject Sitting 'Without' Backrest



Figure 2.4B Subject Sitting 'Without' Backrest Combined with Handgrip Contractions



Figure 2.5 Subject Setup during Sitting 'With' Backrest

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Cerebral oxygenation (top panel) and blood volume (bottom panel) trends in a typical male and female subject during exposure to seated whole-body vibration with and without handgrip exercise



Figure 2.7A. Interaction between Gender and Work Mean Oxygenation Changes (P<0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest).

3. Graph suggests that during Vibration Alone condition, women demonstrated greater oxygenation changes than men. However, men demonstrated greater oxygenation changes during Vibration with Rhythmic Handgrip contractions.



Figure 2.7B. Interaction between Gender and Work Mean Blood Volume Changes (P<0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest).

3. Graph suggests that during both conditions of Vibration Alone and Vibration with Rhythmic Handgrip contractions, men demonstrated greater blood volume changes than women.



Figure 2.8A. Interaction between Gender and Dose Mean Oxygenation Changes (P>0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Backrest and Work).

3. It appears from the above graph that men and women demonstrate distinct patterns of oxygenation changes. However, because of greater varaince obtained, Dose did not influnce the results.



NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Backrest and Work).

3. It appears from the above graph that men and women demonstrate distinct patterns of blood volume changes. However, because of greater varaince obtained, Dose did not influnce the results.

CHAPTER 3

Lumbar Erector Spinae Oxygenation and Blood Volume Responses during Exposure to Seated Whole-body Vibration

According to Guinard (1985), vibration is defined as any continuous or intermittent oscillating mechanical force or motion that affects humans at work through the mediation of anatomical structures and sensory receptors other than the organ of hearing. Whole-body vibration (WBV) is transmitted to the entire body through a supporting structure such as a vehicle seat in a car, tractor, ship, aircraft, and snowmobile or through lifts, escalators in buildings, and from the shop floor. Extensive literature indicates that human exposure to WBV is a risk factor for the development of low-back pain (Bovenzi 1996, Dupuis and Zerlett 1986, Hulshof and van Zanten 1987, Pope 1989, Seidel and Heide 1986). Aircrew and helicopter pilots (Simon-Arndt et al. 1997) and drivers of earthmoving vehicles (Miyashita et al. 1992), agricultural tractor drivers (Boshuizen et al. 1990, Bovenzi and Betta 1994), and buses and transit drivers (Bongers et al. 1988) are some of the occupational groups who are at a greater risk for low-back complaints than the control groups unexposed to WBV.

Establishing dose-response relationships of WBV in humans is important, as driving has become an integral part of life for everybody including the specific occupational groups mentioned above. However, the majority of research has not established a quantitative relationship between exposure (dose) and effects (physiological responses). Some of the reasons include inadequate controls of concomitant exposures to potentially confounding factors, such as prolonged sitting, awkward postures adopted, frequent twisting of the spine during work, handling various controls, and frequent heavy manual material handling.

Primary factors that influence human response to WBV are: the source (vibrating base, vehicle dynamics), interface such as a seat cushion and the host (humans). The host's physical characteristics (e.g. body dynamics, gender, age, weight) and physiological differences (e.g. fitness) also may influence human performance and responses during exposure to WBV (Boff and Lincoln 1988,

Griffin 1990). The most dominant frequency dose identified in vehicles ranges from 3 to 6 Hz (Wilder et al. 1982), with the transfer of maximum mechanical energy to the spine resulting at about 4.5 to 5 Hz (Panjabi et al. 1986, Pope et al. 1998, Wilder et al. 1982).

Electromyographic WBV studies of the paraspinal muscles in the low-back demonstrated the development of localized muscular fatigue in a variety of postural positions and vibration doses (Hansson et al. 1991, Pope et al. 1998, Wilder et al. 1982). It was suggested that this localized fatique could be due to compromised blood flow, resulting in reduced oxygenation and availability of nutrients to these muscles (Hansson et al. 1991, Pope et al. 1990; Zimmerman et al. 1993). This hypothesis, however, has yet to be objectively tested. Using a backrest during driving is common. Several authors have suggested that vibration of a backrest during driving may amplify the vibration transmission to the human body (Cho and Yoon 2001, Griffin 1990, Magnusson et al. 1992, Paddan and Griffin (1988a,b). This raises an important question. Physiologically, does the presence of a backrest increase or decrease lumbar erector spinae muscle oxygenation and blood volume responses? Those employed in occupations with WBV, often work not only in prolonged sitting but also operating manual controls. This demands isometric work of the forearm muscles at varying percentages of maximal voluntary contraction. Hence, it is necessary to understand physiological responses due to rhythmic handgrip during exposure to WBV. Palmer et al. (2000) demonstrated that in general occupational exposure to WBV in men over a one-week period is approximately four times greater than in women. However, research pertaining to differences between genders with respect to physiological responses during exposure to WBV is limited (Bovenzi 1996). Also, the role of physical fitness on responses to WBV has not been studied to date.

Near infra-red spectroscopy (NIRS) is a non-invasive optical technique that has been used to measure relative changes in oxygenation and blood volume levels in skeletal muscle continuously in real-time during rest and work (Bhambhani et al. 1998, Boushel et al. 1998, Chance et al. 1992, Mancini et al.

130

1994, Murthy et al. 1997). Muscle oxygenation, defined as the relative saturation of oxyhemoglobin and oxymyoglobin, depends on the balance between oxygen delivery and oxygen utilization (Mancini et al. 1994). Blood volume is defined as the sum of oxygenated and deoxygenated states of hemoglobin and myoglobin in the illuminated region of the muscle. Detailed description of the NIRS theory, its validation with various established optical and magnetic techniques are provided in Chapter 1. The reliability of NIRS in the upper and lower extremity human skeletal muscles has been reported in the Chapter 1 as well. In establishing the reliability of NIRS in the erector spine muscle during maximal back extension, Maikala et al. (2000) reported significant intra-class correlation coefficients of 0.83 and 0.84 for minimum oxygenation during sitting and standing respectively, and 0.99 for the minimum blood volume during both postures. To date, erector spinae oxygenation and blood volume changes using NIRS have been limited to sub-maximal and maximal isometric contractions (Jensen et al. 1999, McGill et al. 2000, Maikala et al. 2000, Maikala and Bhambhani 1999, Maronitis et al. 2000, Yoshitake et al. 2001). No studies have evaluated the oxygenation and blood volume trends of any skeletal muscle group during exposure to WBV.

The purposes of this study were to examine: (1) the effects of 3 Hz, 4.5 Hz, and 6 Hz vibration dose on the relative changes in oxygenation and blood volume measured by NIRS in the lumbar erector spinae muscle in healthy men and women; (2) the differences in these trends 'With' and 'Without' a backrest exposure to WBV; (3) alterations in these responses with the addition of rhythmic handgrip contractions during WBV, and (4) differences in these responses between genders. It was hypothesized that erector spinae oxygenation and blood volume responses will be similar in both men and women 'With' and 'Without' backrest, and will not be influenced by the exposure to vibration dose within the range of 3 to 6 Hz while performing rhythmic handgrip contractions.

MATERIALS AND METHODS

Subjects

Written informed consent was obtained from 13 healthy men (mean \pm SD: age 24.7 \pm 3.9 yrs, mass 70.8 \pm 12.1 kg, height 171 \pm 7 cm) and 14 healthy women (age 23.9 \pm 3.51 yrs, mass 60.3 \pm 9.0 kg, height 164 \pm 10 cm). Subjects were undergraduate and graduate students recruited from the university population (Table 3.1). They all completed the Revised Physical Activity Readiness Questionnaire (Canadian Society for Exercise Physiology, 1994) to identify contraindications for exercise. All subjects were right hand dominant. The test protocols utilized were approved by the Health Ethics Research Board at this institution (APPENDIX A.1).

Vibrating Base

A steel vibrating base that housed an electro-mechanical motor connected to a cam mechanism (Advanced Therapy Products, Inc., VA, USA) was modified with an external DC speed control device (Figures 2.1A and 2.1B). Varying the speed of this motor from 0 to 100% corresponded to vibrating frequencies of 0 Hz to 6.2 Hz. Detailed description of the vibrating base, placement and calibration of accelerometers were described in the Chapter 2. Average vertical accelerations (in root mean square values [r.m.s]) generated at different doses are also provided in the Chapter 2.

Test Protocols

Each subject completed an arm cranking exercise test in the first session, followed by three WBV tests on separate days (3 Hz, 4.5 Hz, and 6 Hz) in a randomized order.

(1) Assessment of the Peak Oxygen Uptake

In the first testing session, the subject completed an aerobic fitness test on an arm cranking ergometer (Cybex, MET 300, USA). Details of the peak oxygen

132

uptake measurements and the exercise termination criteria are explained in the Chapter 2.

(2) Vibration Tests

The test protocol on each day lasted 30 minutes, and included initial sixminute rest without vibration, eight-minute vibration 'With' (or 'Without') a backrest, four-minute recovery, eight-minute vibration 'Without' (or 'With') a backrest, and four-minute recovery. Subjects were randomly assigned to a posture 'With' or 'Without' a backrest. They were instructed to adopt a comfortable posture when tested 'Without' a backrest (Figures 2.4A and 2.5). Throughout the test, subjects sat in the vibrating chair with the eyes open. A footrest was provided for subjects to place their feet at a comfortable level. During sitting 'With' and 'Without' a backrest, angles at the knee joint and lowback were measured by a goniometer (JAMAR, Clifton, NJ, USA).

(3) Rhythmic Handgrip Measurements

Before the vibration session, right hand grip size was adjusted on the dynamometer (JAMAR, CA, USA) to a position that was comfortable for each subject. During the fifth minute of each vibration condition, the subject performed rhythmic maximal contractions for one minute at a contraction rate of one every five seconds (Figure 2.4B). Contractions were performed with the right hand with the forearm in a horizontal position and the elbow at an angle of 180°. Similar to a method suggested by Schmidt and Toews (1970), throughout the study the dynamometer was calibrated with known weights to check its accuracy identified through the dial reading on the dynamometer.

(4) Oxygenation and blood volume measurements

A continuous dual wave NIRS unit (MicroRunman, NIM Inc., PA, USA) was used to evaluate relative changes in the lumbar erector spinae oxygenation and blood volume during the vibration tests (Figure 2.2). This unit consists of: a superficial near infra-red sensor fitted with six light emitting tungsten lamps

placed 2 cm, 3 cm, and 4 cm part; two photo-diode detectors that absorb light in the near infra-red range between 760 nm and 850 nm; and a display unit that amplifies and displays the absorbency signal generated. Sensor was placed on the erector spinae muscle at 3 cm from the middle of spine to the right side on the third lumbar region [Figure 3.1A] (Maikala et al. 2000). A dark elastic bandage was wrapped around the sensor to secure it in place and minimize any loss of light (Figure 3.1B). A piece of clear plastic was wrapped around the sensor to prevent sweat from distorting the absorbency signal. Erector spinae muscle skinfold thickness at the sensor location was measured twice using a skinfold caliper (Cambridge Scientific Industries, Inc. Maryland, USA) before placing the NIRS sensor on that muscle belly. Then adipose tissue thickness (a sum of fat and skin layer) at the sensor site was calculated as a half of the skinfold thickness. Position of the NIRS sensor at the 3rd Lumbar region from the mid line of the spine was noted to the nearest millimeter, thus ensuring the correct placement of NIRS sensor on each subject for all the testing sessions.

As per the manufacturer's specifications, the sensor was calibrated at 760 nm and 850 nm wavelengths before the test for each subject. The vertical penetration depth of light was set to 4 cm. The difference in absorbency between these two wavelengths indicated the change in oxygenation saturation and the sum of these absorbencies indicated the change in total blood volume. Both oxygenation and blood volume were measured in terms of optical density (OD) that is defined as the logarithmic ratio of intensity of light calibrated at each wavelength to the intensity of measured light at the same wavelength (MicroRunman User Manual, NIM Inc., PA, USA, 1999). Real time data were recorded using the NIRCOM software provided by the manufacturer.

NIRS Data Analysis

The real time NIRS values were averaged every 20 seconds using a customized Microsoft Excel software program specially designed for this study. Baseline oxygenation and blood volume values were recorded during initial rest or recovery from each vibration dose. Minimum values were recorded for each

dose during vibration with and without rhythmic handgrip, and 'With' and 'Without' a backrest. The change (delta value) in oxygenation and blood volume for each condition [dose (3Hz, 4.5 Hz and 6 Hz), backrest (With' and 'Without' a backrest), workload (vibration alone and vibration combined with rhythmic handgrip contraction)] was calculated as the difference between the baseline and minimum values for each phase of the test session. It should be noted that the greater the delta values the greater the decrease in oxygenation and blood volume changes. Oxygenation and blood volume values 30 seconds before the beginning of each WBV session were identified as baseline values, and were used to calculate delta values. All statistical analyses were performed using the delta values calculated for each phase of the study.

Statistical Analysis

A four-way analysis of covariance (ANCOVA), using the peak oxygen uptake as the covariate, gender as a between-subjects factor, and three repeated measures factors (dose, backrest, and workload) with a fully crossed design was used to evaluate the differences in the oxygenation and blood volume responses (measured in OD units). Since peak oxygen uptake is considered to be one of the best indicators of aerobic fitness, it will be used to examine whether the fitness level of the subjects has any influence on the lumbar erector spinae oxygenation and blood volume responses.

To minimize the violation of the assumption of homogeneity of variance, the 'Greenhouse-Geiser' adjustment was used. Since multiple comparisons were made for each dependent variable in the present study, the Bonferroni adjustment for a P value of 0.05 was applied to minimize Type 1 error. The Bonferroni adjustment was calculated as the ratio of 0.05 and number of statistical comparisons (APPENDIX B.2). Thus, statistical significance was considered at P <0.01. Significant F ratios were further analyzed with the Scheffe post hoc multiple comparison test. Also, the relationship between maximal rhythmic handgrip contractions and changes in oxygenation and blood volume responses were evaluated using Pearson product moment correlations. A

Spearman correlation test was used to examine the relationship between logarithmic value of adipose tissue thickness and [minimum and maximum] oxygenation and blood volume responses for the three protocols. Since the adipose tissue thickness of the subjects in the present study was normally distributed, a nonlinear correlation test was used to examine its role on the NIRS measurements. The Statistical Package for Social Sciences SPSS (version 10) was used for all statistical analyses (SPSS Inc., Chicago, USA).

RESULTS

Lumbar Erector Spinae Oxygenation and Blood Volume Trends in Men and Women

A typical profile of the lumbar erector spinae oxygenation and blood volume trends at 4.5 Hz, 'With' and 'Without' a backrest in a male and female subject is shown in Figures 3.2A and 3.2B respectively. It is evident from both figures that as soon as the vibration started 'Without' a backrest, oxygenation and blood volume responses increased for about 20-30 seconds, and then decreased from the baseline (resting) value during the four-minute period. During vibration combined with rhythmic handgrip contractions, oxygenation values decreased to a lower level than vibration alone. As soon as the rhythmic handgrip contractions stopped during vibration, oxygenation (Figure 3.2A) and blood volume (Figure 3.2B) trends returned to the initial vibration level. In the recovery phase, oxygenation and blood volume responses increased towards the baseline values (recovery 1). In the subsequent 'With' backrest condition of vibration period without rhythmic handgrip contractions, similar trends were observed in the oxygenation and blood volume responses with the values less decreasing than those observed during 'Without' a backrest condition.

The lumbar erector spinae oxygenation and blood volume differences during the WBV experimental conditions in both genders are presented in Table 3.2. Statistical analysis of lumbar erector spinae oxygenation and blood volume values for each dose during the different experimental conditions in both men

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

and women revealed: (1) the covariate, namely, peak oxygen uptake, had no significant influence on the erector spinae oxygenation and blood volume responses, (2) the four-way and three-way interactions were not significant implying that both genders responded similarly to the three doses of WBV 'With' and 'Without backrest, and with and without the rhythmic handgrip contractions, (3) the interaction between gender and workload was significant for only oxygenation but not blood volume responses, (4) the interaction between gender and backrest was significant for only blood volume responses but not oxygenation, (5) the main effects of dose and gender were not significant for both oxygenation and blood volume responses, (6) the main effects of workload were significant for both oxygenation and blood volume responses, and (7) the main effects of backrest were significant only for blood volume responses.

It should be noted that the comparisons of the significant two-way interaction were based on the pooled mean values at the three different doses and both genders. Similarly comparison of the main effects was also based on the pooled mean values of the remaining three independent variables in the experimental design. The results of these significant main effects and interactions are presented below.

Comparison between Doses

No significant differences were observed in mean oxygenation and blood volume values for all doses (3Hz: 0.018 ± 0.06 OD, 4.5 Hz: 0.037 ± 0.13 OD, 6Hz, 0.025 ± 0.08 OD, P=0.636). Corresponding comparisons for the blood volume changes (P=0.405) were: 0.005 ± 0.13 OD (3Hz), 0.032 ± 0.14 OD (4.5 Hz), 0.006 ± 0.12 OD (6Hz).

Comparison between 'With' and "without' a Backrest Support

No statistical differences were found in the values of mean oxygenation between 'With' and 'Without' backrest conditions. There was no significant difference in the mean oxygenation responses 'With' and 'Without' backrest conditions (0.029 ± 0.11 OD Vs 0.024 ± 0.08 OD, P=0.271). Interestingly, mean 137 values for blood volume 'Without' a backrest were significantly higher than that of 'With' backrest condition (0.056 ± 0.15 OD Vs -0.028 ± 0.09 OD, P=0.000).

Comparison between Vibration Alone and Vibration Combined with Rhythmic Handgrip Contractions

Significant differences in mean oxygenation and blood volume responses were observed between vibration alone and vibration combined with rhythmic handgrip contractions. Mean oxygenation values were significantly higher during vibration combined with rhythmic handgrip contractions than vibration alone $(0.034 \pm 0.10 \text{ OD Vs} 0.019 \pm 0.09 \text{ OD}, P=0.000)$, Also, mean blood volume values were significantly higher during vibration combined with rhythmic handgrip contractions than vibration alone $(0.020 \pm 0.13 \text{ OD Vs} 0.008 \pm 0.13 \text{ OD})$ P=0.009). Significant interaction was obtained between workload and gender for only oxygenation values (Figure 3.3A). During both conditions of workload, men showed greater mean oxygenation changes (Vibration alone: 0.024 ± 0.12 OD; Vibration combined with handgrip: 048 ± 0.13 OD) than women (Vibration alone: 0.015 ± 0.06 OD; Vibration combined with handgrip: 0.021 ± 0.06 OD, P=0.020). Even though statistically not significant, during vibration alone men demonstrated less mean blood volume changes than women $(0.006 \pm 0.17 \text{ OD Vs} 0.010 \pm 0.08)$ OD, P=0.264). Also, during vibration combined with rhythmic handgrip contractions, men did not show significant difference in mean blood volume changes from women (0.023 \pm 0.16 OD Vs 0.017 \pm 0.08 OD, P=0.264) (Figure 3.3B).

Comparison between Genders

No significant differences in oxygenation and blood volume were found between men and women for all conditions of dose, workload and backrest. However, there was a tendency for the mean oxygenation and blood volume responses to be consistently higher in men. In men, mean values for oxygenation were higher than that of women $(0.036 \pm 0.12 \text{ OD Vs } 0.018 \pm 0.06 \text{ OD, P=0.164})$. Blood volume values were also higher in men compared to women $(0.015 \pm 0.17$ 138 OD Vs 0.014 \pm 0.08 OD, P=0.952). No significant interaction between gender and dose was observed in both oxygenation and blood volume responses. However, men showed greatest values at 4.5 Hz (P>0.05) and women peaked at 6 Hz (P>0.05) (Figures 3.4A and 3.4B). Interestingly, significant interaction between gender and backrest was observed in blood volume responses only, with greater values reported 'Without' backrest condition for both genders (Fig 3.5A and 3.5B).

Correlations between Selected Variables and Oxygenation and Blood Volume

Pearson correlations obtained between maximal rhythmic handgrip contractions and changes in oxygenation and blood volume responses were not significant (P>0.05), suggesting maximal effort did not influence the muscle NIRS measurements (Table 3.3). Further, adipose tissue thickness calculated from erector spinae skinfolds for men and women (mean \pm SD) are: 5.7 \pm 2.5 mm, and 5.8 \pm 2.2 mm, respectively. They were not significantly different from each other (P>0.05). Spearman correlations between adipose tissue thickness and mean changes in oxygenation and blood volume data for both men and women during three protocols are shown in Table 3.4. No significant correlations (P>0.05) were found between adipose tissue thickness at the lumbar erector spinae muscle and oxygenation and blood volume responses.

DISCUSSION

Lumbar Erector Spinae Oxygenation and Blood Volume Trends during Whole-Body vibration

There has been, no studies to date that have examined the lumbar erector spinae oxygenation and blood volume responses during WBV in any species. Hence it is difficult to compare current results with the existing vibration literature. Using photon diffusion principle, NIRS can detect oxygenation and blood volume of skeletal muscle during any state, and in the present study decrease in muscle

139

blood volume and oxygenation during various WBV conditions demonstrates that a reduction in erector spinae oxygen saturation is possible during exposure to WBV (Figures. 3.2A and 3.2B). Such a decrease in oxygenation may also be due to an increase in intramuscular pressure above perfusion pressure, thus limiting blood flow to the working muscle (Rundell et al. 1996).

NIRS signal is treated primarily as a representative of capillary and venous HbO₂ saturation (Mancini et al. 1994a). Since NIRS monitors oxygen saturation at the levels of smaller blood vessels, a decrease in oxygenation and blood volume trends during WBV confirmed deficiency in the vascular supply to the low-back muscles during WBV. During recovery, blood supply (in terms of increase in 'optical' blood volume, thus blood flow) and reoxygenation to these paraspinal muscles returned closer to the baseline. Such an increase in local muscle blood volume can be due to larger capillary volume through local metabolic vasodilation.

Repeated contraction of lumbar erector spinae muscle during WBV in addition to static sitting, might have resulted from vasoconstriction of venous blood, and increase in venous return, thus reflecting a reduction in blood volume. Yoshitake et al. (2001) hypothesized a similar decrease in both oxygenation and blood volume responses observed during maximal back extension in a prone position might be due to ischemic muscle activity in the low-back region, specifically in the muscles that contain fast twitch fibers. Also, Maikala et al. (2000) showed a similar decrease in both oxygenation and blood volume in the right side of lumbar erector spinae muscles during maximal back extension in both sitting and standing postures. McGill et al. (2000) hypothesized a few mechanisms for such a decrease in oxygenation: compromised blood flow in these smaller blood vessels (due to intramuscluar pressure exceeding intravascular pressure), thus reducing blood volume; reduced perfusion and increase in oxygen extraction from the muscle; or skin blood flow. However, NIRS signal primarily monitors muscle not skin oxygenation (Mancini et al. 1994). Also, an imbalance between oxygen supply and uptake may lead to a greater deoxygenation in the muscle, thus reducing muscle oxygen saturation.

However, in the present study, some of the subjects did not show any steep decrease, and few in fact demonstrated an increase in blood volume changes during WBV combined with rhythmic handgrip contractions. This may be due to the diffusion limitation between the capillaries and erector spinae muscle. Boushel et al. (1998) suggested a similar hypothesis during rhythmic handgrip exercise. They found oxygen saturation of venous blood (collected from antecubital space) remained the same (after 15% to 30% contractions) even though a decrease in pH was found. They speculated this unchanged value may also be due to contribution of blood from less active tissues, and flow impeded by the mechanical forces generated during muscle contraction.

Jorgensen (1997) reported that "paravertebral muscles are dominated by slow twitch fibers, with small fiber areas and well-developed capillary network." Junghanns (1990) suggested the relation between the spinal movements and diffusion of nutrition to the intervetebral discs, and attributed exposure to WBV and prolonged sitting (without movement) may lead to muscle fatigue in the lowback. Seroussi et al. (1989) compared muscle fatigue during static sitting and WBV, and concluded that fatigue observed in the lumbar erector spinae muscles after exposure to WBV may affect the load-bearing capacity of the trunk muscles. Several researchers have hypothesized that low-back muscle fatigue might be due to the prolonged cyclic firing of these paraspinal muscles during WBV (Pope et al. 1992, Pope et al. 1998, Pope and DeVocht 1999, Seroussi et al. 1989, Wilder et al. 1982, Zimmerman et al. 1993).

Role of Whole-Body vibration Dose:

Since the mechanical resonance range of the human spine for the seated operator is between 4-8 Hz (International Organization for Standardization 2631/part1 1997), it is evident from the current investigation that dose responses below resonance are different from the responses in the range of human resonance. As the dose increased from 3 to 6 Hz, men showed greatest lumbar erector spinae oxygenation and blood volume values at 4.5 Hz (P>0.05) while women demonstrated greatest values at 6 Hz (P>0.05) (Figures 3.4A and 3.4B).

Several authors suggested the frequency shift from a higher to a lower level in the EMG profile during WBV as an indication of low-back muscle fatigue (Hansson et al. 1991, Pope et al. 1998, Seidel and Heide 1986, Wilder et al. 1982). Based on EMG findings, Hansson et al. (1991) hypothesized that exposure to WBV would result in a reduced arterial supply to the back muscles. The present study confirmed a reduction in both blood volume and oxygenation in the lumbar erector spinae muscles during exposure to WBV (Figures 3.1A and 3.1B), thus suggesting WBV might disturb the blood supply to the erector spinae muscles in the lumbar region.

Seroussi et al. (1989) showed peak-to-peak torque generated at L3 of the erector spinae muscles during vibration was maximum at 4 Hz, and decreased as the frequency increased. These authors also demonstrated that mean and peak-to-peak torque generated during WBV was also significantly higher than static sitting. Hansson et al. (1991) studied the EMG activity of erector spinae muscles in subjects during sitting (at L3 and seventh thoracic) for a period of 5 min with and without WBV at 5 Hz, and demonstrated that vibration increases both the speed and the magnitude of the development of muscle fatigue.

In the present investigation, even though exposure to dose below (3 Hz) and in the human spinal resonance range (4.5 and 6 Hz) did not influence significant difference in oxygenation and blood volume responses. However, the decrease in trend in these local physiological responses compared to sitting without WBV shows the role of prolonged exposure of WBV in muscle desaturation of the human back in both men and women.

Role of Sitting 'With' and 'Without' a Backrest Support:

Mean oxygenation and blood volume values obtained in the present study 'Without' backrest were 20% (P>0.05) and 150% (P<0.05) higher compared to 'With' backrest values. During sitting 'Without' a backrest, it has been speculated that sustained isometric contraction of the paravertebral muscles in maintaining this erect posture for a prolonged period will slow down the pumping mechanism, thus disturbing the nutritional diffusion (Junghanns 1990). This would result in decrease in blood flow (represented by reduction in blood volume changes as shown in Fig. 3.2B), resulting in deprivation of oxygenation availability to these postural muscles. However, non-significant responses in mean oxygenation values during 'With' and 'Without' a backrest may suggest that oxygen availability and utilization were equally balanced in these two conditions, with blood flow compromised to a greater extent during sitting 'Without' backrest as compared to 'With' backrest.

During exposure to WBV for 30 min, Wilder et al. (1982) did not find any statistically significant changes in the EMG activity in both right erector spinae and external oblique muscles at the L3 level due to postural variations (neutral posture, neutral and valsalva position, forward flexion (5°), extension (5°), left and right lateral bend (5°), and maximal left/right axial rotation). In a real driving task on the road, Hosea et al. (1986) evaluated the responses of EMG activities of 12 paraspinal muscles with respect to backrest inclination and lumbar support during a 3.5 h period of car driving. These authors did not find any EMG evidence of fatigue (measured by decrease in mean power frequency and increase in amplitude for equal function) to the duration of WBV exposure. However, minimum EMG activity was reported at 130° inclination of the backrest, 13.5°-18.5° of the seat angle, and 5 cm of the lumbar support.

During 2 min of vertical vibration at 4.5 Hz, the EMG magnitude in the erector spinae muscles was greater for the anterior and neutral postures, with anterior posture being recorded highest than for the posterior lean posture (Zimmerman et al. 1993). These authors demonstrated greater cyclic compression of the spine during anterior lean posture, and hypothesized that during prolonged periods in this position such cyclic compression may result in decreased nutrient diffusion, thus leading to earlier onset of erector spinae muscle fatigue. In a simulated truck driving, Wilder et al. (1994) vibrated subjects for a period of 10 min (1-20 Hz), and based on the surface EMG results demonstrated that upright sitting causes more fatigue than during any other posture. However, seat suspension did not affect the EMG responses, and these authors suggested that such results might be reflective of monitoring for only 143

short-duration. Sitting posture in most of the above EMG studies has used either a slightly flexed posture or subjects were loaded with a small weight so as to register erector spinae activity. Since NIRS identifies oxygenation and blood volume changes in any postural variation during rest and work, we believe it is not necessary to load the subjects with any weights during sitting.

Studies pertaining to muscle oxygenation and blood volume of postural variations during WBV have not been reported to date. However, few studies have reported influence of posture on other skeletal muscle oxygenation and blood volume responses in other activities. Rundell et al. (1996) demonstrated that deoxygenation was greater during low-skating position compared to upright position. These authors also reported lower blood volume changes during low-skating position suggesting that this posture might compromise blood flow to the working muscles, thus limiting oxygen delivery. Binzoni et al. (1997) and Ferrari et al. (1997) demonstrated that changes in calf muscle deoxyhemoglobin and blood volume increased as the bed angle increased, and returned to base line when the bed position was reset, thus suggesting that NIRS was sensitive enough to monitor postural changes. Even though none of the NIRS studies cited above were related to WBV, it can be concluded that posture adopted 'With' and 'Without' a backrest plays a role in determining the changes in oxygenation and blood volume responses, in particular mean blood volume response during WBV.

Role of Workload during Whole-Body Vibration:

During vibration combined with rhythmic handgrip contractions, mean oxygenation and blood volume were 43.3% and 60% respectively higher than vibration alone, suggesting a greater decrease in skeletal muscle oxygenation and blood volume responses when subjects perform physical activity combined with exposure to WBV. From figures 3.2A and 3.2B, it is obvious that oxygenation and blood volume responses decreased from baseline during vibration alone, and continued to decline during vibration combined with work.

Mester et al. (1999) reported that people involved in sports such as sailing, surfing, alpine skiing, horse-back riding, etc., experience considerable

vibration. Their group (Hoffmann et al. 1999) studied the role of isometric contraction during with and without vibration on muscular energy metabolism. Subjects did isometric plantar flexion against a vibrating force generator (at 4 and 8 Hz) in a 4.7 Telsa Magnet. These authors showed that phosphocreatine decreased rapidly by 50% during vibration exposure compared to no exposure, suggesting an increased rate of adenosine triphosphate hydrolysis. Further, they suggested that the minimal level of relative phosphocreatine obtained might depend on both isometric contraction force and vibration frequency, and emphasized the importance of relating this phenomenon to isometric contractions applied during activities such as alpine skiing. Also, Andersson et al. (1974) observed an increase in myoelectric activity in the lumbar muscles with shifting gears during driving. In the current study, a significant deoxygenation in the right side of erector spinae muscle during WBV aione, and during WBV combined with rhythmic handgrip contractions was observed, supporting the similar findings cited above.

In the present study, valsalva maneuver occurred during maximal rhythmic handgrip contractions. This may result in increased intrathoracic pressure subsequently reducing venous return of the blood to the heart (McArdle et al. 1996). However, reduction in blood volume (Fig. 3.2A) and subsequently oxygenation (Fig. 3.2B) during maximal rhythmic handgrip contractions suggests that muscle co-activation is possible even at the erector spinae level when in fact primary muscles involved during such handgrip activity were forearm muscles (Kahn et al. 2000). Furthermore, the additional 'postural load' of sitting for a prolonged period of time might concomitantly affect such decrease in oxygenation and blood volume responses during work as well. Thus, the present study demonstrated decrease in both oxygenation and blood volume responses during the the physical work during WBV may further burden the low-back muscles.

Role of Gender:

In the present study, gender as 'a main effect' did not influence the oxygenation and blood volume responses. Resonance frequencies associated with WBV in the Z-axis (foot-to-head) for humans range from doses of 4 to 8 Hz (International Organization for Standardization 1997). Interestingly, in the present study women showed a systematic decrease in oxygenation and blood volume responses (or greater mean values) with increases in vibration dose up to 6 Hz, while the responses in men reached nadir at 4.5 Hz (Fig 3.4A and 3.4B). Even though these results were not statistically significant, such gender differences may suggest the influence of gender in erector oxygenation and blood volume responses within the resonance frequency range.

Gender interacted with workload and showed significance in the mean oxygenation values but not in mean blood volume values. Mean changes in oxygenation values during WBV alone for men were 38.5% greater than women. and during WBV combined with rhythmic handarip contractions, men showed 56.3% greater values than women. In the present study, since greater mean (oxygenation and blood volume) values are an indication of greater decrease in oxygenation and blood volume responses, such a decrease in phenomenon in men could be attributed to lower percentages of slow twitch (aerobic) fibers, thus having less oxidative capacity and endurance than women (Jorgensen 1997). From Table 3.1, it is evident that maximal grip strength of men are greater (by 34%) than women, suggesting that greater absolute force produced by the men might result in greater change in deoxygenation, however these responses in men were not significantly (P>0.05) correlated to their grip values (Table 3.3). Hence, the present study results may suggest that fatigability of low-back muscles in men is greater than women during exposure to WBV and work. Bilodeau et al. (1992) showed greater EMG responses (median frequency and mean power frequency) in men compared to women as well, and attributed these trends to the lower fast twitch fibers of women in triceps and biceps brachii. Such fiber type content was evidenced in the EMG power spectrum reported in their study. They also attributed such low EMG responses in women to the smaller diameter of fiber size recruited and higher skinfold thickness.

However, NIRS sensitivity to detect the role of slow and fast twitch fibre dominance is poor. Alfonsi et al. (1999) demonstrated that NIRS trends could not successfully distinguish slow or fast muscle fibre composition in the tibialis anterior during ischemia combined with maximal isometric work. Using time resolved spectroscopy, Kuno et al. (1997) showed a very low positive correlation (r < 0.50) between oxygenation in the vastus lateralis muscle and muscle fibre composition. These authors concluded that muscle fibre composition does not affect the availability of oxygen in the muscle during maximal workout. Reasons for mean blood volume values not achieving significance are not clear at present. Does this suggest mean blood volume changes (thus blood flow to these back muscles) will not be affected by both conditions of work even if there exists an imbalance between oxygen delivery and oxygen uptake? Is it due to blood flow from non-exercising muscles? Such types of questions warrant further investigation.

Based on EMG median frequency slopes, Elfving et al. (2000) investigated gender differences during seated back extension. At the fifth lumbar position, men showed a steeper slope than women implying a greater fatigue, however, their results did not reach significance. In the present study, gender interacted with backrest and showed significance in mean blood volume values but not in mean oxygenation values. 'With' backrest, men showed 79% lower mean blood volume changes than women. However, 'Without' backrest, men showed 51% greater mean blood volume values than women. Such discrepancy in only mean blood volume responses between men and women is not clear at present. Similar mean oxygenation responses of men and women during both backrest conditions observed in the present study might be attributed to balance of oxygen uptake with oxygen delivery to these erector spinae muscles.

In the present study, all the subjects were right handed (however, it was not a selection criteria for the current investigation). Interestingly, few subjects demonstrated increase in oxygenation and blood volume trends during handgrip contractions, suggesting a conflicting evidence of their dominant side responses. In our unpublished observations of back muscle oxygenation and blood volume (on either side of spine) during maximal extension tests showed such variable responses as well. Hence, in both genders significant results of mean oxygenation during work, and significant results of mean blood volume responses during postural variation would have been more clear provided the present study would have monitored both sides of the low-back, thus understanding the influence of side difference and side dominance of individual subjects.

Role of Adipose Tissue Thickness on NIRS measurements

van Beekvelt et al. (2001) reported that adipose tissue metabolism is lower than muscle metabolism, and therefore, muscle oxygenation determined using NIRS is underestimated. In the current study, there was no significant difference in adipose tissue thickness at the erector spinae region between men and women (Table 3.1). Also, the adipose tissue thickness in both genders was not significantly correlated to NIRS measurements [at a source-detector separation of 4 cm] among the three vibration tests (Table 3.4). This suggests that at the same source-detector separation of 4 cm, oxygenation and blood volume values obtained during the different experimental conditions were not influenced by adipose tissue thickness. This is in contrast to the findings of van Beekvelt et al. (2001) who demonstrated a significant decrease in oxygenation with increase in adipose tissue thickness.

The discrepancy between these two studies could be due to the NIRS technique used (continuous wave Vs spatially resolved), exercise protocol (WBV Vs maximal rhythmic contractions of forearm flexors), and sample size (11 men and 11 women Vs 44 men and 34 women). Further, Feng et al. (2001) showed that for an adipose thickness of 8 mm, the optimal source-detector distance was 45 mm. These authors also identified the optimal distance of 35-40 mm for an adipose tissue thickness range of 2.5-5 mm and 50 mm for a thickness of 17 mm. However, in the current study, a fixed source-detector distance of 4 cm was

used, irrespective of skinfold thickness which ranged from 7 to 24 mm. It is possible that this could have influenced the findings of the present study.

Role of Fitness

In the present study, level of aerobic fitness did not influence the oxygenation and blood volume responses during WBV. Using the peak oxygen uptake obtained from incremental arm cranking test till exhaustion as a covariate (Table 3.1), the results suggested that erector spinae oxygenation and blood volume responses in men and women were not dependent upon the level of their aerobic fitness. Further, correlations calculated between the relative aerobic capacity and muscle oxygenation and blood volume responses were not significant (P>0.05) (Table 3.5). This is in contradiction to the hypotheses of Boff and Lincoln (1988) and Griffin (1990) who suggested that the physiological responses to WBV may be influenced by the level of fitness. Since oxygenation and blood volume responses monitored in the erector spinae muscles are of 'peripheral' nature, one can also interpret these results in such a way that the level of fitness may not influence the peripheral responses to WBV but may play a role in the central physiological responses.

Conclusions

In the present study, the role of dose, backrest, workload, and gender on erector spinae oxygenation and blood volume responses during seated WBV was investigated. A decrease in vascular supply (observed through decrease in blood volume) to the back muscles was observed in both men and women during WBV compared to sitting without WBV. Similar mean oxygenation values during both backrest conditions suggest equal oxygen supply and utilization. During both vibration alone and vibration combined with rhythmic handgrip contractions, men showed greater mean oxygenation values than women, suggesting greater deprivation of oxygen occurs (without significant concomitant changes in blood volume) in these working muscles during work combined with WBV. Skinfold thickness did not influence oxygenation and blood volume responses in either

gender. Also, controlling for the level of aerobic fitness showed no influence on oxygenation and blood volume responses in both genders.

REFERENCES

- Alfonsi E, Pavesi R, Merlo IM, Gelmetti A, Zambarbieri D, Lago P, Arrigo A, Reggiani C, and Moglia A (1999) Hemoglobin near-infrared spectroscopy and surface EMG study in muscle ischaemia and fatiguing isometric contraction. J Sports Med Phys Fitness 39: 83-92
- Andersson GBJ, Ortengren R, NachemsonA, and Elfstrom G (1974) Lumbar disc pressure and myolelectric back muscle activity during sitting – Studies on a car driver's seat. Scan J Rehab Med 6: 128-133
- 3. Belardinelli R, (1998) Muscle oxygenation kinetics measured by near-infrared spectroscopy during recovery from exercise in chronic heart failure. *G Ital Cardiol* 28: 866-872
- Bhambhani Y, Maikala R, and Buckley S (1998) Muscle oxygenation during incremental arm and leg exercise in men and women. *Eur J Appl Physiol* 78: 422-431
- 5. Bilodeau M, Arsenault AB, Gravel D, and Bourbonnais D (1992) Influence of gender on the EMG power spectrum during an increasing force level. *J Electromyogr Kinesiol* 2(3): 121-129
- Binzoni T, Cerretelli P, Ferrari M, and Delpy D (1997) Metabolic studies of human skeletal muscle by NIRS: possible applications in space research. Int J Sports Med 18: S312-S314
- 7. Boff KR and Lincoln JE (1988) Vibration and display perception. Engineering Data Compendium: Human Perception and Performance. AAMRL, Wright-Patterson Air Force Base, OH.
- 8. Bongers PM, Boshuizen HC, Hulshof CTJ, and Koemeester AP (1988) Longterm sickness absence due to back disorders in crane operators exposed to whole-body vibration. *Int Arch Occup Environ Health* 61: 59-64
- 9. Boshuizen HC, Bongers PN, and Hulshof CTJ (1990) Self-reported back pain in tractor drivers exposed to whole-body vibration. Int Arch Occup Environ Health 62: 109-115
- 10. Boushel R, Pott F, Madsen P, Radegran G, Nowak M, Quistroff B, and Secher N (1998) Muscle metabolism from near infrared spectroscopy during rhythmic handgrip in humans. *Eur J Appl Physiol* 79: 41-48

- 11. Bovenzi M and Betta A (1994) Low-back disorders in agricultural tractor drivers exposed to whole-body vibration and postural stress. *Appl Ergon* 25(4): 231-241
- 12. Bovenzi M (1996) Low back pain disorders and exposure to whole-body vibration in the workpalce. Sem Perinatolgy 20(1): 38-53
- 13. Canadian Society for Exercise Physiology (1994) Revised Physical Activity Readiness Questionnaire – RPARQ. Gloucester, NB, CANADA.
- 14. Chance B, Dait MT, Zhang C, Hamaoka T, and Hagerman F (1992) Recovery from exercise-induced desaturation in the quadriceps muscles of elite competitive rowers. *Am J Phyisol* 262: C766-C775
- 15. Dupuis H and Zerlett G (1986) The effects of whole-body vibration. Springer-Verlag.
- 16. Elfving B, Nemeth G, and Arvidsson I (2000) Back muscle fatigue in healthy men and women studied by electromyography spectral parameters and subjective ratings. Scand J Rehab Med 32: 117-123
- 17. Feng W, Haishu D, Fenghua T, Jun Z, Qing X, and Xianwu T (2001) Influence of overlying tissue and probe geometry on the sensitivity of a near-infrared tissue oximeter. *Physiol Meas* 22: 201-208
- 18. Ferrari M, Binzoni T, and Quaresima V (1997) Oxidative metabolism in muscle. *Phil Trans R Soc Lond* (B) 352: 677-683
- 19. Griffin MJ (1990) Handbook of Human Vibration. Academic Press Limited.
- 20. Guinard JC (1985) Vibration. Chapter 15. In: Patty's Industrial Hygiene and Toxicology. Vol 3b: Biological Responses. Wiley
- 21. Haldeman S (1999) Low back pain current physiologic concepts. Neurol Clinics North Am 17(1): 1-15
- 22. Hansson T, Magnusson M, and Broman H (1991) Back muscle fatigue and seated whole-body vibrations: an experimental study in man. *Clin Biomech* 6(3): 173-178
- 23. Hoffman U, Leyk D, Mester J, Mader A, Spitzenpfeil P, and Wackerhage H (1999) Effects of vibration on muscle energy turnover. Proceedings of the XVIIth Congress of the International Society of Biomechanics, Calgary, CANADA, pp 287
- 24. Hosea TM, Sheldon RS, Delatizky J, Wong MA, and Hsieh C (1986) Myoelectric analysis of the paraspinal musculature in relation to automobile driving. *Spine* 11(9): 928-936

- 25. Hulshof CTJ and van Zanten VOBA (1987) WBV and low-back pain A review of epidemiological studies. *Int Arch Occup Environ Health* 59: 205-220
- 26. International Organization for Standardization 2631 Part 1 (1997) Mechanical vibration and shock Evaluation of human exposure to whole-body vibration Part 1: General Requirements. ISO 2631/1-1997 (E)
- 27. Jensen BR, Jorgensen K, Hargens AR, Nielsen PK, and Nicolaisen T (1999) Physiological response to submaximal isometric contractions of the paravetebral muscles. Spine 24(22): 2332-2338
- 28. Jorgensen K (1997) Human trunk extensor muscle –physiology and ergonomics. Acta Physiol Scand 160 S637: 1-58
- 29. Junghanns H (1990) Clinical applications of normal biomechanical stresses on spinal function. English Language edition. Edited by Hans J Hager. Aspen Publishers, Inc., Maryland, USA.
- 30. Kahn JF, Favriou F, Jouanin JC, and Grucza R (2000) Effect of co-contractions on the cardiovascular response to submaximal static handgrip. *Eur J Appl Physiol* 83: 506-511
- 31. Kuno S, Miwa M, Suga Y, Hisaeda H, Chance B, Katsuta S, and Fukunga T (1997) Relationship between muscle oxygenation during exercise by time resolution spectroscopy and fiber type in human. *Med Sci Sports Exerc* 29(5) S170: Abstract 1016
- 32. McArdle WD, Katch I, and Katch VL (1996) Exercise Physiology, Williams & Wilkins, Baltimore, pp 221
- 33. McGill SM, Hughson RL, and Parks K (2000) Lumbar erector spinae oxygenation during prolonged contractions: implications for prolonged work. *Ergonomics* 43(4): 486-493
- 34. Magnusson M, Pope MH, Lindell V, Svensson M, and Andersson G (1992) Effect of backrest inclination on the transmission of vertical vibrations through the lumbar spine. *Clin Biomech* 8: 5-12
- 35. Maikala R, Farag M, and Bhambhani Y (2000) Low-back muscle oxygenation trends during simple postural variations – Intersession reliability using NIRS. Proceedings of International Ergonomics Conference, San Diego, USA: pp 5.9-5.12

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

- 36. Maikala R and Bhambhani Y (1999) Biceps brachii deoxygenation during cuff ischemia and recovery: Intra and inter session reliability. *Med Sci Sport Exer* S245 31:5 (Abstract)
- 37. Mancini DM, Bolinger L, Kendrick K, Chance B, and Wilson J (1994) Validation of NIRS in humans. J Appl Physiol 77: 2740-2747
- 38. Maronitis AB, Kovacs KM, Splittstoesser RE, and Marras (2000) The effectiveness of whole-body and localized measures of fatigue. Proceedings of the IEA 2000/HFES 2000 Congress, San Diego, USA, pp 5-91-5-94
- 39. Mester J, Spitzenfeil P, Schwarzer J, and Seifriz F (1999) Biological reaction to vibration implications for sport. *J Sci Med Sport* 2(3): 211-226
- 40. MicroRunMan User Manual (1998), NIM Inc., PA, USA.
- 41. Miyashita K, Morioka I, Tanabe T, Iwata H, and Takeda S (1992) Symptoms of construction workers exposed to whole-body vibration and local vibration. *Int Arch Occup Environ Health* 64: 347-351
- 42. Murthy G, Kahan NJ, Hargen AR, and Rempel DM (1997) Forarm muscle oxygenation decreases with low levels of voluntary contraction. *J Orthop Res* 15 : 507-511
- 43. Paddan GS and Griffin MJ (1988a) The transmission of translational seat vibration to the head I. Vertical seat vibration. *J Biomech* 21(3): 191-197
- 44. Paddan GS and Griffin MJ (1988b) The transmission of translational seat vibration to the head II. Horizontal seat vibration. *J Biomech* 21(3): 199-206
- 45. Palmer K, Griffin MJ, Bendall H, Pannett B, and Coggon D (2000) Prevalence and pattern of occupational exposure to whole body vibration in Great Britain: findings from a national survey. *J Occup Environ Med* 57: 229-236
- 46. Panjabi MM, Andersson GBJ, Jorneus L, Hult E and Mattsson L (1986) In vivo measurement of spinal column vibrations. *J Bone Joint Surg* 68A(5): 695-703
- 47. Pope MH (1989) Risk indicators in low-back pain. Ann Med 21(5): 387-392
- 48. Pope MH, Broman H, and Hansson T (1989) The dynamic response of a subject seated on various cushions. *Ergonomics* 32(10): 1155-1166
- 49. Pope MH, Broman H, and Hansson T (1990) Factors affecting the dynamic response of the seated subject. *J Spinal Disord* 3(2): 135-142
- 50. Pope MH and Hansson TH (1992) Vibration of the spine and low back pain. *Clinc* Orhop Rel Res 279: 49-59

- 51. Pope MH, Magnusson M, and Wilder D (1998) Low back pain and whole-body vibration. *Clin Orthop Research* 354: 241-248
- 52. Pope MH and DeVocht JW (1999) The clinical relevance of biomechanics. Neurological Clinics of North America 17(1): 17-41
- 53. Rundell KW, Nioka S and Chance B (1997) Hemoglobin/myoglobin desaturation during speed skating. *Med Sci Sports Exerc* 29(2): 248-258
- 54. Schmidt RT and Toews JV (1970) Grip strength as measured by the Jamar dynamometer. Arch Phys Med Rehab 51: 321-327
- 55. Seidel H and Heide R (1986) Long-term effects of whole-body vibration. A critical survey of the literature. *Int Arch Occup Environ Health* 58: 1-29
- 56. Seroussi RE, Wilder DG, and Pope MH (1989) Trunk muscle electromyography and whole-body vibration. *J Biomech* 22: 219-229
- 57. Simon-Arndt, Yuan, H, and Hourani LL (1997) Aircraft type and diagnosed back disorders in US Navy pilots and aircrew. Aviat Space Environ Med 68(11): 1012-1018
- 58. van Beekvelt MCP, Borghuis MS, van Engelen BGM, Wevers RA, and Colier WNJM (2001) Adipose tissue thickenss affects in vivo quantitative near-IR spectroscopy in human skeletal muscle. *Clin Sci* 101: 21-28
- 59. Wilder D, Woodworth BB, Frymoyer JW, and Pope MH (1982) Vibration and the human spine. *Spine* 7(3): 243-254
- 60. Wilder DG, Pope MH, and Frymoyer JW (1993) The biomechanics of lumbar disc herniation and the effect of overload and instability. *J Spinal Disord* 1(1): 16-32
- 61. Wilder D, Magnusson M, Fenwick J, and Pope M (1994) The effect of posture and seat suspension design on discomfort and back muscle fatigue during simulated truck driving. *Appl Erg*o 25(2): 66-76
- 62. Yoshitake Y, Ue H, Miyazaki M, and Moritani T (2001) Assessment of lower-back muscle fatigue using electromyography, mechanomyography, and near-infrared spectroscopy. *Eur J Appl Physiol* 84: 174-179
- 63. Zimmerman CL, Cook TM, and Goel VK (1993) Effects of seated posture on erector spinae EMG activity during whole body vibration. *Ergonomics* 36(6): 667-675

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Table 3.1: Demographics of subjects

Gender	Age	Body Mass	Sitting Height	Standing Height	Body Mass Index Standing	Maximal Right Hand Grip Strength	Skinfold at Third Lumbar region	Peak Oxygen Uptake during Arm Cranking
	Yrs	Kgs		m	Kg.m ⁻²	Kgs	mm	mL.Kg ⁻¹ .min ⁻¹
Men N=13	24.7 ± 3.9	70.8 ± 12.1 *	0.86 ± 0.05	1.71 ± 0.07*	24.07 ± 3.2	34.22 ± 9.1*	11.31 ± 5.09	24.5 ± 6.6
Women N=14	23.9 ± 3.5	60.26 ± 9.0	0.84 ± 0.08	1.63 ± 0.10	23.0 ± 3.7	22.49 ± 3.6	11.59 ± 4.46	21.4 ± 4.6

*- significant difference between men and women, p<0.05
Table 3.2: Erector Spinae	Oxygenation a	Ind Blood Volume	differences in	Optical Density un	its (Mean ± SD)
in healthy men and wome	n during WBV a	at three different d	oses	•	

GENDER	BACKREST	DOSE		nation *	∆ Blood	volume
		Frequency	WBV	WBV	WBV	WBV
		in Hz	Without Grip	With Grip	Without Grip	With Grip
Men	ON	3	0.020 ± 0.05^{a}	0.036 ± 0.08	-0.063± 0.10 ^b	-0.05 ± 0.10 ^b
	OFF	3	0.031 ± 0.10*	0.046 ± 0.09	0.058 ± 0.22	0.058 ± 0.19
	ON	4.5	0.027 ± 0.09 •	0.060 ± 0.13	-0.049 ± 0.10 ^b	-0.029 ± 0.12 b
	OFF	4.5	0.057 ± 0.23*	0.065 ± 0.24	0.14 ± 0.20	0.14 ± 0.20
	ON	6	-0.002 ± 0.05 *	0.032 ± 0.07	-0.066 ± 0.11 ^b	-0.028 ± 0.12 ^b
	OFF	6	0.014 ± 0.10*	0.046 ± 0.11	0.018 ± 0.18	0.045 ± 0.17
Women	ON	3	-0.004 ± 0.05	0.001 ± 0.05	-0.025 ± 0.08	-0.010 ± 0.08 b
	OFF	3	0.007 ± 0.05	0.012 ± 0.05	0.029 ± 0.07	0.038 ± 0.07
	ON	4.5	0.022 ± 0.09	0.031 ± 0.09	-0.011 ± 0.08	-0.005 ± 0.08 ^b
	OFF	4.5	0.010 ± 0.02	0.025 ± 0.04	0.034 ± 0.06	0.043 ± 0.07
	ON	6	0.028 ± 0.07	0.032 ± 0.09	-0.004 ± 0.05	-0.005 ± 0.04 ^b
	OFF	6	0.025 ± 0.06	0.025 ± 0.07	0.038 ± 0.10 *	0.043 ± 0.12

 * Δ is defined as the difference between 'maximum optical density' [during initial rest without WBV or after recovery from WBV session with a backrest condition (recovery 1)] and 'minimum optical density' during seated vibration with or without rhythmic handgrip contraction. NOTE: All subjects were exposed to 3, 4.5 and 6 Hz doses.

a – significant interaction between work and gender (p < 0.05) b – significant interaction between backrest and gender (p < 0.05)

Table 3.3: Pearson Correlations between Maximal Rhythmic Handgrip Contractions and Erector Spinae [Oxygenation and Blood Volume] responses

DOSE	BACKREST	N	IEN	WOMEN		
Frequency in Hz		Oxygenation	Blood Volume	Oxygenation	Blood Volume	
3	WITH	0.24	-0.38	-0.16	-0.07	
	WITHOUT	0.08	-0.02	0.35	-0.05	
4.5	WITH	0.39	-0.43	0.37	-0.18	
	WITHOUT	-0.05	0.11	0.22	0.23	
6	WITH	0.39	0.27	0.13	-0.28	
	WITHOUT	0.02	-0.02	-0.23	0.44	

Table 3.4: Spearman Correlations between Erector Spinae [Oxygenation and Blood Volume Responses] and Adipose Tissue Thickness

GENDER	BACKREST	DOSE	Oxyge	nation	Blood Volume		
		Frequency in Hz	WBV Without Grip	WBV With Grip	WBV Without Grip	WBV With Grip	
Men	WITH	3	-0.42	-0.38	0.38	0.48	
	WITHOUT	3	-0.04	-0.10	0.03	0.12	
	WITH	4.5	0.00	-0.39	0.15	0.17	
	WITHOUT	4.5	-0.28	-0.21	-0.24	-0.16	
	WITH	6	-0.004	-0.28	0.10	0.06	
	WITHOUT	6	-0.30	-0.41	-0.24	-0.26	
Women	WITH	3	-0.09	-0.15	0.14	-0.05	
	WITHOUT	3	-0.18	-0.29	0.06	-0.04	
	WITH	4.5	-0.19	-0.22	0.09	0.12	
	WITHOUT	WITHOUT 4.5		-0.10	-0.01	-0.05	
	WITH	6	-0.04	-0.05	0.21	0.06	
	WITHOUT	6	0.18	0.16	-0.18	-0.16	

Table 3.5: Pearson Correlations between Erector Spinae [Oxygenation and Blood Volume Responses] and Relative Peak Aerobic Power

GENDER	BACKREST	DOSE	Oxyge	nation	Blood Volume		
		Frequency	WBV	WBV	WBV	WBV	
		in Hz	Without Grip	With Grip	Without Grip	With Grip	
Men	WITH	3	0.51	0.53	-0.46	-0.49	
	WITHOUT	3	-0.01	0.19	-0.02	-0.10	
	WITH	4.5	0.06	0.36	-0.18	-0.15	
	WITHOUT	4.5	0.27	0.22	0.19	0.15	
	WITH	6	0.10	0.35	-0.20	-0.15	
	WITHOUT	6	0.12	0.17	0.10	0.14	
Women	WITH	3	-0.08	-0.03	-0.52	-0.39	
	WITHOUT	3	-0.08	-0.23	-0.39	-0.33	
	WITH	4.5	0.08	0.15	-0.14	-0.22	
	WITHOUT	4.5	0.00	0.03	-0.30	-0.16	
	WITH	6	0.41	0.41	-0.36	-0.42	
	WITHOUT	6	-0.37	-0.47	-0.21	-0.18	



Figure 3.1A NIRS Sensor Placement on the Low-back (The sensor was positioned right side at the level of third vertebra, approximately 3 cm from the spinal column)



Figure 3.1B Tensor Bandage around the Low-back

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 3.2A. Erector Spinae Oxygenation Trends



Erector Spinae Oxygenation (top panel) and blood volume (bottom panel) trends in a typical male [solid line] and female [dotted line] subject during exposure to seated whole-body vibration with and without handgrip exercise

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 3.3A. Interaction between Gender and Work Mean Oxygenation Changes (P<0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest).

3. Graph suggests that during both conditions of WBV Alone and WBV with Rhythmic Handgrip contractions, men demonstrated greater degree of deoxygenation compared to women.



NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest).

3. Graph suggests that during both conditions of WBV Alone and WBV with Rhythmic Handgrip contractions, gender did not influence blood volume changes.



Figure 3.4A. Interaction between Gender and Dose Mean Oxygenation Changes (P>0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Work and Backrest).

3. It appears from the above graph that men and women demonstrate distinct patterns of oxygenation changes. However, because of greater variance obtained, Dose did not inlfuence the results.



NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are

not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Work and Backrest).

3. It appears from the above graph that men and women demonstrate distinct patterns of blood volume changes. However, because of greater variance obtained, Dose did not influence the results.



Figure 3.5A. Interaction between Gender and Backrest Mean Oxygenation Changes (P>0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Work).

3. Graph suggests that presence of Backrest did not influence oxygenation responses in the low-back.



Mean Blood Volume Changes (P<0.01)

NOTE: 1. Due to larger coefficient of variations obtained, standard deviations are not included in the graphs shown above.

2. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Work).

3. Graph suggests that presence of Backrest demonstrated lower blood volume changes in men than women. However, men 'Without' Backrest showed greater blood volume changes than women.

CHAPTER 4

Acute Physiological Responses during Exposure to Seated Whole-body Vibration

Whole-body vibration (WBV), defined as an oscillating motion transmitting to the whole-body, depends primarily on its intensity (frequency, acceleration, amplitude, duration), type of vibration source, and direction of its input to the human body. Human factors such as demographics (height, weight, age, gender), posture adopted, fitness, personality, etc. also influence various psychophysiological, biomechanical and biodynamic responses resulting from the exposure to WBV (Griffin 1990). Several authors have demonstrated variable cardiorespiratory responses during exposure to different vibration doses (from less than 1g to greater than 1g) with a variety of WBV simulators (Bennett et al. 1977, Dupuis and Zerlett 1986, Gaeuaman et al. 1962, Guinard 1985, Hood et al. 1966, Manninen 1984, Pope et al. 1990, Thornton et al. 1984).

Low-frequency WBV exposure in the region of 2 to 20 Hz between the intensities of 0.1 to $0.5q_{r.m.s}$ elicits cardiorespiratory responses similar to moderate exercise, elevating variables such as oxygen uptake, ventilation rate. respiratory frequency, heart rate, and cardiac output (Guinard 1985). Pope et al. (1990) suggested that monitoring physiological variables such as absolute oxygen uptake during WBV is useful, and relevant to the assessment of the energy demands of populations exposed in the industries. Most of the above studies showed that monitoring the cardiorespiratory system provide useful information on the acute effects of WBV exposure. In a comprehensive epidemiological review pertaining to this topic. Seidel and Heide (1986) emphasized that WBV might pose an additional risk factor for ischemic heart disease. However, the review of Wikstrom et al. (1994) suggested that exposure to vibration doesn't significantly increase the risk of diseases in the cardiovascular system. Electromyographic activity of the paravertebral muscles demonstrated low-back muscle fatigue during exposure to WBV (Hansson et al. 1991, Pope et al. 1998, Wilder et al. 1982), subsequently leading to pain and injury of the lower back. With such mixed opinions of various authors, it is

essential to evaluate both central physiological responses and peripheral limitations of the human body during exposure to WBV.

Occupational exposure to WBV usually involves concomitant physical and physiological stresses that may hinder successful human performance. For example, during hovering by helicopter pilots, Thornton et al. (1984) showed higher energy expenditure requirements. Manninen (1984) demonstrated higher heart rate during WBV combined with dynamic work. Also, Bennett et al (1977) reported increase in absolute oxygen consumption and heart rate responses during WBV combined with tracking tasks. However, research pertaining to physiological responses during WBV combined with 'simulated' manual work is still limited. Occupants' posture during prolonged sitting with and without WBV may also influence their physiological responses (Pope et al. 1990, Magnusson et al. 1987). Sitting 'With' and 'Without' backrest support might not only affect human comfort but also load the postural muscles, thus eliciting variable cardiorespiratory and muscular responses. While the number of women employed in occupations involving exposure to WBV has grown considerably in the last century, research pertaining to their physiological responses and comparison with their male counterparts is still lacking (Bovenzi 1996). It has also been suggested that fitness of the subjects plays an important role in determining the effects of WBV exposure on humans (Boff and Lincoln 1988, Griffin 1990). However, there are no studies that have objectively examined its influence on the physiological responses elicited during exposure to WBV.

The purposes of this study were to examine the effects of discrete 3 Hz, 4.5 Hz, and 6 Hz vibration doses on: (1) the acute physiological responses in young healthy men and women; (2) the changes in these physiological responses during exposure to WBV 'With' and 'Without' backrest; (3) the addition of maximal rhythmic handgrip contractions during WBV, and (4) differences in these responses between genders. It was hypothesized that the acute physiological responses will be similar in both men and women 'With' and 'Without' backrest, and will not be influenced by the exposure to vibration dose within the range of 3 to 6 Hz while performing rhythmic handgrip contractions.

METHODS

Subjects

Written informed consent was obtained from 13 healthy men (age 24.7 \pm 3.9 yrs, mass 70.8 \pm 12.1 kg, height 171 \pm 7 cm) and 14 healthy women (age 23.9 \pm 3.51 yrs, mass 60.3 \pm 9.0 kg, height 164 \pm 10 cm). Subjects were undergraduate and graduate students recruited from the university population (Table 4.1). All subjects were right hand dominant. They all completed the Revised Physical Activity Readiness Questionnaire (Canadian Society for Exercise Physiology, 1994) to identify contraindications for exercise. The test protocols utilized were approved by the Human Research Ethics Board at this institution (APPENDIX A.1).

Vibrating Base

A steel vibrating base that housed an electro-mechanical motor connected to a cam mechanism (Advanced Therapy Products, Inc., VA, USA) was modified with an external DC speed control device (Figures 2.1A and 2.1B). Varying the speed of this motor from 0 to 100% corresponded to vibrating frequencies of 0 Hz to 6.2 Hz. Detailed description of vibrating base was provided in the Chapter 2. Average vertical accelerations (in root mean square values) generated at different doses are given in the Chapter 2 (Table 2.2) as well.

Test Protocols

Each subject completed an arm cranking exercise test to determine his/her aerobic fitness in the first session, followed by three WBV tests on separate days (3 Hz, 4.5 Hz, and 6 Hz) in a randomized order.

(1) Assessment of the Peak Oxygen Uptake

In the first testing session, the subject completed an aerobic fitness test on an arm cranking ergometer (Cybex, MET 300, USA). After preparing each subject with the mouthpiece and headgear (Figure 4.1A), the test was initiated

167

with a two-minute rest period. It was followed by two minutes of arm cranking at no load and subsequent increments of 25 Watts every 2 minutes until voluntary exhaustion, or attainment of two or more of the following end points suggested by the American College of Sports Medicine (1993): (1) leveling off in the oxygen consumption (increase of $\leq 100 \text{ ml.min}^{-1}$) with increasing load, (2) age predicted maximal heart rate, equated to 220-age (in years), (3) respiratory exchange ratio of ≥ 1.10 , and (4) rate of perceived exertion ≥ 18 (Borg, 1982).

(2) Vibration Tests

The test protocol on each day lasted 30 minutes, and included initial sixminute rest without vibration, eight-minute vibration 'With' (or 'Without') a backrest, four-minute recovery, eight-minute vibration 'Without' (or 'With') a backrest, and four-minute recovery. Subjects were randomly assigned to a posture 'With' or 'Without' a backrest (Figures 2.4A and 2.5), and were instructed to adopt a comfortable posture when tested 'Without' a backrest. Throughout the test, subjects sat in the vibrating chair with the eyes open. A footrest was provided for subjects to place their feet at a comfortable level. During sitting 'With' and 'Without' a backrest, angles at the knee joint and low-back were measured by a goniometer (JAMAR, Clifton, NJ, USA).

(3) Rhythmic Handgrip Measurements

Before the vibration session, right hand grip size was adjusted on the dynamometer (JAMAR, CA, USA) to a position that was comfortable for each subject. During the fifth minute of each vibration condition, the subject performed rhythmic maximal contractions for one minute at a contraction rate of one every five seconds (Figures 2.4B). Contractions were performed with the right hand with the forearm in a horizontal position and the elbow at an angle of 180°. Similar to a method suggested by Schmidt and Toews (1970), throughout the study the dynamometer was calibrated with known weights to check its accuracy identified through the dial reading on the dynamometer.

(4) Whole-body Physiological Measurements

For all the tests, gas exchange measurements were continuously monitored with an automated metabolic measurement cart (MMC 2900, Sensormedics Corporation, Yorba Linda, CA). The oxygen (O₂) and carbon dioxide (CO₂) analyzers of the cart were calibrated before each test with commercially available precision gases (16% O₂, 4% CO₂, balance Nitrogen; 26% O₂, balance Nitrogen). The pneumotach was calibrated for volume using a 3 liter syringe, as recommended by the manufacturer. Heart rate was recorded using a polar heart rate monitor (Polar Accurex Plus, Finland). Metabolic data were recorded in the mixing chamber mode and averaged over 20 second intervals for further analysis. Oxygen pulse was calculated as the ratio of absolute oxygen uptake and heart rate. Tidal volume was calculated as the ratio of ventilation volume and respiratory frequency.

(5) Measurement of Cardiac Output by CO₂ Rebreathing

According to the indirect Fick equation, cardiac output is determined by the ratio of carbon dioxide produced (VCO₂) to the difference in CO₂ concentration between the mixed venous and arterial blood ($C_{v-a}CO_2$). In a validation study for Q, Marks et al. (1985) showed a significant correlation of 0.82 between the direct Fick method and CO₂ rebreathing technique. Further, Bhambhani et al. (1994) showed a test-retest reliability of 0.89 for the rebreathing technique in males exercising at 60% of their maximal aerobic capacity. The rebreathing maneuver and the calculations to obtain VCO₂/C_{v-a}CO₂ in the above equation are explained below.

During the fourth minute of initial rest without WBV, and during the WBV sessions 'With' and 'Without' backrest, each subject inhaled a mixture of CO_2 and O_2 from a 5 liter anesthesia bag under steady state conditions. During the rest period, the rebreathing mixture consisted of 8% CO_2 and balance O_2 for all the subjects. During exposure to WBV, the rebreathing mixture consisted of 11%-12% CO_2 and balance O_2 . The concentration of CO_2 utilized for rebreathing was determined by the VO₂ during vibration (Jones 1986). Steady state conditions

were ascertained by monitoring the values of absolute oxygen uptake, ventilation rate, heart rate and end tidal CO_2 ($P_{ET}CO_2$) by switching the metabolic cart from the mixing chamber to the breath-by-breath mode prior to the initiation of the rebreathing maneuver.

The subject hyperventilated for approximately 15 sec so as to establish an equilibrium between the CO_2 in the bag mixture and in the lungs (Figure 4.1B). A difference of less than 1 Torr pressure during the first 6-8 sec of the rebreathing maneuver was considered to be indicative of an equilibrium in the CO₂ concentration between the mixed venous blood and alveolar gas. A downstream correction factor (Jones 1986) was applied to this measurement to obtain a more accurate estimate of venous CO₂. The steady state P_{ET}CO₂ obtained prior to the rebreathing maneuver was considered to be indicative of the arterial CO2 level. A hemoglobin concentration of 15.8gm.100ml⁻¹ of blood for men and 13.9gm.100ml⁻¹ of blood for women was assumed for the conversion of CO₂ tension into content (Jones 1986). All of these measurements and calculations (including mixed arterio-venous oxygen difference) were performed using the software available with the metabolic measurement cart (MMC 2900, Sensormedics Corporation, Yorba Linda, CA). Stroke volume was calculated as the ratio of cardiac output and heart rate for each condition (initial rest without WBV, 'With' backrest during WBV, and 'Without' backrest during WBV).

Statistical Analysis

Independent 't' tests were used to compare the physical characteristics of the men and women (Table 4.1). A four-way analysis of covariance (ANCOVA), using the peak oxygen uptake as the covariate, gender as a between-subjects factor, and three repeated measure factors (dose [3 Hz, 4.5 Hz, and 6 Hz], backrest ['With' and 'Without' backrest], and workload [WBV alone and WBV combined with maximal rhythmic handgrip contractions]) with a fully crossed design was used to evaluate the differences in the metabolic responses. A threeway ANCOVA, using the peak oxygen uptake as the covariate, gender as a between-subjects factor, and two repeated measures factors (dose and condition

170

[initial rest without WBV, 'With' backrest during WBV, and 'Without' backrest during WBV]) with a fully crossed design was used to evaluate the differences in the cardiovascular responses.

To minimize the violation of the assumption of homogeneity of variance, the 'Greenhouse-Geiser' adjustment was used. Since multiple comparisons were made for each dependent variable in the present study, the Bonferroni adjustment for a P value of 0.05 was applied to minimize Type 1 error. The Bonferroni adjustment was calculated as the ratio of 0.05 and number of statistical comparisons (APPENDIX B.3). Thus, statistical significance was considered at P <0.01. Significant F ratios were further analyzed with the Scheffe post hoc multiple comparison test. The relationship between maximal rhythmic handgrip contractions and changes in selected physiological responses during WBV were evaluated using the Pearson product moment correlations. Also, relationship between the changes in selected physiological responses during WBV and peak oxygen uptake measured during arm cranking was evaluated using the Pearson correlations. The Statistical Package for Social Sciences SPSS (version 10) was used for all statistical analyses (SPSS Inc., Chicago, USA).

RESULTS

Metabolic and Respiratory Responses

The metabolic and cardiorespiratory responses (mean ± SD) in men and women, 'With' and 'Without' backrest, during the different doses of WBV are summarized in Table 4.2. No significant four-way and three-way interactions were obtained in these responses. This implies that in both genders, the changes observed in these responses at different WBV doses were similar 'With' and 'Without' backrest support. No significant two-way interactions between gender and dose, and dose and workload were obtained for any of these responses. However, the interaction between workload and gender showed significance only for the absolute and relative oxygen uptake, and oxygen pulse. No significant interactions between workload and gender were obtained for the heart rate,

ventilation rate, respiratory frequency and tidal volume. It should be noted that comparisons of these significant interactions between these independent variables were based on the pooled mean values of other remaining factors (i.e., WBV dose and Backrest conditions).

Further, when these physiological responses were controlled for the level of aerobic fitness of the subjects (as a covariate), the effect of aerobic fitness was evident in absolute oxygen uptake (P=0.005), relative oxygen uptake responses (P=0.008), and oxygen pulse (P=0.010) during workload. Pearson correlations between metabolic responses and the peak aerobic capacity during WBV 'With' and 'Without' a backrest support are shown in the Tables 4.5a, 4.5 b and 4.5c.

Comparison between Doses

All subjects showed similar absolute oxygen uptake (L.min⁻¹) responses during the three doses of WBV (3 Hz: 0.289 ± 0.012 , 4.5 Hz: 0.303 ± 0.012 , and 6 Hz: 0.306 ± 0.012 , P=0.331). Also, when oxygen uptake was expressed relative to their body mass (mL.kg.⁻¹.min⁻¹), no significant difference was observed among the doses (3 Hz: 4.48 ± 1.75 , 4.5 Hz: 4.65 ± 1.66 , and 6 Hz: 4.77 ± 1.84 , P=0.285). The heart rate (beats.min⁻¹) was also unaffected by the dose (3 Hz: 85 ± 13 , 4.5 Hz: 88 ± 12 , and 6 Hz: 88 ± 11 , P=0.269). Oxygen pulse (mL.beat⁻¹), defined as the ratio between the absolute oxygen uptake and the heart rate, was similar among all the doses (3 Hz: 3.4 ± 1.3 , 4.5 Hz: $3.4 \pm$ 1.1, and 6 Hz: 3.5 ± 1.2 , P=0.884).

Ventilation rate (L.min⁻¹) showed significant differences only between 3 Hz and 6 Hz (11.3 \pm 5.5 Vs 12.8 \pm 6.2, P=0.005). However, no significant differences were observed in the ventilation rate between 3 Hz and 4.5 Hz (12.5 \pm 9.6, P=0.260), and 4.5 Hz and 6 Hz (P=0.692). Respiratory frequency (breaths.min⁻¹) showed no significant differences among the doses (3 Hz: 23 \pm 7, 4.5 Hz: 22 \pm 7, and 6 Hz: 23 \pm 6, P=0.827). Similar to ventilation rate, a significant difference in tidal volume (L.breath⁻¹) was observed only between 3 Hz and 6 Hz (0.53 \pm

0.26 Vs 0.59 \pm 0.29, P=0.010). However, no significant differences in tidal volume were observed between 3 Hz and 4.5 Hz (0.62 \pm 0.64, P=0.140), and 4.5 Hz and 6 Hz (P=0.692).

Comparison between Sitting 'With' and 'Without' Backrest during Whole-Body Vibration

The following physiological responses were significantly higher during exposure to WBV 'Without' backest: (1) absolute oxygen uptake (0.283 ± 0.018 L.min⁻¹ Vs 0.317 ± 0.012 L.min⁻¹, P=0.000); relative oxygen uptake (4.88 ± 1.77 mL.kg.⁻¹.min⁻¹ Vs 4.37 ± 1.71 mL.kg.⁻¹.min⁻¹, P=0.000); (3) heart rate (89 ± 11 beats.min⁻¹ Vs 85 ± 11 beats.min⁻¹, P=0.000); (4) oxygen pulse (3.53 ± 1.23 mL.beat⁻¹ Vs 3.30 ± 1.16 mL.beat⁻¹, P=0.000); and (5) ventilation rate (12.9 ± 8.5 L.min⁻¹ Vs 11.5 ± 5.8 L.min⁻¹, P=0.010). However, respiratory frequency (22 ± 7 breaths.min⁻¹ Vs 0.61 ± 0.5 L.breath⁻¹, P=0.086) were similar between these conditions.

Comparison between WBV Alone and WBV Combined with Maximal Rhythmic Handgrip Contractions

When subjects performed maximal rhythmic handgrip contractions during WBV, absolute oxygen uptake increased significantly from the WBV alone conditions ($0.365 \pm 0.013 \text{ L.min}^{-1} \text{ Vs} 0.234 \pm 0.065 \text{ L.min}^{-1}$, P=0.000). Similarly, relative oxygen uptake also increased during WBV combined with isometric work ($5.64 \pm 1.8 \text{ mL.kg}^{-1}$.min⁻¹ Vs $3.62 \pm 0.9 \text{ mL.kg}^{-1}$.min⁻¹, P=0.000). During WBV combined with maximal rhythmic handgrip contractions, heart rate values were significantly higher than WBV alone values ($95 \pm 10 \text{ beats.min}^{-1} \text{ Vs} 80 \pm 8 \text{ beats.min}^{-1}$, P=0.000). Oxygen pulse values were significantly higher during maximal rhythmic handgrip contractions as well ($3.88 \pm 1.3 \text{ mL.beat}^{-1} \text{ Vs} 2.96 \pm 0.8 \text{ mL.beat}^{-1}$, P=0.000). Isometric work performed during WBV also resulted in a significant increase in ventilation rate ($14.95 \pm 9.3 \text{ L.min}^{-1} \text{ Vs} 9.41 \pm 2.6 \text{ L.min}^{-1}$,

P=0.000). Respiratory frequencies were not significantly different between the two conditions (23 \pm 6 breaths.min⁻¹ Vs 22 \pm 6 breaths.min⁻¹, P=0.488). However, similar to ventilation rate, tidal volume responses showed significantly higher values during maximal rhythmic handgrip contractions (0.72 \pm 0.56 L.breath⁻¹ Vs 0.44 \pm 0.16 L.breath⁻¹, P=0.000).

Comparison between Genders

Significant interaction between workload and gender was observed for the: absolute and relative oxygen uptake, and oxygen pulse. During WBV alone, men consumed more oxygen than women ($0.256 \pm 0.074 \text{ L.min}^{-1} \text{ Vs } 0.216 \pm 0.051 \text{ L.min}^{-1}$, P=0.008). This gender difference was negated when oxygen uptake was expressed relative to body mass ($3.60 \pm 0.86 \text{ mL.kg}^{-1}$.min⁻¹ Vs $3.63 \pm 0.96 \text{ mL.kg}^{-1}$.min⁻¹, P=0.047). During WBV combined with maximal rhythmic handgrip contractions, men showed significantly higher values for both absolute oxygen uptake ($0.426 \pm 0.015 \text{ L.min}^{-1} \text{ Vs } 0.309 \pm 0.078 \text{ L.min}^{-1}$, P=0.008) and relative oxygen uptake values ($6.10 \pm 1.95 \text{ mL.kg}^{-1}$.min⁻¹ Vs $5.24 \pm 1.60 \text{ mL.kg}^{-1}$.min⁻¹, P=0.001), respectively.

Heart rate responses were similar in both men and women (88 \pm 13 beats.min⁻¹ Vs 86 \pm 10 beats.min⁻¹, P=0.495), with no significant interaction being observed in heart rate between gender and workload (P=0.666). However, gender showed a significant interaction with workload for the oxygen pulse responses. Similar to absolute oxygen uptake, men utilized greater amount of oxygen per heartbeat during WBV alone compared to women (3.2 \pm 1.0 mL.beat⁻¹ Vs 2.7 \pm 0.7 mL.beat⁻¹, P=0.005). A similar trend was observed in both genders during WBV combined with maximal rhythmic handgrip contractions (4.5 \pm 1.5 Vs 3.3 \pm 0.8, P=0.005).

Ventilation rate was not significantly different between men and women $(13.20 \pm 6.8 \text{ L.min}^{-1} \text{ Vs} 11.23 \pm 7.7 \text{ L.min}^{-1}, P=0.176)$, and no significant interaction was observed between gender and workload (P=0.292). Respiratory frequency was similar in both genders (21 ± 6 breaths.min^{-1} Vs 24 ± 7)

174

breaths.min⁻¹, P=0.103) under both conditions of work (P=0.762). No significant differences in tidal volume were observed between men and women (0.66 \pm 0.34 L.breath⁻¹ Vs 0.50 \pm 0.5 L.breath⁻¹, P=0.047), and no significant interaction was observed between gender and workload as well (P=0.400).

Cardiovascular Responses at Three Conditions

Effect of WBV on the cardiovascular responses in men and women at the three doses and during the three conditions (initial rest with no WBV, 'With' backrest during WBV, and 'Without' backrest during WBV) are summarized in Table 3. No significant three-way or two-way interactions were obtained for the cardiac output (L.min⁻¹), heart rate (beats.min⁻¹), stroke volume (mL.beat⁻¹), and mixed arterio-venous oxygen difference (mL%) responses. This implies that in both genders, the changes observed in these responses at different WBV doses were similar during rest, 'With' and 'Without' backrest. The results of the significant main effects are explained below. It should be noted that comparisons of these significant effects were based on the pooled mean values of the remaining factors (gender and backrest conditions).

Comparison between Doses

All the cardiovascular responses were similar during exposure to the three doses of WBV: cardiac output (3 Hz: 6.0 ± 1.1 L.min⁻¹, 4.5 Hz: 5.9 ± 1.3 L.min⁻¹, and 6 Hz: 6.2 ± 1.5 L.min⁻¹, P=0.393); heart rate (3 Hz: 78 ± 1 beats.min⁻¹, 4.5 Hz: 77 ± 2 beats.min⁻¹, and 6 Hz: 78 ± 1 beats.min⁻¹, P=0.234); stroke volume (3 Hz: 79 ± 19 mL.beat⁻¹, 4.5 Hz: 77 ± 17 mL.beat⁻¹ and 6 Hz: 81 ± 20 mL.beat⁻¹, P=0.462); and mixed arterio-venous oxygen difference (3 Hz: 6.3 ± 1.2 mL%, 4.5 Hz: 6.4 ± 1.1 mL% and 6 Hz: 6.4 ± 1.0 mL%, P=0.842).

Comparison among Three Conditions

Significant difference in cardiac output was obtained between sitting 'With' and 'Without' backrest during WBV ($5.8 \pm 1.4 \text{ L.min}^{-1} \text{ Vs } 6.3 \pm 1.3 \text{ L.min}^{-1}$, P=0.000). However, no significant difference was observed between: either rest ($6.0 \pm 1.2 \text{ L.min}^{-1}$, P=0.312) and 'With' backrest conditions; or rest and 'Without' 175

backrest conditions (P=0.104). Heart rate responses showed significant differences among all conditions (P=0.000). During sitting 'Without' backrest, heart rate values were higher (84 ± 1 beats.min⁻¹) than both sitting 'With' backrest (78 ± 1 beats.min⁻¹) and rest 'Without' WBV conditions (71 ± 1 beats.min⁻¹). Stroke volume responses showed significant differences between sitting 'With' and 'Without' backrest conditions (73 ± 18 mL.beat⁻¹ Vs 78 ± 16 mL.beat⁻¹, P=0.006). These values were also significantly higher during rest (85 ± 19 mL.beat⁻¹) than both 'With' backrest (P=0.000), and 'Without' backrest conditions (P=0.012). However, mixed arterio-venous oxygen difference showed no significant difference among these three conditions (rest: 6.36 ± 0.16 mL%, 'With' backrest: 6.37 ± 0.15 mL%, and 'Without' backrest: 6.41 ± 0.15 mL%, P=0.633).

Comparison between Genders

Cardiac output was significantly higher in men than women (6.7 \pm 1.4 L.min⁻¹ Vs 5.4 \pm 0.9 L.min⁻¹, P=0.000). However, gender did not influence heart rate responses (men: 78 \pm 2 beats.min⁻¹ Vs women: 77 \pm 2 beats.min⁻¹, P=0.956). Stroke volume responses in men were significantly higher than women (82 \pm 19 mL.beat⁻¹ Vs 71 \pm 13 mL.beat⁻¹, P=0.000). Mixed arterio-venous oxygen difference also showed significant difference between men and women (men: 6.8 \pm 1.1 mL% Vs women: 6.0 \pm 1.0 mL%, P=0.010).

Correlations between Selected Variables and Physiological Responses

Correlation coefficients between maximal handgrip contractions and metabolic responses are summarized in Table 4.4. Majority of the responses correlated with handgrip contractions were absolute and relative oxygen uptake, oxygen pulse, ventilation rate, and tidal volume (P<0.05). However, these correlations were not consistent between genders among three vibration tests.

DISCUSSION

Whole-body Physiological responses

In general, the metabolic and respiratory responses observed during WBV alone were greater than those observed during initial rest before WBV or recovery after each WBV exposure. During WBV combined with maximal rhythmic handgrip contractions, these responses were greater than that observed during WBV alone (Table 4.2). However, as soon as the WBV stopped, all the metabolic and respiratory responses reached baseline in less than a minute. In a majority of the subjects, cardiac output and stroke volume responses were lowest during sitting 'With' backrest. Mixed arterio-venous oxygen differences were not consistent during a variety of WBV conditions. However, heart rate responses during sitting 'Without' backrest were greater than both 'With' backrest during WBV and initial rest without WBV (Table 4.3).

Role of WBV Dose

In the present study, the effects of dose on physiological responses obtained (Table 4.2 and Table 4.3) were in accordance with those reported in the literature (Gaeuman et al. 1962, Hood et al. 1966, Hoover and Ashe 1962, Pope et al. 1990). However, only two physiological responses (ventilation rate and tidal volume) were significantly affected by the dose in the present study. Ventilation rate was 12% greater during 6 Hz than that obtained at 3 Hz. Since respiratory frequency was similar for all the doses, and tidal volume was significantly high during 6 Hz than at 3 Hz, increases in ventilation rate were attributed to changes in tidal volume only (Hoover and Ashe 1962, Gaeuman et al. 1962).

During exposure to doses of 2, 4, 6, 8, and 11 Hz (at accelerations of 0.03 and 2.88g, and amplitudes of 1.6 mm and 3.2 mm), Hoover and Ashe (1962) reported respiratory responses similar to the present study (Table 4.2). In their subjects, ventilation rate ranged from 7 to 13 L.min⁻¹, tidal volume from 0.44 to 0.84 liters.breath⁻¹, and respiratory frequency from 16 to 20 breaths.min⁻¹. Furthermore, they reported significant differences in respiratory responses at only 6 Hz (at 3.2

mm vibration amplitude). They concluded that changes in tidal volume resulted in an increase in ventilation rate and the changes observed are similar to those seen in hyperventilation. In another study, Gaeuman et al. (1962) exposed their subjects to a variety of doses and accelerations (at a fixed amplitude of 6.4 mm) and also concluded that exposure to WBV induces hyperventilation. They suggested that if subjects were unrestrained in their sitting posture, they were able to adjust or position their body with respect to different WBV conditions through voluntary and involuntary muscle guarding, thus damping the WBV, and subsequently reducing its transmission to human body parts. These authors hypothesized that such postural adjustments during WBV may eventually result in an increase in metabolic activity, which if continued for extended periods at higher doses (6-15 Hz), could lead to fatigue. However, in the present study metabolic responses during exposure to the three doses were identical (Table 4.2). The subjects were unrestrained during the WBV exposures as well. Also, vibration amplitude was almost double (6.1mm - 6.6 mm) to that of cited in Hoover and Ashe (1962), and were similar to that of Gaeuman et al. (1962). Furthermore, accelerations obtained (Chapter 2, Table 2.2) were approximately less than half of the values reported by Hoover and Ashe (1962).

Similar to the studies reported above, Hood et al. (1966) also demonstrated an increase in ventilation rate during WBV. Their findings supported the hypothesis that the physiological changes observed were due to maximal stretching of muscle spindles and reflex muscular contractions under optimal vibratory conditions. They emphasized that greater mechanical movements of the human body during WBV, specifically of the extremities, may result "in a mild circulatory stress, suggesting patients with severe cardiovascular disease should avoid rough means of transportation".

The oxygen uptake responses reported in the present study (Table 4.2) are also in accordance with the 'light physical activity' values reported in the literature (McArdle et al. 1996, Pope et al. 1990). Pope et al. (1990) exposed seated subjects to intermittent (at 5 Hz) and continuous (0, 2.5, 5, and 8 Hz) WBV, and studied the effect of posture (relaxed and erect) on absolute oxygen uptake responses. Similar to the present study, their results showed an increase in oxygen uptake during WBV (0.24 L.min⁻¹ to 0.29 L.min⁻¹) when compared to sitting rest without WBV. Also, similar to the present study, they did not observe any significant increase in oxygen uptake with the increase in WBV dose. Based on the low oxygen uptake responses, these authors concluded that changes observed in physiological responses during WBV resembled light machining work.

During exposure to WBV in a semi supine restrained position (with doses of 2 to 12 Hz in steps of 2 Hz increments; and accelerations of 0.6g and 1.2g), Hood et al. (1966) demonstrated significant cardiorespiratory changes in the range of 8 -10 Hz (at 1.2g). Such changes were accompanied by increased mixed arterialvenous oxygen difference, decreased peripheral vascular resistance index, and a slight increase in stroke index, suggesting physiological responses were similar to a typical semi-supine 'exercise' pattern. At 0.6g and at frequencies above and below 8-10 Hz range, these authors attributed the small increase in cardiac index (without increased heart rate or oxygen consumption) to the mechanism of overperfusion relative to the recovery values. This suggested that at lower intensities of WBV a venous pump mechanism is usually activated. However, unlike the significant differences obtained in cardiovascular responses in Hood et al. (1966), the cardiovascular responses such as cardiac output, stroke volume, and mixed arterio-venous oxygen difference obtained in the present study (Table 4.3) were unchanged during exposure to 3 Hz, 4.5 Hz, and 6 Hz. This discrepancy between the two studies may be due to the different dose and intensity combinations of WBV, and the postures adopted during exposure to WBV.

Role of Sitting 'With' and 'Without' a Backrest during WBV

When leaning against the backrest, subjects tend to transfer most of their body weight to the backrest, thus reducing the load on the low-back imposed by the upper body weight (Andersson et al. 1991). During WBV 'Without' backrest, absolute and relative oxygen uptake significantly increased by 11% and 12%, respectively than 'With' backrest condition. Further, a small but significant increase of 4.5% was observed in the heart rate during sitting 'Without' backrest

condition. Since oxygen pulse is calculated as the ratio of oxygen uptake and heart rate, this value was 7% higher during WBV 'Without' backrest. Ventilation rate was 11% higher 'Without' backrest than 'With' backrest as well. Interestingly, neither respiratory frequency nor tidal volume influenced the increase in ventilation rate.

Few authors have reported similar oxygen uptake and heart rate responses during postural variation in the absence of WBV (Sato and Tanaka 1973). and during WBV (Magnusson et al. 1987, Pope et al. 1990). As the posture becomes more upright, venous return to the heart will decrease due to an increase in gravitational force, thereby lowering cardiac output (Sato and Tanaka 1973). Consequently there would be an increase in the heart rate to maintain cardiac output (mediated through baroreceptor mechanism, with changes being observed in distribution of blood and peripheral resistance). However, in the present study, cardiac output during sitting without backrest (in an erect posture) was 8% greater than sitting with backrest, suggesting that 'WBV' might have stimulated such a reverse response. An increase in cardiac output might have also been facilitated by an increase in both the heart rate and stroke volume responses. Further, Sato and Tanaka (1973) attributed the increase in heart rate during upright posture to the static muscle contractions necessary to sustain the adopted posture. This would increase the intramuscular tension leading to an inefficient venous pumping mechanism, subsequently resulting in reduced venous return to the heart. They reported a correlation coefficient of 0.81 between relative oxygen uptake and heart rate responses during various postures.

Magnusson et al. (1987) monitored oxygen uptake in subjects 'Without' backrest support during: a) sitting (without WBV) in a relaxed symmetric posture; b) sitting erect without WBV; c) axial trunk rotation to the left without WBV; and d) WBV plus axial rotation, respectively. Their results indicated that axial rotation and WBV exposure (to a constant frequency of 5 Hz and acceleration 2.0 m/sec^2) increases the oxygen uptake by 32% compared to non-vibration and non-twisted tests. During exposure to intermittent (at 5 Hz) and continuous WBV

(0, 2.5, 5, and 8 Hz), Pope et al. (1990) found an increase in oxygen uptake in the erect posture compared to the relaxed posture. These results essentially suggest more energy is expended in maintaining an erect posture during sitting 'Without' backrest condition. Based on these findings, Pope et al. (1990) postulated that muscle contraction to maintain a posture leads to muscle fatigue, resulting in shifting loads on to ligaments and discs.

Role of Workload during Whole-Body Vibration

The responses observed in the present study during WBV combined with maximal rhythmic handgrip contractions can be related to WBV studies that combined physical and psychological work during WBV (Bennett et al. 1977, Grether et al. 1971 and 1972, Manninen 1984, 1985, and 1988, Thomton et al. 1984, Webb et al. 1981). In the present study, both absolute and relative oxygen uptake responses showed approximately 36% increase during WBV combined with maximal rhythmic handgrip contractions (Table 4.2). Further, heart rate and oxygen pulse values were 16% and 24% higher during WBV combined with maximal rhythmic handgrip contractions, respectively. The increase in ventilation rate (37%) paralleled the increase in oxygen uptake values. However, respiratory frequencies were similar during both conditions, suggesting that the increase in ventilation rate was primarily due to an increase in tidal volume (39%).

Bennett et al. (1977) demonstrated that performance of a motor task (tracking) during WBV might be affected by the physiological cost (heart rate and oxygen consumption). They showed an increase in oxygen consumption (0.31 L.min⁻¹ to 0.42 L.min⁻¹) with WBV exposure, and suggested that a combination of WBV and tracking tasks may lead to a higher metabolic cost. Additionally, they also classified subjects into internal and external 'personality' types and showed that internal subjects, who believed in themselves and were more successful in performing any type of task, demonstrated less performance decrements accompanied by greater physiological cost. Thus, internal subjects showed greater heart rate responses than external subjects during a motor task combined with WBV.

Webb et al. (1981) also studied the role of personality type in performance and physiological cost during WBV. 'Internal' subjects who assume their effort in a given task as a personal challenge, showed less performance decrements (indicated by less tracking errors) and greater physiological cost (measured by heart rate responses) than 'external' subjects. They attributed the significant results of internal subjects' responses to higher level of job satisfaction, suggesting "...conditions which foster higher job satisfaction may cause people to have greater confidence in their own influence over events". These authors also hypothesized that internal subjects show similar characteristics of type A behavior pattern (greater competitiveness and enthusiasm to control the environment), and are often linked with coronary heart disease, thus suggesting long-term effects may be psychosomatic.

Grether et al. (1971 and 1972) studied the effect of environmental stressors namely heat, noise, and WBV (of 5 Hz at peak 0.5g), individually and/or in combination, on psychological (tracking, choice reaction time. voice communication, mental arithmetic, visual acuity, and subjective severity ratings on a 7 point scale), and physiological responses (body temperature, heart rate, weight loss, and biochemical urine analysis). Interestingly, with the exception of mental arithmetic tests and subjective ratings, their results showed lower values in physiological and performance measurements during exposure to combined stressors. They concluded that additive or synergistic interactions among the three stressors were absent. However, subjective stress severity ratings increased with the number of stressor combinations. Moreover, Thornton et al. (1984) showed that handling controls during hovering required the greatest energy expenditure. The oxygen uptake for aviation pilots during level flight was almost 50% higher than that of sitting at rest. Even though the mental demand during flying at low levels was greater, they did not find significant differences in energy expenditure between flying at cruising altitude and flying at low level.

Manninen (1984) exposed subjects to WBV alone, noise alone or a combination of these two during dynamic work. Subjects intermittently pulled a handle attached to a pneumatic piston-cylinder mechanism with one hand. The

workload was adjusted to correspond to 2, 4, and 8 Watts. The author demonstrated an increase in heart rate values during all exposures and in all exposure combinations when compared to the control values, suggesting that the change in heart rate was not only accomplished with the dynamic work but also with the quality of noise and WBV exposure. Consistent with the findings of Bennett et al. (1977) and Webb et al. (1981), Manninen (1985) showed that mental load caused by 'competition' increased the temporary hearing threshold, the heart rate, and systolic blood pressure when subjects were simultaneously exposed to noise or a combination of noise and WBV. In a related study, Manninen (1988) showed that duration of exposure and temperature differences in the environment enhanced the changes induced by shorter exposure periods and smaller differences in temperature. A rise in the hearing threshold values was very strongly correlated with a rise in the heart rate, systolic blood pressure and hemodynamic index values (a product of heart rate and systolic blood pressure divided by one thousand), especially when the subjects were sitting and performing light, self paced work while exposed to WBV and noise.

Thus from the above studies, it is quite evident that combined stressors such as noise, light, temperature, physical work, and psychological load combined with WBV, elicit variable physiological responses in humans. As Bennett et al. (1977) stated "...cognitive factors in personality are involved in attempts to maintain performance during vibration stress and that the level of performance achieved may be associated with the physiological cost incurred".

Role of Gender

During WBV alone and WBV combined with maximal rhythmic handgrip contractions, men showed 15% and 27% higher absolute oxygen uptake values than women, respectively (Figure 4.2A). When oxygen uptake was normalized relative to body mass, gender differences were absent during the WBV alone condition (Figure 4.2B). However, during WBV combined with maximal rhythmic handgrip contractions, men showed a 14% higher relative oxygen uptake value than women (Figure 4.2B). Since heart rate values were not significantly different

between genders, the difference in oxygen pulse obtained during both conditions of workload were attributed to oxygen uptake responses only (Figure 4.3). Ventilation rate and respiratory frequency were similar but tidal volume was 24% higher in men compared to women. The latter was most likely due to differences in body size between the genders. Similarly, men showed greater grip strength than women (34%), which might have induced the greater increase in physiologic responses during WBV combined with maximal rhythmic handgrip contractions. However, correlation coefficients (Table 4.4) were not consistent among three vibration tests 'With' and 'Without' backrest.

To our knowledge, no WBV study to date has differentiated physiological responses between men and women. Pope et al. (1990) demonstrated differences in resting oxygen uptake between men and women; however, they did not differentiate gender responses during WBV. In the present study, cardiac output, stroke volume, and mixed arterio-venous oxygen difference was 19%, 13%, and 12% higher in men than women, respectively. However, gender per se did not influence heart rate responses. This phenomenon suggests that an increase in cardiac output could be attributed purely to an increase in stroke volume. Also, the higher absolute oxygen uptake in men could be the result of increase in both cardiac output and mixed arterio-venous oxygen difference.

Role of Fitness

The peak aerobic capacity during arm cranking in women were less but not significantly different from men. Despite the low metabolic rates during WBV, the effect of aerobic fitness was evident only in men for absolute oxygen uptake (r=0.73, P=0.005), and relative oxygen uptake responses (r=0.73, P=0.008) during handgrip contractions. Further, fitness did not influence any other physiological responses. These findings are also consistent with the postulations made by Boff and Lincoln (1988) and Griffin (1990) that physical fitness does influence exposure to WBV. However, they did not report any WBV study that has identified the importance of aerobic fitness. At present it is not clear why such an intra- and inter-gender differences in correlations exist in these vibration tests.

A few studies have reported that during 'heavy work' such as repetitive lifting, aerobic fitness did not influence low-back complaints (Battie et al. 1989, Kujala et al. 1996). In contrast, several authors (Cady et al. 1979, Craig et al. 1998, Davis et al. 1982, Williford et al. 1999) showed a strong correlation between lack of physical fitness and injury. Factors such as work demanding high fitness in fire fighting (Cady et al. 1979, Davis et al. 1982, Williford et al. 1999) and high frequency lifting in some industries (Craig et al. 1998) might have been the reasons for such a discrepancy in the role of fitness. However, in the present study, based on the acute physiological responses obtained, the results suggest that the level of aerobic fitness in participating subjects does play a significant role during even 'light work' experienced in seated WBV.

CONCLUSION

In the present study, the role of dose, backrest, workload, and gender on whole-body acute physiological responses during seated WBV were investigated. No significant increase in metabolic responses was observed in men and women during exposure to discrete doses 3 Hz, 4.5 Hz, and 6 Hz. However, only ventilation rate and tidal volume responses reached significance during 3 and 6 Hz. The responses during sitting 'Without' backrest were significantly higher than 'With' backrest condition suggesting that postural muscular load imposed by static sitting might result in higher metabolic load. WBV alone showed much lower increase in physiological responses than WBV combined with maximal rhythmic handgrip contractions, suggesting WBV combined with physical workload can influence physiological responses in both men and women. Additionally, gender played a significant role during WBV combined with maximal rhythmic handgrip contractions in absolute and relative oxygen uptake, and oxygen pulse responses only. These responses suggest if subjects were

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

exposed to WBV combined with physical workload, they would experience greater metabolic responses.

Cadiovascular responses were not influenced by the WBV dose. However, 'With' backrest, cardiac output showed lowest responses than either rest without WBV or 'Without' backrest during WBV. During WBV 'Without' backrest, heart rate values were higher than both sitting 'With' backrest during WBV and initial rest without WBV conditions. Higher stroke volume values were also reported 'Without' backrest than 'With' backrest conditions. However, mixed arteriovenous oxygen difference was not influenced by any condition. Men showed significantly highest cardiac output, stroke volume and mixed arterio-venous oxygen difference responses than women. Aerobic fitness played a significant role in determining the absolute and relative oxygen uptake, and oxygen pulse values during workload.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

REFERENCES

- 1. American College of Sports Medicine (1993) Resource manual for guidelines for exercise testing and prescription. Lea & Febiger, Philadelphia, USA.
- Andersson GBJ, Chaffin DB, Pope MH (1991) Occupational biomechanics of the lumbar spine. In Occupational low back pain. Assessment, Treatment, and Prevention. Edited by Pope MH, Andersson GBJ, Frymoyer JW, Chaffin DB. Mosby Year Book, Chicago, pp: 20-43.
- Battie MC, Bigos SJ, Fisher LD, Hansson TH, Nachemson AL, Spengler DM, Wortley MD, Zeh JH (1989) A prospective study of the role of cardiovascular risk factors and fitness in industrial back pain complaints. *Spine* 14(2): 141147
- Bennett MD, Webb R, Withey WR (1977) Personality, performance and physiological cost during vibration. Proceedings of the Physiological Society: 75-76
- 5. Bhambhani Y, Norris S, Bell G (1994) Prediction of stroke volume from oxygen pulse measurements in untrained and trained men. *Can J Appl Physiol* 19: 49-59
- Boff KR, Lincoln JE (1988) Vibration and display perception. In: Engineering Data Compendium: Human Perception and Performance. AAMRL, Wright-Patterson Air Force Base, OH, pp 2064-2066
- 7. Borg G (1982) Psychophysical bases of perceived exertion. *Med Sci Sports Exerc* 29: 402-409
- Bovenzi M (1996) Low back pain disorders and exposure to whole-body vibration in the workpalce. Semin Perinatol 20(1): 38-53
- 9. Cady LD, Bischoff DP, O'Connell ER, Thomas PC, Allan JH (1979) Strength and fitness and subsequent back injuries in firefighters. *J Occ Med* 21(4): 269-272
- 10. Canadian Society for Exercise Physiology (1994) Revised Physical Activity Readiness Questionnaire – RPARQ. Gloucester, NB, CANADA.
- 11. Craig BN, Congleton JJ, Kerk CJ, Lawler JM, McSweeney KP (1998) Correlation of injury occurrence data with estimated maximal aerobic capacity and body composition in a high-frequency manual material handling task. *Am J Ind Hyg Assoc J* 59:25-33
- Davis PO, Dotson C, Santa Maria DL (1982) Relationship between simulated fire fighting tasks and physical performance measures. Med Sci Sport Exerc 14: 65-71

- 13. Dupuis H and Zerlett G (1986) The Effects of Whole-body Vibration. Springer-Verlag.
- 14. Dupuis H, Hartung E, Haverkamp M (1991) Acute effects of transient vertical whole-body vibration. Int Arch Occup Environ Health 63: 261-265
- 15. Gaeuman JV, Hoover GN, Ashe WF (1962) Human vibration exposure. Aerospace Med 33: 469-474
- 16. Grether W F, Harris C S, Mohr G C, Nixon C W, Shalburn M Sommer H C, Thaler V H, Reghte J H (1971) Effects of combined heat, noise and vibration stress on human performance and physiological functions. *Aerospace Med* 42(10): 1092-1097
- Grether W F, Harris C S, Ohlbaum M, Sampson PA, Guignard JC (1972) Further study of combined heat, noise and vibration stress. *Aerospace Med* 43(6): 641-645
- 18. Griffin MJ, Whitham EM, Parsons KC (1982) Vibration an comfort I: Translational set vibration. *Ergonomics* 26: 603-630
- 19. Griffin MJ (1990) Vibration and discomfort. In: *Handbook of Human Vibration*. New York, Academic Press, pp 43-123
- 20. Guinard JC (1985) Vibration. Chapter 15. In: Patty's Industrial Hygiene and Toxicology. Vol. 3b: Biological Responses. Wiley.
- Hansson T, Magnusson M, Broman H (1991) Back muscle fatigue and seated whole-body vibrations: an experimental study in man. *Clin Biomech* 6(3): 173-178
- 22. Hood W B, Murray R H, Urschel C W, Bower J A, Clark J G (1966) Cardio pulmonary effects of whole-body vibration in man. *J Appl Physiol* 21(6) : 1725-1731.
- 23. Hoover GN and Ashe WF (1962) Respiratory response to whole-body vertical vibration. *Aerospace Med* 33: 980-984
- 24. Jones NL (1986) Clinical exercise testing. Saunders, Toronto, pp 193-194
- 25. Kujala UM, Taimela S, Viljanen T, Jutila H, Viitasalo JT, Videman T, Battie MC (1996) Physical loading and performance as predictors of back pain in healthy adults A 5-year Prospective study. *Eur J Appl Physiol* 73:452-458
- 26. McArdle WD, Katch I, Katch VL (1996) Exercise Physiology, Williams & Wilkins, Baltimore, pp 159

- 27. Magnusson M, Pope MH, Lindell V, Svensson M, Andersson G (1987) The metabolic cost of occupational exposures. *Ergonomics* 30 (1): 55-60
- 28. Manninen O (1984) Hearing threshold and heart rate in men after repeated exposure to dynamic muscle work, sinusoidal Vs stochastic whole body vibration and stable broad band noise. Int Arch Occup Environ Health 54: 19-32
- 29. Mannien O (1985) Cardiovascular changes and hearing threshold shifts in men under complex exposures to noise, whole-body vibrations, temperature and competition type psychic load. *Int Arch Occup Environ Health* 56: 251-274
- 30. Manninen O (1988) Changes in hearing, cardiovascular functions, haemodynamics, upright body sway, urinary catecholamines and their correlates after prolonged successive exposures to complex environmental conditions. *Int Arch Occup Environ Health* 60: 249-272
- 31. Marks C, Katch VL, Rocchini A, Rosenthal A (1985) Validity and reliability of cardiac output by CO₂ rebreathing. *Sports Med* 2: 432-446
- 32. Paddan GS, Griffin MJ (1988) The transmission of translational seat vibration to the head I. Vertical seat vibration. *J Biomech* 21(3): 191-197
- 33. Panjabi MM, Andersson GBJ, Jorneus L, Hult E, Mattsson L (1986) In vivo measurement of spinal column vibrations. *J Bone Joint Surg* 68A(5): 695-703
- 34. Pope MH, Wilder DG, Jorneus L, Broman H, Svensson M, Andersson GBJ (1987) The response of the seated human to sinusoidal vibration and impact. *Am* Soc Mech Eng J Biomech Eng 109(4): 279-284
- 35. Pope MH, Magnusson M, Lindell V, Svensson M, Andersson GBJ (1990). The measurement of oxygen uptake under whole-body vibration. *Iowa Orthop* J 10: 85-88
- 36. Pope MH, Magnusson M, Wilder D (1998) Kappa Delta Award Low back pain and whole-body vibration. *Clin Orthop Research* 354: 241-248
- 37. Sato M and Tanaka S (1973) Postural Effect on the Relation between Oxygen Consumption and Heart Rate. *J Human Ergology* 2: 21-30
- 38. Schmidt RT and Toews JV (1970) Grip strength as measured by the Jamar dynamometer. Arch Phys Med Rehab 51: 321-327
- 39. Seidel H and Heide R (1986) Long term effects of whole-body vibration: a critical survey of the literature. Int Arch Occup Environ Health 58: 1-26
- 40. Seidel H, Harazin B, Pavlas K, Sroka C, Richter J, Bluthner R, Erdmann U, Grzesik J, Hinz B, Rothe R (1988) Isolated and combined effects of prolonged

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

exposures to noise and whole-body vibration on hearing, vision and strain. Int Arch Occup Environ Health 61: 95-106

- 41. Thornton R, Brown GA, Higenbottam C (1984) The energy expenditure of helicopter pilots. Avi Space Envi Med 55: 746-750
- 42. Weinstein J, Pope M, Schmidt R, Seroussi R (1988) Neuropharmacologic effects of vibration on the dorsal root ganglion An animal model. *Spine* 13(5): 521-525
- 43. Wikstrom BO, Kjellberg A, Landstrom U (1994) Health effects of long-term occupational exposure to WBV: a review. Int J Ind Ergo 14: 273-292
- 44. Wilder D, Woodworth BB, Frymoyer JW, Pope MH (1982) Vibration and the human spine. *Spine* 7(3): 243-254
- 45. Williford HN, Duey WJ, Olson MS, Howard R, Wang N (1999) Relationship between fire fighting suppression tasks and physical fitness. Ergonomics 42(9): 1179-1186

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Table 4.1A:	Demographics	of subjects
-------------	--------------	-------------

Gender	Age	Body Mass	Sitting Height	Standing Height	Body Mass Index Standing	
	Yrs	Kgs	m	m	Kg. m ⁻²	
Men N=13	24.7 ± 3.9	70.8 ± 12.1	0.86 ± 0.05	1.71 ± 0.07 •	24.0 ± 3.2	
Women N=14	23.9 ± 3.5	60.3 ± 9.0	0.84 ± 0.08	1.63 ± 0.10	23.0 ± 3.7	

^a – significant difference between Men and Women (p < 0.05)

Table 4.1B: Demographics of subjects

Gender		Neck Dimer	nsion	Maximal Right Hand	Peak Oxygen Uptake	
	Length C6 ^Ψ		Occipital ^Ψ	Grip Strength	Arm Cranking	
	Cm	Cm	Cm	Kgs	mL.Kg ⁻¹ .min ⁻¹	
Men N=13	12.6 ± 1.4	37.8 ± 3.2*	37.5 ± 2.8 *	34.2 ± 9.1 °	24.5 ± 6.6	
Women N=14	12.3 ± 1.5	32.4 ± 1.7	31.5 ± 1.8	22.5 ± 3.6	21.4 ± 4.6	

^{Ψ} Diameter measured at the placement of accelerometer at C6 and at Occipital regions

* - significant difference between Men and Women (p < 0.05)

Variable	Gender	Back-			WBV	Dose		
		rest	3 Hz Alone	3 Hz + Grip	4.5 Hz Alone ^c	4.5 HZ + Grip	6 Hz Alone ^c	6 Hz + Grip
Oxygen Uptake, L.min ⁻¹	Men	On	0.207 ± 0.032	0.405 ± 0.016	0.249 ± 0.068	0.412 ± 0.013	0.241 ± 0.078	0.411 ± 0.013
		Off	0.261 ± 0.063	0.421 ± 0.016	0.287 ± 0.080	0.454 ± 0.015	0.276 ± 0.090	0.454 ± 0.014
ļ	Women	On	0.194 ± 0.050	0.285 ± 0.064	0.200 ± 0.048	0.286 ± 0.067	0.214 ± 0.030	0.304 ± 0.078
		Off	0.223 ± 0.055	0.326 ± 0.093	0.224 ± 0.061	0.327.± 0.087	0.237 ± 0.040	0.323 ± 0.076
Oxygen Uptake, ml.kg ⁻¹ .min ⁻¹	Men ^d	On®	2.98 ± 0.6	5.72 ± 2.01	3.52 ± 0.73	5.87 ± 1.81	3.4 ± 0.93	5.94 ± 2.13
		Off	3.71 ± 0.7	5.93 ± 2.0	4.05 ± 0.78	6.38 ± 1.74	3.91 ± 1.03	6.7 ± 2.20
	Women	On	3.28 ± 0.9	4.80 ± 1.2	3.30 ± 0.8	4.90 ± 1.5	3.62 ± 1.0	5.2 ± 1.7
		Off	3.78 ± 1.1	5.6 ± 1.9	3.8 ± 1.1	5.5 ± 1.5	4.0 ± 0.8	5.5 ± 1.7
Heart Rate, beats.min ¹	Men	On®	76±9	91 ± 12	79 ± 10	95 ± 12	79±6	92 ± 13
		Off	81 ± 13	96±14	84 ± 10	99 ± 12	83±7	99±11
	Women	On	75±7	91±9	77±5	91±4	78±5	94±6
L		Off	79±7	94±8	81±6	95±6	84 ± 10	97 ± 8
Oxygen Pulse, mi.beat ^{**}	Men ^a	On	2.8 ± 0.7	4.5 ± 1.8	3.2 ± 0.9	4.3 ± 1.3	3.1 ± 1.0	4.5 ± 1.5
		Off	3.3 ± 1.1	4.5 ± 1.8	3.4 ± 1.0	4.6 ± 1.4	3.4 ± 1.1	4.6 ± 1.3
	Women	On®	2.6 ± 0.7	3.2 ± 0.7	2.6 ± 0.6	3.1 ± 0.7	2.8 ± 0.6	3.4 ± 0.8
		Off	2.9 ± 0.8	3.5 ± 1.1	2.8 ± 0.8	3.5 ± 0.8	2.9 ± 0.6	3.4 ± 0.8
Ventilation Rate, I.min	Men	On®	8.0 ± 1.4 *	16.0 ± 9.0^{a}	9.9 ± 2.5	14.3 ± 7.8	9.7 ± 2.6	18.1 ± 9.3
		Off	10.1 ± 2.7*	15.2 ± 7.8 *	10.9 ± 2.8	16.5 ± 6.7	10.6 ± 2.6	19.2 ± 7.6
	Women	On	8.4 ± 2.5*	11.3 ± 2.8*	8.5 ± 2.7	12.0 ± 3.9	9.2 ± 2.4	13.1 ± 5.3
		Off	8.8 ± 2.2*	12.8 ± 4.9*	9.4 ± 3.2	18.4±3	9.8 ± 1.8	13.2 ± 5.2
Respiratory frequency,	Men	On	19 ± 5 ^a	21±6*	22 ± 8	18±7	22 ± 6	24±6
Breaths.min ⁻¹		Off	$22 \pm 6^{*}$	21±4ª	23±8	21±5	20 ± 6	22 ± 5
	Women	On	24 ± 7*	25±9*	23±6	23±6	24±6	24±5
		Off	25 ± 9*	25±9*	24 ± 8	23±6	24±4	23 ± 6
Tidal Volume, L.breath"	Men	On	0.44 ± 0.1	0.81 ± 0.4	0.54 ± 0.2	0.93 ± 0.5	0.47 ± 0.17	0.80 ± 0.4
		Off	0.46 ± 0.11	0.76 ± 0.4	0.54 ± 0.26	0.84 ± 0.31	0.57 ± 0.25	0.90 ± 0.4
	Women	On	0.35 ± 0.06	0.49 ± 0.15	0.38 ± 0.12	0.56 ± 0.25	0.39 ± 0.06	0.56 ± 0.22
		Off	0.38 ± 0.10	0.53 ± 0.20	0.41 ± 0.13	0.95 ± 1.6	0.43 ± 0.11	0.64 ± 0.28

Table 4.2: Metabolic and Respiratory responses during WBV (Mean ± SD)

NOTE: All subjects were exposed to 3, 4.5 and 6 Hz doses. ^a – significant difference between 3 Hz and 6 Hz when all other main effects were pooled (p < 0.01) ^b – significant difference between WBV 'with' and 'without' backrest when all other main effects were pooled (p < 0.01), ^c – significant difference between vibration alone and vibration combined with rhythmic handgrip (except for Respiratory frequency) when all other main effects were pooled (p < 0.01)

significant difference between gender and work (p < 0.01)</p>
Variable	Dose	T	Men		Women		
		Rest without WBV	WBV 'with' Backrest	WBV 'without' Backrest	Rest without WBV	WBV 'with' Backrest	WBV 'without' Backrest
Cardiac Output, I.min ^{**}	3 Hz	6.5 ± 1.1*	6.5 ± 1.0 ^{eb}	6.7 ± 0.9*	5.6 ± 0.9	5.3 ± 1.3 °	5.7 ± 0.5
Heart Rate, beats. min ⁻¹		70 ± 8 ^{cd}	76±9 ^b	81 ± 12	71 ± 9 ^{cd}	79±7°	87 ± 7
Stroke Volume, ml.beat ⁻¹		95 ± 21 and	86 ± 13 ^{ab}	84 ± 14 *	79 ± 16 ^{cd}	60 ± 17 b	72±9
Arterio-venous oxygen difference, ml%		6.50 ± 1.1 *	6.89 ± 1.3 *	6.9 ± 1.3 *	5.93 ± 1.0	5.88 ± 1.1	5.82 ± 1.1
Cardiac Output, I.min ⁻¹		6.7 ± 1.5*	6.2 ± 1.3 ^{ab}	6.7±0.9°	5.4 ± 0.9	4.8 ± 0.9 b	5.7 ± 0.8
Heart Rate, beats. min ⁻¹	4.5 Hz	72 ± 10 ^{&}	79 ± 10 ^b	84 ± 10	69 ± 8 [∞]	77±5°	81±6
Stroke Volume, ml.beat ⁻¹		94 ± 24 acd	78 ± 15 **	80 ± 9°	79 ± 13 ^{cd}	63 ± 12 ^b	70 ± 9
Arterio-venous oxygen difference, ml%		6.83 ± 1.2*	6.50 ± 1.2 ^a	6.64 ± 1.1	6.11 ± 1.2	6.14 ± 1.0	6.18 ± 0.9
Cardiac Output, I.min ⁻¹		6.9±1.1*	6.8 ± 1.9 *	7.5 ± 2.2*	5.1 ± 0.5	5.4 ± 0.6 ^b	5.9 ± 0.9
Heart Rate, beats. min ⁻¹	6 Hz	72 ± 5 ^{cd}	79±6 ^b	83 ± 7	71 ± 6 ^{∞1}	78±5 ^b	84 ± 10
Stroke Volume, ml.beat ⁻¹		97 ± 16 acd	86 ± 23 ^{sb}	90 ± 26 *	73 ± 9 ^{cd}	70 ± 9 ^b	71±14
Arterio-venous oxygen difference, ml%		6.90 ± 1.0*	6.80 ± 1.0 [®]	6.91± 1.0*	5.91 ± 1.1	5.95 ± 0.8	5.98 ± 0.8

Table 4.3: Cardiovascular responses during WBV (Mean ± SD)

^a – significant difference between men and women when all other main effects were pooled (p < 0.01),
 ^b – significant difference between conditions WBV 'With' and 'Without' backrest when all other main effects were pooled (p < 0.01)
 ^c – significant difference between conditions rest and WBV 'With' backrest when all other main effects were pooled (p < 0.01)
 ^d – significant difference between conditions rest and WBV 'Without' backrest when all other main effects were pooled (p < 0.01)

GENDER	Selected Variables	3 Hz		4.5 Hz		6 Hz	
		With' Backrest	Without' Backrest	With' Backrest	Without' Backrest	With' Backrest	Without' Backrest
Men	Oxygen Uptake, L.min	0.81 [♥]	0.69 [•]	0.38	0.63 ^{\mathcal{P}}	0.42	0.24
	Oxygen Uptake, mL.kg ⁻¹ .min ⁻¹	0.57 [♥]	0.44	-0.01	0.35	0.04	-0.12
	Heart Rate, beats.min ⁻¹	-0.23	-0.52	0.20	0.23	0.21	-0.01
	Oxygen Pulse, mL.beat ⁻¹	0.83 [♥]	0.76♥	0.27	0.54	0.34	0.21
	Ventilation Rate, L.min ⁻¹	0.60 [♥]	0.52	0.46	0.57*	0.37	0.52
	Respiratory frequency, breaths.min ⁻¹	-0.10	0.02	-0.22	-0.13	-0.41	-0.49
	Tidal Volume, L.breath ⁻¹	0.62 [♥]	0.46	0.51	0.62♥	0.48	0.74♥
Women	Oxygen Uptake, L.min ⁻¹	-0.20	0.65 ^Ψ	0.54 ^Ψ	0.14	0.40	0.46
	Oxygen Uptake, mL.kg ⁻¹ .min ⁻¹	0.27	0.67*	0.65 [♥]	0.28	0.44	0.54 ^Ψ
	Heart Rate, beats.min ⁻¹	-0.08	0.02	-0.21	-0.23	-0.08	0.16
	Oxygen Pulse, mL.beat ⁻¹	-0.18	0.56 ^{\mathcal{P}}	0.60 [♥]	0.21	0.39	0.38
	Ventilation Rate, L.min ⁻¹	-0.12	0.69 [*]	0.58 [¥]	-0.21	0.45	0.30
	Respiratory frequency, breaths.min ⁻¹	-0.21	-0.01	-0.11	0.35	-0.07	0.08
	Tidal Volume, L.breath ⁻¹	0.04	0.59 [♥]	0.36	-0.26	0.50	0.28
						L	

Table 4.4: Pearson Correlations between Metabolic responses and Maximal Rhythmic Handgrip Contractions during WBV

*- Significance at P< 0.05



Figure 4.1A Subject Setup for Metabolic Measurements



Figure 4.1B Noninvasive Cardiac Output Using CO₂ Rebreathing Technique



Figure 4.2A. Interaction between Gender and Work Absolute Oxygen Uptake Changes

NOTE: 1. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest). Also, values represent mean \pm SD from 13 men and 14 women. 2. Graph suggests that during both conditions of workload, men demonstrated greater changes in oxygen uptake compared to women.

a - significant difference between men and women during WBV alone

b - significant difference between men and women during WBV with Work Rthythmic Handgrip





NOTE: 1. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest). Also, values represent mean \pm SD from 13 men and 14 women. 2. Graph suggests that during both conditions of workload, men demonstrated greater changes in oxygen uptake compared to women.

a - significant difference between men and women during WBV alone

b - significant difference between men and women during WBV with Work Rthythmic Handgrip





NOTE: 1. Values for the graphs are based on the pooled mean values of remaining independent variables (Dose and Backrest). Also, values represent mean \pm SD from 13 men and 14 women.

2. Graph suggests that during both conditons of workload, men showed greater differences in oxygen pulse compared to women.

a - significant difference between men and women during WBV alone

b - significant difference between men and women during WBV combined with Rhythmic Handgrip

CHAPTER 5

Cardiorespiratory and Biceps Brachii Oxygenation and Blood Volume Comparisons during Arm Cranking and Task-specific Pushing-Pulling

Pushing and pulling is a manual material handling activity, and is defined as the application of one or two handed force on an object or person, with feet either stationary (Chaffin et al. 1983, Gagnon et al. 1992, Garg et al. 1978) or moving (Lee et al. 1991, Snook and Ciriello 1991). For example, pushing and pulling loads from the shelves (Gagnon et al. 1992), floor mopping or snow shovelling can be treated as effort applied with feet stationary (Franklin et al. 1995, Søgaard et al. 1996). Cart pushing and pulling is an example with feet moving (Snook and Ciriello 1991). In pushing, the applied force is directed away from the body, and in pulling, the resistance is directed towards the body (van der Beek et al. 2000). Similar to a repetitive lifting and/or lowering activity, repetitive pushing and pulling is also accompanied by isometric muscle activity in the trunk and upper extremities of workers leading to localized fatigue and increased cardiovascular stress (Hoozemans et al. 1998). Further, this activity is often implicated in both low-back and upper extremity disorders (Bernard and Fine 1997, OSHA Draft 1999).

In the field of Occupational Health and Safety, it is standard practice to evaluate the worker's fitness for work using traditional incremental aerobic protocols on a treadmill, cycle ergometer or arm ergometer, instead of task-specific maximal or submaximal aerobic protocols. These diagnostic protocols, though informative, may or may not identify the individual's actual physical capacity to do the job due to the different movements involved in the tasks compared to standardized running, cycling, or arm cranking protocols (Garg et al. 1992, Khalil et al. 1985, McConnell et al. 1984, Sharp et al. 1988, Waters et al. 1993). A recent study by Nindl et al. (1998) showed a modest correlation in peak oxygen uptake values between repetitive lifting and treadmill running (r = 0.73 and r = 0.60 for men and women, respectively). Further, Sharp et al. (1988) estimated maximal aerobic power of repetitive lifting as 78% of treadmill, 89% of

cycling and 125% of arm cranking values, whereas for various lifting tasks, Khalil et al. (1985) reported maximal aerobic power as 91% of cycling values. Furthermore, depending on the weight of the load lifted or lowered, and rate of the activity performed, average lifting and lowering maximal aerobic power values were 19% to 47% lower than the cycling values (Petrofsky and Lind 1978). However, to our knowledge comparison of peak aerobic values during maximal pushing and/or pulling to the standardized protocols have not been reported to date.

It is evident from the findings cited that there is considerable variation between standardized and task-specific protocols for the measurement of aerobic power. Moreover, these differences emphasize that the exercise mode selected to measure aerobic power should simulate the specific task to be performed in the occupational setting (Sharp et al. 1988). Few physiological studies have reported energy expenditure during pushing and pulling with feet stationary (Esmail et al. 1995, Garg et al. 1978). Gender comparisons during this activity are also scarce in the literature. In a regression model of pushing and pulling of loads at different heights, Garg et al. (1978) demonstrated higher energy expenditure in women compared to men. Esmail et al. (1995) established both biomechanical and physiological normative data in healthy males and females during a push-pull task on a work simulator for a period of 4 min. Their results showed significant differences between genders in their relative maximal aerobic capacity and energy expenditure per unit amount of work done (men showed 19% lower values than women).

Near infra-red spectroscopy (NIRS) is a non-invasive optical technique that has been used to measure relative changes in oxygenation and blood volume levels in skeletal muscle continuously in real-time during rest and work (Bhambhani et al. 1998, Boushel et al. 1998, Chance et al. 1992, Mancini et al. 1994, Murthy et al. 1997). Muscle oxygenation, defined as the relative saturation of oxyhemoglobin and oxymyoglobin, depends on the balance between oxygen delivery and oxygen utilization (Mancini et al. 1994). Blood volume is defined as the sum of oxygenated and deoxygenated states of hemoglobin and myoglobin in

the illuminated region of the muscle. Detailed description of the NIRS theory, its validation with various established optical and magnetic techniques are provided in Chapter 1. The reliability of NIRS in the upper and lower extremity human skeletal muscles has also been reported in Chapter 1. Under ischemic conditions induced by 250 mm Hg blood pressure in the biceps brachii at rest, Maikala and Bhambhani (1999) showed correlations of 0.86 to 0.96 (between sessions), and 0.95 to 0.97 (within sessions) for oxygenation measurements.

To our knowledge, no studies to date have evaluated the whole-body physiological and oxygenation and blood volume trends of the biceps brachii muscle group during repetitive pushing and pulling, and compared these responses with a standardized incremental arm cranking protocol. Hence, the purposes of this study were to compare the: (1) whole-body physiological responses during incremental pushing-pulling (with feet stationary) and incremental arm cranking till volitional exhaustion; (2) oxygenation and blood volume responses of the biceps brachii muscle during both protocols, and (3) differences in these responses between genders. It was hypothesized that: (1) there would be no significant differences in the whole-body physiological responses between the two modes of exercise, (2) the biceps brachii oxygenation and blood volume trends would be similar during these exercise modes, and (3) there would be no significant gender differences in these responses.

MATERIALS AND METHODS

Subjects

Written informed consent was obtained from 11 healthy men (age 23.7 \pm 4.5 yrs, mass 77.5 \pm 13.8 kg, height 176 \pm 8 cm) and 11 healthy women (age 24.5 \pm 3.51 yrs, mass 57.6 \pm 9.4 kg, height 163 \pm 7 cm). Subjects were undergraduate and graduate students recruited from the university population (Table 5.1). They all completed the Revised Physical Activity Readiness Questionnaire (Canadian Society for Exercise Physiology, 1994) to identify 200

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

contraindications for exercise. The test protocols utilized were approved by the Human Research Ethics Board at this institution (APPENDIX A.2).

Description of the Baltimore Therapeutic Equipment (BTE) Work Simulator:

The BTE work simulator (Baltimore Therapeutic Equipment Co., Maryland, USA) is a multi-purpose instrument that can be used for the measurement of upper-extremity strength and endurance during functional tasks (Figure 5.1). With various customized attachments provided by the manufacturer, this simulator quantifies the work performed, force exerted (in terms of torque), and the power output during various simulated occupational activities. It is widely used in the rehabilitation centers by occupational and exercise therapists for static and dynamic testing in both work evaluation and work hardening.

The work simulator essentially consists of three components. First, a multi-purpose exercise head that is mounted on an arm that extends from the vertical column. Force output from the head is measured using a built-in strain gauge. This exercise head with an electronically controlled variable-resistance mechanism is adjustable to a vertical height of 33 cm to 167.6 cm. Second. various customized attachments provided by the manufacturer can be used to simulate common tasks in the occupational setting. For pushing-pulling without feet moving attachment # 171 is used. This attachment is a lever consisting of a straight handle with an articulating joint producing linear motion. The resistance can be applied in one (either away from or towards the subject) or both directions. For the current study, resistance was applied in both directions during the incremental pushing-pulling protocol. Third, a computer software program was developed for conversion of mechanical loads generated during work into power units. By using the manufacturer's conversion chart, one can convert the torque generated into the load (in-lbs) to be pushed or pulled. This information is displayed on the console and can be used to control the work rate. The software provided with the BTE work simulator calculates the force output in pounds and the power output in watts. Before each testing session, the BTE work simulator was balanced in such a way that the strain gauge in the exercise head was in synchrony with the electronics in the control console (Powell et al. 1991).

Thereafter, the simulator was calibrated as per the manufacturer instructions (BTE Operator's Manual, 1992).

Reliability of the BTE Work Simulator

To date. only few studies have compared the measurements obtained from the BTE work simulator with real (field) measurements. Beaton et al. (1995) tested the grip strength using both the hydraulic Jamar dynamometer and the Intra-class correlation coefficient between the Jamar BTE work simulator. instrument at 90° and the BTE work simulator was 0.87. Pearson's correlation coefficients at 45° and 90° elbow flexion were identical (0.90). Kennedy and Bhambhani (1991) studied the test-retest reliability and criterion validity of the BTE work simulator during three intensities (light, medium, and heavy) of real and simulated work. Correlation coefficients between trials of the same task for oxygen consumption ranged from r=0.74-0.87 and for heart rate from r=0.59-0.78. Higher correlation coefficients were observed for light work (oxygen consumption: r=0.81 and 0.83; heart rate: r=0.88 and 0.95), followed by heavy work (oxygen consumption: r=0.68 and 0.75; heart rate: r=0.91 and 0.90) and medium work (oxygen consumption: r=0.56 and 0.52; heart rate: r=0.91 and 0.92) respectively.

Test Protocols

Each subject completed a cuff ischemia test on the first day, and an incremental arm cranking exercise and pushing-pulling test on two separate days in a random order.

(1) Cuff Ischemia Measurements

The purpose of the cuff ischemia test was to establish minimum and maximum oxygenation and blood volume values for the biceps brachii muscle at rest. The subject sat in a chair with the arm resting on the armrest, thus minimizing any extraneous muscular activity (Figure 5.3A). A blood pressure cuff was placed around the right upper arm just above the NIRS sensor on the biceps. Initial baseline values were monitored for two minutes. The cuff was

inflated to 250 mm Hg pressure for a period of six minutes so as to induce ischemia within the muscle at rest. The cuff was then deflated to restore blood flow to the muscle and recovery measurements were recorded for four minutes.

(2) Incremental Arm Cranking

In a separate testing session, the subject completed an incremental test on an arm cranking ergometer (Cybex, MET 300, USA). The axis of the ergometer crankshaft was adjusted to shoulder height of each seated subject. Position of the handle was adjusted in such a way that when the arms were fully extended, a slight flexion was present at the elbow joint. The legs and trunks were not restrained and their feet were flat on the ergometer base. After fitting each subject with a mouthpiece and headgear (Figure 5.3B), the test was initiated with a two-minute rest period. It was followed by two minutes of arm cranking at no load and subsequent increments of 25 watts every 2 minutes at 50 crank rotations per minute (rpm) until voluntary exhaustion, or attainment of two or more of the end points suggested by the American College of Sports Medicine (1993): (1) leveling off in the oxygen consumption (increase of \leq 100 ml.min⁻¹) with increasing load, (2) age predicted maximal heart rate, equated to 220-age (in years), (3) respiratory exchange ratio of ≥1.10, and (4) rate of perceived exertion (RPE) \geq 18 (Borg, 1982). Protocol was terminated with 2 minutes of active [cranking at zero load] and 4 minutes of passive recovery. During the last 30 s at each workload, the overall RPE was recorded using the Borg scale. Bar-Or and Zwiren (1975) and Davis et al. (1976) reported test-retest reliability coefficients of 0.94 and 0.92 respectively for the peak oxygen uptake at an arm cranking cadence of 50 rpm.

(3) Incremental Pushing-Pulling

In a separate testing session, each subject completed an incremental test on the BTE work simulator. After fitting the subject with a mouthpiece and headgear, the height of the push-pull lever #171 was adjusted to each subject's elbow crease in the standing upright posture (Figure 5.4). The right hand was

used as the main grasping hand (adopting a power grip) in this two-handed effort. The elbows were unlocked and subjects were discouraged from trunk rotation, swaying the body, and moving their lower extremities during pushingpulling. They were instructed to put their right leg (extended at the knee) backward at a comfortable distance, and the left leg forward [with a slight knee flexion]. The left toe was placed on a marker that was in line with the handle being pulled [similar to a posture adopted in Ayoub and McDaniel (1974)]. The subject was not permitted to lift the heels off the floor throughout the testing session.

The test was initiated with a two-minute rest period and was followed by two minutes of repetitive pushing-pulling at no load in the horizontal direction in a sagittal plane towards (Figure 5.5A) and away from the body (Figure 5.5B). Distance of pushing-pulling was restricted by a delimiter fixed on the circumference of the BTE work simulator wheel at a fixed distance of 10 cm. A push-pull cadence of 50 repetitions per minute was used for this protocol. Thereafter, the resistance was increased by 30 in-lb every two minutes until voluntary exhaustion or attainment of two or more of the following end points suggested above by the American College of Sports Medicine (1993). Protocol was terminated with 2 minutes of active [cranking at zero load] and 4 minutes of passive recovery.

(4) Cardiorespiratory Measurements

Gas exchange measurements were continuously monitored with an automated metabolic measurement cart (MMC Horizon, Sensormedics Corporation, Yorba Linda, CA). The oxygen (O_2) and carbon dioxide (CO_2) analyzers of the cart were calibrated before each test with commercially available precision gases (16% O_2 , 4% CO_2 , balance Nitrogen). The pneumotach was calibrated for volume using a three-liter syringe, as recommended by the manufacturer. Heart rate was recorded using a polar heart rate monitor (Polar Accurex Plus, Finland). Expired gases were analyzed in the mixing chamber mode. Calibration was verified at the end of each test to ensure accuracy of the

data. Peak oxygen pulse was calculated as the ratio of peak absolute oxygen uptake and peak heart rate, and peak ventilatory equivalent for oxygen as the ratio of peak ventilation volume and the peak absolute oxygen uptake. Furthermore, ratios of peak absolute oxygen uptake to the power output (metabolic efficiency), and relative oxygen uptake to the power output generated (relative metabolic efficiency) during each incremental exercise were also calculated.

(5) Biceps Brachii Oxygenation and Blood Volume Measurements

A continuous dual wave NIRS unit (MicroRunman, NIM Inc., PA, USA) was used to evaluate relative changes in the biceps brachii oxygenation and blood volume during both arm cranking and pushing-pulling tests. This unit consists of: a superficial near infra-red sensor fitted with six light emitting tungsten lamps placed 2 cm, 3 cm, and 4 cm apart; two photo-diode detectors that absorb light in the near infrared range at 760 nm and 850 nm; and a display unit that amplifies and displays the absorbency signal generated (Figure 2.2). The sensor was placed on the right biceps brachii (Figure 5.2), approximately 6-8 cm from the elbow crease (Jensen-Urstad et al. 1995). A piece of clear plastic was wrapped around the sensor to prevent sweat from distorting the absorbency signal. A dark elastic bandage was wrapped around the sensor to secure it in place and minimize any loss of light. Biceps brachii muscle skin fold thickness at the sensor location was measured using a Lange skinfold caliper (Cambridge Scientific Industries, Inc. Maryland, USA) before placing the NIRS sensor on the muscle belly. The adipose tissue thickness (a sum of fat and skin layer) at the sensor site was calculated as half of the skinfold thickness. Position of the NIRS sensor on the muscle belly was noted to the nearest millimeter, thus ensuring the correct placement of NIRS sensor on each subject for all the testing sessions.

As per the manufacturer's specifications, the sensor was calibrated at 760 nm and 850 nm wavelengths before each session. The source-to-detector separation that also determines the vertical penetration depth of light was set to 4 cm. The difference in absorbency between these two wavelengths indicated the

change in oxygen saturation and the sum of these absorbencies indicated the change in total blood volume. Both oxygenation and blood volume were measured in terms of optical density (OD), defined as the logarithmic ratio of the intensity of light calibrated at each wavelength to the intensity of measured light at the same wavelength (MicroRunman User Manual, NIM Inc., PA, USA, 1999). Real time data were recorded using the NIRCOM software provided by the manufacturer.

NIRS Data Analysis

All the real time NIRS values were averaged every 20 seconds using a customized Microsoft Excel software program specially designed for this study. Baseline values were recorded during initial 2 minutes rest. Minimum oxygenation and blood volume values for each subject were recorded during cuff ischemia. Maximum values were recorded during recovery from cuff ischemia. Further, minimum oxygenation values for each subject were recorded during both incremental protocols. Minimum blood volume values were recorded at the baseline [last 20 seconds] prior to the beginning of each exercise mode. Maximum oxygenation and blood volume were recorded during the recovery from exercise.

The range in oxygenation and blood volume responses for all the three protocols were calculated as the difference between the minimum and the maximum OD values. For each incremental protocol, muscle oxygenation and blood volume were calculated at 20, 40, 60 and 80% of the peak oxygen uptake obtained for that specific test. The oxygenation values obtained at these percentages were expressed as a percentage of the oxygenation range observed during each incremental session and recovery. The equation developed by Belardinelli et al. (1995) was used for this purpose:

Percent change in the biceps muscle oxygenation = $[(y-M) + (m-M)] \times 100$

where 'y': is the OD value corresponding to the time interval at which the specific percentage of peak oxygen uptake was obtained; 'M' was the minimum OD

observed during the incremental test; and 'm' was the maximum OD obtained during recovery from incremental work (Figure 1.4).

Statistical Analysis

Independent 't' tests were used to compare the physical characteristics of the men and women (Table 5.1). A two-way analysis of variance (ANOVA) with gender as a between-subjects factor, and mode (arm cranking and pushing-pulling) as a repeated measures factor was used to evaluate the differences in the peak biomechanical, cardiorespiratory, and perceptual responses. Also, a two-way ANOVA, with gender as a between-subjects factor, and protocol (arm cranking; pushing-pulling; and cuff ischemia) as a repeated factor was used to compare oxygenation and blood volume responses. To minimize the violation of the assumption of homogeneity of variance, the 'Greenhouse-Geiser' adjustment was used. Since multiple comparisons were made for each dependent variable in the present study, the Bonferroni adjustment for a P value of 0.05 was applied to minimize Type 1 error. The Bonferroni adjustment was calculated as the ratio of 0.05 and number of statistical comparisons (APPENDIX B.4). Thus, statistical significance was considered at P <0.01. Significant F ratios were further analyzed with the Scheffe post hoc multiple comparison test.

Regression analysis was used to examine the relationship of whole-body oxygen uptake and percent change in biceps brachii oxygenation responses at equal percentages of peak oxygen uptake between the two exercise modes for both genders. A Spearman correlation test was used to examine the relationship between logarithmic value of adipose tissue thickness and [minimum and maximum] oxygenation and blood volume responses for the three protocols. Since the adipose tissue thickness of the subjects in the present study was normally distributed, a nonlinear correlation test was used to examine its role on the NIRS measurements. The Statistical Package for Social Sciences SPSS (version 10) was used for all statistical analyses (SPSS Inc., Chicago, USA).

RESULTS

Peak Biomechanical, Physiological, and Perceptual Responses

Comparison between Incremental Exercise Modes:

The peak biomechanical, physiological, and perceptual responses during incremental arm cranking and pushing-pulling in the men and women are reported in Table 5.2. In general, most of the peak physiological responses were significantly higher during arm cranking compared to pushing-pulling in both men and women. A significant interaction between exercise mode and gender was obtained only for power output (Watts) implying power output during arm cranking was significantly higher (by 79%) than pushing-pulling, with men demonstrating significantly greater power output (30%) during both exercise modes (P<0.01). Absolute oxygen uptake (L.min⁻¹) during arm cranking was 24% greater than that attained during pushing-pulling (P<0.01). When oxygen uptake values were corrected for subjects' body weight, the relative oxygen uptake (mL.min⁻¹.Kg⁻¹) during arm cranking was 22% greater than pushing-pulling (P<0.01). The lower power output generated during pushing-pulling resulted in significantly greater ratios for the absolute peak oxygen uptake to the power output (by 72%); and relative peak oxygen uptake to the power output (by 74%). when compared to arm cranking (P<0.01).

Heart rate (beats.min⁻¹) during pushing-pulling was 91% of arm cranking but was not significantly different (P>0.05). Oxygen pulse, calculated as the ratio of oxygen uptake and heart rate (mL.beat⁻¹) was approximately 22% higher during arm cranking compared to pushing-pulling (P<0.01). Further, ventilation rate (L.min⁻¹) and respiratory frequency (breaths.min⁻¹) were approximately 22% and 17% greater during arm cranking compared to pushing-pulling (P<0.01). However, tidal volume (L.breath⁻¹), a ratio of ventilation rate and respiratory frequency (breaths.min⁻¹), did not demonstrate any significant difference between the exercise modes (P>0.05). Ventilatory equivalent for oxygen did not demonstrate significant difference between these exercise modes (P>0.05). Respiratory exchange ratio, a measure indicating the ratio of carbon dioxide

produced to oxygen consumed, was significantly greater by 5% during arm cranking (P<0.05). Unlike the physiological measures that showed greater responses during arm cranking, the rating of perceived exertion was significantly higher during pushing-pulling (P<0.05).

Comparison between Genders

The peak absolute oxygen consumption during pushing-pulling was approximately 74% (in men) and 79% (in women) of the respective values attained during arm cranking (P<0.01). Gender did not influence the oxygen consumption responses when it was corrected for body weight (P>0.05). Similarly, ratios of absolute and relative oxygen uptake to power output were not significantly different between genders (P>0.05). Peak heart rate did not demonstrate any significant difference between men and women during both modes of exercise (P>0.05). However, oxygen pulse was significantly higher [pushing-pulling - 30%, arm cranking - 45%] in men compared to women (P<0.01). Ventilation rate was significantly higher in men (37%) than in women due to a significantly greater (37%) tidal volume (P<0.01). However, no gender differences were observed for the respiratory frequency, ventilatory equivalent for oxygen, respiratory exchange ratio, and rating of perceived exertion (P>0.05).

Biceps Brachii Oxygenation and Blood Volume Trends

The oxygenation and blood volume trends during cuff ischemia and both incremental exercise modes in a typical male and female subject are shown in Figures 5.6 and 5.7, respectively.

Cuff Ischemia

During the cuff ischemia protocol, both oxygenation and blood volume decreased continuously with time. Further, these trends reached a full desaturation at the end of four to five minutes, suggesting a complete occlusion of blood flow to this muscle. During recovery both these variables increased very rapidly and exceeded their resting baseline values indicating a supersaturation in the biceps.

Incremental Arm Cranking

At the onset of the two-minute warm-up period, there was an immediate increase in oxygenation for a brief period in half-of the subjects followed by a systematic decrease in oxygenation. However, the rest of the subjects demonstrated a decrease in oxygenation at the onset of exercise. As the workload increased, oxygenation rapidly decreased (in all the subjects) and demonstrated a leveling off at approximately 80% of their work intensity (in most of the subjects). The blood volume trends showed a systematic increase with work rate with a leveling off as the peak oxygen uptake was attained.

During the recovery period from arm cranking, oxygenation and blood volume increased rapidly and showed three distinct patterns. In majority of the subjects, both oxygenation and blood volume trends increased beyond the baseline demonstrating hyperaemia for a brief period. By the end of the recovery period, only few subjects approached the baseline values for both oxygenation and blood volume values, and some did not reach the baseline.

Incremental Pushing-Pulling

During the warm-up phase of pushing-pulling, the oxygenation and blood volume trends were similar to those observed during arm cranking. However, in majority of the male subjects, unlike the oxygenation trends during arm cranking, the oxygenation trends decreased very rapidly with increasing workload. Only a few subjects demonstrated a leveling off pattern that was observed in the arm cranking protocol. However, at their maximum workload, there was a greater decrease in oxygenation than that observed in arm cranking. Blood volume trends increased significantly with increasing workload, with leveling off at the maximal workload. During the recovery from pushing-pulling, oxygenation and blood volume values increased rapidly and showed the three distinct trends observed during arm cranking.

Comparison of Biceps Brachil Oxygenation and Blood Volume Responses

Comparisons among Cuff Ischemia and Incremental Exercise Modes

No significant interaction between the three exercise protocols and gender was observed for the oxygenation and blood volume responses. It should be noted that comparison of the main effects was based on the pooled values of the remaining independent variable in the experimental design. Highest mean deoxygenation values were obtained during cuff ischemia (Table 5.3), however, they were significantly greater (by 62%) compared to the arm cranking protocol only (P<0.05). Interestingly, pushing-pulling demonstrated 42% more deoxygenation in the biceps than during arm cranking (P<0.01). Post-exercise mean reoxygenation values from arm cranking were the highest, but not significantly different from the other protocols (P>0.05). Further, maximal oxygenation obtained after arm cranking was greater (by 62%) than that obtained after pushing-pulling (P<0.05).

Based on the percent of cuff ischemia values (Table 5.3), significant differences between pushing-pulling and arm cranking were observed for minimum and maximum oxygenation values (P<0.05). Minimum oxygenation for arm cranking and pushing-pulling were 36% and 78% of that obtained during cuff ishemia, respectively. Further, maximum oxygenation during arm cranking and pushing-pulling were 183% and 86% of that obtained during cuff ischemia, respectively.

In the blood volume comparisons for both men and women, cuff ischemia showed the greatest minimum blood volume response, and was significantly greater than both arm cranking (by 74%) and pushing-pulling (by 88%) (P<0.05). However, the minimum blood volume values obtained at rest between the incremental protocols were similar. The maximum blood volume responses obtained after incremental exercise were highest during arm cranking, and were 44% greater than pushing-pulling (P<0.05), and 55% greater than cuff ischemia (P<0.05). When compared to arm cranking, deoxygenation cost, calculated as the ratio of minimum oxygenation and absolute oxygen uptake (OD. L⁻¹ min⁻¹), was significantly higher (by 34%) during pushing-pulling (P<0.05). Also,

correlations of NIRS measurements (Table 5.4) among the three protocols were not significant (P>0.05).

Comparison between Genders

Men showed higher deoxygenation values than women during the three protocols but none of the values were significantly different (P>0.05) (Table 5.3). All other responses were also not significantly different between genders (P>0.05). Further, gender did not influence the oxygenation and blood volume responses when expressed as a percentage of cuff ischemia (P>0.05). Deoxygenation ratio was also similar between genders (P>0.05).

Relationship between Whole-body Oxygen Uptake and Percent Change in Oxygenation

Based on the regression analysis and R^2 values, the relationship between absolute oxygen uptake and percent change in biceps brachii oxygenation at different percentages of peak aerobic capacity was best described by a linear trend. The regression analysis shown in the Figure 5.8, demonstrated that there were significant inverse relationships between percent change in biceps brachii oxygenation and percent of peak oxygen uptake during each exercise mode in both genders. In men, the R^2 values were 0.95 and 0.99 during pushing-pulling and arm cranking, respectively. In women, the R^2 values were 0.91 and 0.98 during pushing-pulling and arm cranking, respectively. ANOVA showed that there was no significant interaction between exercise mode and gender for these percent change in biceps brachii oxygenation values, suggesting that the rate of decline in biceps oxygenation for a given change in absolute oxygen uptake was similar between men and women during both pushing-pulling and arm cranking.

Correlations between Skinfold and Oxygenation and Blood Volume

Adipose tissue thickness calculated from biceps skinfolds for men and women (mean \pm SD) were: 2.8 \pm 1.5 mm and 4.0 \pm 1.8 mm, respectively. They were not significantly different from each other (P>0.05). Spearman correlations

between adipose tissue thickness and mean changes in oxygenation and blood volume data for both men and women during the three protocols are shown in Table 5.5. A significant correlation was found between adipose tissue thickness of men at the biceps and maximum oxygenation (r=-0.61, P<0.05) and maximum blood volume (r=0.70, P<0.05) during recovery from pushing-pulling. However, correlations were not significant (P>0.05) between adipose tissue thickness and other NIRS variables for both genders among the three protocols.

DISCUSSION

Comparison between Incremental Exercise Modes

Whole-body Physiological, Perceptual and Biomechanical Responses

The physiological responses observed in the present study during incremental arm cranking are comparable to those reported in previous investigations (Bar-Or and Zwiren 1975, Bhambhani et al. 1998, Davis et al. 1976, Price and Campbell 1997, Warren et al. 1990, Washburn and Seals 1984). Sawka (1986) in an extensive review, indicated that the physiological responses during upper body exercise were influenced by gender, populations studied, fitness levels, workload increments, type of exercise protocol, crank cadence, etc. In the current study, most of the physiological responses obtained during arm cranking were significantly greater than those observed during pushing-pulling, demonstrating that arm cranking is physiologically more strenuous on the upper body.

Some of the biomechanical factors that can account for these large physiological differences are: (1) pushing-pulling was performed in a different geometric plane compared to arm cranking; (2) to-and-fro horizontal motion was employed during pushing-pulling to generate maximum torque in the specified range of motion, whereas a circular motion away from the body was adopted during arm cranking; (3) length of the lever arm from the crankshaft and range of motion during both exercise modes were different; (4) differences in force exertion to generate the required power output - holding the pushing-pulling lever in the semi-supine position vs holding the arm crank handle in the prone position during arm cranking exercise; (5) asynchronous activity during arm cranking vs synchronous activity during pushing-pulling; and (6) biomechanical advantage of work performance in standing during pushing-pulling vs sitting during arm cranking.

In addition to the biomechanical differences between pushing-pulling and arm cranking explained above, one can argue that in the physiological responses differences might be due to the postural differences in the work performed during the two exercise modes. During arm cranking the position of the arms was closer to the level of the heart, whereas during pushing-pulling the arms were positioned on the lever below the heart level. The influence of arm position on the physiological responses during upper body work is controversial. Pendergast et al. (1979) demonstrated that the decrease in the peak oxygen uptake with the arms positioned above the heart level during exercise might be due to lower blood perfusion of the arm muscles. During occupational activities, Astrand et al. (1968) showed greater heart rate responses when subjects hammered nails into ceiling compared to across the wall (at the head level) or on the bench (closer to the waist level). However, Cummins and Gladden (1983) demonstrated no significant difference in physiological responses (oxygen uptake, heart rate, ventilation rate, cardiac output, stroke volume, and arterio-venous oxygen difference) during submaximal and maximal arm work with arms positioned above, at, and below the heart level.

Bhambhani et al. (1994) and Esmail et al. (1995) also showed no significant difference in physiological responses between pushing-pulling at the elbow height and repetitively pulling down an overhead attachment on the BTE work simulator. It is likely that the discrepancy amongst these studies might be due to the type of arm work investigated. Further, less isometric effort may be required in holding the crank handle during arm cranking compared to a much greater static and postural component during hammering above the overhead (Cummins and Gladden 1983). Based on these results, it is likely that arm

position during upright exercise has a lower impact on the physiological responses compared to the amount of isometric work involved in the exercise.

Research pertaining to the physiological responses during arm cranking while standing is limited. Boucahrd et al. (1979) and Intranont (1983) reported that the peak oxygen uptake of men during arm cranking in the standing position was significantly lower than the values attained during upright and supine cycling, treadmill walking/running, bench stepping and lifting at various heights. In the current study, the peak oxygen uptake for men during incremental arm cranking in the seated position was less than that reported by Boucherd et al. (1979) and Intranont et al (1983). Based on these cross sectional observations, one can argue that the peak oxygen uptake during arm cranking while standing is greater than that observed during arm cranking while sitting or pushing-pulling while standing.

In the current study, power output achieved during arm cranking was 79% greater than during pushing-pulling, with men generating 30% greater power output than women during both exercise modes. The biomechanical differences suggested above might have resulted in such a large difference in the power output between the exercise modes. When the results of the men and women were pooled, the peak oxygen uptake values during pushing-pulling were 32% lower compared to arm cranking. As a result, the oxygen cost per unit amount of power generated [ratio of absolute oxygen uptake and power output] was 72% higher (P<0.01) during pushing-pulling compared to arm cranking. In other words, pushing-pulling was metabolically less efficient than arm cranking.

The National Institute for Occupational Safety and Health (NIOSH 1981) proposed that in order to avoid undue fatigue, the time-weighted average oxygen uptake during an eight-hour work day should not exceed 33% of the individual's aerobic capacity obtained on a treadmill running test. Later, they modified the limit to 33% of repetitive lifting either below or above the waist (Waters et al. 1993). Based on this proposed limit and the data from Table 5.2, a healthy young male with an arm cranking peak oxygen uptake of 2.36 l.min⁻¹ should not exceed an average oxygen uptake of 0.78 l.min⁻¹ (33%) during task-specific pushing-

pulling. However, based on peak oxygen uptake results obtained from pushingpulling (1.74 I.min⁻¹), this limit equates to 44% of pushing-pulling, which exceeds the NIOSH standard by 11%. This difference may result in premature fatigue over an eight-hour work day and may increase the susceptibility to injury. A similar discrepancy in repetitive lifting and standardized aerobic protocols were also reported in previous investigations (Petrofsky and Lind 1978, Sharp et al. 1988).

Further, eight out of 22 subjects in the present study had similar peak oxygen uptake during arm cranking, however, their oxygen uptake values were different during identical pushing-pulling tasks. Sharp et al. (1988) suggested that factors such as skill-level, task-specific techniques, and physiological or morphological differences among individuals play an important role in work performance. Health care professionals such as ergonomists, exercise physiologists and occupational therapists who work in the area of Occupational Health and Safety should be aware of the limitations of using peak oxygen uptake values in setting limits for performance during an eight-hour work day.

Interestingly, in the current study, the rating of perceived exertion was the only variable that demonstrated significantly greater responses during pushing-pulling. Most of the subjects complained a much greater discomfort in the arms during pushing-pulling. However the present study did not investigate local rating of perceived exertion responses. According to the review of Sawka (1986): "perceptual sensitivity to process the physiological information may be enhanced during small muscle mass (upper body) exercise". In the current study, factors such as mode of grip during pushing-pulling, recruitment of muscles other than smaller muscles, and prolonged standing posture throughout the session might have resulted in greater perceptual responses during pushing-pulling. It should also be noted that the time to reach exhaustion by our subjects during maximal incremental arm cranking sessions (8 to 10 minutes). The longer duration of 'monotonous' work in the standing posture might have played a

significant role in the greater rating of perceived exertion responses reported during pushing-pulling.

Biceps Brachii Oxygenation and Blood Volume Responses During Incremental Exercise

In differentiating the importance of exercise mode selection, Sawka et al. (1983) commented: "protocol selection may be particularly important for upper body exercise where peak oxygen uptake is probably limited by peripheral factors (i.e., local fatigue and muscle perfusion) rather than central circulatory factors". Pushing-pulling and arm cranking involve predominantly upper body work (OSHA Draft 1999, Sawka 1986). In order to evaluate the peripheral responses during arm cranking and pushing-pulling, NIRS was used to evaluate the oxygenation and blood volume trends during these two exercise modes.

From Table 5.1, it is evident that peak oxygen uptake values during arm cranking were greater compared to pushing-pulling. Also, the correlation coefficients of the peak oxygen uptake between the two exercise modes for all the subjects was significant (r=0.67, P<0.05). However, the correlation between these two modes for minimum oxygenation was not significant (r=0.05, P>0.05), suggesting that peripheral responses during these two exercise modes were not similar. In this study, cardiac output measurements were not undertaken to evaluate the central contribution. However, the results of biceps brachii deoxygenation suggested that there was a significant difference in the peripheral contribution between these two exercise modes. This further strengthens our argument that development of task-specific aerobic protocols are essential at the workplace.

At the onset of Exercise

In both exercise modes during the rest-to-exercise period for 2 min, majority of the subjects demonstrated a rapid increase in oxygenation and blood volume for at least 40 to 60 seconds. At the onset of exercise, the muscle pump is activated with the first contraction and relaxation phase, leading to an

exponential increase (within 10 seconds of exercise) in the blood flow to the active muscles (Hughson et al. 1996). Few authors reported similar oxygenation trends during dynamic exercise (Belardinelli et al. 1995, Bhambhani et al. 2000, 1998). These authors suggested that such an increase in oxygenation might be due to an increase in blood flow to the active 'focal' region of the muscle, resulting in redistribution of intramuscular blood volume within the biceps. Furthermore, the role of increase in skin blood flow at the onset of exercise was also suggested for such an increase in oxygenation and blood volume trends (Maehara et al. 1997). However, Mancini et al. (1994) demonstrated that absorption range of NIRS signals were not influenced by the skin blood flow, and concluded that these signals essentially monitor the skeletal muscle and not skin oxygenation.

Few subjects demonstrated a decrease in oxygenation with a simultaneous increase in blood volume for a brief period at the onset of both exercise modes [and decrease in blood volume in three female subjects during arm cranking]. Bhambhani et al. (1998), Chance et al. (1992), and Jensen-Urstad et al. (1995) reported similar trends at the onset of exercise. To this effect, Bhambhani et al. (1998) hypothesized that zero and/or 25 Watts workload during an initial period of upper body exercise may be considerably high for few subjects to overcome, thus resulting in an increased deoxygenation. Accordingly, at the onset of submaximal arm cranking exercise, Jensen-Urstad et al. (1995) attributed a similar decrease in oxygenation to a transient imbalance between oxygen supply to the biceps brachii and oxygen uptake, thus resulting in higher anaerobic metabolism. Tamaki et al. (1994) suggested a similar hypothesis during the first contraction of repetitive biceps curls in body builders. These authors demonstrated a relative lack of oxygen supply (reflecting a decrease in oxygenation) and attributed this phenomenon to the unexpectedly high proportion of slow-twitch fibers in these athletes. Schmalbarch and Kaminecka (1974) showed that the biceps brachii is predominantly rich in slow twitch motor units.

It has been demonstrated that at the onset of exercise, cardiac output increases more rapidly than oxygen uptake (Davies et al. 1972, Yoshida and

Whipp 1994). Further, a rapid increase of muscle blood flow at the onset of exercise has also been reported (Bangsbo 2000, Walloe and Wesche 1988). To this effect, Hughson et al. (1996) demonstrated that oxygen uptake in the muscle depends on the supply of oxygen (aided by increase in blood flow) at the onset of exercise. These authors also showed that the increase in forearm blood flow kinetics was much slower during rhythmic squeezing of a lever above the heart level compared to the responses observed below the heart level.

However, similar to the findings of other authors (Boushel et al. 1998, Chance et al. 1992, Grassi et al. 1999, Tamaki et al. 1994), few subjects in the current study showed a decrease in blood volume at the onset of exercise as well. Chance et al. (1992) attributed such a decrease in blood volume at no load cycling to redistribution of blood in the lower limb. Further, during biceps curls Tamaki et al. (1994) reported that "venous blood in the muscle was squeezed out by the muscle milking action and venous return was enhanced after the first contraction, thus temporarily decreasing the muscle blood volume". Moreover, Boushel et al. (1998) attributed reduction in blood volume during the rest-toexercise transition to the muscle pump directing blood to the heart.

During the Incremental Effort

As the workload increased during arm cranking, biceps muscle oxygenation started decreasing steadily as indicated in Figures 1A and 2A. This trend is consistent with that reported by Bhambhani et al. (1998) in healthy men and women. Such a systematic decrease in oxygenation with increase in work intensity during both exercise modes implies a greater release of oxygen by hemoglobin/myoglobin via the Bohr effect (Belardinelli et al. 1995, Chance et al. 1992). However, the myoglobin dissociation curve does not exhibit the Bohr effect, hence its overall role in determining muscle deoxygenation is relatively minor (Mancini et al. 1994). To this effect, Stringer et al. (1994) and Wasserman et al. (1991) suggested that metabolic acidosis during exercise was essential for skeletal muscle deoxygenation to take place. In other words, the presence of hydrogen ions associated with lactate accumulation was necessary for oxygen to be released from oxygenated hemoglobin in the skeletal muscle via the Bohr effect. Therefore, a steady decrease in the venous oxygen saturation in the muscle with increase in workload might suggest an increase in the oxygen demand exceeding the muscle saturation, thus resulting in an increase in deoxygenation (reflecting increased oxygen extraction from the biceps).

In theory, the degree of deoxygenation measured by NIRS is attributed to the amount of oxygen released by both oxyhemoglobin and oxymyoglobin (Mancini et al. 1994a). However, the current NIRS technique is unable to differentiate between the amount of deoxygenation that is attributed to hemoglobin and myoglobin because the absorbency signals of hemoglobin and myoglobin overlap in this region (Chance et al. 1992). During knee extension exercise, Richardson et al. (1995) demonstrated that myoglobin desaturation occurred rapidly and reached a maximal value at approximately 50% of maximal oxygen consumption. Recently, Tran et al. (1999) showed that the decline in oxygenation obtained with the NIRS signal in the gastrocnemius muscle closely matched myoglobin desaturation measured from the nuclear magnetic resonance spectra.

During maximal exercise, greater mean deoxygenation values were observed during pushing-pulling compared to arm cranking in both genders. This was despite the fact that the peak oxygen uptake was significantly lower during pushing-pulling. In fact, when the ratio between the minimum deoxygenation and peak oxygen uptake was calculated (herein refereed to as the deoxygenation cost), the values were significantly higher during pushing-pulling, implying a greater peripheral limitation during this exercise mode.

In the current study, a cuff ischemia protocol (Maikala and Bhambhani 1999) was used to establish vascular occlusion and obtain a fully reduced state (minimum) of the biceps oxygenation and blood volume values. When the minimum oxygenation values of the incremental exercise protocols were expressed relative to the cuff ischemia value, pushing-pulling values were significantly higher compared to arm cranking (78% vs 36%). Chance et al. (1992) reported that the application of cuff ischemia to the vastus laterlais muscle

following maximal rowing exercise increased the maximum deoxygenation value by only 3%, implying the vastus lateralis muscle was nearly maximally deoxygenated at the point of exhaustion. The fact that the biceps brachii muscle was not maximally deoxygenated in comparison to cuff ischemia protocol during these two upper body exercise modes clearly indicate a peripheral limitation to performance during dynamic work in men and women. Interestingly, maximum oxygenation resulting from post-arm cranking was significantly higher than pushing-pulling but not with respect to cuff ischemia values.

During both exercise modes muscle blood volume increased systematically with increasing power output, with a leveling off as the peak oxygen uptake was attained. This trend suggests that during upper body exercise, muscle oxygen extraction is matched by the elevated blood flow, and therefore by an equivalent increase in the volume of arterialized blood (Boushel et al. 2000). This is in contrast with the observations of Grassi et al. (1999) who demonstrated that the blood volume in the vastus lateralis increased until approximately 60% to 65% of the maximal oxygen uptake during leg cycling, and remained constant until the peak oxygen uptake was attained.

It has been well established that peak oxygen uptake during leg exercise is limited by the rate of oxygen delivery to the active muscles, whereas during arm cranking it is limited by the reduced blood flow to the 'smaller' active muscle mass (Sawka 1986). However, a proportionate increase in blood volume with the increase in workload suggests that blood flow to biceps is greater during upper body exercise compared to legs during cycling exercise. To this effect, Sawka (1986) cited the observations of Clausen (1973) who found equal absolute muscle blood flow during both arm cranking and leg cycling at the same oxygen uptake. Sawka (1986) further stated that " ..since upper body exercise uses a smaller skeletal muscle mass". The continual increase in muscle blood volume in the biceps during incremental pushing-pulling and arm cranking exercise is consistent with these observations. Similar to the maximal oxygenation values reported, higher maximum blood volume responses were also reported for arm cranking compared to pushing-pulling. These findings indicate that the oxygen requirements of the exercising muscle after the termination of the test may be influenced by the exercise mode selected, and also may be dependent on the biomechanical and postural variations between these exercise modes described earlier.

Comparison between Genders:

Peak Cardiorespiratry Responses

Men demonstrated significantly greater (30%) peak power output during both exercise modes. As s result, the peak absolute oxygen uptake was also significantly higher in men (29% during pushing-pulling and 33% during arm cranking). This gender difference was maintained when the peak oxygen uptake was corrected for body mass. These observations during arm cranking are consistent with those of previous investigations (Bhambhani et al. 1998, Falkel et al. 1986, Washburn and Seals 1984). However, when the metabolic efficiency (oxygen uptake/power output) was calculated, no significant gender differences were observed for either pushing-pulling or arm cranking. This is consistent with the findings of Esmail et al. (1995) who reported no significant gender differences during submaximal simulated work during pushing-pulling on the BTE. Overall, these observations suggest that any gender differences in muscle fatigue are not related to the metabolic efficiency of the tasks performed. In the current study, the men and women did not reach their age-predicted maximal heart rate. During pushing-pulling the peak heart rate was approximately 88% in women and 90% in men of age-predicted maximal heart rate, while during arm cranking the values were approximately 90% in women and 95% in men. These values are consistent with the average value of 93% reported by Sawka (1986) in his comprehensive review of studies pertaining to upper body exercise.

Since heart rate was not influenced by either gender or exercise mode, greater gender differences in oxygen pulse in men during each exercise mode were due to the influence of peak oxygen uptake responses only. Since the peak oxygen uptake is closely related to the cardiac output, the lower peak oxygen uptake in the women was most likely due to their lower cardiac output (Wells 1991). During both exercise modes, men demonstrated significantly greater (37%) ventilation rates compared to women, with arm cranking values being significantly higher (23%) compared to pushing-pulling. Similar gender differences during arm cranking were reported by other investigators (Bhambhani et al. 1998, Washburn and Seals 1984). However, no significant gender differences were observed for the ventilatory equivalent for oxygen which is an index of the economics of ventilation. These observations suggest that the ventilatory stress during upper body exercise is similar in both genders, and corroborates previous findings (Bhambhani et al. 1998; Falker et al. 1986).

Biceps Brachii Oxygenation and Blood Volume Responses

It has been well established that women have a smaller upper body muscle mass (Hettinger 1961, Ikai and Fukunga 1968, Miller et al. 1993, Sharp 1994), and have lower strength for upper body muscle groups (Laubach 1976, Miller et al. 1993). In the current study, men showed no significant differences in the oxygenation and blood volume responses compared to women during both cuff ischemia and the two incremental exercise modes. Further, the deoxygenation cost was also similar between men and women during both exercise modes, suggesting similar oxygen extraction capacity of the biceps muscle in both genders. Shephard et al. (1988) demonstrated that women have less muscle volume in the arms than men, and suggested that

"female-to-male ratio of peak oxygen uptake tended to rise as the volume of active muscle was reduced".

Additionally, women have 10-14% less hemoglobin than men, resulting in a less oxygen carrying capacity (Wells 1991). However, it is not clear why gender did not influence oxygenation and blood volume responses in these two upper body exercise modes.

The relationship between percent change in oxygenation calculated from the Belardinelli equation (Belardinelli et al. 1995) and the absolute oxygen uptake at different percentages of peak aerobic power (20%, 40%, 60% and 80%) in both men and women are shown in the Figure 5.8. Significant inverse relationships were observed during the two exercise modes for both genders. The common variance between these two variables was high, ranging between 95% and 99% in men, and 91% and 98% in women, with values being slightly higher during arm cranking compared to pushing-pulling. No significant differences were observed between the slopes of these regression equations, suggesting that the rate of deoxygenation in the biceps muscle was similar in both sexes during two incremental exercise modes. Bhambhani et al. (1998) reported a similar finding during arm cranking exercise in men and women.

Role of Adipose Tissue Thickness on NIRS measurements

van Beekvelt et al. (2001) reported that adipose tissue metabolism is lower than muscle metabolism, hence muscle oxygenation determined using NIRS is underestimated. In the current study, there was no significant difference between men and women for the adipose tissue thickness at the biceps (Table 5.1). In the flexor digitorum super-ficialis muscle during rest and sustained isometric handgrip contractions, van Beekvelt et al. (2001) demonstrated a significant decrease in muscle oxygen consumption in both men and women with increase in the thickness of adipose tissue (r=-0.70, P<0.05). These investigators showed a significantly lower muscle oxygenation in women than in men during both rest and submaximal isometric contractions. In the current investigation, adipose tissue thickness in women was not significantly correlated to NIRS measurements [at a source-detector separation of 4 cm] during the three protocols (Table 5.4). However, a significant negative correlation was obtained in men for maximum oxygenation responses during recovery from pushing-pulling (r=-0.61, P<0.05). This may suggest that at the same source-detector separation of 4 cm, the greater the thickness of adipose tissue the lower the oxygenation values obtained during recovery from the incremental pushing-pulling activity. However, regression analysis between adipose tissue thickness and maximum oxygenation in men did not achieve statistical significance:

[=0.29-0.45*log(adipose tissue thickness), r=-0.61, P=0.18]

Further, this trend in men was not significant either after cuff ischemia or during recovery from arm cranking. These findings confirm that the correlations obtained in the current investigation occurred by chance.

In contrast to significant decrease in oxygenation with increase in adipose tissue thickness reported in van Beekvelt et al. (2001), the current investigation did not demonstrate a significant correlations with minimum oxygenation during three protocols. Additionally, van Beekvelt et al. (2001) found a significant but a weak correlation of 0.29 between forearm blood flow and adipose tissue thickness, suggesting an increase in blood flow with increase in the thickness of adipose tissue. This is in agreement with Coppack et al. (1990, 1999) who demonstrated a higher blood flow in adipose tissue compared with that in muscle. In the current investigation, a significant positive correlation (r=0.70, P<0.05) was also obtained between the adipose tissue thickness and maximal blood volume in men during recovery from pushing-pulling. This suggests that the greater the thickness of adipose tissue, the higher the blood volume (and blood flow during recovery). However, regression analysis between adipose tissue thickness and blood volume did not achieve statistical significance:

[=0.08+0.66* log(adipose tissue thickness), r=0.70, P=0.07] Also, this trend in men was not evident during the other exercise modes. These

findings further confirm that the correlations obtained in the current investigation occurred by chance. The influence of skin blood flow on the NIRS blood volume signal may arise as well. However, with a source-to-detector separation of 40-50 mm, Hampson and Piantadosi (1988) showed that NIRS measurements are not affected by skin blood flow, and attributed such findings to a small volume of skin relative to muscle between the NIRS optodes, and the low concentration of cyctochorme oxidase in skin. In the current investigation, the source to detector distance was set to 40 mm for each subject during the three protocols. Other authors also reported negligible influence of skin blood flow on NIRS signals (Mancini et al. 1994, Piantadosi et al. 1986).

A significant negative correlation for maximum oxygenation and positive correlation for maximum blood volume observed in the current investigation may be the result of chance alone. Women also participated in the current investigation, and if such a significant relationship truly exists for men, correlations should be reliable for women as well [especially when the adipose tissue thickness was similar between genders]. However, as evident from the Table 5.5, this was not the case in the current investigation. Furthermore, calculating common variance for the *r* value of -0.61 results in 37% common variance, suggesting that approximately one third of the variability in the maximum oxygenation during recovery from pushing-pulling can only be explained by variance in the adipose tissue thickness (or vice versa). Hence, the remaining 63% of the variance in maximum oxygenation during recovery from pushing-pulling might be influenced by other factors. Similarly, 49% of the variability in the maximum blood volume changes during recovery from pushing-pulling can be explained by variance in the adipose tissue thickness (or vice versa), and the remaining 51% of the variance might be attributed to prediction error.

It is well-established that a significant correlation between two variables does not imply causality (Cody and Smith 1987, Vincent 1999). Some of the data points in the Figures 5.9A and 5.9B are extreme, and that might have caused larger correlation coefficients as well. The discrepancy observed in the current investigation and the study of van Beekvelt et al. (2001) might be due to their larger sample size of 44 men and 34 women. Also, the exercise protocols used and quantification of NIRS signals were different between these two studies. Hence based on the current limited sample size (11 men and 11 women), the fact that adipose tissue thickness and maximum oxygenation and blood volume during recovery from pushing-pulling are related does not mean that a change in one is necessary to produce a corresponding change in other.

Further, Feng et al. (2001) showed that for an adipose thickness of 8 mm, the optimal source-detector distance was 45 mm. These authors have also identified the optimal distance of 35-40 mm for an adipose tissue thickness range of 2.5-5 mm, and 50 mm for a thickness of 17 mm. However, irrespective of the differences in the subjects' variation in the skinfolds the current investigation

used a fixed source-detector distance. This might have resulted in non-significant differences between men and women as well.

CONCLUSION

In evaluating a client's functional capacity, McConnell et al. (1984) suggested the maximal cardiorespiratory and hemodynamics responses are a function of exercise adopted, and do not represent the client's actual maximal consumption. These authors further stated that classification based on such a "not true" maximal aerobic capacity could result in assigning the clients' to either lower or higher fitness categories for the work demand. Sawka (1986) also stressed the importance of protocol selection for the upper body work. suggesting the peak aerobic capacity during this type of activity is limited by peripheral factors rather than the central circulatory factors. In the present study, in both men and women the task-specific pushing-pulling cardiorespiratory responses were lower than a standardized arm cranking test results. Such a current finding is similar to the pervious investigations (Khalil et al. 1985, Nindl et al. 1998, Petrofsky and Lind 1978, Sharp et al. 1988) that have compared standardized exercise protocols with the task-specific ones. The current investigation also demonstrated that pushing-pulling is metabolically less efficient than arm cranking.

The most interesting finding of the present study is the greater peripheral limitation during pushing-pulling compared to the corresponding arm cranking exercise responses, indicating upper extremity muscles are influenced by the exercise mode. Further, the rate of perceived exertion was also greater during pushing-pulling compared to arm cranking. Biomechanical and postural differences such as sitting vs standing, postural stability, mode of grip, the range of motion, position of hands below and above the heart level, and the forces being applied by the upper extremities are some of the reasons that can be attributed to the physiological differences between these two exercise modes. Also, adipose tissue thickness did not influence NIRS measurements.

REFERENCES

- 1. American College of Sports Medicine (1993) Resource manual for guidelines for exercise testing and prescription. Lea & Febiger, Philadelphia, USA.
- 2. Astrand I, Guharay A, and Wahren J (1968) Circulatory responses to arm exercise with different arm positions. *J Appl Physiol* 25: 528-532
- 3. Ayoub MM and McDaniel JW (1974) Effects of operator stance on pushing and pulling tasks. *AllE Transactions* 185-195
- 4. Bangsbo J (2000) Muscle oxygen uptake in humans at the onset of and during intense exercise. Acta Physiol Scand (168): 457-464
- 5. Bar-Or O and Zwiren LD (1975) Maximal oxygen consumption test during arm exercise reliability and validity. *J Appl Physiol* 38(3): 424-426
- Beaton DE, O'Driscoll SW, and Richards RR (1995) Grip strength testing using the BTE work simulator and the Jamar dynamometer: a comparative study. BTE. J Hand Surgery 20(2): 293-298
- Belardinelli R, Barstow TJ, Porszasz J, and Wasserman K (1995) Change in skeletal muscle oxygenation during incremental exercise measured with NIRS. *Eur J Appl Physiol* 70: 487-492
- Bernanrd B and Fine LJ (1997) Musculoskeletal disorders and workplace factors.
 (ed) Beranrd B, US Dept. of Health and Human Services, Public Health Services, CDC, DHHS (NIOSH) Publication # 97-141, Cincinnati, OHIO, USA.
- 9. Bhambhani Y, Esmail S, and Brintnell S (1994) The Baltimore therapeutic equipment work simulator: biomechanical and physiologic norms for three attachments in healthy men. *Am J Occ Ther* 48: 19-25
- Bhambhani Y, Maikala R, and Buckley S (1998) Muscle oxygenation during incremental arm and leg exercise in men and women. *Eur J Appl Physiol* 78: 422-431
- 11. Bhambhani Y, Maikala R, and Esmail S (2001) Oxygenation trends in vastus lateralis muscle during incremental and intense anaerobic cycle exercise in young men and women. *Eur J Appl Physiol* 84 : 547-556
- 12. Borg G (1982) Psychophysical bases of perceived exertion. Med Sci Sports Exerc 29: 402-409
- 13. Bouchard C, Godbout P, Mondor JC, and Leblanc C (1979) Specificity of maximal aerobic power. Eur J Appl Physiol 40: 85-93
- 14. Boushel R, Pott F, Madsen P, Radegran G, Nowak M, Quistroff B, and Secher N (1998) Musice metabolism from near infrared spectroscopy during rhythmic handgrip in humans. *Eur J Appl Physiol* 79: 41-48
- 15. Boushel R, Landberg H, Green S, Skovgaard D, Bulow J, and Kjaer M (2000) Blood flow and oxygenation in peritendinous tissue and calf muscle during dynamic exercise in humans. *J Physiol* 524(1): 305-313
- 16. BTE Operators Manual (1992). Baltimore Theraupeutic Equipment, Maryland.
- 17. Canadian Society for Exercise Physiology (1994) Revised Physical Activity Readiness Questionnaire – RPARQ. Gloucester, NB, CANADA.
- 18. Chaffin D, Andres RO, and Garg A (1983) Volitional postures during maximal push/pull exertions in the sagittal plane. *Human Factors* 25(5): 541-550
- 19. Chance B, Dait MT, Zhang C, Hamaoka T, and Hagerman F (1992) Recovery from exercise-induced desaturation in the quadriceps muscles of elite competitive rowers. *Am J Phyisol* 262: C766-C775
- 20. Clausen JP (1973) Muscle blood flow during exercise and its significance for maximal performance. In Keul J (ed.) *Limiting factors of physical performance*. Stuttgart: Thieme, pp. 253-266.
- 21. Cody RP and Smith JK (1987) Applied Statistics and the SAS programming language. Elsevier Scinece Publishers, New York, NY, pp: 70-85
- 22. Coppack SW, Fisher RM, Gibbons GF, et al. (1990) Postprandial substrate deposition in human forarm and adipose tissues in vivo. *Clin Sci* 79: 229-248
- 23. Coppack SW, Persson M, Judd RL, and Miles JM (1999) Glycerol and nonesterified fatty acid metabolism in human muscle and adipose tissue in vivo. *Am J Physiol* 276: E233-E240
- 24. Costes F, Barthelemy JC, Feasson L, Busso T, Geyssant A, and Denis C (1996) Comparison of muscle near-infrared spectroscopy and femoral blood gases during steady-state exercise in humans. *J Appl Physiol* 80(4): 1345-1350
- 25. Cummins TD and Gladden LB (1983) Responses to submaximal and maximal arm cycling above, at, and below heart level. *Med Sci Sports Exerc* 15: 295-298
- 26. Davies CTM, DiPrampero PE, and Gerretelli P (1972) Kinetics of cardiac output and respiratory gas exchange during exercise and recovery. *J Appl Physiol* 32: 618-625

- 27. Davis JA, Vodak P, Wilmore JH, Vodak J, and Kurtz P (1976) Anaerobic threshold and maximal aerobic power for three modes of exercises. *J Appl Physiol* 41: 544-550
- 28. Esmail S, Bhambhani Y, and Brintnell S (1995) Gender differences in work performance on the Baltimore Therapeutic Equipment work simulator. *Am J Occu Therapy* 49: 405-411
- 29. Falkel JE, Sawka MN, Levine L, Pimental NA, and Pandolf KB (1986) Upperbody exercise performance: comparison between women and men. *Ergonomics* 29: 145-154
- 30. Feng W, Haishu D, Fenghua T, Jun Z, Qing X, and Xianwu T (2001) Influence of overlying tissue and probe geometry on the sensitivity of a near-infrared tissue oximeter. *Physiol Meas* 22: 201-208
- 31. Franklin BA, Hogan P, Bonzheim K, Bakalyar D, Terrien E, Gordon S, and Timmis GC (1995) Cardiac demands of heavy snow shoveling. JAMA 273(11): 880-882
- 32. Gagnon M, Beaugrand S, and Authier M (1992) The dynamics of pushing loads onto shelves of different heights. *Int J Ind Ergs* 9: 1-13
- 33. Garg A, Chaffin DB, and Herrin GD (1978) Prediction of metabolic rates for manual material handling jobs. *Am Ind Hyg Assoc J* 39(8): 661-674
- 34. Garg A, Rodgers SH, and Yates JW (1992) The physiological basis for manual lifting. In: Kumar S (eds). Advances in Industrial Ergonomics and Safety IV, Taylor & Francis.
- 35. Grassi BM, Quaresima V, Marconi C, Ferrari M, and Cerretelli P (1999) Blood lactate accumulation and muscle deoxygenation during incremental exercise. *J Appl Physiol* 87: 348-355
- 36. Hampson NB and Piantadosi CA (1988) Near infrared monitoring of human skeletal muscle oxygenation during forearm ischemia. J Appl Physiol 64: 2449-2457
- 37. Hettinger T (1961) Physiology of strength. Charles Thomas, Springfield, Illionois, pp 8-12
- 38. Homma S, Fukunaga T, and Kagaya A (1996) Influence of adipose tissue thickness on near infrared spectroscopic signal in the measurement of human muscle. *J Biomed Opt* 1: 418-424

- 39. Hoozemans MJM, van Der Beek AJ, Frings-dressen MHW, van Dijk JH, and Van Der Wode (1998) Pushing and pulling in relation to musculoskeletal disorders: a review of risk factors. *Ergonomics* 41: 757-781
- 40. Hughson RL, Shoemaker JK, Tschakovsky ME, and Kowalchuk JM (1996) Dependence of muscle VO2 on blood flow dynamics at onset of forearm exercise. J Appl Physiol 81(4): 1619-1626
- 41. Intaranont K (1983) Evaluation of anaerobic threshold for lifting tasks. An unpublished PhD Dissertation, Texas Tech University, Lubbock, Texas.
- 42. Ikai M and Fukunga T (1968) Calculation of muscle strength per unit crosssectional area of human muscle by means of ultrasonic measurements. Int Zeitschrift Ang Phys 26: 26-32
- Jensen-Urstad M, Hallback I, and Sahlin K (1995) Effect of hypoxia on muscle oxygenation and metabolism during arm exercise in humans. *Clin Physiol* 15: 27-37
- 44. Kennedy L and Bhambhani Y (1991) The BTE work simulator: reliability and validity at three work intensities. Arch Phy Med Rehab 72(7): 511-516
- 45. Khalil TM, Genaidy AM, Asfour SS, and Vinciguerra T (1985) Physiological limits in lifting. Am Ind Hyg Assoc J 46: 220-224
- 46. Laubach LL (1976) Comparative muscular strngth of men and women: a review of the literature. Aviat Space Envi Med 47: 534-542
- 47. Lee KS, Chaffin DB, Herrin GD, and Waikar, AM (1991) Effect of handle height on lower-back loading in cart pushing and pulling. *Applied Ergonomics* 22(2): 117-123
- 48. McArdle WD, Katch I, and Katch VL (1996) Exercise Physiology, Williams & Wilkins, Baltimore, pp 221
- 49. McConnell TR, Swett DD, Jeresaty RM, Missri JC, and Al-Hani AJ (1984) The hemodynamic and physiologic differences between exercise modalities. *J Sports Med* 24: 238-245
- 50. Maehara K, Riley M, Galassetti P, Barstow TJ, and Wasserman K (1997) Effects of hypoxia and carbon monoxide on muscle oxygenation during exercise. *Am J Resp Crit Care Med* 155: 229-235
- 51. Maikala R and Bhambhani Y (1999) Biceps brachii deoxygenation during cuff ischemia and recovery: Intra and inter session reliability. *Med Sci Sport Exer* S245 31:5 (Abstract)

- 52. Mancini, DM, Bolinger L, Li H, Kendrick K, Chance B, and Wilson JR (1994) Validation of near-infrared Spectroscopy in Humans. *J Appl Physiol* (77): 2740-2747
- 53. MicroRunMan User Manual (1998), NIM Inc., PA, USA.
- 54. Miller AEJ, MacDougall JD, and Tarnoplosky MA (1993) Gender differences in strength and muscle fiber characteristics. *Eur J Appl Physiol* 66: 254-262
- 55. Murthy G, Kahan NJ, Hargen AR, and Rempel DM (1997) Forarm muscle oxygenation decreases with low levels of voluntary contraction. *J Orthop Res* 15: 507-511
- 56. Nindl BC, Sharp MA, Mello RP, Rice VJ, Murphy MM, and Patton JF (1998) Gender comparison of peak oxygen uptake: repetitive box lifting versus treadmill running. *Eur J Appl Physiol* 77: 112-117
- 57. OSHA Draft (1999) OSHA CFR Part 1910 Ergonomics program: proposed rule. Department of Labor, Occupational Safety and Health Administration Federal Register, 64 (225): 65768-66078
- 58. Pendegast DP, Cerretelli P, and Rennie DW (1979) Aerobic and glycolytic metabolism in arm exercise. *J Appl Physiol* 47: 754-760
- 59. Petrofsky JS and Lind AR (1978) Comparison of metabolic and ventilatory responses of men to various lifting tasks and bicycle geometry. *J Appl Physiol* 45: 60-63
- 60. Piantadosi CA, Hemstreet TA, and Jobsis-VanderVliet FF (1986) Near-infrared spectrophotometric monitoring of oxygen distribution to intact brain and skeletal muscle tissues. *Crit Care Med* 14: 698-706
- 61. Powell DM, Christopher A, Zimmer BA, Antoine MM, Baruch LD, Bellian KT, Morgan RF, and Edlich RF (1991) Computer analysis of the performance of the BTE work simulator. J Burn Care Rehabil 12: 250-256
- 62. Price MJ and Campbell IG (1997) Determination of peak oxygen uptake during upper body exercise. *Ergonomics* 40: 491-499
- 63. Richardson RS, Noyszewski EA, Kendrick KF, Leigh JS, and Wagner PD (1995) Myoglobin O_2 desaturation during exercise: evidence of limited O_2 transport. J Clin Invest 96: 1916-1926
- 64. Sanchez J, Monod H, and Chabaud F (1979) Effects of Dynamic, Static and Combined work on Heart Rate and Oxygen Consumption. *Ergonomics* 8: 935-943

- 65. Sawka MN, Foley ME, Pimental NA, and Pandolf KB (1983) Physiological factors affecting upper body aerobic exercise. *Ergonomics* 26: 639-646
- 66. Sawka MN (1986) Physiology of upper body exercise. In: Pandolf KB (ed) Exercise and sport sciences reviews, vol 14. MacMillan, New York, pp 175-211
- 67. Schmalbruch H and Kaminiecka Z (1974) Fibre types in human brachial biceps muscle. *Exp Neurol* 44: 313-328
- 68. Sharp M (1994) Physical fitness and occupation performance of women in the US Army. Work 4(2): 80-92
- 69. Sharp MA, Harman E, Vogel JA, Knapik JJ, and Legg SJ (1988) Maximal aerobic capacity for repetitive lifting: Comparison with three standard exercise testing modes. *Eur J Appl Physiol* 57:753-760
- 70. Shephard RJ, Bouhlel E, Vanderwalle H, and Monod H (1988) Muscle mass as a factor limiting physical work. *J Appl Physiol* 64: 1472-1479
- 71. Snook SH and Ciriello VM (1991) The Design of Manual Handling Tasks: Revised Tables of Maximum Acceptable Weights and Forces. *Ergonomics* 34(9): 1197-1213
- 72. Sogarrd K, Fallentin N, and Nielsen J (1996) Work load during floor cleaning. The effects of cleaning methods and work technique. *Eur J Appl Physiol* 73: 73-81
- 73. Stringer WW, Wasserman K, Casaburi R, Porszasz J, Maehara K, and French W (1994) Lactic acidosis as a facilitator of oxyhemoglobin during ewxercise. *J Appl Physiol* 76: 1462-1467
- 74. Tamaki T, Uchiyama S, Tamura T, and Nakano S (1994) Change in muscle oxygenation during weight-lifting exercise. *Eur J Appl Physiol* 68: 465-469
- 75. Tran TK, Saliasuta N, Kreutzer U, Hurd R, Chung Y, Mole P, Kuno S, and Jue T (1999) Comparative analysis of NMR and NIRS measurements of intracellular PO2 in human skeletal muscle. *Am J Physiol* 276 (45): R1682-R1690
- 76. van Beekvelt MCP, Borghuis MS, van Engelen BGM, Wevers RA, and Colier WNJM (2001) Adipose tissue thickenss affects in vivo quantitative near-IR spectroscopy in human skeletal muscle. *Clin Sci* 101: 21-28
- 77. van der Beek AJ, Klurer BD, Frings-Dresen MH, and Hoozemans MJ (2000) Gender differences in exerted forces and physiological load during pushingpulling of wheeled cages by postal workers. *Ergonomics* 43: 269-281
- 78. Vincent WJ (1999) Statisitcs in Kinesiology. Human Kinetics, second edition, Champaign, IL., pp: 87-99

- 79. Walloe L and Wesche J (1988) Time course and magnitude of blood flow changes in the human quadriceps muscles during and following rhythmic exercise. J Physiol Lond 405: 257-274
- 80. Warren GL, Cureton KJ, Dengel DR, Graham RE, and Ray CA (1990) Is the gender difference in peak VO₂ greater for arm than leg exercise? *Eur J Appl Physiol* 60: 149-154
- Washburn RA and Seals DR (1984) Peak oxygen uptake during arm cranking for men and women. J Appl Physiol 56: 954-957
- 82. Wasserman K, Hansen JE, and Sue DI (1991) Facilitation of oxygen consumption by lactic acidosis during exercise. *NIPS* 6: 29-34
- 83. Waters TR, Putz-Anderson V, Garg A, and Fine LJ (1993) Revised NIOSH equation for the design and evaluation of manual lifting tasks. *Ergonomics* 36:749-776
- 84. Wells CL (1991) Women, sport and performance: A physiological perspective (2nd Edition). Champaign, IL: Human Kinetics.
- 85. Yamamoto K, Niwayama M, Lin L, Shiga T, Kudo N, and Takahashi M (1998) Accurate NIRS measurement of muscle oxygenation by correcting the influence of a subcutaneous layer. *Proc SPIE Int Soc Opt Eng* 3194: 166-173
- 86. Yoshida T and Whipp BJ (1994) Dynamic asymmetries of cardiac output transients in response to muscular exercise in man. *J Physiol Lond* 480: 355-359

Gender	Age	Body Mass	Height	Body Mass Index	Biceps Skinfold Thickness	PUSH-PULL HEIGHT
	Yrs	Kgs	m	Kg.m ⁻²	mm	cm
Men N=11	23.7 ± 4.5	77.5 ± 13.8 ^ª	1.76 ± 0.08 •	24.6 ± 3.3 °	5.6 ± 3.0	119±7*
Women N=11	24.5 ± 4.0	57.6 ± 9.4	1.63 ± 0.06	21.4 ± 2.4	8.0 ± 3.6	111±6

Table 5.1: Demographics of subjects (Mean \pm SD)

^a – significant difference between men and women, p<0.05

Table 5.2: Peak Physiological, Perceptual and Biomechanical Responses during Incremental Pushing-Pulling and Incremental Arm Cranking in Healthy Men and Women (Mean \pm SD)

VARIABLE	GENDER	ACTIVITY			
		PUSH-PULL	ARM CRANKING		
Power output (W)	Men	32 ± 8 ^{acd}	161 ± 32 °		
	Women	25 ± 8 *	114 ± 20		
Oxygen Uptake, L.min ⁻¹	Men	1.74 ± 0.51^{ac}	2.36 ± 0.51 °		
	Women	1.23 ± 0.32*	1.56 ± 0.50		
Oxygen Uptake, mL.Kg ⁻¹ .min ⁻¹	Men	22.7 ± 6.1 •	30.3 ± 3.8		
	Women	20.9 ± 6.4 *	27.3 ± 8.5		
Oxygen Uptake/Peak Power,	Men	55.2 ± 11.0*	14.8 ± 3.3		
(mL.min ⁻¹ .Watt ⁻¹)	Women	52.3 ± 12.0*	14.2 ± 4.6		
Oxygen Uptake/Peak Power,	Men	0.76 ± 0.32 *	0.19 ± 0.04		
(mL.Kg ⁻¹ .min ⁻¹ .Watt ⁻¹)	Women	0.84 ± 0.25*	0.25 ± 0.08		
Heart Rate, beats.min ⁻¹	Men	174 ± 16	182 ± 15		
	Women	171 ± 14	176 ± 16		
Oxygen Pulse, mL.beat ⁻¹	Men	10.11 ± 3.1 **	13.12 ± 3.1 °		
	Women	7.10 ± 1.7*	8.9 ± 3.0		
Ventilation Rate, L.min ⁻¹	Men	90.9 ± 28.0 **	124.6 ± 31.6 °		
(BTPS)	Women	62.4 ± 21.3*	73.6 ± 19.3		
Ventilatory Equivalent	Men	54.2 ± 14.3	53.8 ± 10.7		
for Oxygen	Women	51.3 ± 13.6	49.2 ± 14.0		
Tidal Volume, L.breath ¹	Men	1.9 ± 0.4 °	2.1 ± 0.6 °		
	Women	1.2 ± 0.3	1.3 ± 0.3		
Respiratory Frequency	Men	48 ± 13 ^b	58 ± 6		
breaths.min ⁻¹	Women	49±11 ^b	57 ± 14		
Respiratory Exchange Ratio	Men	1.07 ± 0.07 b	1.15 ± 0.04		
	Women	1.05 ± 0.08 ^b	1.10 ± 0.07		
Rating of Perceived Exertion	Men	18.8 ± 1.0 •	18.2 ± 1.5		
	Women	18.7 ± 1.2 ^b	17.5 ± 1.6		

*-significant difference between arm cranking and pushing-pulling (p < 0.01)

^b-significant difference between arm cranking and pushing-pulling (p <0.05)

^c-significant difference between men and women (p < 0.01)

^a – significant interaction between gender and exercise [arm cranking and pushing-pulling] (p < 0.05)

GENDER	TASK	Oxygenation			Blood Volume		
		Min	Max	Range	Min	Max	Range
Men	Push-Puil	-0.212 ± 0.170°	0.115 ± 0.221 °	0.330 ± 0.300 ^d	-0.005 ± 0.290 ^b	0.339 ± 0.249 °	0.340 ± 0.240 tc
	Arm Cranking	-0.121 ± 0.181 *	0.235 ± 0.251	0.356 ± 0.145 d	0.112 ± 0.344*	0.678 ± 0.221*	0.566 ± 0.382
	Cuff Ischemia	-0.270 ± 0.172	0.168 ± 0.214	0.438 ± 0.262 °	-0.317 ± 0.223	0.314 ± 0.086	0.631 ± 0.203
	Push-Pull - %of Cuff Ischemia	78 ± 98 °	$50\pm64^{\circ}$	75±114	2 ± 130	107 ± 290 °	60 ± 118 °
	Arm Cranking - %of Cuff Ischemia	45 ± 105	140 ± 117	81 ± 55	35 ± 154	215 ± 256	89 ± 188
	Deoxygenation Cost [®] [Push-Pull], OD.L ⁻¹ .min ⁻¹	-12 ± 33 °					
	Deoxygenation Cost [®] (Arm Cranking), OD.L ⁻¹ .min ⁻¹	-5 ± 36					
Women	Push-Pull	-0.150 ± 0.132 °	0.075 ± 0.080 °	0.220 ± 0.140	0.061 ± 0.126 •	0.335 ± 0.229 °	0.270 ± 0.160 bc
	Arm Cranking	-0.050 ± 0.100*	0.205 ± 0.127	0.255 ± 0.103	0.047 ± 0.161 •	0.523 ± 0.231 *	0.476 ± 0.236
	Cuff Ischemia	-0.193 ± 0.173	0.090 ± 0.038	0.284 ± 0.175	-0.274 ± 0.173	0.232 ± 0.104	0.506 ± 0.249
	Push-Pull - %of Cuff Ischemia	78 ± 76 °	104 ± 129 °	78 ± 80	22 ± 73	144 ± 220 °	58 ± 64 °
	Arm Cranking - %of Cuff Ischemia	26 ± 58	227 ± 334	90 ± 60	17 ± 93	225 ± 222	94 ± 94
	Deoxygenation Cost ^e [Push-Pull], OD.L ⁻¹ .min ⁻¹	-12 ± 41 °					
	Deoxygenation Cost [®] [Arm Cranking], OD.L ⁻¹ .min ⁻¹	-3 ± 20					

 Table 5.3: Biceps Brachil Oxygenation and Blood Volume responses during Incremental Pushing-Pulling and Incremental Arm Cranking in Healthy

 Men and Women in Optical Density (Mean ± SD)

Range is defined as the difference between Maximum Optical Density (Max) and Minimum Optical Density (Min). ^a-significant difference between arm cranking and cuff ischemia (p < 0.05); ^b-significant difference between pushing-pulling and cuff ischemia (p < 0.05); ^c-significant difference between arm cranking and pushing-pulling (p < 0.05); ^d-significant difference between men and women (p < 0.01) ^c- Deoxygenation Cost is defined as the ratio of muscle deoxygenation and whole-body oxygen uptake during specific incremental exercise.

Table 5.4: Correlations of Biceps Brachii Oxygenation and Blood Volume Responses among Three Protocols (n=22)

Variable	Protocols					
Oxygenation	Cuff Ischemia Vs Pushing-pulling	Cuff Ischemia Vs Arm Cranking	Pushing-pulling Vs Arm Cranking			
Minimum	0.23	0.32	0.05			
Maximum	0.23	-0.05	0.02			
Range	0.18	0.23	-0.20			
Blood Volume	Cuff Ischemia Vs Pushing-pulling	Cuff Ischemia Vs Arm Cranking	Pushing-pulling Vs Arm Cranking			
Minimum	-0.28	0.41	0.12			
Maximum	0.12	0.32	0.45*			
Range	0.03	0.42	0.05			

^a-significant difference at P<0.05

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Table 5.5: Spearman Correlation Coefficients between Adipose Tissue Thickness and Biceps Brachii Oxygenation and Blood Volume responses in Healthy Men and Women

GENDER TASK		Oxyge	onation	Blood Volume	
		Min	Max	Min	Max
Men	Push-Pull	0.04	-0.61 ^a	0.60	0.70 ^a
	Arm Cranking	-0.10	-0.40	0.20	0.34
	Cuff Ischemia	-0.56	-0.21	-0.55	-0.25
Women	Push-Pull	0.01	-0.08	-0.19	-0.18
	Arm Cranking	0.08	-0.06	0.19	0.57
	Cuff Ischemia	-0.11	-0.18	0.45	-0.57

^a - significant difference at P<0.05



Figure 5.1 BTE Work Simulator for Repetitive Pushing-Pulling



Figure 5.2 NIRS Sensor Placement on the Biceps Brachii



Figure 5.3A Subject Setup for Cuff Ischemia



Figure 5.3B Subject Setup for Arm Cranking



Figure 5.4 Subject Stance during Pushing-Pulling



Figure 5.5A Subject during Pulling Phase



Figure 5.5B Subject during Pushing Phase



Figure 5.6A: Biceps Oxygenation Trends in a Typical Female



Figure 5.6B: Biceps Blood Volume Trends in a Typical Female

Min - Minimum Oxygenation (or Blood Volume) value obtained during specific Protocol Max - Maximum Oxygenation (or Blood Volume) value obtained during recovery from specific Protocol

Biceps brachii oxygenation (top panel) and blood volume (bottom panel) trends in a typical female subject during cuff ischemia, arm cranking and pushing-pulling, and recovery



Figure 5.7A: Biceps Oxygenation Trends in a Typical Male



Min - Minimum Oxygenation (or Blood Volume) value obtained during specific Protocol Max - Maximum Oxygenation (or Blood Volume) value obtained during recovery from specific Protocol

Biceps brachii oxygenation (top panel) and blood volume (bottom panel) trends in a typical male subject during cuff ischemia, arm cranking and pushing-pulling and recovery





247







Figure 5.9B Correlaion between Adipose Tissue Thickness and Maximum Blood Volume during Recovery (Pushing-Pulling)

CHAPTER 6

Lumbar Erector Spinae Oxygenation and Blood Volume Comparisons during Arm Cranking and Task-specific Pushing-Pulling

Pushing and pulling is a manual material handling activity, and is defined as the application of one or two handed force directed away from the body [pushing] and directed towards the body [pulling] on an object or person, with feet either stationary (Chaffin et al. 1983, Gagnon et al. 1992, Garg et al. 1978) or moving (Lee et al. 1991, Snook and Ciriello 1991). Pushing and pulling of loads from the shelves (Gagnon et al. 1992), floor mopping or shovelling of snow (Franklin et al. 1995, Søgaard et al. 1996), and cart pushing and pulling (Snook and Ciriello 1991) are some of the activities that involve substantial amount of muscle activity in the trunk and upper extremities of workers (Bernard and Fine 1997).

The whole human body is involved in repetitive pushing and pulling of loads. In addition to the upper extremities, a significant amount of effort is required from the trunk musculature in stabilizing the body. Hence, it is important to understand the local physiological responses in the trunk muscles during pushing and pulling activities at the workplace. In a critical review by the National Institute for Occupational Safety and Health (NIOSH), Bernard and Fine (1997) reported the evidence for low-back related discomfort, pain, and injuries at the workplace during activities such as lifting, pushing and pulling, carrying, whole-body vibration, etc. In the province of Alberta, CANADA during 1995 and 1996, twenty nine-percent of total compensation claims were related to the lower back resulting mostly from overexertion injuries due to lifting, pushing and pulling (Alberta Labour 1997). Further, in a warehouse distribution study in the province of Quebec, CANADA, Lortie and Baril-Gingras (1998) reported pushing and pulling accounted for 69.9% and 46.4% of material handling, respectively.

Several studies have documented significant differences in aerobic power between standardized exercise and task-specific protocols (Khalil et al. 1985, Petrofsky and Lind 1978, Nindl et al. 1998, Sharp et al. 1988). However, to the best of our knowledge, comparison of local oxygenation and blood volume responses during maximal pushing and/or pulling to these standardized protocols has not been reported to date. Investigations related to gender responses during pushing and pulling are also very limited. Esmail et al. (1995) demonstrated that torque, work, and power output generated by men are significantly greater compared to women. However, when these values were corrected for their body mass, gender didn't influence the results. In another investigation, Ayoub and McDaniel (1974) reported that men had higher static push and pull horizontal forces in the sagittal plane compared to women.

Near infra-red spectroscopy (NIRS) is a non-invasive optical technique that has been used to measure relative changes in oxygenation and blood volume levels in skeletal muscle continuously in real-time during rest and work (Bellardinelli et al. 1995, Bhambhani et al. 1998a, Boushel et al. 1998, Chance et al. 1992, Mancini et al. 1994). Muscle oxygenation, defined as the relative saturation of oxyhemoglobin and oxymyoglobin, depends on the balance between oxygen delivery and tissue oxygen utilization (Mancini et al. 1994). Blood volume is defined as the sum of oxygenated and deoxygenated states of hemoglobin and myoglobin in the illuminated region of the muscle. Detailed description of the NIRS theory, its validation with various established optical and magnetic techniques are provided in Chapter 1. The reliability of NIRS in the upper and lower extremity human skeletal muscles has been reported in Chapter 1 as well. In establishing the reliability of NIRS in the erector spine muscle during maximal back extension, Maikala et al. (2000) reported significant correlation coefficients of 0.83 and 0.84 for minimum oxygenation during sitting and standing respectively, and 0.99 for the minimum blood volume during both postures. To date, erector spinae oxygenation and blood volume changes using NIRS have been limited to sub-maximal and maximal isometric contractions (Jensen et al. 1999, McGill et al. 2000, Maikala et al. 2000, Maikala and Bhambhani 1999, Maronitis et al. 2000, Yoshitake et al. 2001).

To the best of our knowledge, no studies to date have evaluated the oxygenation and blood volume trends of lumbar erector spinae muscle during

repetitive pushing and pulling, and compared these responses with a standardized incremental arm cranking protocol. Hence, the purposes of this study were to compare the: (1) lumbar erector spinae oxygenation and blood volume differences curing incremental pushing-pulling (with feet stationery) and incremental arm cranking till volitional exhaustion; and (2) differences in these responses between genders. It was hypothesized that: (1) the lumbar erector spinae oxygenation and blood volume trends would be similar between the two modes of exercise, and (2) there would be no significant gender differences in these responses.

MATERIALS AND METHODS

Subjects

Written informed consent was obtained from 11 healthy men (age 23.7 ± 4.5 yrs, mass 77.5 ± 13.8 kg, height 176 ± 8 cm) and 11 healthy women (age 24.5 ± 3.5 yrs, mass 57.6 ± 9.4 kg, height 163 ± 7 cm). They all completed the Revised Physical Activity Readiness Questionnaire (Canadian Society for Exercise Physiology, 1994) to identify contraindications for exercise. The test protocols utilized were approved by the Human Research Ethics Board at this institution (APPENDIX A.2).

Description of the Baltimore Therapeutic Equipment (BTE) Work Simulator:

The BTE work simulator (Baltimore Therapeutic Equipment Co., Maryland, USA) is a multi-purpose instrument that can be used for assessing upperextremity strength and endurance during functional tasks (Figure 5.1). This simulator consists of three components: a) exercise head; b) task-specific attachments; and c) computer software to convert the mechanical resistance to respective functional units. For pushing-pulling without feet moving, attachment # 171 is used. This attachment is a lever consisting of a straight handle with an articulating joint that result in linear motion when connected to the exercise head. Mechanical resistance in the linear motion can be controlled manually in both push and pull directions. Force output from the head is measured using an inbuilt strain gauge. The torque generated is calculated from the force output and the distance traveled by the lever, and displayed on the console. Before each testing session, the BTE was balanced in such a way that the strain gauge in the exercise head was synchronized with the electronic components in the control console (Powell et al. 1991), and calibrated as per the manufacturer's instructions (BTE Operator's Manual, 1992). Greater details of the BTE and its reliability in measuring various isometric and isokinetic activities were reported in Chapter 5.

Test Protocols

Each subject completed a back muscle endurance test on the first day, and incremental arm cranking exercise test and pushing-pulling test on two separate days in a random order.

(1) Back Muscle Endurance Measurements

For establishing the minimum and the maximum oxygenation and blood volume values for the lumbar erector spinae muscle of each subject, the Biering-Sørensen test for back muscle endurance was adapted (Biering-Sørensen 1984). The NIRS sensor was placed at the 3rd lumbar vertebra (Maikala et al. 2000) and a dark elastic bandage was wrapped around the lower back during standing posture (Figures 3.1A and 3.1B). The subject was then asked to lie prone on an adjustable plinth, with the anterior border of the iliac crest adjusted to the edge of a marker on the plinth (Figure 6.1A). One pillow was placed on the mid-thigh and another beneath the tibialis region with two elastic straps positioned at mid-thigh and mid-calf region (Ciriello and McGorry 2000). These straps were tightened so that the subjects' lower body was completely restrained at those positions (Figure 6.1B).

An adjustable rope with a weight attached at its edge was lowered to the position between the subject's scapula to serve as a reference point for the trunk alignment. Two minutes of NIRS data were collected for baseline measurements.

Just five seconds before the endurance test began, a countdown was started and the section of the plinth supporting only the upper body was lowered at the final countdown. The subject was instructed to place the arms across their chest throughout the endurance session and maintain contact with the reference point for as long as possible (Figure 6.2). When the subject was unable to maintain contact with this reference point the test was terminated, and the plinth support to the upper body was restored. Endurance time to the nearest second was recorded. Thereafter, the subjects were allowed to recover for four minutes in the prone position, at the end of which the session was terminated.

(2) Incremental Arm Cranking

For this session, the subject completed an incremental aerobic test on an arm cranking ergometer (Cybex, MET 300, USA). The axis of the ergometer crankshaft was adjusted to shoulder height of each seated subject. Position of the handle was adjusted in such a way that when the arms were fully extended, a slight flexion was present at the elbow joint. The legs and trunks were not restrained and their feet were flat on the ergometer base (Figure 5.3B). The test was initiated with a two-minute rest period. It was followed by two minutes of arm cranking at no load and subsequent increments of 25 Watts every 2 minutes at 50 crank rotations per minute (rpm) until voluntary exhaustion, or attainment of two or more of the end points suggested by the American College of Sports Medicine (1993): (1) leveling off in the oxygen consumption (increase of \leq 100 ml.min⁻¹) with increasing load, (2) age predicted maximal heart rate, equated to 220-age (in years), (3) respiratory exchange ratio of \geq 1.10, and (4) rate of perceived exertion (RPE) \geq 18 (Borg, 1982). During the last 30 s at each workload, the overall RPE was recorded using the Borg scale. Protocol was terminated with 2 minutes of active [cranking at zero load] and 4 minutes of passive recovery.

(3) Incremental Pushing-Pulling

For this session, each subject completed an incremental aerobic test on the BTE work simulator. Before beginning the session the height of the push-pull lever #171 was adjusted to each subject's elbow crease in the standing upright posture (Figure 5.4). The right hand was used as the main grasping hand (adopting a power grip) in this two-handed effort. The elbows were unlocked and the subject was discouraged from trunk rotation, swaying the body, and moving the lower extremities during pushing-pulling. The subject was instructed to put the right leg (extended at the knee) backward at a comfortable distance, and the left leg forward with a slight knee flexion so that the toe was placed on a marker that was in line with the handle being pulled [similar to a posture adopted in Ayoub and McDaniel (1974)]. Further, the subject was not permitted to lift the heels off the floor, and a nonskid flooring was used for this testing session.

The test was initiated with a two-minute rest period and was followed by two minutes of repetitive pushing-pulling in the sagittal plane at no load in the horizontal direction (Figures 5.5A and 5.5B). The distance of pushing-pulling was restricted by a delimiter fixed on the circumference of the BTE work simulator wheel at a fixed distance of 10 cm. A cadence of 50 push-pulls per minute was used for this protocol. Thereafter, the resistance was increased by 30 in-Ib every two minutes until voluntary exhaustion or attainment of two or more of the following end points suggested above by the American College of Sports Medicine (1993). Protocol was terminated with 2 minutes of active [cranking at zero load] and 4 minutes of passive recovery. Total force output in pounds and the power output in Watts for each subject were obtained from the printer attached to the BTE work simulator.

(4) Lumbar Erector Spinae Oxygenation and Blood Volume measurements

A continuous dual wave NIRS unit (MicroRunman, NIM Inc., PA, USA) was used to evaluate relative changes in the lumbar erector spinae oxygenation and blood volume during both arm cranking and pushing-pulling tests. This unit consists of: a superficial near infra-red sensor fitted with six light emitting

tungsten lamps placed 2 cm, 3 cm, and 4 cm apart; two photo-diode detectors that absorb light in the near infrared range between 760 nm and 850 nm; and a display unit that amplifies and displays the absorbency signal (Figure 2.2). The sensor was placed on the right erector spinae muscle at the 3rd lumbar vertebra, approximately 3 cm from the midline of the spine (Maikala et al. 2000) [Figure 3.1A]. A piece of clear plastic was wrapped around the sensor to prevent sweat from distorting the absorbency signal. A dark elastic bandage was wrapped around the sensor to secure it in place and minimize any loss of light (Figure 3.1B). Muscle skin fold thickness at the sensor location was measured twice using a Lange skinfold caliper (Cambridge Scientific Industries, Inc. Maryland, USA) before placing the NIRS sensor on the muscle belly. The adipose tissue thickness (a sum of fat and skin layer) at the sensor site was calculated as half the skinfold thickness. Position of the NIRS sensor at the 3rd lumbar region from the mid line of the spine was noted to the nearest millimeter, thus ensuring the correct placement of NIRS sensor on each subject for all the testing sessions.

As per the manufacturer's specifications, the sensor was calibrated at 760 nm and 850 nm wavelengths before the test for each subject. The light sourcedetector separation was set to 4 cm. The difference in absorbency between these two wavelengths indicated the change in oxygenation and the sum of these absorbencies indicated the change in total blood volume. Both oxygenation and blood volume were measured in terms of optical density (OD) which is defined as the logarithmic ratio of intensity of light calibrated at each wavelength to the intensity of measured light at the same wavelength (MicroRunman User Manual, NIM Inc., PA, USA, 1999). Real time data were recorded using the NIRCOM software provided by the manufacturer.

NIRS Data Analysis

All the real time NIRS values were averaged every 20 seconds using a customized Microsoft Excel program specially designed for this study. Baseline values were recorded during the initial rest. Minimum oxygenation and blood volume values of muscle tissue for each subject were recorded during both

incremental protocols and the Sørensen test. Maximum values were recorded during recovery from all these protocols as well. Furthermore, the range for oxygenation and blood volume responses for three protocols were calculated as the difference between the maximum and minimum values. Deoxygenation cost was calculated as the ratio of minimum oxygenation and peak oxygen uptake during each incremental exercise test.

Statistical Analysis

Independent 't' tests were used to compare the physical and performance characteristics of the men and women (Table 6.1A and 6.1B). A two-way analysis of variance (ANOVA), with gender as a between-subjects factor, and protocol (arm cranking; pushing-pulling; and the Sorensen test) as a repeated measure factor was used to compare the oxygenation and blood volume responses (measured in OD units). Also, a three-way ANOVA, with gender as a betweensubjects factor, muscle (lumbar erector spinae and biceps brachii) and exercise mode (arm cranking and pushing-pulling) as repeated measure factors was used to compare the oxygenation and blood volume responses between these two muscles. To minimize the violation of the assumption of homogeneity of variance, the 'Greenhouse-Geiser' adjustment was used. Since multiple comparisons were made for each dependent variable in the present study, the Bonferroni adjustment for a P value of 0.05 was applied to minimize Type 1 error. The Bonferroni adjustment was calculated as the ratio of 0.05 and number of statistical comparisons (APPENDIX B.5). Thus, statistical significance was considered at P < 0.01. Significant F ratios were further analyzed with the Scheffe post hoc multiple comparison test.

Pearson product moment correlations were calculated to examine the relationship between minimum, maximum, and range of NIRS measurements among the three protocols. A Spearman correlation test was used to examine the relationship between logarithmic value of adipose tissue thickness and oxygenation and blood volume responses during three protocols. Since there was no normal distribution in the adipose tissue thickness of the subjects in the

present study, a nonlinear Spearman correlation test was used to examine its role on the NIRS measurements. The Statistical Package for Social Sciences SPSS (version 10) was used for all statistical analyses (SPSS Inc., Chicago, USA).

RESULTS

Peak Lumbar Erector Spinae Oxygenation and Blood Volume Trends

The oxygenation and blood volume trends during both incremental exercise modes and the Sørensen test in a typical subject are shown in the Figures 6.3 and 6.4.

The Sørensen Test

At the onset of the test majority of the subjects demonstrated a decrease in both oxygenation and blood volume for at least 20 to 30 seconds, and only six subjects showed both increase in oxygenation and blood volume for a brief period of 15-20 seconds. However, during the reminder of the isometric test, majority of the subjects showed a steady decline followed by a leveling off in the deoxygenation at the end of contraction period of 60-90 seconds, suggesting a complete restriction of blood flow to the erector spinae muscle. During this contraction period, a rapid increase in blood volume was observed with increase in holding time.

From the onset to the end of the contraction period, two men and one woman demonstrated an increase in oxygenation and blood volume with increase in the duration of the endurance test. Further, only two subjects demonstrated decrease in blood volume with increase in contraction period. During recovery, oxygenation and blood volume reached baseline values within a minute in most of the subjects, with some of them demonstrating a supersaturation in both these variables.

Incremental Arm Cranking

In 70% of the subjects, there was an immediate increase in oxygenation for a brief period of 30-40 seconds during the warmup period followed by a

257

decrease in oxygenation. However, the rest of the subjects demonstrated a decrease in oxygenation at the onset of exercise. As the workload increased, oxygenation followed two distinct trends. It increased during lighter loads in most of the subjects and leveled off as the workload increased. A rapid decrease in oxygenation resulted at the final workload. However, few subjects demonstrated a decrease in oxygenation and leveling off at the initial loads, followed by a steady decrease in oxygenation as the work intensity increased. In 40% of the subjects blood volume trends showed an increase at the onset of exercise, and others demonstrated a decrease in blood volume for a brief period. Blood volume trends in most of the subjects did not change at lighter loads, but leveled off with increase in workload. Only few subjects demonstrated increase in blood volume during submaximal work increments. At their maximal workloads, blood volume either leveled off or decreased farther than the previous stages.

During the recovery from arm cranking, oxygenation and blood volume values increased rapidly and showed three distinct patterns. In majority of the subjects, both oxygenation and blood volume trends increased beyond the baseline demonstrating hyperaemia for a brief period. This hyperaemic response was not evident in some subjects, while in others these trends did not recover to the baseline values.

Incremental Pushing-Pulling

At the onset of the warmup period, half of the subjects showed a rapid decrease and the other half demonstrated an increase in both oxygenation and blood volume trends for a period of 30-60 seconds. As the workload increased, half of the subjects demonstrated a steady and rapid decrease in oxygenation, reaching maximum deoxygenation at the final workload. However, 30% of the subjects showed an initial decrease in oxygenation for a brief period and leveled off during the initial loads, followed by a rapid decrease observed at the final workloads. The rest of the subjects showed an initial increase in oxygenation during lighter workloads, leveling off, and a rapid decrease at the final workload. In the majority of the subjects, blood volume trends leveled off at initial loads,

followed by compromised blood flow with increase in workload. During the recovery from pushing-pulling, oxygenation and blood volume values increased rapidly and showed three distinct trends. In the majority of the subjects, both oxygenation and blood volume trends did not reach the baseline. However, some subjects reached the baseline values and few demonstrated hyperaemia.

Comparison among Incremental Exercise Modes and the Sørensen test

No significant interaction between the three exercise protocols and gender was observed in both oxygenation and blood volume responses. It should be noted that comparison of the main effects was based on the pooled values of the remaining independent variable in the experimental design. Minimum and maximum oxygenation values were not significantly different among the three protocols (P>0.01). However, the range for oxygenation during the Sørensen test was significantly lower by 35% and 42% compared to arm cranking and pushingpulling, respectively (P<0.01) (Table 6.2). There was no significant difference in the range for oxygenation values between arm cranking and pushing-pulling (P>0.01). When NIRS measurements obtained during incremental exercise modes were expressed as the percent of maximal desaturation obtained during the Sørensen test values (Table 6.2), no significant differences between pushingpulling and arm cranking were observed for minimum, maximum and range for oxygenation values (P>0.01). Additionally, no significant difference in deoxygenation cost was observed between pushing-pulling and arm cranking (P>0.01).

Minimum and maximum blood volume comparisons among the three protocols were also not significantly different (P>0.01). However, the range for blood volume values were the highest during arm cranking, with values being 72% greater than the Sørensen test (P<0.01). Additionally, the pushing-pulling values were significantly greater (by 65%) compared to the Sørensen test (P<0.01). However, there was no significant difference in the range for blood volume values between arm cranking and pushing-pulling (P>0.01). Based on the percent of the Sørensen test values (Table 6.2), no significant differences

259

between pushing-pulling and arm cranking were observed for minimum, maximum and range for blood volume values (P>0.01).

Comparison between Genders

Gender did not influence either oxygenation or blood volume responses among the three protocols investigated (P>0.01). Further, when the oxygenation and blood volume responses were expressed as a percentage of the Sørensen test, gender differences did not achieve statistical significance (P>0.01). Also, deoxygenation cost was not significantly different between genders (Table 6.2) (P>0.01).

Correlations among Three Protocols and NIRS measures

To relate peak oxygen uptake and minimum oxygenation between the two incremental exercise modes, specific correlations are reported in Table 6.3A. It is evident from this table that significant correlations were not obtained for minimum oxygenation between the arm cranking and pushing-pulling modes (r=0.11, P>0.05). Relationship between NIRS measurements during incremental arm cranking, pushing-pulling and the Sørensen test are given in Table 6.3B. A significant correlation was observed between the minimum and range for oxygenation during pushing-pulling (r=-0.93, P<0.05). This suggests that significant difference in range for oxygenation between pushing-pulling and the Sørensen test was attributed mainly to the minimum oxygenation during pushingpulling rather than corresponding maximum oxygenation values (r=0.53, P>0.05). A similar conclusion for the range in oxygenation cannot be made due to nonsignificant correlations between arm cranking and the Sørensen test (Table 6.3). Based on the ANOVA statistics, significant correlations for the range in the Sørensen test corresponded to minimum (r=0.45, P<0.05) and maximum (r=0.51, P<0.05) blood volume during pushing-pulling compared to corresponding arm cranking values.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Correlations between Adipose Tissue Thickness and Oxygenation and Blood Volume

Adipose tissue thickness calculated from the right side of erector spinae skinfolds at 3^{rd} lumbar region for men and women (mean \pm SD) were: 5.4 ± 1.5 mm, and 6.0 ± 2.3 mm, respectively. They were not significantly different from each other (P>0.05). Spearman correlations between adipose tissue thickness and mean changes in oxygenation and blood volume data for both men and women during the three protocols are shown in Table 6.4. A significant correlation was found between adipose tissue thickness and minimum oxygenation (r=-0.70, P<0.05) for men during the Sørensen test. However, correlations were not significant (P>0.05) between adipose tissue thickness and other NIRS variables for both genders among three protocols.

DISCUSSION

In a skeletal muscle, a complete range of oxygenation conditions is necessary to establish both the 'minimum' (0% oxygenation) and the 'maximum' (100% oxygenation) spectral values (Piantadosi 1993). Such a range helps in evaluating the role of dynamic exercise in influencing the 'true' hypoxic values of the specific muscle under study. This is usually obtained during cuff ischemia ('minimum' condition) for a period of 6 to 8 min to the specific muscle followed by subsequent hyperaemia indicating 'maximum' condition (Chance et al. 1992). However, occluding arterial supply to the trunk muscles is not possible. In the current investigation, a well-established back muscle endurance test was used to identify both minimum and maximum oxygenation and blood volume limits for the lumbar erector spinae muscle, which were used for comparison with incremental exercise protocols.

The Sørensen Test

The Biering-Sørensen method, also known as the "Sørensen test" has been utilized to evaluate static back muscle endurance in healthy populations

261

(Biering-Sørensen 1984). In the current investigation, most of the subjects demonstrated a steep decline in oxygenation till 60-90 seconds and leveled off after that period, suggesting a compromised blood flow in the erector spinae muscles during isometric contraction. However, the blood volume decreased rapidly in the beginning of isometric contractions, and then increased with holding time. These blood volume trends were not in agreement with maximal trunk extensions (Jensen et al. 1999, Maikala et al. 2000, Yoshitake et al. 2001).

Recently, Yoshitake et al. (2001) used the isometric trunk extension in prone position to establish the minimum and maximum lumbar erector spinae oxygenation and blood volume responses in healthy men at 0° and 15° of trunk extension for 60 seconds with reference to the horizontal plane. They demonstrated that oxygenation and blood volume were significantly reduced to a greater extent during 15° compared to 0° of trunk extension. However, in the present study subjects were asked to perform maximum voluntary contractions at 0° of trunk extension till exhaustion, and as the contraction period increased the majority demonstrated a rapid muscle desaturation, resulting in restriction of blood flow due to a greater intramuscular pressure in the erector spinae muscle. Several authors also reported a similar decrease in oxygenation trends during isometric back extension in both standing and sitting postures (Jensen et al. 1999, McGill et al. 2000, Maikala et al. 2000).

It is well established that at all exercise intensities, blood flow to the working muscles varies with changes in intramuscular pressure (Saltin et al. 1998). During back extension in a standing posture, Jensen et al. (1999) demonstrated a decrease in the lumbar erector spinae muscle oxygenation that remained unchanged after a period of 30 seconds. This suggests that a local homeostatic adjustment over a period may result in constant capillary oxygen tension thus further preventing decrease in hemodynamics in these paraspinal muscles (Jensen et al. 1999). Under such circumstances, these authors reported that at an intramuscular pressure range of 30-40 mm Hg, oxygenation in the paraspinal muscles was significantly reduced suggesting a greater demand for oxygen compared to oxygen supply. Maikala et al. (2000) demonstrated a similar

decrease in oxygenation and blood volume trends during both sitting and standing low-back extension postures. However, these authors did not report the intramuscular pressure values in the lumbar erector spinae muscles.

Based on the oxygenation responses from a seated isometric low-back extension, McGill et al. (2000) postulated that a decrease in oxygenation during an isometric contraction might be due to compromised blood flow in the smaller blood vessels because the intramuscular pressure exceeds the intravascular pressure as the duration of contraction increases. Another possible mechanism suggested by these authors is reduced perfusion and a concomitant increase in oxygen extraction from the muscle during the contraction period (McGill et al. 2000). However, these authors did not report the blood volume responses.

Incremental Exercise Modes

Several authors reported the involvement of shoulder and trunk muscles during arm cranking (Davis et al. 1976, Nag 1986, Sawka 1986, Toner et al. 1983). Constant gripping of the crank handle during arm cranking in a sitting posture results in an additional isometric component for torso stabilization (Sawka 1986). According to Davis et al. (1976), during the early stages of incremental arm cranking, involvement of arm muscles is greater, however as the workload increases the possibility of non-uniform recruitment of shoulder and trunk muscles results, reaching maximal values at peak workloads. Using electromyography, Nag (1986) demonstrated that the erector spinae muscles were minimally active during arm cranking but suggested that rhythmic component of trunk muscle activity is greater in arm work compared to leg work such as cycling.

During arm cranking at a constant load of 76 Watts, Toner et al. (1983) suggested that in addition to the linear increase of arm percent muscle volume with increase in oxygen uptake, the linear increase in oxygen uptake was also proportional to torso stabilization of the upper body. However, at a workload of 109 Watts, these authors demonstrated a curvilinear relationship suggesting the involvement of excessive body movements. Furthermore, from the current

investigation, the oxygenation and blood volume trends (Figures 6.3 and 6.4) in the lumbar erector spinae muscles suggest that during arm cranking, not only the upper extremity muscles (Chapter 5) but also the trunk muscles were being recruited, particularly at the maximal effort. As the workload increased, lumbar erector spinae oxygenation trends in the majority of the subjects during arm cranking showed either a smaller decrease or leveling off in oxygenation, suggesting recruitment of this muscle may not be influential during lighter workloads. However, at the final workloads when subjects were reaching exhaustion, there was a rapid decline in oxygenation indicating the importance of trunk muscle recruitment and stabilization during maximal effort.

Research pertaining to the erector spinae oxygenation and blood volume during low-back intensive occupational activities (e.g., repetitive lifting and lowering) is very limited (Maikala et al. 1999, Maronitis et al. 2000). As well, NIRS trends in the erector spinae oxygenation and blood volume during task-specific pushing-pulling have not been reported. In the current investigation, majority of men and women demonstrated a leveling off in both oxygenation and blood volume trends at initial loads, followed by a decrease in blood volume to the lowback muscles with simultaneous and greater decrease in oxygenation to these muscles. Such a decrease in oxygen saturation in smaller blood vessels of the lumbar erector spinae muscle during pushing-pulling suggests an imbalance between oxygen supply and demand during repetitive contraction of the trunk muscles.

However, a decrease in oxygenation and blood volume in the trunk extensor muscles during repetitive pushing-pulling is similar to the incremental arm cranking responses obtained in the present study. Several pushing-pulling studies have demonstrated the involvement of trunk muscles in both the sitting and standing postures (Ayoub and McDaniel 1977, Chaffin et al. 1983, Gagnon et al. 1992, Nadeau and Gagnon 1996, Troup and Chapman 1969). However, none of these previous studies were of repetitive nature, i.e., rather than repetitively pushing-pulling the lever in a stepwise fashion as investigated in the present study, these studies investigated the static push and/or pull forces
applied either for a single instant or frequently for a brief period. Further, the application of force in these investigations was only in one direction (either pushing forward or pulling towards the body) per activity.

As the workload increased during pushing-pulling, muscle oxygenation and blood volume started decreasing steadily with an increase in workload (Figures 6.3 and 6.4). Such a systematic decrease in oxygenation with increase in work intensity implies a greater release of oxygen by hemoglobin/myoglobin via the Bohr effect (Belardinelli et al. 1995, Chance et al. 1992). However, the myoglobin dissociation curve does not exhibit the Bohr effect, hence its overall role in determining muscle deoxygenation is relatively minor (Mancini et al. 1994). To this effect, Stringer et al. (1994) and Wasserman et al. (1991) suggested that metabolic acidosis during exercise was essential for skeletal muscle deoxygenation to take place. In other words, the presence of hydrogen ions associated with lactate accumulation was necessary for oxygen to be released from oxygenated hemoglobin in the skeletal muscle via the Bohr effect. Therefore, a steady decrease in the oxygen saturation in the muscle with increase in workload might suggest an increase in the oxygen demand exceeding the muscle saturation, thus resulting in an increase in deoxygenation (reflecting increased oxygen extraction from the lumbar erector spinae muscle).

In theory, the degree of deoxygenation measured by NIRS is attributed to the amount of oxygen released by both oxyhemoglobin and oxymyoglobin (Mancini et al. 1994). However, the current NIRS apparatus is unable to differentiate between the amount of deoxygenation that is attributed to hemoglobin and myoglobin due to the overlap of the absorbency signals of these chromophores in the near infra-red region (Chance et al. 1992). During knee extension exercise, Richardson et al. (1995) demonstrated that myoglobin desaturation occurs rapidly and reaches a maximal value at approximately 50% of maximal oxygen consumption. Recently, Tran et al. (1999) showed decline in the oxygenation obtained with the NIRS signal in the gastrocnemius muscle closely matches myoglobin desaturation from the nuclear magnetic resonance spectra.

Based on the lumbar erector spinae oxygenation and blood volume results obtained from arm cranking and pushing-pulling (Table 6.2), it is safe to say that trunk muscles respond similarly during both incremental exercise modes, and the postural difference (sitting Vs standing) might not influence in the muscle oxygenation and blood volume responses. This finding is in agreement with that of Troup and Chapman (1969) who demonstrated that the applied extensor forces during pushing-pulling were greater in sitting compared to standing, however, the forces transmitted by the erector spinae (turning moment) were of similar magnitude during both postures. These authors hypothesized that during standing, the lumbar spine is extended and the erector spinae is shortened, thus providing a relatively greater mechanical advantage in transmitting extensor forces than during the sitting posture. However, during sitting the lumbar spine is flexed; thus a smaller mechanical advantage in the sitting posture is compensated for by a shorter lever arm between the pelvis and upper limbs. resulting in a turning moment similar to that of the standing position (Troup and Chapman 1969). Hence in both postures identical forces exerted by the erector spinae might have led to similar oxygenation and blood volume responses during both pushing-pulling and arm cranking.

In the current study, subjects performed incremental exercises at their comfortable height, i.e., during arm cranking the shaft was adjusted to their shoulder level, and during pushing-pulling subjects performed at their respective waist heights (Table 6.1A). It has been previously demonstrated that pushing-pulling at comfortable heights is less strenuous on the lower-back (Chaffin et al. 1983). Thus, similar oxygenation and blood volume responses can also be attributed to adjustable workstation height during human performance. Deoxygenation cost during pushing-pulling was similar to arm cranking, suggesting that oxygen desaturation [or tissue oxygen consumption] in the lumbar erector spinae muscle per unit of oxygen uptake is similar during both task-specific and standardized protocols.

Comparison between Genders

Gender did not influence any of the oxygenation and blood volume responses among the three protocols. However, endurance (holding) time during the Sørensen test was significantly greater in women (by 32%) than in men (Table 6.1A), and is in agreement with several studies (Biering-Sørensen 1984, Jorgensen 1997, Mayer et al. 1995). Some of the reasons that might be attributed to greater endurance time in women are: a) upper-body weight is much lower in women than men (Jorgensen 1997), and b) women have greater percentages of slow twitch fibers in the trunk musculature, and large relative cross-sectional area of slow twitch and type A fast twitch fibers compared to men (Jorgensen 1997). Also, Mayer et al. (1995) hypothesized that load-to-maximum voluntary contraction ratio in women will be greater than men. However, in the current investigation both oxygenation and blood volume responses were not significantly different between men and women during the Sørensen test. This suggests, irrespective of greater holding time in women, both genders reported similar desaturation during trunk extension.

Men reported greater deoxygenation values but were not significantly different (P>0.05) compared to women during pushing-pulling. Even though not significant, a greater decrease in deoxygenation in men can be attributed to less oxidative capacity due to lower percentages of slow twitch fibers compared to women (Jorgensen 1997). However, such a trend was reversed for women with greater deoxygenation cost (P>0.05) during arm cranking. These results essentially suggest that even though men demonstrate greater whole-body oxygen uptake (Table 6.1B), their peripheral limitations at the erector spinae level are similar to women (Table 6.2).

Relationship between Peak Oxygen uptake and Minimum deoxygenation:

From Table 6.1, it is evident that peak oxygen uptake during arm cranking values were greater compared to pushing-pulling. Also, correlations of peak oxygen uptake between these exercise modes for the total sample was significant (r=0.67, P<0.05). However, the correlation between these two

exercise modes for minimum oxygenation was not significant (r=0.11, P>0.05), suggesting that peripheral responses during these two exercise modes are not similar. Since cardiac outputs were not investigated in the current study, and based on the lumbar erector spinae deoxygenation one can confirm that there is a significant difference in peripheral contribution between two incremental exercise modes. This finding (in addition to the findings suggested in Chapter 5) further strengthens our argument that developments of task-specific aerobic protocols are essential at the workplace.

The Optimal Test in Establishing Minimum and Maximum Physiological Limits for the Erector Spinae Muscle

In the present study, contrary to our assumption that maximum deoxygenation would result during the Sørensen test, greatest deoxygenation (P>0.05) was observed during pushing-pulling (Table 6.2A). However, this value was not significantly different among the three protocols. Further, maximum oxygenation values were also similar during three protocols. Interestingly, the range for oxygenation values showed significantly higher responses during both pushing-pulling and arm cranking separately, compared to the Sørensen test. From Table 6.4, it is evident that oxygenation range (based on the total sample size of 22 subjects) for pushing-pulling was significantly correlated to corresponding minimum than maximum oxygenation values (r=-0.93, P<0.05), suggesting greater deoxygenation during pushing-pulling can be a better measure among the three protocols in predicting the minimum value for the lumbar erector spinae muscle.

Furthermore, oxygenation and blood volume responses during isometric endurance test demonstrated in the present study might also suggest that unlike cuff ischemia protocols (see also Chapter 5) that have been adopted to establish fully oxidized and reduced states for specific skeletal muscles such as biceps brachii (Maikala and Bhambhani 1999) and vastus lateralis muscle (Bhambhani et al. 1998b, Chance et al. 1988, 1992), the Sørensen test might not be a good protocol in establishing minimum and maximum calibration limits for the erector spinae muscle. Questions that then arise are: why is the Sørensen test not a true protocol in establishing these limits? Are there any alternatives in establishing fully oxidized and reduced states for the lumbar erector spinae muscle?

Since the torque generated at the center of lumbopelvic joint is equivalent to the load lifted [the weight of the unsupported upper body only] and the lever arm from the center of joint to the center of gravity of the upper body, Mayer et al. (1995) criticized that during the Sørensen test subjects do not perform maximal voluntary contractions. Thus, these authors concluded that this test might evaluate a load equivalent to less than 50% of the subjects' maximal voluntary contraction. Further, Hultman et al (1993) demonstrated that in healthy subjects the cross sectional muscle area of the erector spinae at the 3rd lumbar region was not significantly related to static back muscle endurance, but was highly correlated with isokinetic trunk muscle strength.

Other techniques for evaluating endurance of the trunk muscles include: (1) pulling against a dynamometer in a standing position, thus measuring endurance 'dynamically' through the full range of motion rather than 'statically' in determining the back muscle endurance (Graves et al. 1990, Jorgensen 1997); (2) using force feedback during back extension at higher levels of force during positions such as standing, semi-standing or sitting (Elfving et al. 2000). Based on the present results (Table 6.2A), it may be also suitable to adopt task-specific protocols to establish minimum and maximum physiological limits for the trunk extensor muscles. Research needs to be conducted to evaluate the muscle oxygenation and blood volume responses during muscle oxygenation and blood volume responses during these methods to fully understand the localized responses elicited by these tests.

Interestingly, three subjects (two men and one women) showed an increase in oxygenation and blood volume during the Sørensen test. Initially it was speculated that such a response might be due to the side dominance of trunk muscles. Previous studies have shown conflicting evidence regarding right and left side EMG profiles of back muscles during isometric contractions (Elfving et al. 2000, Mannion et al. 1997, Merletti et al. 1994, Tsuboi et al. 1994). However during both incremental protocols, the oxygenation and blood volume

trends in these subjects were similar to all other subjects. Such conflicting evidence during the Sørensen test may be due to the asymmetrical trunk rotation in adjusting the left and right side of their trunk during the maximal effort (Tsuboi et al. 1994). These authors also hypothesized that such a discrepancy in the asymmetry may be due to "compensatory hypertrophy or atrophy resulting from specific muscle training or as a result of scoliosis which may cause uneven mechanical strain on the muscles". Hence, we believe such an increase in oxygenation and blood volume trends observed in three subjects during maximal back extension might be due to the inconsistent trunk rotation only.

Role of Adipose Tissue Thickness on NIRS measurements

van Beekvelt et al. (2001) reported that adipose tissue metabolism is lower than muscle metabolism hence muscle oxygenation determined using NIRS is underestimated. In the current study, there was no significant difference between men and women for the adipose tissue thickness at the L3 region of the erector spinae muscle (Table 6.1A). In the flexor digitorum super-ficialis muscle during rest and sustained isometric handgrip contractions, van Beekvelt et al. (2001) demonstrated a significant decrease in muscle oxygen consumption in both men and women with increase in the thickness of adipose tissue (r=-0.70, P<0.05). These authors further showed significant lower muscle oxygenation in women than in men during both rest and submaximal isometric contractions. Additionally, these authors found a significant but a weak correlation of 0.29 between forearm blood flow and adipose tissue thickness, suggesting an increase in blood flow with increase in the thickness of adipose tissue.

In the current investigation, adipose tissue thickness in women was not significantly correlated to any of the NIRS measurements among the three protocols (Table 6.4). However, a significant negative correlations was obtained in men for minimum oxygenation responses during the Sørensen test (r=-0.70, P<0.05). The regression between adipose tissue thickness and minimum oxygenation in men was:

[=0.57-0.95*log(adipose tissue thickness), r=-0.70, P=0.01]

270

This suggests that at the same source-detector separation of 4 cm, greater the thickness of adipose tissue lower the oxygenation values obtained during the Sørensen test. However, this correlation of minimum oxygenation in men was not significant during either arm cranking or pushing-pulling. Further, any other NIRS measurements in men were not significantly correlated among three protocols.

Such a significant negative correlation for minimum oxygenation during the Sørensen test in the current investigation may be a result of chance alone. Women also participated in the current investigation, and if such a significant relationship truly exists for men, correlations should be reliable for women as well [especially when the adipose tissue thickness was similar between genders]. However, as evident from the Table 6.4, this is not the case in the current investigation. Furthermore, calculating common variance for the r value of -0.70 results in 49% common variance, suggesting that 49% of the variability in the minimum oxygenation during the Sørensen test can only be explained by variance in the adipose tissue thickness (or vice versa). Hence, the remaining 51% of the variance might be influenced by other factors attributed to prediction error.

The major limitation of analyzing the correlation coefficients is that a significant correlation does not imply causality (Cody and Smith 1987, Vincent 1999). Some of the data points in the Figures 6.5 are extreme, and that might have caused larger correlation coefficients as well. Further, discrepancy observed in the current investigation and the study of van Beekvelt et al. (2001) might be due to their larger sample size of 44 men and 34 women. Also, factors such as exercise protocols used, specific muscle under investigation, and quantification of NIRS signals were different between these two studies. Hence based on the current limited sample size (11 men and 11 women), the fact that adipose tissue thickness and minimum oxygenation during the Sørensen test are related does not mean that a change in one variable will necessarily produce a corresponding change in other.

One question may also arise here, i.e., role of skin blood flow in the contamination of NIRS blood volume signal. With source-to-detector separation

of 40-50 mm, Hampson and Piantadosi (1988) showed that NIRS measurements are not affected by skin blood flow, and attributed such findings to a small volume of skin relative to muscle between the NIRS optodes, and the low concentration of cyctochorme oxidase in skin. In the current investigation, the source to detector distance was set to 40 mm for every subject during three protocols. Other authors also reported negligible influence of skin blood flow on NIRS signals (Mancini et al. 1994, Piantadosi et al. 1986). Further, Feng et al. (2001) showed that for an adipose thickness of 8 mm, the optimal source-detector distance was 45 mm. These authors have also identified the optimal distance of 35-40 mm for an adipose tissue thickness range of 2.5-5 mm, and 50 mm for a thickness of 17 mm. However, irrespective of the differences in the subjects' variation in the skinfolds, the current investigation used a fixed source-detector distance, which might have resulted in non-significant differences between men and women as well.

Further, Chance et al. (1992) reported a total pathlength of light in muscles as 7-10 cm that corresponded to a penetration depth of 25-30 mm. Using an ultrasound imaging technique, Jensen et al. (1999) measured the distance from the skin surface to the lumbar erector spinae muscle at the NIRS sensor site, and reported a distance of 9.0 ± 0.3 mm (at the cranial aspect) and 9.3 ± 0.5 mm (at distal aspect), respectively. Thus, measurements obtained with the existing NIRS setup (e.g., source-detector separation, photon penetration depth) in the current investigation are well within the range for the lumbar muscle.

CONCLUSION

In the present study, erector spinae oxygenation and blood volume responses were similar in men and women during both task-specific pushingpulling and arm cranking, indicating peripheral limitations at the trunk musculature were not influenced by the exercise mode. These results agree with our predicted hypotheses: (1) the erector spinae oxygenation and blood volume trends would be similar between the two modes of exercise, and (2) there would

272

be no significant gender differences in these responses. Further, deoxygenation cost during pushing-pulling and arm cranking are also similar.

Greater desaturation values were reported in men but were not significant compared to responses in women during both exercise modes, suggesting even though men can generate greater power output, their peripheral limitations at the erector spinae level are similar to women. Deoxygenation cost in men and women were also similar suggesting, men and women can extract similar amounts of oxygen from the lumbar erector spinae muscle. However, the most interesting finding of the present study is the greatest desaturation during incremental exercise modes compared to the back muscle endurance 'Sørensen test'. Based on the significant correlations for the range in oxygenation between pushing-pulling and the Sørensen test, maximum deoxygenation was obtained during pushing-pulling which may suggest faster fatiguability of these muscles during task-specific protocols. Simultaneous comparisons of NIRS measures in biceps and the lumbar erector spinae muscles demonstrated that blood volume increased in biceps with increase in the workload, however, compromised blood flow from limited oxygen delivery was evident in the erector spinae muscle. The role of adipose tissue thickness on NIRS measures was not consistent in both genders during the three exercise test protocols.

REFERENCES

- 1. Alberta Labor (1997) Occupational injury and diseases in Alberta Lost-time claims and Claim rates 1996 Survey. Alberta Labor, Edmonton, CANADA.
- 2. American College of Sports Medicine (1993) Resource manual for guidelines for exercise testing and prescription. Lea & Febiger, Philadelphia, USA.
- 3. Ayoub MM and McDaniel JW (1974) Effects of operator stance on pushing and pulling tasks. *AIIE Transactions* 185-195
- 4. Belardinelli R, Barstow TJ, Porszasz J, and Wasserman K (1995) Change in skeletal muscle oxygenation during incremental exercise measured with NIRS. *Eur J Appl Physiol* 70: 487-492
- Bernanrd B and Fine LJ (1997) Musculoskeletal disorders and workplace factors.
 (ed) Beranrd B, US Dept. of Health and Human Services, Public Health Services, CDC, DHHS (NIOSH) Publication # 97-141, Cincinnati, OHIO, USA.
- Bhambhani Y, Maikala R, and Buckley S (1998a) Muscle oxygenation during incremental arm and leg exercise in men and women. *Eur J Appl Physiol* 78: 422-431
- 7. Bhambhani Y, Maikala R, Jeon J, and Bell G (1998b) Reliability of tissue deoxygenation during cuff ischemia in healthy males. Annual Scientific Conference-Canadian Society for Exercise Physiology: Abstract page 3
- 8. Biering-Sørensen F (1984) Physical measurements as risk indicators for low back trouble over a one-year period. *Spine* 9: 106-119
- 9. Borg G (1982) Psychophysical bases of perceived exertion. Med Sci Sports Exerc 29: 402-409
- 10. Boushel R, Pott F, Madsen P, Radegran G, Nowak M, Quistroff B, and Secher N (1998) Muscle metabolism from near infrared spectroscopy during rhythmic handgrip in humans. *Eur J Appl Physiol* 79: 41-48
- 11. Chaffin D, Andres RO, and Garg A (1983) Volitional postures during maximal push/pull exertions in the sagittal plane. *Human Factors* 25(5): 541-550
- 12. Canadian Society for Exercise Physiology (1994) Revised Physical Activity Readiness Questionnaire – RPARQ. Gloucester, NB, CANADA.
- 13. Chance B, Dait MT, Zhang C, Hamaoka T, and Hagerman F (1992) Recovery from exercise-induced desaturation in the quadriceps muscles of elite competitive rowers. *Am J Phyisol* 262: C766-C775

- 14. Chance B, Nioka S, Kent J, McCully K, Fountain M, Greenfeld R, and Holtom G (1988) Time-resolved spectroscopy of hemoglobin and myoglobin in resting and ischemic muscle. *Anal Biochem* 174: 698-707
- 15. Ciriello VM and McGorry R (2000) Effects of preloaded isokinetic versus pure isokinetic training on dynamic and static lumbar muscle strength and endurance. *J Occ Rehab* 10(4): 257-269
- 16. Cody RP and Smith JK (1987) Applied Statistics and the SAS programming language. Elsevier Scinece Publishers, New York, NY, pp: 70-85
- 17. Davis JA, Vodak P, Wilmore JH, Vodak J, and Kurtz P (1976) Anaerobic threshold and maximal aerobic power for three modes of exercises. *J Appl Physiol* 41: 544-550
- 18. Elfving B, Nemeth G, and Arvidson I (2000) Back muscle fatigue in healthy men and women studied by electromyography spectral parameters and subjective ratings. *Scan J Rehab Med* 32: 117-123
- 19. Falkel JE, Sawka MN, Levine L, Pimental NA, and Pandolf KB (1986) Upperbody exercise performance: comparison between women and men. *Ergonomics* 29: 145-154
- 20. Feng W, Haishu D, Fenghua T, Jun Z, Qing X, and Xianwu T (2001) Influence of overlying tissue and probe geometry on the sensitivity of a near-infrared tissue oximeter. *Physiol Meas* 22: 201-208
- 21. Franklin BA, Hogan P, Bonzheim K, Bakalyar D, Terrien E, Gordon S, and Timmis GC (1995) Cardiac demands of heavy snow shovelling. *JAMA* 273(11): 880-882
- 22. Gagnon M, Beaugrand S, and Authier M (1992) The dynamics of pushing loads onto shelves of different heights. *Int J Ind Ergs* 9: 1-13
- 23. Garg A, Chaffin DB, and Herrin GD (1978) Prediction of metabolic rates for manual material handling jobs. *Am Ind Hyg Assoc J* 39(8): 661-674
- 24. Graves JE, Pollock ML, Carpenter DM, Leggett SH, Jones A, McMillan M, and Fulton M (1990) Quantitative assessment of full range-of-motion isometric lumbar extension strength. *Spine* 15: 289-294
- 25. Hampson NB and Piantadosi CA (1988) Near infrared monitoring of human skeletal muscle oxygenation during forearm ischemia. *J Appl Physiol* 64(6): 2449-2457

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

- 26. Hultman G, Nordin M, Saraste H, and Ohlsen H (1993) Body composition, endurance, strength, cross-sectional area, and density of MM erector spinae in men with and without low back pain. *J Spinal Disorders* 6(2): 114-123
- 27. Jensen BR, Jorgensen K, Hargens AR, Nielsen PK, and Nicolaisen T (1999) Physiological response to submaximal isometric contractions of the paravetebral muscles. Spine 24(22): 2332-2338
- 28. Jorgensen K (1997) Human trunk extensor muscles physiology and ergonomics. *Acta Phyisiol Scand* 160 S637: 1-58
- 29. Khalil TM, Genaidy AM, Asfour SS, and Vinciguerra T (1985) Physiological limits in lifting. *Am Ind Hyg Assoc J* 46: 220-224
- 30. Lee KS, Chaffin DB, Herrin GD, and Waikar AM (1991) Effect of handle height on lower-back loading in cart pushing and pulling. *Applied Ergonomics* 22(2): 117-123
- 31. Lortie M and Baril-Gingras G (1998) Box handling in the loading and unloading of vans. Int J Occ Safety Ergo 4(1): 3-18
- 32. McGill SM, Hughson RL, and Parks K (2000) Lumbar erector spinae oxygenation during prolonged contractions: implications for prolonged work. *Ergonomics* 43(4): 486-493
- 33. Maikala R, Farag M, and Bhambhani Y (2000) Low-back muscle oxygenation trends during simple postural variations – Intersession reliability using NIRS. Proceedings of International Ergonomics Conference, San Diego, USA: pp 5.9-5.12
- 34. Maikala R and Bhambhani Y (1999) Biceps brachii deoxygenation during cuff ischemia and recovery: Intra and inter session reliability. *Med Sci Sport Exer* S245 31:5 (Abstract)
- 35. Mancini DM, Bolinger L, Kendrick K, Chance B, and Wilson J (1994) Validation of NIRS in humans. *J Appl Physiol* 77: 2740-2747
- 36. Mannion AF, Conolly B, Wood K, and Dolan P (1997) The use of the surface EMG power spectral analysis in the evaluation of back muscle function. *J Rehabil Res Dev* 34: 427-439
- 37. Maronitis AB, Kovacs KM, Splittstoesser RE, and Marras W (2000) The effectiveness of whole-body and localized measures of fatigue. Proceedings of the IEA 2000/HFES 2000 Congress, San Diego, USA, pp 5-91-5-94

- 38. Mayer T, Gatchel R, Betancur J, and Bovasso E (1995) Trunk musice endurance measuremen – isometric contrasted to isokinetic testing in normal subjects. Spine 920-927
- 39. Merlett R, De Luca CJ, and Sathyan D (1994) Electrically evoked myoelectric signals in back muscles: effect of side dominance. *J Appl Physiol* 77(5): 2104-2114
- 40. MicroRunMan User Manual (1998), NIM Inc., PA, USA.
- 41. Nadeau S and Gagnon M (1996) Work and energy transfers in maximal pushing of loads. Int J Ind Ergo 17: 221-234
- 42. Nag PK (1986) Influence of posture on muscle contraction behaviour in arm and leg ergometry. In: Corlett N et al. (ed) The Ergonomics of Working Postures, vol 14. MacMillan, New York, pp 175-211
- 43. Nindl BC, Sharp MA, Mello RP, Rice VJ, Murphy MM, and Patton JF (1998) Gender comparison of peak oxygen uptake: repetitive box lifting versus treadmill running. *Eur J Appl Physiol* 77: 112-117
- 44. Petrofsky JS and Lind AR (1978) Comparison of metabolic and ventilatory responses of men to various lifting tasks and bicycle geometry. *J Appl Physiol* 45: 60-63
- 45. Piantadosi CA (1993) Absorption spectroscopy for assessment of mitochondrial function in vivo. *Methods in Toxicology* vol. 2: Academic Press, pp.107-126
- 46. Piantadosi CA, Hemstreet TA, and Jobsis-VanderVliet FF (1986) Near-infrared spectrophotometric monitoring of oxygen distribution to intact brain and skeletal muscle tissues. *Crit Care Med* 14: 698-706
- 47. Powell DM, Christopher A, Zimmer BA, Antoine MM, Baruch LD, Bellian KT, Morgan RF, and Edlich RF (1991) Computer analysis of the performance of the BTE work simulator. *J Burn Care Rehabil* 12: 250-256
- Richardson RS, Noyszewski EA, Kendrick KF, Leigh JS, and Wagner PD (1995) Myoglobin O2 desaturation during exercise: evidence of limited O2 transport. J Clin Invest 96: 1916-1926
- 49. Saltin B, Radegran G, Koskolou MD, and Roach R (1998) Skeletal muscle blood flow in humans and its regulation during exercise. *Acta Phyisiol Scand* 162: 421-436
- 50. Sawka MN (1986) Physiology of upper body exercise. In: Pandolf KB (ed) Exercise and sport sciences reviews, vol 14. MacMillan, New York, pp 175-211

277

- 51. Sharp MA, Harman E, Vogel JA, Knapik JJ, and Legg SJ (1988) Maximal aerobic capacity for repetitive lifting: Comparison with three standard exercise testing modes. *Eur J Appl Physiol* 57:753-760
- 52. Snook SH and Ciriello VM (1991) The Design of Manual Handling Tasks: Revised Tables of Maximum Acceptable Weights and Forces. *Ergonomics* 34(9): 1197-1213
- 53. Søgarrd K, Fallentin N, and Nielsen J (1996) Work load during floor cleaning.
 The effects of cleaning methods and work technique. *Eur J Appl Physiol* 73: 73-81
- 54. Stringer WW, Wasserman K, Casaburi R, Porszasz J, Maehara K, and French W (1994) Lactic acidosis as a facilitator of oxyhemoglobin during exercise. *J Appl Physiol* 76: 1462-1467
- 55. Tran TK, Saliasuta N, Kreutzer U, Hurd R, Chung Y, Mole P, Kuno S, and Jue T (1999) Comparative analysis of NMR and NIRS measurements of intracellular PO2 in human skeletal muscle. *Am J Physiol* 276 (45): R1682-R1690
- 56. Troup JDG and Chapman AE (1969) The strength of the flexor and extensor muscles of the trunk. J Biomech 2: 49-62
- 57. Tsuboi T, Satou T, Egawa K, Izumi Y, and Miyazaki M (1994) Spectral analysis of electromyogram in lumbar muscles: fatigue induced endurance contraction. *Eur J Appl Physiol* 69: 361-366
- 58. van Beekvelt MCP, Borghuis MS, van Engelen BGM, Wevers RA, and Colier WNJM (2001) Adipose tissue thickenss affects in vivo quantitative near-IR spectroscopy in human skeletal muscle. *Clin Sci* 101: 21-28
- 59. Vincent WJ (1999) Statisitcs in Kinesiology. Human Kinetics, second edition, Champaign, IL., pp: 87-99
- 60. Wasserman K, Hansen JE, and Sue DI (1991) Facilitation of oxygen consumption by lactic acidosis during exercise. *NIPS* 6: 29-34
- 61. Yoshitake Y, Ue H, Miyazaki M, and Moritani T (2001) Assessment of lower-back muscle fatigue using electromyography, mechanomyography, and near-infrared spectroscopy. *Eur J Appl Physiol* 84: 174-179

Gender	Age	Body Mass	Height	Body Mass Index	Back Muscle Skinfold at L3	Back Muscle Endurance Time	Push-Pull Height
[Yrs	Kgs	m	Kg.m ^{-z}	mm	Sec	cm
Men N=11	23.7 ± 4.5	77.5 ± 13.8 •	1.76 ± 0.08 *	24.6 ± 3.3 °	10.8 ± 3.0	139 ± 53 °	119±7*
Women N=11	24.5 ± 4.0	57.6 ± 9.4	1.63 ± 0.06	21.4 ± 2.4	12.0 ±4.5	203 ±72	111±6

Table 6.1A: Demographics of subjects (Mean ± SD)

^a – significant difference between men and women, p<0.05 b – significant difference between men and women, p<0.01

Table 6.1B: Demographics of subjects (Mean ± SD)

Gender	Peak Oxy L.	/gen Uptake min ⁻¹	Peak Power output Watts			
	Push-Pull	Arm Cranking	Push-Pull	Arm Cranking		
Men N=11	1.74 ± 0.51 ^{ab}	2.36 ± 0.51*	$32 \pm 8^{\circ}$	161 ± 32		
Women N=11	1.23 ± 0.32 ^b	1.56 ± 0.50	25 ± 8	114 ± 20		

^a – significant difference between men and women, p<0.01
 ^b – significant difference between Arm cranking and Pushing-pulling, p<0.01
 ^c – significant difference between gender and exercise mode [Arm cranking and Pushing-pulling], p<0.01

GENDER	TASK	[Oxygenation		T	Blood Volume	
		Min	Max	Range¥	Min	Max	Range¥
Men	Push-Pull	-0.302 ± 0.440	0.045 ± 0.185	0.347 ± 0.542 •	-0.314 ± 0.526	0.188 ± 0.448	0.503 ± 0.405
	Arm Cranking	-0.177 ± 0.178	0.172 ± 0.210	0.350 ± 0.282*	-0.328 ± 0.484	0.395 ± 0.688	0.723 ± 0.583*
	Sorensen Test	-0.087 ± 0.245	0.056 ± 0.331	0.140 ± 0.202	0.083 ± 0.322	0.092 ± 0.281	0.094 ± 0.308
	Push-Pull, %Sorensen Test	347 ± 180	80 ± 56	250 ± 268	[•] 378 ± 163	204 ± 150	535 ± 131
	Arm Cranking, %Sorensen Test	203 ± 138	307 ± 63	250 ± 139	395 ± 150	429 ± 244	769 ± 189
	Deoxygenation Cost ^a [Push-Puli], OD.L ⁻¹ .min ⁻¹	-17 ± 86					
	Deoxygenation Cost ^a [Arm Cranking], OD.L ⁻¹ .min ⁻¹	-8 ± 35					
Women	Push-Pull	-0.166 ± 0.196	0.086 ± 0.125	0.252 ± 0.221 ^b	-0.195 ± 0.406	0.279 ± 0.478	0.474 ± 0.341 ^b
	Arm Cranking	-0.223 ± 0.618	0.147 ± 0.470	0.370 ± 0.504 *	-0.158 ± 0.369	0.348 ± 0.378	0.507 ± 0.472 *
	Sorensen Test	-0.093 ± 0.195	0.018 ± 0.230	0.110 ± 0.133	-0.077 ± 0.206	0.251 ± 0.450	0.328 ± 0.485
	Push-Pull, %Sorensen Test	178 ± 100	477 ± 54	227 ± 169	253 ± 197	111 ± 106	145 ± 70
	Arm Cranking, %Sorensen Test	240 ± 317	816 ± 204	336 ± 387	205 ± 179	138 ± 84	155 ± 97
	Deoxygenation Cost ^e [Push-Pull], OD.L ⁻¹ .min ⁻¹	-13 ± 38					
	Deoxygenation Cost ^a [Arm Cranking], OD.L ⁻¹ .min ⁻¹	-14 ± 123					

Table 6.2: Erector Spinae Oxygenation and Blood Volume responses during Incremental Pushing-Pulling and Incremental Arm Cranking in Healthy Men and Women in Optical Density (Mean ± SD)

Range Ψ is defined as the difference between Maximum Optical Density (Max) and Minimum Optical Density (Min): *-significant difference between Arm cranking and Sorensen test (p <0.01): *-significant difference between Pushing-pulling and Sorensen test (p <0.01): *-significant difference between Arm cranking and Pushingpulling (p <0.01): *- Deoxygenation Cost is defined as the ratio of muscle deoxygenation and whole-body oxygen uptake during specific incremental exercise. Table 6.3A: Correlations of Erector Spinae Oxygenation and Blood Volume Responses among Three Protocols (n=22)

Variable	Protocols								
Oxygenation	Sorensen Vs Pushing-pulling	Sorensen Vs Arm Cranking	Pushing-pulling Vs Arm Cranking						
Minimum	-0.58 °	0.07	0.11						
Maximum	0.25	0.12	-0.31						
Range	0.31	0.37	0.27						
Blood Volume	Sorensen Vs Pushing-pulling	Sorensen Vs Arm Cranking	Pushing-pulling Vs Arm Cranking						
Minimum	-0.30	-0.10	0.30						
Maximum	0.21	0.21	0.35						
Range	0.07	0.26	0.51 *						

*-significant difference at P<0.05

							_				_	_			_			
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18
1		0.11	0.07	0.52	-0.22	-0.26	-0.65	-0.18	-0.52	0.45	-0.25	0.25	-0.21	-0.50	-0.56	-0.58	-0.30	-0.69
2			-0.58	-0.23	-0.25	-0.59	-0.33	-0.93	-0.24	0.10	0.42	0.14	-0.62	-0.06	0.08	-0.64	-0.61	-0.01
3	1			0.34	0.10	0.80	0.22	0.52	0.04	0.32	-0.27	0.24	0.53	-0.20	0.27	0.29	0.10	0.10
4	1				-0.31	0.12	0.31	0.07	-0.24	0.16	-0.50	0.23	0.21	-0.52	-0.21	0.10	-0.02	-0.34
5						0.25	-0.03	0.59	0.29	0.25	0.43	-0.51	0.60	0.62	0.01	0.41	0.23	0.34
6	1						0.40	0.60	0.63	0.32	-0.11	-0.03	0.68	0.05	0.50	0.43	0.21	0.48
7								0.27	0.37	-0.36	-0.16	-0.07	0.42	0.09	0.45	0.73	0.31	0.46
8	1								0.31	0.09	-0.20	-0.31	0.74	0.29	-0.06	0.69	0.60	0.14
9									•	0.13	0.17	-0.37	0.43	0.34	0.47	0.34	0.22	0.67
10											0.30	-0.10	0.43	0.07	0.12	-0.37	-0.28	0.18
11		•										-0.30	0.13	0.68	0.27	-0.11	-0.42	0.45
12													-0.31	-0.49	0.25	-0.24	-0.23	-0.41
13														0.35	0.21	0.68	0.28	0.40
14															0.21	0.30	0.38	0.51
15																0.12	-0.08	0.78
16																	0.51	0.26
17																		0.07
<u> </u>	,																	
		• ••••••						40 14				A C	stiss					
1 – M	Inimum	Uxygen	Buon — A	im Cran	King				nimum t		NUMO – A			•				
2 – M	2 - Minimum Oxydenation - Pushing-builing 11 - Minimum Blood Volume - Pushing-builing																	

12 - Minimum Blood Volume - Sorensen Test

13 - Maximum Blood Volume - Arm Cranking 14 - Maximum Blood Volume - Pushing-pulling

15 - Maximum Blood Volume - Sorensen Test 16 - Range Blood Volume - Arm Cranking

17 - Range Blood Volume - Pushing-pulling

18 - Range Blood Volume - Sorensen Test

Table 6.3B: Correlations of Erector Spinae Oxygenation and Blood Volume Responses among Three Protocols (n=22)

- 2 Minimum Oxygenation Pushing-pulling
- 3 Minimum Oxygenation Sorensen Test
- 4 Maximum Oxygenation Arm Cranking
- 5 Maximum Oxygenation Pushing-pulling
- 6 Maximum Oxygenation Sorensen Test
- 7 Range Oxygenation Arm Cranking
- 8 Range Oxygenation -- Pushing-pulling
- 9 Range Oxygenation Sorensen Test

Significance at P < 0.05

Gender	Protocol	Oxyge	nation	Blood Volume		
		Min	Max	Min	Max	
Men	Push-Pull	0.26	-0.46	0.06	0.23	
	Arm Cranking	0.15	-0.32	-0.36	-0.19	
	Sorensen Test	-0.70 ª	-0.46	0.07	0.13	
Women	Push-Pull	0.30	-0.24	0.24	0.19	
	Arm Cranking	0.39	0.26	-0.09	0.28	
	Sorensen Test	0.40	-0.18	-0.02	-0.46	
· • · • •					I	

 Table 6.4: Spearman Correlation Coefficients between Adipose Tissue Thickness and Erector Spinae Oxygenation and

 Blood Volume responses in Healthy Men and Women

*-significant difference at P<0.05



Figure 6.1A Subject Setup for the Sorensen Test



Figure 6.1B Securing the Subject on the Plinth



Figure 6.2 Subject Performing the Maximal Isometric Extension Test



Figure 6.3A: Erector Spinae Oxygenation Trends in a Typical Female



Figure 6.38: Erector Spinae Blood Volume Trends in a Typical Female

Min - Minimum Oxygenation (or Blood Volume) value obtained during specific Protocol Max - Maximum Oxygenation (or Blood Volume) value obtained during recovery from specific Protocol

Erector spinae oxygenation (top panel) and blood volume (bottom panel) trends in typical female subject during sorensen test, arm cranking and pushing-pulling, and recovery



Figure 6.4A: Erector Spinae Oxygenation Trends in a Typical Male



Min - Minimum Oxygenation [or Blood Volume] value obtained during specific Protocol Max - Maximum Oxygenation [or Blood Volume] value obtained during recovery from specific Protocol

Erector spinae oxygenation (top panel) and blood volume (bottom panel) trends in a typical male subject during sorensen test, arm cranking and pushing-pulling, and recovery



Thickness and Minimum Oxygenation during the Sorensen Test

CHAPTER 7

General Discussion

Study #1: Dose - Response Effects of Whole-body Vibration (WBV):

Occupations which involve exposure to WBV are often associated with factors such as awkward (and often prolonged) posture, constant attention during driving, operating various controls and hand-held levers, etc. These activities combined with WBV can be both physically demanding and mentally challenging. Such combined exposures have been implicated in both low-back disorders (Pope et al. 1998), and development of neck pain leading to headaches (Scutter et al. 1997). Hence, it is important to understand the energetics of specific tissues, in particular the 'focal' changes in cerebral and muscle regions that may be recruited simultaneously during exposure to WBV.

In this study, Near Infra-red Spectroscopy (NIRS), was used to examine the acute changes in oxygenation and blood volume simultaneously in the lumbar erector spinae muscle (as an indirect measure of evaluating physical workload) and pre-frontal cortex region (as an indirect measure of evaluating psychological workload) during exposure to 3 Hz, 4.5 Hz and 6 Hz WBV in healthy men and women. Open circuit spirometry was also used to monitor the cardiorespiratory measurements during these exposures. The subjects were asked to perform a motor task (maximal rhythmic handgrip contractions for a period of one minute) during WBV.

Exposure to WBV elicited an increase in cerebral oxygenation and blood volume with a concomitant decrease in these variables in the erector spinae muscle in both genders. These changes were not dependent upon the WBV dose, but were greater without the backrest support. Addition of handgrip work elicited an increase in cerebral activation (indicated by increases in both cerebral oxygenation and blood volume responses [Chapter 2, Figure 2.6]) with a concomitant decrease in the lumbar erector spinae muscle oxygenation (and blood volume responses in majority of subjects [Chapter 3, Figure 3.2]). Men demonstrated significantly greater mean deoxygenation values than women

during work combined with WBV, suggesting in men greater deprivation of oxygen occurs (without significant changes in muscle blood volume) in the erector spinae muscles. However, the greater cerebral and muscle oxygenation and blood volume responses in men were not significantly correlated to their maximal grip strength.

These observations raise several interesting questions. Why is an increase in cerebral oxygenation and blood volume associated with a concomitant decrease in lumbar erector spinae oxygenation and blood volume during WBV, and vice versa? Why does such an inverse relationship exist between these two tissues? Does an increase in cerebral activation observed in the right prefrontal cortex in the present study demonstrate a physiological pathway from the muscle to the brain during WBV?

During WBV, a decrease in substance P [in the dorsal root ganglion, a major output structure of the frontal cortex and considered as the 'brain' of the functional spinal unit (Weinstein 1991)] has been considered to be an indication of increased axonal transport in the brain. Based on animal WBV studies, several authors reported an increase of axonal transport in the frontal cortex due to WBV, thus affecting the neuropeptides (e.g. substance P) present in the sensory parts of the nervous system (Nakamura et al. 1994 and 1988, Weinstein 1991, Weinstein et al. 1988). The present findings (Chapter 2 of study #1) demonstrated that there was an increase in the 'focal' cerebral oxygenation and blood volume in the prefrontal cortex during all conditions of WBV (compared to rest without WBV or post-WBV period), suggesting an increase in neuronal activity with a subsequent increase in perfusion to the prefrontal cortex (Villringer and Chance 1997). This indirect evidence suggests a possible afferent feedback to the cerebral cortex during exposure to WBV in humans. It appears, therefore, that the increase in synaptic activity resulting from exposure to WBV elicits an increase in cerebral oxygenation and blood volume in the pre-frontal cortex which can be monitored non-invasively using NIRS.

Spinal reflex studies have demonstrated a decrease in human sensorimotor performance during WBV, suggesting a possible feedback

290

mechanism induced by vibration via the proprioceptive and central nervous systems (Gauthier et al. 1981, Martin et al. 1984, Roll et al. 1980). Martin et al. (1984) regarded the role of WBV as a combination of synchronous vibration applied 'locally' and simultaneously to several muscle tendons. The current observations in the specific paraspinal muscle suggest that WBV induces a 'localized vasoconstrictor stimulus' in the regions of the lumbar erector spinae muscle. Consequently, the reduction in blood flow would result in a lower blood volume and greater deoxygenation in this area monitored by NIRS. Based on electromyographic investigations, Hansson et al. (1991) hypothesized that there was a lack of blood flow to the back muscles during exposure to WBV. Kauppila and Tallroth (1993) investigated postmortem lumbar arteriorgrams of subjects with and without low-back pain in the lumbar and middle sacral arteries, and demonstrated that "insufficient arterial blood flow may be an underlying factor for low-back symptoms". Thus, the current evidence (Chapter 3 of study #1), demonstrates that the decrease in oxygenation may be the result of a compromised blood flow to the paraspinal muscles during WBV, and supports the use of NIRS in evaluating these changes.

It has been shown that 'localized' vibration (e.g. hand-arm vibration during a drilling task in 'construction' occupations) results in vasoconstriction of blood vessels in the hand thereby enhancing peripheral circulatory disturbances. This condition is termed "Vibration-induced White Finger" or "Raynaud Phenomenon" (Griffin 1990). Martin and Park (1997) showed that such localized vibration triggers the tonic vibration reflex in active muscles. However, to date there are no studies that have demonstrated the tonic vibration reflex in the trunk muscles during WBV. It could be argued that since exposure to WBV is a 'cumulative' of local vibrations being transmitted from the seat, floor, backrest, body segments, etc., a mechanism similar to the 'Raynaud Phenomenon' might occur 'locally' at the paraspinal muscles during prolonged exposure to WBV which, thus resulting in a decrease in muscle oxygenation and blood volume in this area.

Several authors have demonstrated an increase in the erector spinae muscle activity during cycling or repetitive loading due to WBV results in a greater recruitment of motor units in this muscle group (Pope et al. 1998, Seroussi et al. 1989, Wilder et al. 1982). Based on the current investigations of Study #1 (Chapters 2 and 3), this increase in muscular activity might lead to increased cortical activation. It should be noted that the increased acceleration measured at C6 in the current study (Chapter 2, Table 2.2) indicates considerable vibration being transmitted from the vibrating base ('With' and 'Without' a backrest) to the head. It can be argued that in contrast to the vasoconstrictor stimulus observed in the lumbar erector spinae muscle (demonstrated through decrease in oxygenation and blood volume), WBV elicits a cerebral 'vasodilator stimulus' as a result of increased vibration transmission to the head.

Another factor that should be considered is the cerebral oxygen extraction factor. Based on human positron emission tomography studies of somatosensory and visual stimulation, Fox and Raichle (1984) and Fox et al. (1988) demonstrated a greater proportion of increase in cerebral blood flow and cerebral glucose utilization, but a very small increase in the cerebral oxygen uptake. Such an imbalance in these variables suggests that less oxygen is being removed from the blood; i.e., a low cerebral oxygen extraction factor. This phenomenon might result in vasodilation and opening of precapillaries (thus maintaining autoregulation), accompanied by an increase in both cerebral blood volume and oxygenation trends. In contrast, based on the trunk muscle NIRS trends observed in the Chapter 3, it could be argued that the oxygen extraction in the muscle is considerably greater than that occurring simultaneously in the brain during WBV. This could be the result of vasoconstriction in the 'focal' muscle during this period, which would compromise blood flow and therefore result in reduced blood volume and oxygenation. Along these lines, Jorgensen (1997) stated that: "muscle contraction may increase the intramuscular pressure and thereby cause restriction of the blood flow which limits the rate of energy supply".

The reverse cerebral and muscle trends observed concomitantly during exposure to WBV may suggest a redistribution of blood [the 'stealing' phenomenon as addressed by McAllister (1997)] to the brain to match the needs of neuronal activity due to WBV. Similarly, Schumacker and Cain (1987) proposed that optimization of blood flow occurs in such a way that the regional redistribution in the tissues matches its metabolic activity, thereby minimizing the possibility of overperfusing some organs. Moreover, Sibbald et al. (2000) suggested a positive correlation between perfusion heterogeneity and heterogeneous tissue metabolic demand, and concluded that:

"..heterogeneity of perfusion and oxygen delivery is a physiological property at all levels of the heterogeneously distributed metabolic demands. Tight coupling between the two indicates autoregulation. The ultimate site of sensing and controlling presumably is the consumer, the cell itself".

It is conceivable, therefore, that during WBV a differential outflow of blood to these two regions in need (i.e., cerebral and lumbar erector spinae muscle) at the same time might have resulted in the contrasting trends observed in the oxygenation and blood volume responses. If such an argument is tenable, such a dual and out-of-phase phenomenon suggests that the human body is trying to maintain homeostasis while matching the needs of oxygen demand at the 'focal' lumbar erector spinae muscle region to the oxygen delivery to the brain at the same time. To further strengthen these arguments, Gandevia (1999) lucidly stated that "muscles are the servants of the brain and the motoneurone constitutes the final common path for output from the central nervous system".

The increases in the acute physiological responses as a result of exposure to 3 to 6 Hz WBV (Chapter 4) are similar to those observed during light physical activity (McArdle et al. 1996, Pope et al. 1990), suggesting that the whole-body physiological system is not heavily stressed under these conditions. In general, men showed significantly greater physiological responses as a result of exposure to WBV than women. The subjects demonstrated significantly greater physiological responses during sitting 'Without' backrest support compared to sitting 'With' back support. However, the highest physiological responses were observed during WBV combined with handgrip contractions compared to WBV alone, suggesting a combination of physical work and

exposure to WBV results in a greater cardiovascular stress and higher metabolic cost.

Changes in the mean cerebral and erector spinae oxygenation and blood volume responses were not dependent upon the peak oxygen uptake in either gender, suggesting that the level of aerobic fitness did not influence the tissue responses to WBV. However, despite the low metabolic rates reported during WBV, the effect of aerobic fitness was evident on the absolute and relative oxygen uptake responses, and oxygen pulse during isometric work. These observations suggest that high levels of aerobic fitness can buffer the increase in metabolic cost during exposure to WBV, most likely by acting on the oxygen transport system. As well, the degree of deoxygenation of the lumbar erector spinae muscle was not influenced by adipose tissue thickness in men and women.

Implications for Ergonomists:

Seated humans exhibit spinal resonance in the frequency range of 4.5 to 5.5 Hz, which falls within the range of 3 to 6 Hz usually experienced in the working and vehicular environments. The findings of the acute exposures to 3, 4.5 and 6 Hz in this study have several important implications that should be considered by ergonomists.

1. Development of dose response guidelines based on changes in cerebral and muscle oxygenation and blood volume:

The present findings indicated that the increases in the cardiorespiratory and metabolic responses during exposure to WBV were quite small and did not place excessive stress on the cardiorespiratory system. However, the muscle oxygenation and blood volume changes measured by NIRS indicated that blood volume and oxygenation to lumbar muscles decreased during exposure to WBV, thereby making them more susceptible to fatigue. To this effect, electromyographic studies have demonstrated low-back muscle fatigue during exposure to WBV. Similarly, cerebral responses increased during WBV suggested an increased axonal transport in the brain. Although these acute responses provide useful information, as an ergonomist, it is essential to study the effects of chronic exposure (hours per shift to days and years of exposure) to WBV to determine the long term effects on these tissues. These results will help health professionals to develop dose-response guidelines for populations exposed to WBV in various industries; information that is currently lacking in the widely used ISO document (ISO 2631/part 1, 1997).

2. Recommendations for the use of backrest in vehicles:

Cerebral oxygenation and blood volume responses were not influenced by the presence or absence of a backrest. However, erector spinae oxygenation and blood volume decreased to a greater extent during sitting 'Without' backrest compared to sitting 'With' backrest. As well, higher cardiorespiratory responses were observed during sitting 'Without' a backrest. These observations highlight the importance of utilizing a backrest in transportation vehicles to support the postural muscles during prolonged exposure to WBV.

3. Design of manual controls in occupations where exposure to whole-body vibration is prevalent:

Many of the occupations in which individuals are exposed to WBV necessitate manual work such as the operation of controls, gripping tools, etc. Based on the present findings, addition of rhythmic isometric work significantly increased the cardiorespiratory stress, elevated cerebral oxygenation and blood volume, and reduced muscle oxygenation and blood volume to a greater extent compared to the absence of such work. These factors should be considered by ergonomists in designing workstations, recommending non-vibration tools to be used by employees, etc. in order to minimize undue fatigue and reduce the risk of injury.

4. Development of specific guidelines for men and women:

The absolute changes in the cardiorespiratory responses, cerebral and muscle oxygenation and blood volume were greater in men compared to women. As well, the acceleration values obtained at the cervical region were higher in men than women. These findings suggest a differential gender response to WBV, implying that it may be necessary to develop specific guidelines for men and women.

5. Inclusion of regular physical activity into the lifestyle.

In the present study, the peak aerobic fitness of the subjects attenuated the acute cardiorespiratory responses during exposure to WBV, but not the tissue oxygenation and blood volume responses during handgrip contractions. This suggests that improved aerobic fitness is important even for light work performed during WBV, especially when the exposure is for the greater part of the working day. It is important, therefore for ergonomists and occupational health and safety professionals to encourage workers to participate in employee fitness programs in order to minimize the stress associated with such activities.

Study #2: Comparison Between Arm Cranking and Repetitive Pushing-Pulling:

Pushing-pulling is an important task in many occupations. Research has suggested that individuals performing such an activity over extended periods may be more susceptible to low back problems (Bernard and Fine 1997, Burdorf and Sorock 1997). A reduction in localized blood flow has been implicated in this disorder. In this study the cardiorespiratory responses during pushing-pulling and arm cranking were compared using open circuit spirometry in healthy men and women. As well, NIRS, was used to evaluate the acute changes in oxygenation and blood volume in the biceps brachii muscle groups which were directly involved in both these exercise modes, and the lumbar erector spinae muscle which was an accessory muscle involved in the exercise.

The peak cardiorespiratory responses (absolute and relative oxygen uptake, oxygen pulse, respiratory exchange ratio, ventilation rate, and respiratory frequency) were significantly higher during arm cranking compared to pushingpulling. Power output during arm cranking was significantly higher (by 79%) than pushing-pulling, with men demonstrating significantly 30% greater power output during both exercise modes. The lower power output generated during pushingpulling resulted in significantly greater ratios of peak oxygen uptake to peak power output (by 72%) compared to arm cranking. In general, men showed greater cardiorespiratory responses (absolute oxygen uptake, oxygen pulse, ventilation rate, tidal volume) than women. However, gender did not influence the relative oxygen consumption, heart rate, respiratory frequency, ventilatory equivalent for oxygen, respiratory exchange ratio, rating of perceived exertion, and ratios of absolute and relative oxygen uptake to power output during both modes of exercise.

The oxygenation trends in the biceps brachii and erector spinae muscles decreased significantly during both exercise modes in both genders. During both exercise modes the oxygen demand was greater in the primary muscles (biceps), resulting in a greater blood volume in this region. However, the blood volume responses in the biceps brachii and erector spinae muscles demonstrated opposing trends. From the Figures 5.6B and 5.7B, it is evident that as the workload increased, biceps brachii blood volume increased. However, blood volume in the lumbar erector spinae decreased at the onset of exercise (Figures 6.3B and 6.4B) and demonstrated a compromised blood flow (in majority of the subjects) with increasing workload. This difference in blood volume profiles during the same exercise suggests that muscle co-contraction is possible (Kahn et al. 2000), and the blood flow to individual muscle tissue differs based on the oxygen demand. According to Sibbald et al. (2000), tissue perfusion and metabolic demand are heterogeneously distributed.

The increase in biceps brachii blood volume could be attributed to an increase in perfusion pressure (Boushel et al. 2000a). It appears that some of the increased blood volume in the biceps muscle during these two exercise modes may be due to a redistribution from the postural muscles, thereby making them more susceptible to fatigue. To this effect, Boushel et al. (2000b) demonstrated variability in the magnitude of blood flow in different muscles of the same muscle group. However, the biceps and lumbar erector spinae are two different muscle groups, and based on the heterogeneity in the muscle blood volume trends (Figures 5.6B, 5.7 B and Figures 6.3B, 6.4B), it is likely that oxygen supply and demand is regulated depending on the muscle location and

muscle group (biceps and lumbar erector spinae). To this effect, Vallet (1998) stated that:

"According to the metabolic theory of microvascular regulation, metabolic vasodilation in tissues with relatively high metabolic rates competes with sympathetic vasoconstrictor tone, thereby adjusting the balance between local tissue oxygen supply and demand."

Further, in explaining the redistribution of blood flow during acute dynamic exercise, McAllister (1997) stated significant increase in blood flow to the active muscle and heart. However, blood flow to the non-exercising skeletal muscle decreases significantly. This author also stated that such a redistribution of blood flow, thus increased cardiac output is achieved through vasoconstriction in these inactive organs, and vasodilation in the active muscles. This might have resulted in reverse erector spinae blood volume trends compared to the biceps blood volume trends. However, the range for blood volume responses calculated in both biceps (Table 5.3) and lumbar erector spinae muscles (Table 6.2) was not significantly different suggesting that regulation of blood flow to the individual muscle tissues at the end of exercise is similar.

Despite the lower cardiorespiratory responses during pushing-pulling, a greater degree of deoxygenation was observed in the biceps during this mode compared to arm cranking, suggesting a greater peripheral limitation in the biceps during pushing-pulling. This limitation was evident in the lumbar erector spinae muscle as well, further strengthening the argument that preference for task-specific aerobic protocols over standardized arm cranking protocols are essential at the workplace. Adipose tissue thickness of the biceps brachii and lumbar erector spinae did not influence the muscle oxygenation and blood volume measurements in both genders.

Implications for Ergonomists

Currently, many occupational fitness screening programs routinely use standardized exercise protocols for assessing cardiorespiratory fitness of potential applicants and employees. It is highly recommended that these screening protocols be specific to the job in question so as to increase the validity of the findings. In many physically demanding occupations such as the firefighters, police force and emergency medical services, physical ability tests that best reflect job requirements are also undertaken as part of the screening program. The results of the present study highlight the importance of using task-specific protocols, and demonstrate several points that have implications for ergonomists involved in occupational fitness testing programs.

1. Assessment of the peak oxygen uptake during arm cranking may overestimate an individuals' functional work capacity:

The current results indicated that the peak oxygen uptake during arm cranking was significantly higher than that attained during pushing-pulling in both genders. This difference between the two exercise modes could be due to both biomechanical and physiological factors that could influence performance. Therefore, ergonomists should be aware that recommendations for job suitability based on arm cranking values could be erroneous if the job requirement involved extensive pushing-pulling.

2. Relative contribution of the peripheral system:

The current findings demonstrated that the deoxygenation cost of the biceps brachii muscle (deoxygenation per unit amount of work) during pushing-pulling was significantly greater than that attained during arm cranking (i.e. pushing-pulling was less efficient than arm cranking). This suggests a greater peripheral contribution and limitation during pushing-pulling compared to arm cranking. Therefore, individuals employed in occupations that involve extensive pushing-pulling will most likely be predisposed to earlier fatigue than what an arm cranking test would suggest.

3. Recruitment of accessory muscles:

Epidemiological studies have suggested that individuals employed in jobs that involve extensive pushing-pulling have a greater incidence of low back problems. The current evidence indicates that during simulated pushingpulling in the laboratory, muscle oxygenation and blood volume to the erector spinae muscle are compromised. This suggests that these accessory muscles may be predisposed to fatigue.

General Limitations of the Present Studies:

All the subjects who participated in these studies were young, healthy volunteers from a university population and were not a representative sample of the actual workforce in a typical industry. Factors such as age and job experience that can affect work performance could influence the generalizability of these findings to the workplace. To this effect, Mital (1986) demonstrated that in general, responses of volunteers from a non-industrial workforce (such as student populations] to task variables [frequency of lift, lifting height, and size of carton lifted) were similar to those observed on an industrial workforce during repetitive lifting. However, they also showed that the lifting capabilities [maximum acceptable lifting weight] of such representative populations were significantly lower compared to experienced industrial workers. This implies that there is considerable adaptation in the physical work capacity of experienced workers, and therefore, the current observations on untrained student volunteers should be interpreted with caution. It is recommended that future research comparing these acute physiological and hemodynamic responses (muscle and cerebral) between novice subjects and experienced industrial workers be designed to evaluate whether there are any inherent differences between these populations. As well, replication of these findings by independent researchers is important to increase the impact of these findings.

Another limitation of the present findings could be the measurements obtained from the NRS instrumentation. The large variance in the muscle and cerebral hemodynamic responses in the present research [Table 2.3 in Chapter 2, Table 3.2 in Chapter 3, Table 5.3 in Chapter 5, and Table 6.2 in Chapter 6] is similar to that previously reported (Bhambhani et al. 1998, 2000, and 2001, Rundell et al. 1997). Whether this variance is an inherent nature of NIRS measurements is still not clear. It could be due to new (and may be relevant) hemodynamic behavior not identifiable by the existing oxygenation and blood

300
volume measurements. Giller et al. (2000) suggested a similar explanation for the high variance in cerebral hemodynamic responses measured by transcranial Doppler ultrasound during rhythmic handgrip contractions. Using a larger sample size in these studies may have reduced the large variance in the NIRS measurements obtained in the present research. It is suggested that sample sizes for future studies, which incorporate both cardiorespiratory, and tissue hemodynamic measurements be calculated on the basis of statistical power that can be obtained with the NIRS variables of interest.

In the workplace, employees are exposed to a variety of environmental factors that can affect their work performance and the associated physiological responses. The current observations under controlled laboratory conditions can mask the true nature of job demands of workers who are exposed to whole body vibration or pushing/pulling in conjunction with other extraneous variables at the workplace. Keyserling (2000) suggested that laboratory studies are "generally incapable of proving a relationship between exposure and injury". Further, the subjects participating in these laboratory studies were instructed to maintain specific postures to suit the experimental testing procedures. However, in the real world there usually is considerable flexibility in the job demands and workplace design that allows workers to adopt postures that suit their comfort, which could influence the physiological responses during these tasks. Nevertheless, Keyserling (2000) stressed the importance of laboratory studies for providing valuable scientific insights into how the human body responds to risk factors in the workplace.

REFERENCES

- Bhambhani Y, Maikala R, and Buckley S (1998) Muscle Oxygenation during Incremental Arm and Leg Exercise in Men and Women. Eur J Appl Physiol (78): 422-431
- 2. Bhambhani Y, Kawchuk C, Burnham R, Justin J, and Maikala R (2000) Quadriceps muscle deoxygenation during functional electrical stimulation in adults with spinal cord injury. Spinal Cord 38: 630-638
- Bhambhani Y, Maikala R, and Esmail S (2001) Oxygenation Trends in Vastus Lateralis Muscle during Incremental and Intense Anaerobic Cycle Exercise in Young Men and Women. Eur J Appl Phsiol
- 4. Boushel R, Langberg H, Green S, Skovgaard, Bulow J, and Kjaer M (2000a) Blood flow and oxygenation in peritendinous tissue and calf muscle during dynamic exercise in humans. *J Physiol (London)* 524(1) : 305-313
- Boushel R, Langberg H, Olesen J, Nowak M, Simonsen L, Bulow J, and Kjaer M (2000b) Regional blood flow during exercise in humans measured by nearinfrared spectroscopy and indocynanine green. *J Appl Physiol* 89 : 1868-1878
- Bernard B and Fine LJ (1997) Musculoskeletal disorders and workplace factors.
 (ed) Beranrd B, US Dept. of Health and Human Services, Public Health Services, CDC, DHHS (NIOSH) Publication # 97-141, Cincinnati, OHIO, USA.
- 7. Burdorf A and Sorock G (1997) Positive and negative evidence of risk factors for back disorders. *Scan J Work Environ Health* 23: 243-256
- 8. Fox PT and Raichle ME (1984) Stimulus rate determines regional brain blood flow in straite cortex. Ann Neurol 17: 303-305
- 9. Fox PT, Raichle ME, Mintun MA, and Dence C (1988) Nonoxidative glucose consumption during focal physiologic neural activity. *Science* 241: 462-464
- 10. Gandevia SC (1999) Mind, muscles, and motoneurones. J Sci Med Sport 2(3): 167-180
- 11. Gauthier GM, Roll JP, Martin B, and Harlay F (1981) Effects of whole-body vibrations on sensory motor system performance in man. *Aviat Space Envi Med* 52(8): 473-479
- 12. Giller CA, Giller AM, Cooper CR, and Hatab MR (2000) Evaluation of the cerebral hemodynamic response to rhythmic handgrip. J Appl Physiol 88: 2205-2213

- 13. Griffin MJ (1990) Vibration and discomfort. In: Handbook of Human Vibration. New York, Academic Press, pp 43-123
- 14. Hansson T, Magnusson M, and Broman H (1991) Back muscle fatigue and seated whole-body vibrations: an experimental study in man. *Clin Biomech* 6(3): 173-178
- 15. International Organization for Standardization 2631 Part 1 (1997) Mechanical vibration and shock Evaluation of human exposure to whole-body vibration Part 1: General Requirements. ISO 2631/1-1997 (E)
- 16. Jorgensen K (1997) Human trunk extensor muscles physiology and ergonomics. Acta Physiol Scand Suppl 160: 1-58
- 17. Kahn JF, Favriou F, Jouanin JC, and Grueza R (2000) Effect of Co-Contractions on the Cardiovascular Response to Submaximal Static Handgrip. *Eur J Appl Physiol* 83: 506-511
- 18. Kauppila LI and Tallroth K (1993) Postmortem angiographic findings for arteries supplying the lumbar spine: Their relationship to low-back symptoms. *J Spinal Disord* 6(2): 124-129
- 19. Keyserling WM (2000) Workplace risk factors and occupational musculoskeletal disorders, Part 1: A review of biomechanical and psychophysical research on risk factors associated with low-back pain. Am Ind Hyg Assoc J 61: 39-50
- 20. McAllister RM (1997) Adaptations in control of blood flow with training: splanchnic and renal blood flows. *Med Sci Sports Exerc* 30(3): 375-381
- 21. McArdle WD, Katch I, Katch VL (1996) Exercise Physiology, Williams & Wilkins, Baltimore, pp 159
- 22. Martin BJ, Roll JP, and Gauthier GM (1984) Spinal reflex alterations as a function of intensity and frequency of vibration applied to the feet of seated subjects. *Aviat Space Envi Med* 55(1): 8-12
- 23. Martin BJ and Park H (1997) Analysis of the tonic vibration reflex: influence of vibration variables on motor unit synchronization and fatigue. *Eur J Appl Physiol* 75: 504-511
- 24. Mital A (1986) Comparison of lifting capabilities of industrial and non-industrial populations. Proceedings of the Human Factors Society 30th Annual Meeting.
- 25. Nakamura H, Moroji T, Nagase H, Okazawa T, and Okada A (1994) Changes of cerebral vasoactive intestinal polypeptide and somatostatin-like immunoreactivity

induced by noise and whole-body vibration in the rat. Eur J Appl Phyisiol 68: 62-67

- 26. Nakamura H, Okada A, Nohara S, and Nakamura H (1988) The effect of whole body vibration on the cerebral metabolites in rats. Aerospace Science – Proceedings of the 1st Nihon University International Symposium on Aerospace Medicine, Nihon University Press, Japan, pp 153-156
- 27. Pope MH, Magnusson M, and Wilder D (1998) Kappa Delta Award Low back pain and whole-body vibration. *Clin Orthop Research* 354: 241-248
- Pope MH, Magnusson M, Lindell V, Svensson M, Andersson GBJ (1990). The measurement of oxygen uptake under whole-body vibration. *Iowa Orthop* J 10: 85-88
- 29. Roll JP, Martin B, Gauthier GM, and Ivaldi FM (1980) Effects of whole-body vibration on spinal reflexes in man. *Aviat. Space Environ Med* 51: 1227-1233
- 30. Rundell KW, Nioka S, and Chance B (1997) Hemoglobin/myoglobin desatuartion during speed skating. *Med Sci Sports Exerc* 29: 248-258
- 31. Schumacker PT and Cain SM (1987) The concept of a critical oxygen delivery. Int Care Med 13: 223-229
- 32. Scutter S, Turker KS, and Hall R (1997) Headaches and neck pain in farmers. Aust J Rural Health 5(1): 2-5
- 33. Seroussi RE, Wilder DG, and Pope MH (1989) Trunk Muscle Electromyography and Whole Body Vibration. *J. Biomech* 22: 219-229
- 34. Sibbald WJ, Messmer K, and Fink MP (2000) Roundtable conference on tissue oxygenation in acute medicine, Brussels, Belgium, 14-16 March 1998. *Int Care Med* 26: 780-791
- 35. Vallet B (1998) Vascular reactivity and tissue oxygenation. Int Care Med 24: 3-11
- 36. Villringer A and Chance B (1997) Non-invasive optical spectroscopy and imaging of human brain function. *Trends Neurosci* 20:435-441
- 37. Weinstein J, Pope M, Schmidt R, and Seroussi R (1988) Neuropharmacologic effects of vibration on the dorsal root ganglion An Animal Model. *Spine* 13(5): 521-525
- 38. Weinstein J (1991) Neurogenic and nonneurogenic pain and inflammatory mediators. Orthopedic Clinics of North America 22: 235-244
- 39. Wilder D, Woodworth BB, Frymoyer JW, and Pope MH (1982) Vibration and the human spine. Spine 7(3): 243-254

APPENDIX A

CONSENT FORMS AND INFORMATION SHEETS

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



UNIVERSITY OF ALBERTA CONSENT FORM

Part 1 (to be completed by the Principal Investigator):

Title: Dose Response Effects of Whole Body Vibration on Acute Physiological Responses in Healthy Men and Women

Principal Investigator(s): Dr. Yagesh Bhambhani

Co-Investigator(s): Ram Maikala, Ph.D Student, Faculty of Rehabilitation Medicine; 7248

Part 2 (to be completed by the research subject):

Do you understand that you have been asked to be in a research study?	Yes	No
Have you read and received a copy of the attached Information Sheet?	Yes	No
Do you understand the benefits and risks involved in taking part in this research study?	Yes	No
Have you had an opportunity to ask questions and discuss this study?	Yes	No
Do you understand that you are free to refuse to participate or withdraw from the study at any time? You do not have to give a reason and it will not affect your care. (Use wording appropriate to your subject group)	Yes	No
Has the issue of confidentiality been explained to you? Do you understand who will have access to your records?	Yes	No
Do you want the investigator(s) to inform your family doctor that you are participating in this research study? If so, please provide your doctor's name: (N.B. This question is optional).	Yes	No

This study was explained to me by:

I agree to take part in this study.

Signature of Research Participant

Witness

Printed Name

Printed Name

I believe that the person signing this form understands what is involved in the study and voluntarily agrees to participate.

Date

Signature of Investigator or Designee

Date

THE INFORMATION SHEET MUST BE ATTACHED TO THIS CONSENT FORM AND A COPY GIVEN TO THE RESEARCH SUBJECT

Faculty of Rehabilitation Medicine

2-64 Corbest Hall + University of Alberts + Edmanson + Canada + T6G 2G4 Telephone: (780) 492-2499 + Faz: (780) 492-1626

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



UNIVERSITY OF ALBERTA INFORMATION SHEET

Project Title: Dose Response Effects of Whole Body Vibration on Acute Physiological Responses in Healthy Men and Women

Investigator(s):

Dr. Y. Bhambhani, Professor 3-73 Corbett Hall Department of Occupational Therapy University of Alberta Tel: 492 - 7248 Rammohan V. Maikala, PhD Candidate 1-94 Corbett Hall Work Physiology Lab University of Alberta Tel: 492 – 0404

(1) Purpose

Vibration is movement of the body that occurs when a person sits in a car or bus. A dose of vibration may cause pleasure, pain, or motion sickness. This depends on the type of motion and activities of the person. It can also depend on other things in the environment. Small doses of vibration increase the physiological stress on the body in the same way as walking or jogging. Exposure to vibration for several years may cause physical and mental changes.

Some researchers think that the body's response to vibration may depend upon the person's physical fitness. Others feel that vibration affects men and women differently. However, there are no studies that have tested these theories.

In this study, we will test: a) the effects of vibration on the oxygen levels of the back muscles and brain in healthy males and females, and b) examine the relationship of these changes with the aerobic fitness levels of the subjects.

(2) Study Requirements

This study will last two weeks. During this time you will be tested four times. Each test will last approximately 45 minutes. These tests will tell us how well your heart, lungs, arm muscles and brain are working. The four tests are: arm cranking till exhaustion, 16 minutes of seated vibration at 3 Hz, 4.5 Hz and 6 Hz with rest in between vibration. During vibration, you will be asked to squeeze a grip strength instrument with your right hand for one minute. You will be also asked to do a deep breathing for 15 seconds during these vibration tests. You will be shown how the testing is done by a research assistant before you volunteer for the study.

(b) Testing Methodology

In the first test, you will be asked to exercise with arms for approximately 12 to 15 minutes. The test will be started with no load for two minutes. After that, the load will be increased every two minutes until you are tired and cannot exercise any more. After the test, you will be asked to continue exercising with no load for two minutes and then rest for four minutes. You can stop exercising at any time during the test if you feel you cannot continue or do not want to continue.

Faculty of Rehabilitation Medicine

2-64 تنتخط Hall • University of Alberta • Edmonson • Canada • TGG 2G4 Telephone: (780) 492-2499 • Fast (780) 492-1626



In the next three tests you will be asked to sit on a vibrating chair for 15 minutes. The chair will be vibrated at three different levels (3 Hz, 4.5 Hz or 6 Hz). These levels are similar to that you feel while sitting in a car or truck that is going over a bumpy road at a slow speed. During all the exercise tests we will place a sterilized rubber tube in your mouth. This is like a snorkel that is used for underwater swimming. Your nose will be closed with a nose clip. The tube will be connected to a machine that will measure how hard you are breathing and how much oxygen you are using. We will place four sticky pieces of tape on your chest. These tapes will be connected with some wires to a machine. This will tell us how your heart is working during the exercise tests. We will also place a cuff around your left arm to measure your blood pressure while you are exercising. We will place elastic bandages around the lower back and forehead. These elastic bandages have a light source in them. They will measure how fast the muscles and brain utilize oxygen.

All these machines have been checked and they are working properly. Only a research assistant who knows how to use them will test you. If we see anything unusual while you are being tested we will stop the test immediately.

(4) Risks

During the arm cranking test you will: feel tired, have difficulty in breathing, and sweat. You may feel nauseous and dizzy during or after the exercise tests. This usually happen when a person exercises and will last for a few minutes. During any maximal exercise test, there is a possibility that you may have a problem with the heart. In this study, you will be allowed to volunteer only if you pass the screening process. During the tests, your responses will be continuously monitored. If anything unusual is observed the test will be stopped immediately. Over the last 10 years, we have safely conducted approximately 500 exercise tests using this procedure.

Studies that have examined the effects of vibration at these levels have shown no harm. During the vibration tests, your heart rate and breathing will increase as it does while walking at a comfortable speed.

Research studies that have tested volunteers during arm exercise and vibration have shown that they can be safely tested using these procedures.

(5) Benefits

If you volunteer for this study, you will get important information on the fitness level of the heart, lungs and muscles. This study will help us understand how vibration affects the brain, heart, lungs, and muscles in healthy men and women.

(6) Confidentiality

All information will be held confidential except when professional codes of ethics and/or legislation require reporting. Any report published as a result of this study will not identify you by name. The study data will be kept for at least seven years after the study is completed in a secure area accessible only by research team.

Faculty of Rehabilitation Medicine

2-64 Corbert Hall + University of Alberts + Edmonson + Canada + TGC 2C4 Telephone: (780) 492-2499 + Faz: (780) 492-1626



(7) Freedom to Withdraw

You are free to withdraw from the research study at any time. If any knowledge gained from this or any other study becomes available which could influence your decision to continue in this study, you will be promptly informed.

(8) Additional Contact

If you have any questions or concerns about this study, please contact Dr. Paul Hagier, Associate Dean of Graduate Studies, Faculty of Rehabilitation Medicine; Tel: 492-0841, email: paul.hagier@ualberta.ca.

.

Faculty of Rehabilitation Medicine

2-64 Carbox Hall • University of Alberts • Edmannes • Canada • T6G 3G4 Telephone: (780) 492-2499 • Fas: (780) 492-1626



CONSENT FORM

Part 1 (to be completed by the Principal Investigator):

Title: Tissue Oxygenation and Metabolism Comparisons during Incremental Pushing and Pulling and Arm Cranking

Principal Investigator(s): Dr. Yagesh Bhambhani

Co-Investigator(s): Ram Maikala, Ph.D Student, Faculty of Rehabilitation Medicine; 7248

Part 2 (to be completed by the research subject):

I believe that the person signing this for agrees to participate.	rm understands what	is involved in the study and ve	ohuntarily	
Printed Name		Printed Name		
Signature of Research Participant	Date	Witness		
I agree to take part in this study.				
This study was explained to me by: _				
Do you want the investigator(s) to infor participating in this research study? If	rm your family docto so, please provide yo (N.B. This ques	r that you are nur doctor's name: tion is optional).	Yes	No
Has the issue of confidentiality been ex who will have access to your records?	plained to you? Do	you understand	Yes	No
Do you understand that you are free to the study at any time? You do not have your care. (Use wording appropriate to	refuse to participate to give a reason and your subject group)	or withdraw from I it will not affect	Yes	No
Have you had an opportunity to ask qu	estions and discuss th	ais study?	Yes	No
Do you understand the benefits and risk research study?	ks involved in taking	part in this	Yes	No
Have you read and received a copy of t	he attached Informat	ioa Sheet?	Yes	No
Do you understand that you have been	asked to be in a resea	irch study?	Yes	No

Signature of Investigator or Designee

Date

THE INFORMATION SHEET MUST BE ATTACHED TO THIS CONSENT FORM AND A COPY GIVEN TO THE RESEARCH SUBJECT

Faculty of Rehabilitation Medicine

2-64 Carbox Hall • University of Alberta • Edmonton • Canada • 76G 2G4 Telephone: (780) 492-2499 • Fas: (780) 492-1626



INFORMATION SHEET

Project Title: Tissue Oxygenation and Metabolism Comparisons during Incremental Pushing and Pulling and Arm Cranking

Investigator(s):

Dr. Y. Bhambhani, Professor 3-73 Corbett Hall Department of Occupational Therapy University of Alberta Tel: 492 - 7248 Ram Maikala PhD Candidate 1-94 Corbett Hall University of Alberta Tel: 432-7943/ 492-0404

(1) Purpose

Pushing and pulling is moving any material without walking for some distance. It can be one hand or two-handed work like shoveling snow. Some researchers think that the body's response to pushing and pulling may depend upon the person's physical fitness. Others feel that pushing and pulling affects men and women differently. However, there are no studies that have tested these theories.

In this study, we will test: a) the effects of pushing and pulling and arm cranking on the oxygen levels of the back and arm muscles in healthy males and females.

(2) Study Requirements

This study will last three weeks. During this time you will be tested three times. Each test will last approximately 45 minutes. You will be tested for your back muscle endurance on the first day. This test will tell you how long you can hold your back when your upper body is unsupported. For this test, you will be asked to lye on your stomach on a bed for 2 minutes. A weight attached to a rope will be lowered to touch the portion between your shoulders. At the end of 2 minutes, a portion of the bed supporting your upper body is lowered. You will be asked to hold your arms across the chest and keep in contact with the weight as long as you can. When you are unable to be in contact with the rope, test will be stopped and your upper body will be supported for 4 minutes. Next two tests will tell us how well your heart, lungs, arm and back muscles working. These two tests are: arm cranking till exhaustion, and one hand pulling and pushing till exhaustion. You will be shown how the testing is done by a research assistant before you volunteer for the study.

During the first test, a blood pressure cuff will be folded around your dominant hand and blood flow will be stopped for a brief period. You will be asked to sit in a chair during this blood pressure test. This test will be useful for finding your minimum and maximum oxygen values in the arm at rest.

Faculty of Rehabilitation Medicine

2-64 Corbert Hall • University of Alberta • Edmonton • Canada • 76G 3G4 Telephone: (780) 492-2499 • Fuz: (780) 492-1626



(b) Testing Methodology

In the first test, you will be asked to exercise with arms for approximately 8 to 12 minutes. The test will be started with no load for two minutes. After that, the load will be increased every two minutes until you are tired and cannot exercise any more. After the test, you will be asked to continue exercising with no load for two minutes and then rest for four minutes. You can stop exercising at any time during the test if you feel you cannot continue or do not want to continue.

In the next four tests you will be asked to push and pull an attachment on the work simulator at your comfortable height using your dominant hand. The work simulator is adjustable for any height. During all the exercise tests we will place a sterilized rubber tube in your mouth. This is like a snorkel that is used for underwater swimming. Your nose will be closed with a nose clip. A tube will be connected to a machine that will measure how hard you are breathing and how much oxygen you are using. We will place four sticky pieces of tape on your chest. These tapes will be connected with some wires to a machine. This will tell us how your heart is working during the exercise tests. We will place elastic bandages around the lower back and forearm. These elastic bandages have a light source in them. They will measure how fast the muscles and utilize oxygen. For measuring your back endurance, you will be asked to rest your body on a couch with lower half of your body strapped to the couch. When instructed, you are asked to raise your upper body horizontal to the couch as long as you can.

All these machines have been checked and they are working properly. Only a research assistant who knows how to use them will test you. If we see anything unusual while you are being tested we will stop the test immediately.

(4) Risks

During the arm cranking test you will: feel tired, have difficulty in breathing, and sweat. You may feel nauseous and dizzy during or after the exercise tests. This usually happen when a person exercises and will last for a few minutes. During any maximal exercise test, there is a possibility that you may have a problem with the heart. In this study, you will be allowed to volunteer only if you pass the screening process. During the tests, your responses will be continuously monitored. If anything unusual is observed the test will be stopped immediately. Over the last 10 years, we have safely conducted approximately 500 exercise tests using this procedure. During the blood pressure cuff test, you may feel dizzy for few minutes but will go away as soon as the cuff is opened.

Research studies that have tested volunteers during arm exercise and pushing and pulling have shown that they can be safely tested using these procedures.

(5) Benefits

If you volunteer for this study, you will get important information on the fitness level of the heart, lungs and muscles. This study will help us understand how pushing and pulling affects the heart, lungs, and muscles in healthy men and women.

Faculty of Rehabilitation Medicine

2-64 Corbert Hall • University of Alberta • Edmonton • Canada • T6G 2G4 Telephone: (700) 452-2459 • Fac: (700) 452-1626



(6) Confidentiality

All information will be held confidential except when professional codes of ethics and/or legislation requir reporting. Any report published as a result of this study will not identify you by name. The study data will be kept for at least seven years after the study is completed in a secure area accessible only by research team.

(7) Freedom to Withdraw

You are free to withdraw from the research study at any time. If any knowledge gained from this or any other study becomes available which could influence your decision to continue in this study, you will be promptly informed.

(8) Additional Contact

If you have any questions or concerns about this study, please contact Dr. Paul Hagler, Associate Dean of Graduate Studies, Faculty of Rehabilitation Medicine; Tel: 492-0841, email: paul.hagler@ualberta.ca.

Faculty of Rehabilitation Medicine

2-64 Cashen Hell • University of Alberts • Edmonton • Canada • T6G 2G4 Telephone: (780) 492-2499 • Faz: (780) 492-1626 Physical Asthray Readiness Quantumaine - PAR-Q (purlant 1964)

•



Regular physical activity is fun and healthy, and increasingly more people are starting to become more active every day. Being more ~sive is very sale for most people. However, some people should check with their doctor before they start becoming much more /sically active.

If you are planning to become much more physically active than you are now, start by answering the seven questions in the box below. If you are between the ages of 15 and 69, the PAR-O will tell you if you should check with your doctor before you start. If you are over 68 years of age, and you are not used to being very active, check with your doctor.

Common series is your best guide when you answer these questions. Please read the questions carefully and answer each one honesely: check YES or NO.

	.NO		
	۵	1.	Has your doctor ever said that you have a heart condition and that you should only do physical activity recommended by a doctor?
	0	2.	Do you feel pain in your chest when you do physical activity?
	3. In the past month, have you had chest pain when you were not doing physical activity?		
	1 4. Do you tose your belance because of dizziness or do you ever lose consciousness?		
	5. Do you have a bone or joint problem that could be made worse by a change in your physical activity?		
0	٥	6.	Is your doctor currently prescribing drugs (for example, water pills) for your blood pressure or heart condition?
		7.	Do you know of <u>any other reason</u> why you should not do physical activity?
lf _ou answ	vere	ed	YES to one or more questions Talk with your ductor by phone or in person BEPORE you start becaming much more physically active or BEPORE you have a Binasa approisal. Tell your ductor about the PAR-Q and which questions you answered YES. • You may be able to do any activity you want — as long as you start slowly and build up gradually. Or, you may need to restrict your activities to those which are safe for you. Talk with your doctor about the hinds of activities you wish to participate in and follow higher advice. • Find out which examinely programs are safe and heighd for you.
Nyou and oure that	NO to all questions Nyou are used to hencestly to all PAR-Q questions, you can be reasonably sure that you can: • start becoming much more physically active — begin slowly and build up gradually. This is the safest and escient way to ge. • take part in a lineas appraisal — this is an excellent way to determine your basis linease as that you can plan the beat way for you to bre- gradually. The lineas appraisal — this is an excellent way to determine your basis linease as that you can plan the beat way for you to bre- gradually. The lineas appraisal — this is an excellent way to determine your basis linease as that you can plan the beat way for you to bre- gradually. The safe and a famous appraisal — this is an excellent way to determine your basis linease as that you can plan the beat way for you to bre- gradually. The safe and the break way for you to bre- gradually. The safe and the beat way for you to bre- gradually. The safe and the beat way for you to bre- gradually. The safe and the b		
• 986 • 99 (• 1080 990 • 605	partin partin r basic roly.	y, 11 a Ann Annai	is is the safest and essient way to pe. see approved — this is an excellent way to determine a so that you can plan the best way for you to the Construction of the safe way for you to the safe way for you to the Construction of the safe way for you to the s
- star - take yes ech	partin partin r basic roly. her com	y. Ti a Bina Binas PAR-	ile is the safest and essient way to ge. ses approleal — this is an excellent way to determine a so that you can plan the best way for you to the The Canadan Sectory for Econdes Physically, Heath Canada, and their agents assume no foldity for persons who undersite physical eachily, and the questionneirs, sensel your desar plus to physical eachily.
• other • year • year other • year • year	partin partin r basic roly. Aur con	y. The Bins	Is is the safest and essient way to pa. The approval — this is an excellent way to determine to as that you can plan the beat way for you to the The Canadan Sectory to Econtra Physically, Hauth Canada, and that sports assume no leading to parson who undership physical early, ord The Canadan Sectory to Econtra Physically, Houth Canada, and that sports assume no leading to parson who undership physical early, ord The Canadan Sectory to Econtra Physically, Houth Canada, and that sports assume no leading to parson who undership physical early, ord You are encouraged to copy the PAR-Q but only if you use the entire form
· diar · you oth · the you oth · the · the · the · the · the · the · the · the · the · t	produced opport in r basels redy. <u>No all fin</u> Aur com	y. The Bines	It is the safert and essient way to pa. The special — this is an excellent way to determine a so that you can plan the best way for you to the The Constan Sectory for Constan Physiclegy, Nexth Consta, and that special essame no failing for persons who understa physical estably, and the questionnain, second part dester play to physical activity. You are encouraged to copy the PAR-Q but only if you use the entire form and given to a person balance he or the personaes in a physical activity program or a finance approach, the section may be used for legal or
· offer · w (· take years offer · take offer · take · offer · take · offer · take · offer · o	partin partin r basis raly, <u>ta atita</u> Aur cam he partin he partin he parti	y. Th a Minist Minister PARA panta pata pa	As is the assist and essient way to permission as approximate the second of the second
- dan - yeu - yeu - yeu - gen -	portal porta porta reaction Aurona Aurona Aurona Aurona Aurona	y. The filmest	It is the safest and essient way to pa. The special — this is an excellent way to determine a so that you can plan the best way for you to the The Constan Secury to Eventso Physicage, Nexth Consta, and that special activity for persons who understap physical catholy, and The Constan Secury to Eventso Physicage, Nexth Consta, and that special activity for persons who understap physical catholy, and the quantizerism, security to Eventso Physicage Activity on the PAR-Q built only If you use the entire form and grave to a person before to or the personants is a physical activity program or a three answered to my full satisfaction.
	partial partia r basis roly. <u>a stilue</u> for com he <i>PAR</i> - he partial here st	y, 11 a line line patiet patiet patiet patiet patiet patiet patiet patiet	As is the estivat and essivel way to per as approved — this is an excellent way to determine to a softet you can plan the beat way for you to the The Canadan Sectory to Econtra Physicage, iterato Canada, and that agants assume no failing to persons who understood physical earthly, and The Canadan Sectory to Econtra Physicage, iterato Canada, and that agants assume no failing to persons who understood and completed this questionnaire. Any questions I had were answered to my full satisfaction. OATE
	rectual part in r basic raty. Mar can he part he part	y, T) a Biner Biner PAR- parti	And a specifical and assist way to pa. The approval — this is an excellent way to determine a so that you can plan the best way for you to the Constant Sectory for Exercise Physicage, Nexth Consta, and this agents assume as failing for any full sector play the physical activity. You are encouraged to copy the PAR-Q but only if you use the entire form and grins to a penal balancias for or the participant is a physical activity program or a these approved, this section may be used for logist or anderstood and completed this questionnaire. Any questions I had were answered to my full satisfaction. DATE

APPENDIX B

BONFERRONI ADJUSTMENTS

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

APPENDIX B.1

Chapter 2

Table 2.2: Postural and acceleration differences during WBV at different doses in healthy men and women

 \propto = 0.05 ÷ by 4 (# of comparisons) = 0.0125 level \approx 0.01^{Ψ}

Table 2.3: Cerebral Oxygenation and Blood Volume differences during WBV at different doses in healthy men and women

 \propto = 0.05 ÷ by 4 (# of comparisons) = 0.0125 level \approx 0.01^{Ψ}

APPENDIX B.2

Chapter 3

Table 3.2: Erector Spinae Oxygenation and Blood Volume values

 \propto = 0.05 ÷ by 4 (# of comparisons) = 0.0125 level \approx 0.01^{Ψ}

APPENDIX B.3

Chapter 4

Table 4.2: Metabolic and Respiratory responses during Whole-body Vibration

 $\alpha = 0.05 \div$ by 7 (# of comparisons) = 0.007 level $\approx 0.01^{\Psi}$

Table 4.3: Cardiovascular responses during Whole-body Vibration

 $\alpha = 0.05 + by 4$ (# of comparisons) = 0.0125 level $\approx 0.01^{\Psi}$

APPENDIX B.4

Chapter 5

Table 5.2: Peak Physiological, Perceptual and Biomechanical Responses during Incremental Pushing-Pulling and Incremental Arm Cranking in Healthy Men and Women

 $\infty = 0.05 \div$ by 13 (# of comparisons) = 0.003 level $\approx 0.01^{\Psi}$

Table 5.3: Biceps Brachii Oxygenation and Blood Volume responses

 $\alpha = 0.05 + by 7$ (# of comparisons) = 0.007 level $\approx 0.01^{\Psi}$

APPENDIX B.5

Chapter 6

Table 6.2: Erector Spinae Oxygenation and Blood Volume responses during Incremental Pushing-Pulling and Incremental Arm Cranking in Healthy Men and Women

 $\propto = 0.05 + \text{by 7}$ (# of comparisons) = 0.007 level $\approx 0.01^{\Psi}$

^{Ψ} For multiple comparisons, any \propto value obtained less than \propto =0.01 [for a smaller sample size] can be approximated to \propto = 0.01 (*Personal Communication, Dr. Todd Rogers, Director, Centre for Research in Applied Measurement and Evaluation, University of Alberta, Edmonton, CANADA*).